

City Research Online

City, University of London Institutional Repository

Citation: Mazzoni, M. (2024). Entanglement and symmetries in one-dimensional quantum field theories. (Unpublished Doctoral thesis, City, University of London)

This is the accepted version of the paper.

This version of the publication may differ from the final published version.

Permanent repository link: https://openaccess.city.ac.uk/id/eprint/35398/

Link to published version:

Copyright: City Research Online aims to make research outputs of City, University of London available to a wider audience. Copyright and Moral Rights remain with the author(s) and/or copyright holders. URLs from City Research Online may be freely distributed and linked to.

Reuse: Copies of full items can be used for personal research or study, educational, or not-for-profit purposes without prior permission or charge. Provided that the authors, title and full bibliographic details are credited, a hyperlink and/or URL is given for the original metadata page and the content is not changed in any way.

Entanglement and symmetries in one-dimensional quantum field theories



Michele Mazzoni City, University of London Department of Mathematics

Thesis submitted for the degree of $Doctor \ of \ Philosophy$ $23^{rd} \ October \ 2024$

La mia vita è nella fretta La mia strada assai ristretta La mia casa è una cantina La mia vita è in officina

Il lavoro a me mi stende E per giunta non mi rende Per comprarmi una maglietta Ho venduto la mia Lambretta

Quando prendo lo stipendio In gelati me lo spendo I gelati sono buoni Ma costano milioni

—Skiantos, Gelati

LIST OF PUBLICATIONS

During my PhD, I co-authored the following works, which have either been published or are currently under peer review:

- [1] Capizzi, Luca and Castro-Alvaredo, Olalla A. and De Fazio, Cecilia and Mazzoni, Michele and Santamaria-Sanz, Lucia "Symmetry resolved entanglement of excited states in quantum field theory. Part I. Free theories, twist fields and qubits" *JHEP*, 2022, 127(2022). https://doi.org/10.1007/JHEP12(2022)127
- [2] Capizzi, Luca and De Fazio, Cecilia and Mazzoni, Michele and Santamaria-Sanz, Lucia and Castro-Alvaredo, Olalla A. "Symmetry resolved entanglement of excited states in quantum field theory. Part II. Numerics, interacting theories and higher dimensions" *JHEP*, 2022, 128(2022). https://doi.org/10.1007/JHEP12(2022)128
- [3] Capizzi, Luca and Mazzoni, Michele and Castro-Alvaredo, Olalla A. "Symmetry resolved entanglement of excited states in quantum field theory. Part III. Bosonic and fermionic negativity" *JHEP*, 2023, 74(2023). https://doi.org/10.1007/JHEP06(2023)074
- [4] Castro-Alvaredo, Olalla A. and Mazzoni, Michele "Two-point functions of composite twist fields in the Ising field theory" J. Phys. A: Math. Theor., 2023, 56, 124001. https://doi.org/10.1088/1751-8121/acbe82
- [5] Capizzi, Luca and Mazzoni, Michele "Entanglement asymmetry in the ordered phase of many-body systems: the Ising field theory", *JHEP*, 2023, 144(2023). https://doi.org/10. 1007/JHEP12(2023)144
- [6] Mazzoni, Michele and Travaglino, Riccardo and Castro-Alvaredo, Olalla A. "Expectation Values of Conserved Charges in Integrable Quantum Field Theories out of Thermal Equilibrium", arXiv preprint arXiv:2402.14788, 2024. https://doi.org/10.48550/arXiv.2402.14788
- [7] Travaglino, Riccardo and Mazzoni, Michele and Castro-Alvaredo, Olalla A. "Generalised

Hydrodynamics of $T\bar{T}$ -Deformed Integrable Quantum Field Theories", JHEP, 2024, 90(2024). https://doi.org/10.1007/JHEP08(2024)090

• [8] Capizzi, Luca and Mazzoni, Michele "Entanglement content of kink excitations", arXiv preprint arXiv:2409.03048, 2024. https://doi.org/10.48550/arXiv.2409.03048

This thesis is based on the works [1-5].

Contents

			P	age			
A	Acknowledgements						
D	Declaration of Authenticity						
A	bstra	ıct		xi			
1	Intr	roduct	ion	1			
	1.1	Entan	glement in many-body quantum systems	3			
	1.2	Bipart	tite entanglement in $1 + 1D$ quantum field theories	9			
	1.3	The re	ole of internal symmetries	15			
	1.4	Struct	ture of the thesis	21			
P	ART	I : S	symmetry-resolved entanglement of localised excitations in massive				
	con	plex f	ree theories	23			
2	\mathbf{Syn}	nmetry	y-resolved entropy of excited states	24			
	2.1	Introd	luction and summary of results	24			
	2.2	Charg	ged moments and SREE of free theories: form factor approach	29			
		2.2.1	Composite branch-point twist field factorisation	29			
		2.2.2	Complex free boson	32			
		2.2.3	Complex free fermion	35			
		2.2.4	Symmetry-resolved entanglement entropies	37			
	2.3	The q	ubit picture	40			
		2.3.1	Charged moments of multi-qubit states	40			
		2.3.2	SREE of multi-qubit states	42			
	2.4	Gener	alisations to interacting theories and higher dimensions	45			
		2.4.1	Magnon states	45			
			2.4.1.1 One-magnon states	46			
			2.4.1.2 Two-magnon states	47			
		2.4.2	Twist operator approach	49			

			2.4.2.1 Excited states and operator algebra	50
			2.4.2.2 Replica construction for symmetry-resolved entanglement	53
			2.4.2.3 Single-particle state	54
	2.5	Nume	rical results	56
		2.5.1	1D lattice Fermi gas	56
		2.5.2	Complex free boson and 1D harmonic chain	59
			2.5.2.1 The wave-functional method	60
			2.5.2.2 The harmonic chain	62
			2.5.2.3 Numerical results	65
	2.6	Conclu	uding remarks	68
	App	endices		69
	2.A	Comp	lex free boson computation	69
		2.A.1	Single-particle excited states	69
		2.A.2	Free boson $(k = 1, n = 2)$	71
		2.A.3	Multi-particle excited states	72
		2.A.4	Free boson $(k = 2, n = 2)$	73
	2.B	Comp	lex free fermion computation	74
		2.B.1	Single-particle excited states	74
		2.B.2	Multi-particle excited states	76
	$2.\mathrm{C}$	Finite	-volume two-point function in the ground state	79
	2.D	Trace	calculations via wave-functional method	82
		2.D.1	Zero flux	82
		2.D.2	Non-trivial flux insertion	85
3	Syn	nmetry	v-resolved negativity of excited states	87
	3.1	Introd	luction	87
	3.2	Qubit	computation	92
		3.2.1	Single bosonic excitation	92
		3.2.2	Single fermionic excitation	94
		3.2.3	Multiple distinct excitations	95
		3.2.4	Multiple identical excitations	96
	3.3	Twist	operator approach	97
		3.3.1	Single bosonic excitation	98
		3.3.2	Single fermionic excitation	103
			3.3.2.1 Single fermionic excitation via replica diagonalisation 1	105
		3.3.3	Bosonic state with multiple distinct excitations	106

		3.3.3.1 An example: $k = n = 2$	108
	3.4	Numerics	109
		3.4.1 The method	110
		3.4.2 Lattice Fermi gas	111
	3.5	Concluding remarks	112
	App	endices	114
	3.A	Combinatorics	114
		3.A.1 Non-vanishing strings	114
		3.A.2 Combinatorial counting of strings	115
	3.B	Mathematical identities	116
		3.B.1 Generalised Lucas polynomials and a proof of equation (3.63)	116
		3.B.2 Product formulae	118
D			100
P	ART	II: Twist operators and entanglement measures in the Ising field theory	120
4	Two	p-point function of composite twist fields in the disordered phase of the Ising	
	field	l theory	121
	4.1	Introduction	121
	4.2	Field content of the Ising model and form factors	124
		4.2.1 Two-point function and cumulant expansion	127
		4.2.2 Two-particle contribution	128
	4.3	Higher particle contributions: closed formulae	129
		4.3.1 Recursive formulae	130
		4.3.2 Main result of this section	132
	4.4	Conformal dimension from the cumulant expansion	132
		4.4.1 Exponential factors	133
		4.4.2 Short-distance behaviour of the cumulant expansion	134
		4.4.3 Leading contribution to $c_{2k}^{(1)}(\ell;n)$	135
		4.4.4 Leading contribution to $c_{2k}^{(2)}(\ell) + c_{2k}^{\mu}(\ell) \dots \dots \dots \dots \dots \dots \dots \dots \dots$	136
	4.5	Analytic continuation to $n \in \mathbb{R}^{\geq 1}$	138
		4.5.1 Analytic continuation of leading short-distance contributions	138
		4.5.2 Analytic continuation of the <i>n</i> -derivative	141
	4.6	Concluding remarks	143
	App	endices	143
	4.A	Summation formulae	143
5	Ent	anglement asymmetry in the ordered phase of the Ising field theory	147

Re	References				
6	Out	look	183		
		duality	180		
	$5.\mathrm{C}$	Exact results for the Rényi entropies via corner transfer matrix and Kramers-Wannier			
	$5.\mathrm{B}$	Useful inequalities	178		
	5.A	Two-particle form factors of composite twist fields	176		
	App	endices	176		
	5.6	Concluding remarks	175		
		5.5.1 Zero-entanglement states	174		
	5.5	Further generalisations	173		
		5.4.3 Entanglement asymmetry	170		
		5.4.2 Two-particle form factors	169		
		5.4.1 Zero-particle form factors	168		
		ferromagnetic phase of Ising QFT	167		
	5.4	Form factors for the composite twist fields and entanglement asymmetry in the			
		5.3.3 Rényi entropy and its analytic continuation	165		
		5.3.2 Two-particle form factors	163		
		5.3.1.2 Vanishing one-particle form factors and further constraints	162		
		5.3.1.1 Zero-particle form factors	162		
		5.3.1 Standard twist fields	160		
	5.3	Form factor program for the twist fields in the Ising ferromagnetic phase	156		
		5.2.2 Entanglement asymmetry via twist operators	155		
	0.2	5.2.1 Characterisation of the twist operators	151		
	5.2	Twist operators and entanglement asymmetry	151		
	5.1	Introduction	147		

ACKNOWLEDGEMENTS

Writing a PhD thesis is a walk in the park: be careful where you lay your feet. If I have not stepped on any really big one, the merit goes to my supervisor. Olalla, my warmest gratitude to you for having been a wonderful supervisor and human being, always providing the guidance and support I needed throughout my PhD. A second, huge thank you goes to Benjamin, who so many times had the right answers to my questions about Physics. I want to express my gratitude to the colleagues (and friends) I had the pleasure to work or share ideas with: Luca, Riccardo, Cecilia, Lucia, Federico, Giuseppe, Friedrich, David. I truly learnt a lot from all of you. I am also very grateful to all the staff at the Department of Mathematics of City, University of London, for creating a friendly and warm environment where PhD students feel heard: I hope things will keep up this way and I will always remember City with a smile.

Like most things, a walk in the park is more pleasant when you are with a loved one, and did I have the best company. To start with, I want to thank Fabio, Diogo, Beth, Elli, Elohim, Damiano, Maddalena, Laura: it is not obvious that office mates should be friends, and I am so happy I can call you so. To James, Aurora, Matteo, Didem, Robbie, Kim, Ottavia, Ami: each one of you has been a fundamental person for me in the past three years (some of you for more than that). You are my home in London and I love you. As for my *actual* home in London, my deepest gratitude to Alan and Lorenzo for being great friends and really the best housemates I could have hoped for. Finally, from the bottom of my heart, I want to thank my family, Cristina, Paolo and Marta. You have always been there for me, the greatest support has come from you. Thank you.

DECLARATION OF AUTHENTICITY

This report contains genuine work conducted originally by the author. The work presented herein has not been submitted and/or accepted for the award of any other degree or diploma in any university. To the best of my knowledge and belief, this thesis contains no materials previously published or written by other person, except where due references has been made.

ABSTRACT

In this thesis, we investigate the interplay between entanglement and internal symmetries in 1+1D quantum field theories (QFTs). Although the notion of symmetry-resolved entanglement has been widely studied in various systems, its behavior in excited states of massive QFTs remains largely unexplored. In the first part of the thesis, we focus on theories with global U(1) symmetry and compute the symmetry-resolved entanglement entropy and logarithmic negativity of zero-density excited states of the massive free complex boson and Dirac fermion. We find that the excess of symmetry-resolved entropy (and negativity) of these states with respect to the ground state is largely independent on the details of the excited state and it has a very simple dependence on the U(1) charge. We test our results numerically on a one-dimensional Fermi gas and on a one-dimensional harmonic chain, and we propose an interpretation of our formulae in terms of simple multi-qubit states. Next, we generalise the field-theoretic computation of symmetry-resolved entanglement measures to higher-dimensional, non-integrable field theories using semi-local twist operators, which are defined through their commutation relations with an algebra of local observables in a QFT. In the second part of the thesis, we turn our attention to the one-dimensional massive Ising QFT, which possesses a \mathbb{Z}_2 symmetry. The ground state in the paramagnetic (disordered) phase of the theory is symmetric, and its \mathbb{Z}_2 -resolved entanglement entropy can be obtained from a two-point function of composite twist fields. We provide an exact expression for the cumulant expansion of this two-point function. In contrast, the ferromagnetic (ordered) phase features two \mathbb{Z}_2 -breaking vacua. The extent to which the symmetry is broken can be quantified by a relative entropy measure known as entanglement asymmetry. By making use of twist operators, we develop a method to compute entanglement asymmetry in massive 1+1D QFTs with discrete internal symmetry and apply this approach to the Ising model.

CHAPTER

ONE

INTRODUCTION

Over the past decades, advances in experimental technologies have led to extensive exploration of the role of entanglement in quantum many-body systems [9]. In particular, the study of entanglement measures in quantum field theory (QFT), which started in the 1990s with the seminal works [10, 11] and was then crucially extended by Calabrese and Cardy [12], brought results to the attention of a much wider scientific community. These theoretical results, in conjunction with numerical and analytical work in integrable spin chain models [13–16], revealed how certain entanglement measures, such as entanglement entropy [17], display universal scaling at conformal critical points [18]. This observation has many implications, a very important one being that computing the entanglement entropy of a pure state is one of the most numerically effective ways of determining if the theory is critical.

A recent development in this field is the growing interest in a type of entanglement termed symmetry-resolved entanglement. In the context of conformal field theory (CFT), a definition of this quantity was given in [19], where it was related to correlation functions of generalised (or composite) branch-point twist fields (CBPTF). In the context of entanglement, such fields were first introduced in [12] as associated with conical singularities in conformal maps and in [18, 20, 21] as symmetry fields associated to cyclic permutation symmetry in 1+1D QFT (both critical and gapped). The basic idea is that in theories that posses an underlying symmetry (say U(1) symmetry in a complex free boson theory or in sine-Gordon theory) entanglement can be expressed as a sum over contributions from different symmetry sectors. A strong motivation to study the symmetry-resolved entanglement measures is that the different contributions to the total entropy are experimentally measurable [22–24] and can be related to the operationally accessible entanglement in quantum computing [25–27]. Prior to the works presented in this thesis, several results existed for the symmetry-resolved entanglement of symmetric ground states of CFT, massive QFT and spin chains [28], as well as for low-lying excited states of critical theory, but not for excited states of massive QFTs¹.

Alongside with progress in the study of symmetry-resolved entanglement measures, in the past two years growing attention has been devoted to the notion of entanglement asymmetry. Introduced in [29], entanglement asymmetry provides the first information-based probe of symmetry breaking: it quantifies how much a certain state breaks an underlying internal symmetry of the system, spontaneously or explicitly. Initially, such quantity was used to analyse the restoration (or lack thereof) of a U(1) symmetry in the quench dynamics of quantum spin chains [29–31]. Shortly after these first works, entanglement asymmetry studies were conducted on CFT [32], critical spin chains with larger symmetries [33] and black-hole radiation [34]. The largest part of the research regarding entanglement asymmetry has thus far been devoted to the dynamics of the entanglement asymmetry, whereas a characterisation of this measure in a field-theoretic setting and for more complex symmetry groups is mostly still lacking.

In part I and part II of this thesis, we contribute to filling the two large gaps in the literature that we mentioned above. Namely, in part I we characterise the symmetry-resolved entanglement content of excited states of massive QFTs, and in part II we introduce a field-theoretic framework for the study of entanglement asymmetry in ordered phases of massive field theories. The two parts of the thesis can also be distinguished according to the type of symmetry we treat. In part I, we deal with theories that possess an internal U(1) symmetry, namely the complex boson and the Dirac fermion. In part two of the thesis, on the other hand, we consider the massive Ising field theory, the most paradigmatic quantum field theory featuring discrete \mathbb{Z}_2 symmetry. Chapter 4, in which we derive exact formulae for a correlation function in the disordered phase of the Ising QFT, is included in part II of thesis according to this criterion. Among our main contributions in the two parts of this thesis is the introduction of a class of operators, that we call *twist operators* and that generalise objects known in 1 + 1D dimensions as branch-point twist fields. We will define twist operators in Chapter 2 and make a consistent use of them throughout the rest of the thesis.

The remaining part of this Introduction contains an essential review of the theoretical foundations necessary for the following chapters: the definition and main properties of entanglement measures, the replica model in 1 + 1D quantum field theories, the definition of symmetry-resolved entanglement and entanglement asymmetry. A more detailed outline of the thesis structure concludes this chapter.

¹Throughout this thesis, the expression "massive QFT", without further specification, refers to either a free or interacting local QFT with a mass gap.

1.1 Entanglement in many-body quantum systems

Entanglement is arguably the quantum phenomenon that irremediably marks the departure from deterministic, classical physics. This is, at least, the opinion of someone who might have had a say in the matter [35]. Entanglement is essentially present when there are correlations of an intrinsic quantum nature between two parts of a system prepared in a certain state. Thus, entanglement depends on the state and on the way the system is partitioned. Quantum correlations can be revealed by a measurement on one of the two parts of the system, which causes the collapse of the wavefunction according to the Copenhagen interpretation of quantum mechanics [36]. The non-locality of the collapse was at the heart of a debate about the *completeness* of quantum mechanics as a fundamental theory of reality, a debate that culminated in the famous 1935 article by Einstein, Podolski and Rosen [37]. In this paper, the authors proposed that quantum correlations between distant particles should be explained by means of some local hidden variables: our ignorance about the probability distribution associated with these variable would then explain the most counter-intuitive features of quantum mechanics. This position on the fundamental properties of quantum mechanics is sometimes referred to as *local realism*. However, if there are local hidden variables of a purely classical nature, then the possible results of a measurement should satisfy certain inequalities, the first of which were derived by Bell in 1964 [38]. The violation of Bell inequalities was crucially proved by Alain Aspect and collaborators in a series of experiments that led to the 2022 Nobel Prize in Physics [39–41]. The experimental proof that there can be no local hidden variables marks the end of local realism: quantum mechanics is complete and quantum entanglement cannot be explained in terms of classical correlations.

With the developments in quantum technology achieved in the 1980s and 1990s, entanglement started to be considered a resource in the field of information and communication protocols. Just to mention two examples, it is possible to encode in a single *qubit* (quantum bit) the information carried by two classical bits, as long as the qubit belongs to an entangled pair: this is known as *dense coding* [42]. Moreover, by exploiting entanglement it is possible to achieve *quantum teleportation* [43], i.e. transferring a state between two observers (without neither of them knowing what the state actually is).

There are different ways of measuring entanglement, depending on the different points of view that may be adopted. For instance, one may be interested in measuring the entanglement between two spins at different sites in a quantum chain, ignoring the rest of the system: this is an example of pairwise entanglement, which can be measured for instance by the entanglement of formation [9, 44]. Conversely, one may be interested in measuring how entangled are two large subsystems of a larger system: in this case, one talks of entanglement of regions. This is the point of view we will adopt in this thesis.

We give the following definition of entanglement. Let $\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_B$ be the Hilbert space corresponding to a total system partitioned into subsystems A and B. Let $\{|\phi_i\rangle_A\}$, $\{|\chi_i\rangle_B\}$ be basis of \mathcal{H}_A and \mathcal{H}_B respectively. Then a state:

$$|\psi\rangle = \sum_{i,j} \alpha_{i,j} |\phi_i\rangle_A \otimes |\chi_j\rangle_B \in \mathcal{H}, \quad \alpha_{i,j} \in \mathbb{C}, \quad \sum_{i,j} |\alpha_{i,j}|^2 = 1,$$
(1.1)

is said to be separable if there exist some $|\phi\rangle_A \in \mathcal{H}_A$, $|\chi\rangle_B \in \mathcal{H}_B$ such that

$$|\psi\rangle = |\phi\rangle_A \otimes |\chi\rangle_B. \tag{1.2}$$

A state is entangled with respect to a given bipartition if it is not separable.

In quantum many-body systems, it is often convenient to adopt the viewpoint of density operators. A density operator, or density matrix, is a map $\rho : \mathcal{H} \to \mathcal{H}$ that encodes all the characteristics of a state. For a state $|\psi\rangle \in \mathcal{H}$, the density matrix is defined as the projector:

$$\rho = |\psi\rangle\langle\psi|. \tag{1.3}$$

A state for which the density matrix can be written as a projector, i.e. every state defined by a vector as in (1.1), is called a *pure state*. However, the true advantage of the density matrix formalism is that density matrices incorporate a notion of classical probability, which is present if the state of the system is a statistical mixture. Indeed, suppose that one wants to describe an ensemble of states $\{|\psi_i\rangle \in \mathcal{H}\}$ in which each state is associated to a certain probability p_i . The correct way to represent this ensemble is via the density operator:

$$\rho = \sum_{i} p_i |\psi_i\rangle \langle \psi_i|, \quad \sum_{i} p_i = 1.$$
(1.4)

This is called a *mixed state*. The paradigmatic example of a mixed state is given by the usual Gibbs ensemble (or thermal state), which in quantum statistical mechanics is described by a density matrix:

$$\rho_{\beta} = \frac{e^{-\beta H}}{Z} = \sum_{n} \underbrace{Z^{-1} e^{-\beta E_n}}_{p_n} |n\rangle \langle n|, \quad \sum_{n} p_n = 1,$$
(1.5)

where $\{|n\rangle\}$ are the energy eigenstates. The Gibbs ensemble displays all the natural requirements of a density operator, namely, it is a positive semi-definite, Hermitian operator of unit trace:

$$\rho = \rho^{\dagger}, \quad \langle \psi | \rho | \psi \rangle \ge 0 \quad \forall | \psi \rangle \in \mathcal{H}, \quad \text{Tr}_{\mathcal{H}} \rho = 1.$$
(1.6)

Every operator with these properties can be written in the form (1.4). Additionally, density matrices of pure states (and of pure states only) are idempotent, $\rho^2 = \rho$: for this to hold, all the probabilities in (1.4) must be zero except for one $p_i = 1$, and the state reduces to (1.3) for a certain $|\psi\rangle$.

In the state ρ , the expectation value of a local observable represented by an operator $\mathcal{O} : \mathcal{H} \to \mathcal{H}$ is given by $\operatorname{Tr}_{\mathcal{H}}(\rho \mathcal{O})$. However, in systems characterised by a large number of degrees of freedom, it often occurs that an observer can access only a portion of the total system. Suppose the observer is interested in the expectation value of a local operator $\mathcal{O}_{\mathcal{A}} : \mathcal{H}_A \to \mathcal{H}_A$. This is easily embedded in the algebra of operators acting on the total space by defining $\mathcal{O} := \mathcal{O}_{\mathcal{A}} \otimes \mathbb{1}_B$. Then, by picking an orthonormal basis $\{|e_i, e_j\rangle := |e_i\rangle_A \otimes |e_j\rangle_B\}$ for the total Hilbert space, the expectation value of \mathcal{O} reads:

$$\langle \mathcal{O} \rangle = \operatorname{Tr}_{\mathcal{H}}(\rho(\mathcal{O}_{\mathcal{A}} \otimes \mathbb{1}_{B}))$$
$$= \sum_{i,j,i',j'} \langle e_{i}, e_{j} | \rho | e_{i'}, e_{j'} \rangle \langle e_{i'}, e_{j'} | (\mathcal{O}_{\mathcal{A}} \otimes \mathbb{1}_{B}) | e_{i}, e_{j} \rangle = \operatorname{Tr}_{\mathcal{A}}(\rho_{A}\mathcal{O}_{A}).$$
(1.7)

The quantity

$$\rho_A = \mathrm{Tr}_{\mathcal{H}_{\mathcal{B}}}\rho,\tag{1.8}$$

is the reduced density matrix (RDM) of the subsystem A. It encodes the description of the system according to an observer who can only access a portion of the total space: the degrees of freedom associated to the complementary region B describe then the environment, and are traced out. When taking the partial trace of a pure state $\rho = |\psi\rangle\langle\psi|$, one may end up with a state ρ_A which is mixed. Namely, this happens when (and only when) $|\psi\rangle$ is not separable. Therefore, the mixedness of the reduced density matrix is a criterion for the presence of entanglement in a pure state of the total system: the state $|\psi\rangle \in \mathcal{H}_A \otimes \mathcal{H}_B$ is separable if and only if its reduced density matrix has only one non-vanishing eigenvalue. We do not specify which of the two density matrices: indeed, as a consequence of the Schmidt decomposition, the two operators ρ_A and ρ_B have the same spectrum, and can only differ in the number of their zero eigenvalues.

The inspection of the reduced density matrix is sufficient to reveal whether a state is separable, however, it does not provide information about the amount of entanglement in that state. To do so, one needs to define a *measure of entanglement*. In particular, for most of this thesis we will be interested in quantifying the bipartite entanglement of pure states. The reason is that these states play a prominent role in many-body systems: they describe the (non-degenerate) ground state of spin chains and of zero-temperature QFTs. Also the particular excitations we will consider in Chapter 2, i.e. zero-density excited states of quantum field theories, are pure states. There are several ways to define entanglement measures, but all good measures should be *entanglement monotones*. An entanglement monotone is a functional E from the set of quantum states to the real numbers, which satisfies the fundamental properties [44]:

- non-negativity: $E(\rho) \ge 0$ for every state ρ , and $E(\rho) = 0$ if ρ is separable.
- monotonicity under LOCC: if the state σ is obtained from ρ via local operations and classical communications (LOCC) then $E(\rho) \ge E(\sigma)$.

The notion of LOCC can be given a precise mathematical formulation by means of Kraus operators [45]. Here, we only remark that requiring E to be non-increasing under LOCC is fundamental for at least two reasons. First, from a technological point of view, local quantum operations and classical communications between separate subsystems are necessary for the implementation of protocols such as quantum teleportation: asking that the state which is teleported does not become more entangled during the process seems quite obvious. Second, LOCC operations can be employed to distinguish between classical and quantum correlations: classical correlations are the ones that can be generated by acting on a certain quantum system via LOCC.

The prototypical entanglement monotone for pure states in a bipartite quantum system is the entanglement entropy. To define the entanglement entropy, we start from the von Neumann entropy, i.e. the quantum analog of the Shannon entropy of classical information theory [46]. For a state $\rho: \mathcal{H} \to \mathcal{H}$ (pure or mixed), the von Neumann entropy:

$$S[\rho] := -\mathrm{Tr}(\rho \log \rho), \tag{1.9}$$

has (among others) the following properties:

- $S[\rho] \ge 0$ for every state ρ and $S[\rho] = 0$ if and only if ρ is pure.
- If $\dim(\mathcal{H}) = N$, $\max_{\rho} S[\rho] = \log N$, obtained for the maximally mixed state $\rho = \frac{1}{N} \mathbb{1}_{\mathcal{H}}$.
- It is concave: if $\sum_i \lambda_i = 1$, $S[\sum_i \lambda_i \rho_i] \ge \sum_i \lambda_i S[\rho_i]$.
- It is sub-additive: if $\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_B$ and ρ_A , ρ_B are the two RDMs, then $S[\rho] \leq S[\rho_A] + S[\rho_B]$. The equality holds only if $\rho = \rho_A \otimes \rho_B$.

The entanglement entropy is the von Neumann entropy of the reduced density matrix:

$$S_A = -\mathrm{Tr}_{\mathcal{H}_A}(\rho_A \log \rho_A). \tag{1.10}$$

The von Neumann entropy vanishes on pure states, but the reduced density matrix ρ_A is pure if and only if the full state ρ is separable. This means that the entanglement entropy of a bipartition is zero if and only if ρ is not entangled. Moreover, because the two reduced density matrices have the same non-vanishing eigenvalues, $S_A = S_B$. In the following chapters, we will equivalently refer to the von Neumann or entanglement entropy of a bipartition.

The entanglement entropy is the most ubiquitous measure of entanglement, being widely used in quantum many-body physics [9], quantum information theory [47], quantum and conformal field theory [12, 18, 20, 48], black-hole physics and holography [10, 11, 49]. There are several reasons behind the popularity of entanglement entropy in condensed matter and low-dimensional QFT. First of all, it recently became experimentally accessible through measurements of a strictly related quantity, the Rényi entropy. The Rényi entropy of order α , or α th Rényi entropy, for $\alpha \in \mathbb{R}^+$ and $\alpha \neq 1$, is defined as [50]:

$$S_{\alpha}[\rho_A] := \frac{1}{1-\alpha} \log \operatorname{Tr}(\rho_A^{\alpha}), \qquad (1.11)$$

and it reproduces the entanglement entropy in the limit $\alpha \to 1$:

$$S_A = \lim_{\alpha \to 1} S_\alpha[\rho_A]. \tag{1.12}$$

The authors of [22] were able to measure the second Rényi entropy in a trapped gas of ultra-cold bosonic atoms by letting two many-body wavefunctions interfere². In the next section, we will show how the Rényi entropies can be computed in a *replica version* of a quantum field theory in 1 + 1D: the "replica trick" will then be extensively used throughout the entire thesis.

Other than the experimental and computational advantages, the entanglement entropy is a prime measure of entanglement because it encodes universal features and scaling properties of a theory. The universality stems from the fact that the ground state entanglement entropy (as well as the thermal entropy) of a critical system, described for instance by a two-dimensional conformal field theory, contains information about the central charge of such theory, and thus identifies a certain universality class: we will provide some more details about this in the next section. The scaling properties, on the other hand, are related to the celebrated *area law* [51]. This is the statement that the entanglement entropy S_A of a bipartite system generally scales with the (generalised) area of the surface ∂A that separates the two subsystems A and B. In D dimensions, this means:

$$S_A \sim |\partial A| \sim \ell^{D-1},\tag{1.13}$$

where ℓ is a characteristic length scale associated with either of the two subsystems. The physical intuition behind this law is that the entanglement entropy is due to those pairs of entangled particles for which one particle is in A and one is in B. For systems with short-range correlations, the number of such pairs is roughly proportional to the area of the separating surface. The validity of the area law

²For the sake of precision, the quantity that was experimentally measured is the quantum purity $\text{Tr}(\rho_A^2)$, the logarithm of which is proportional to $S_2[\rho_A]$.

for the ground state of one-dimensional gapped spin chains (i.e. with finite correlation length $\xi < \infty$) was proved by Hastings in [52], whereas gapless spin chains, which are described by 1 + 1D CFTs in the continuum limit, violate the area law. A good review about the area law of entanglement entropy can be found in [53].

We conclude this section by introducing a different measure of entanglement. We mentioned that the entanglement entropy is a good measure for bipartite systems in a pure state. When the state is mixed, the entropy is no longer an adequate measure of entanglement because it fails to discern true quantum correlations from the statistical correlations present in the initial ensemble. A similar scenario occurs in many-body systems in two paradigmatic cases: when the initial state is thermal, as in equation (1.5), or when the state itself is pure, but the system is not bipartite, i.e. the subsystems A and B are not complementary. In this thesis, we focus on the second situation. Let us consider a tripartite Hilbert space:

$$\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_B \otimes \mathcal{H}_C, \tag{1.14}$$

where the subsystem C plays the role of the environment, and suppose we want to measure the entanglement between subsystems A and B in a pure state ρ of the total Hilbert space. The reduced density matrix $\rho_{A\cup B} = \text{Tr}_{\mathcal{H}_C}\rho$ is mixed. A necessary condition for this state to be separable is that its partial transpose $\rho_{A\cup B}^{T_B}$ has no negative eigenvalues (this is known as the Peres-Horodecki criterion [54, 55]). The quantity $\rho_{A\cup B}^{T_B}$ is defined by transposing the matrix only with respect to the subsystem B: if $\{|e_i, e_j\rangle := |e_i\rangle_A \otimes |e_j\rangle_B\}$ is an orthonormal basis of $\mathcal{H}_A \otimes \mathcal{H}_B$, then

$$\langle e_i, e_j | \rho_{A \cup B}^{T_B} | e_k, e_l \rangle := \langle e_i, e_l | \rho_{A \cup B} | e_k, e_j \rangle.$$
(1.15)

According to the Peres-Horodecki criterion, the *logarithmic negativity* [56, 57]:

$$\mathcal{E}[\rho_{A\cup B}] := \log \operatorname{Tr}(|\rho_{A\cup B}^{T_B}|), \qquad (1.16)$$

where we introduced the trace norm of a matrix, $\text{Tr}|A| := \text{Tr}\sqrt{A^{\dagger}A}$, is a good indicator of entanglement of the state $\rho_{A\cup B}$ (and an entanglement monotone). Indeed, it is possible to show that $\rho_{A\cup B}^{T_B}$ has real spectrum, but in general it is not positive semi-definite, and the trace norm of the partial transpose can be equivalently expressed as the sum of all its singular values. Therefore, a logarithmic negativity $\mathcal{E} > 0$ necessarily indicates that $\rho_{A\cup B}^{T_B}$ has at least one negative eigenvalue. This means that the subsystems A and B are entangled. We postpone a more detailed discussion of the logarithmic negativity to Chapter 3.

1.2 Bipartite entanglement in 1 + 1D quantum field theories

In this section, we review the replica method for the computation of bipartite entanglement entropy in (1 + 1)-dimensional QFTs. As we show below, in a replica theory a central role is played by some fields which implement cyclic permutations of replicas and display semi-locality properties with respect to the fundamental fields of the theory: these are the branch-point twist fields (BPTF). In general, twist fields are associated to internal symmetries of a theory, and they first appeared in the context of orbifold CFT, where they describe the propagation of bosonic strings in a background described by a \mathbb{Z}_N orbifold [58, 59]. In statistical field theory, a prominent example of twist fields is given by the order and the disorder fields of the Ising field theory [60–62], which are associated to the \mathbb{Z}_2 symmetry of the model and are semi-local with respect to the fermion field. The replica method for the computation of bipartite entanglement in CFT was employed for the first time in the pioneering works [10, 11]. These results were then extended in [12], where the entanglement of an interval in a replica theory of a CFT (at finite temperature and in the ground state) was computed by means of correlation functions of some primary operators of the CFT. The full picture was finally unveiled in [20], where the case of massive integrable QFT was treated: here, the branch-point twist fields were first defined in relation to the symmetry of the replica theory under permutation of copies. These fields are the massive counterpart of the CFT primary operators introduced in [12], i.e. the fields that flow to the CFT branch-point twist fields in the UV limit. The form-factor program for the BPTF was also developed in [20], and extended to theories with a boundary in [63]. We refer to [64] for a definition of twist fields associated to an internal continuous symmetry. In the following review, we mostly follow [12], [20] and [18].

Let us consider the ground state $|\psi\rangle \in \mathcal{H}_A \otimes \mathcal{H}_B$ of a one-dimensional QFT, where $A = [x_1, x_2]$, $B = (-\infty, x_1) \cup (x_2, \infty)$. We aim to compute the bipartite entanglement entropy of this state. This is done by first obtaining an expression for the Rényi entropy (1.11) for $\alpha = n \in \mathbb{N}$, n > 1, and then performing an analytic continuation over real values of n. The density matrix $\rho = |\psi\rangle\langle\psi|$ is obtained by taking the limit $\beta \to \infty$ of the thermal state (1.5), whose matrix elements can be expressed by an Euclidean path integral as:

$$\langle \varphi_1 | \rho_\beta | \varphi_2 \rangle = \frac{1}{Z(\beta)} \int_{\varphi(x,0)=\varphi_1(x)}^{\varphi(x,\beta)=\varphi_2(x)} \mathcal{D}\varphi \, e^{-S_E[\varphi]}, \quad Z(\beta) = \int_{\varphi(x,0)=\varphi(x,\beta)} \mathcal{D}\varphi \, e^{-S_E[\varphi]}, \tag{1.17}$$

where we assume that $\varphi(x,\tau)$ is the fundamental scalar field in the theory and the Euclidean action is $S_E[\varphi] = \int_0^\beta d\tau \int_{-\infty}^\infty dx \mathcal{L}[\varphi](x,\tau)$. The reduced density matrix is obtained by splitting the field φ in the components φ_A and φ_B defined on the regions A, B respectively and tracing out the degrees of freedom φ_B :

$$\langle \varphi_{1,A} | \rho_{\beta,A} | \varphi_{2,A} \rangle = \frac{1}{Z(\beta)} \int \mathcal{D}\varphi_B \int \mathcal{D}\varphi \, e^{-S_E[\varphi]}, \qquad (1.18)$$



Figure 1.1 Replica manifold \mathcal{R}_n for n = 3. The branch cuts in each sheet are along the interval $A = [x_1, x_2]$. Picture taken from [20].

where the boundary conditions for the integration in $\mathcal{D}\varphi$ are $\varphi(x,0) = \varphi_{1,A}(x)$ and $\varphi(x,\beta) = \varphi_{2,A}(x)$ for $x \in A$, $\varphi(x,0) = \varphi(x,\beta) = \varphi_B(x)$ for $x \in B$. From the expression above, after taking the zero-temperature limit, it is straightforward to compute:

$$\operatorname{Tr}\rho_{A}^{n} = \int_{\mathrm{BC}} \left[\prod_{i=1}^{n} \mathcal{D}\varphi\right] \prod_{i=1} \langle \varphi_{i,A} | \rho_{\beta,A} | \varphi_{i+1,A} \rangle = \frac{Z_{n}}{(Z_{1})^{n}}, \qquad (1.19)$$

where $\varphi_{n+1} \equiv \varphi_1$ and the presence of the factor $(Z_1)^n$ ensures that $\text{Tr}\rho_A = 1$. In the above expression we introduced the partition function

$$Z_n = \int \prod_i \left[\mathcal{D}\varphi_i \right]_{\mathcal{R}_n} e^{-S_{E,\mathcal{R}_n}[\{\varphi_i\}]},\tag{1.20}$$

over the *n*-sheeted Riemann surface \mathcal{R}_n . This surface is the *replica manifold*, and its structure is dictated by the boundary conditions (BC) imposed on the path integral, which we specify below. \mathcal{R}_n consists in *n* sheets which are flat everywhere except at the branch points $(x_1, 0)$ and $(x_2, 0)$. The branch cuts extending between these points (that is, on the interval *A*) cyclically connect the sheets, as depicted in Figure 1.1.

The replica action $S_{E,\mathcal{R}_n}[\{\varphi_i\}]$ in equation (1.20) is given by:

$$S_{E,\mathcal{R}_n}[\{\varphi_i\}] = \int_{\mathcal{R}_n} \mathrm{d}\tau \mathrm{d}x \,\mathcal{L}^{(n)}[\{\varphi_i\}](x,\tau), \quad \mathcal{L}^{(n)}[\{\varphi_i\}] = \sum_{i=1}^n \mathcal{L}[\varphi_i]. \tag{1.21}$$

Thus, the total energy of the theory is the sum of the single-replica energies, which allows interpreting the sheets as independent copies of the theory. The fields φ_i , i = 1, ..., n, each of which is defined on a different copy, are not interacting, being connected only through the boundary conditions:

$$\mathcal{C}(x_1, x_2): \begin{cases} \varphi_i(x, 0^+) = \varphi_{i+1}(x, 0^-), & \forall x \in [x_1, x_2] \\ \varphi_i(x, 0^+) = \varphi_i(x, 0^-), & \forall x \notin [x_1, x_2] \end{cases}, \quad i = 1, \dots, n, \quad n+1 \equiv 1$$
(1.22)

By explicitly employing these conditions, together with the fact that the Lagrangian density $\mathcal{L}^{(n)}[\{\varphi_i\}]$ is insensitive to the structure of the manifold, we can rewrite Z_n as a partition function on \mathbb{R}^2 :

$$Z_n = \int_{\mathcal{C}(x_1, x_2)} \prod_i \left[\mathcal{D}\varphi_i \right]_{\mathbb{R}^2} e^{-S_{E, \mathcal{R}^2}[\{\varphi_i\}]}.$$
 (1.23)

The equality between the two expressions (1.20) and (1.23) reflects the presence of a symmetry in the replica version of the theory. In fact, the replica partition function is invariant under the full group S_n of replica permutations. As evident from the boundary conditions (1.22), among the permutations a special role is played by the subgroup \mathbb{Z}_n of cyclic permutations. The branch-point twist field \mathcal{T} and its Hermitian conjugate $\tilde{\mathcal{T}}$ are the symmetry fields associated to the generator σ of cyclic permutations of replicas and to its inverse σ^{-1} , respectively:

$$\mathcal{T} = \mathcal{T}_{\sigma}, \quad \sigma: \quad i \mapsto i+1 \mod n, \quad \tilde{\mathcal{T}} = \mathcal{T}_{\sigma^{-1}}, \quad \sigma^{-1}: \quad i \mapsto i-1 \mod n.$$
(1.24)

The fields \mathcal{T} and $\tilde{\mathcal{T}}$ are *defined* by the property of implementing the boundary conditions (1.22). Namely, if $\mathcal{O}_i(x,\tau)$ is a local observable defined on the *i*th sheet, its replica correlation function in the theory described by the Lagrangian density \mathcal{L} is:

$$\langle \mathcal{O}_i(x,\tau)\dots\rangle_{\mathcal{L};\mathcal{R}_n} = \frac{\langle \mathcal{T}(x_1,0)\tilde{\mathcal{T}}(x_2,0)\mathcal{O}_i(x,\tau)\dots\rangle_{\mathcal{L}^{(n)};\mathbb{R}^2}}{\langle \mathcal{T}(x_1,0)\tilde{\mathcal{T}}(x_2,0)\dots\rangle_{\mathcal{L}^{(n)};\mathbb{R}^2}}.$$
(1.25)

Pictorially, one can think of \mathcal{T} , $\tilde{\mathcal{T}}$ as generating semi-infinite branch cuts at their insertion points, along which the permutations (1.24) are realised. These fields, being associated to a global symmetry of the theory, are local with respect to the replica action. However, they do not have vanishing equal-time commutators with the fields φ_i . Indeed, it follows from (1.25):

$$\varphi_i(x)\mathcal{T}(y) = \begin{cases} \mathcal{T}(y)\varphi_{i+1}(x) & y > x \\ \mathcal{T}(y)\varphi_i(x) & y < x \end{cases} \quad \text{and} \quad \varphi_i(x)\tilde{\mathcal{T}}(y) = \begin{cases} \tilde{\mathcal{T}}(y)\varphi_{i-1}(x) & y > x \\ \tilde{\mathcal{T}}(y)\varphi_i(x) & y < x \end{cases} \quad (1.26)$$

Semi-local fields with the properties outlined above are not unique, but they can be fixed by imposing that \mathcal{T} , $\tilde{\mathcal{T}}$ correspond to the lightest, spinless primary fields of the underlying CFT (of

central charge c), with scaling dimension[12, 20, 58, 59]:

$$d_{\mathcal{T}} = 2\Delta_{\mathcal{T}} = 2\Delta_{\tilde{\mathcal{T}}}, \quad \Delta_{\mathcal{T}} = \frac{c}{24}\left(n - \frac{1}{n}\right).$$
 (1.27)

Thus, by employing the CFT normalisation of these fields and equation (1.25) one can finally express:

$$\operatorname{Tr}\rho_{A}^{n} = \frac{Z_{n}}{(Z_{1})^{n}} = \zeta_{n} \epsilon^{4\Delta \tau} \langle \mathcal{T}(x_{1}, 0) \tilde{\mathcal{T}}(x_{2}, 0) \rangle_{\mathcal{L}^{(n)}; \mathbb{R}^{2}}, \qquad (1.28)$$

where the correlator is on the replicated ground state of the theory³. In the expression above, ϵ is a short-distance cutoff, which can be identified for instance with a lattice spacing $\epsilon \sim a$, while ζ_n is a non-universal normalisation factor such that $\zeta_1 = 1$, consistently with the fact that for n = 1 \mathcal{T} and $\tilde{\mathcal{T}}$ reduce to the identity operator, and $\frac{d\zeta_n}{dn}\Big|_{n=1} = 0$. The *n*th Rényi entropy directly follows from the previous equation, while the entanglement entropy is obtained as:

$$S_A = -\lim_{n \to 1} \frac{\partial}{\partial n} \left[\epsilon^{4\Delta \tau} \langle \mathcal{T}(x_1, 0) \tilde{\mathcal{T}}(x_2, 0) \rangle \right].$$
(1.29)

We mention that analytically continuing the replica index n to real values usually requires some care. We shall come back to this at the end of Chapter 4.

A replica picture for the computation of logarithmic negativity in QFT was devised in [65, 66]. In these works, the authors were able to obtain the $R\acute{e}nyi$ negativities:

$$\mathcal{E}_n[\rho_{A\cup B}^{T_B}] = \log \operatorname{Tr}\left(\rho_{A\cup B}^{T_B}\right)^n, \quad \text{for} \quad n \in \mathbb{N}, \quad n > 1,$$
(1.30)

for a subsystem $A \cup B$ consisting of two disjoint intervals $A = [x_1, x_2], B = [x_3, x_4]$, by means of a four-point function of BPTFs:

$$\mathcal{E}_{n}[\rho_{A\cup B}^{T_{B}}] = \log\left[\xi_{n}\epsilon^{8\Delta_{\mathcal{T}}}\langle \mathcal{T}(x_{1})\tilde{\mathcal{T}}(x_{2})\tilde{\mathcal{T}}(x_{3})\mathcal{T}(x_{4})\rangle\right],\tag{1.31}$$

where the correlator is again taken on the replicated ground state of the theory. The replica manifold associated to the computation of the above correlation function is more complicated than the one depicted in Figure 1.1, as there are in this case two branch cuts in each copy which cyclically connect the sheets (see for instance [65] for a pictorial representation). We remark that the logarithmic negativity (1.16) is obtained from the expression (1.30) by analytically continuing n to the reals and taking the limit $n \to 1$ when n is even. Indeed, the nth Rényi entropy has a different expression in terms of the singular values $|\lambda_i|$ of the partial transpose $\rho_{A\cup B}^{T_B}$ according to the parity of n, as can be seen from direct inspection: if n_e is an even positive integer and $n_o > 1$ is an odd positive integer,

³However, as we will discuss in Chapter 2, this relation is valid even in zero-density excited state

then

$$\mathcal{E}_{n_e}[\rho_{A\cup B}^{T_B}] = \log\left[\sum_{\lambda_i>0} |\lambda_i|^{n_e} + \sum_{\lambda_i<0} |\lambda_i|^{n_e}\right], \quad \mathcal{E}_{n_o}[\rho_{A\cup B}^{T_B}] = \log\left[\sum_{\lambda_i>0} |\lambda_i|^{n_o} - \sum_{\lambda_i<0} |\lambda_i|^{n_o}\right], \quad (1.32)$$

and the trace norm in (1.16) is correctly reproduced only by taking the limit $\lim_{n_e \to 1} \mathcal{E}_{n_e}[\rho_{A \cup B}^{T_B}]$.

We now come back to the bipartite entanglement entropy and briefly discuss the main findings for the ground state of a 1+1D QFT. In a critical theory, for an interval A of length ℓ , S_A is proportional to the logarithm of ℓ [11–14]:

$$S_A \sim \frac{c}{3} \log \frac{\ell}{\epsilon}, \quad \text{for} \quad \epsilon \ll \ell \ll \xi,$$
 (1.33)

where c is the central charge of the CFT, ϵ is a short-distance cutoff and ξ is the correlation length of the system. The above expression violates the area law (1.13), which predicts that for one-dimensional systems the bipartite entropy is independent of the size⁴. The situation is more complex in D > 1, where some systems characterised by a divergent correlation length satisfy the area law [67], whereas the violation persists in other cases, see for instance [68].

On the other hand, one-dimensional off-critical systems satisfy the area law. These systems are described by gapped quantum spin chains or, in the continuum limit limit, by massive QFTs. In this case, the entropy S_A of an interval saturates to a constant value in the limit of large ℓ :

$$S_A \sim \frac{c}{3} \log \frac{\xi}{\epsilon}, \quad \text{for} \quad \epsilon \ll \xi \ll \ell,$$
 (1.34)

where the correlation length is typically given by the inverse of a characteristic mass scale in the theory, $\xi = m^{-1}$. The off-critical behaviour of the bipartite entropy was derived for 1 + 1D massive quantum field theories in [12, 20], and observed in several gapped spin chains, see for instance [69–73].

We conclude this section by mentioning how the subleading corrections to the entanglement saturation (1.34) can be obtained in (relativistic) massive 1 + 1D QFTs. This is done by a spectral expansion of the branch-point twist field two-point function in (1.29). In one-dimensional quantum field theories, a basis for the Hilbert space of the theory is provided by the asymptotic multi-particle states, labelled by the rapidities θ_j of the particles in the theory together with their quantum numbers μ_j :

$$|\theta_1, \dots, \theta_n\rangle_{\mu_1, \dots, \mu_n},\tag{1.35}$$

where the rapidity $\theta \in \mathbb{R}$ of an on-shell particle of mass m parametrises its energy and momentum as $E(\theta) = m \cosh \theta, P(\theta) = m \sinh \theta$. In a replica version of a QFT, the quantum numbers $\mu_j = (i_j, a_j)$

⁴However, consistently with the area law, when A is a disjoint union of intervals (see e.g. [12, 18]), S_A is proportional to the number of boundary points, which is precisely the value of $|\partial A|$ in D = 1

account for internal degrees of freedom a_j as well as the replica indices $i_j \in \{1, \ldots, n\}$. Moreover, the rapidities in the above state are usually ordered such that $\theta_1 > \cdots > \theta_n$ if the state is incoming, i.e. far in the past before any interaction, and $\theta_1 < \cdots < \theta_n$ if the state is far in the future, after any interaction. Using the expansion in asymptotic states, one can write [20]:

$${}^{n}\langle 0|\mathcal{T}(x_{1})\tilde{\mathcal{T}}(x_{2})|0\rangle^{n} = \sum_{k=1}^{\infty}\sum_{\mu_{1},\dots,\mu_{k}}\int \frac{\mathrm{d}\theta_{1}\dots\mathrm{d}\theta_{k}}{k!(2\pi)^{k}} e^{-\ell\sum_{j=1}^{k}m_{a_{j}}\cosh\theta_{j}}|{}^{n}\langle 0|\mathcal{T}(0)|\theta_{1},\dots,\theta_{k}\rangle_{\mu_{1},\dots,\mu_{k}}|^{2}.$$
(1.36)

In the above equation, $\ell = |x_2 - x_1|, |0\rangle^n$ is the replica ground state and the quantity:

$$F^{\mathcal{T}|\mu_1,\dots,\mu_K}(\theta_1,\dots,\theta_k) :=^n \langle 0|\mathcal{T}(0)|\theta_1,\dots,\theta_k\rangle_{\mu_1,\dots,\mu_k},$$
(1.37)

is the k-particle form factor of the branch-point twist fields \mathcal{T} . In general, form factors are matrix elements of local or semi-local operators of a field theory, and they are used to characterise its operator content. Form factors can be defined in every 1 + 1D QFT, as long as the asymptotic states (1.35) are provided together with a description of the interaction in terms of scattering amplitudes between those states. However, it is only in *integrable* quantum field theories (IQFT) that form factors can be exactly obtained in a systematic way⁵. The reason behind the prominent role of integrability is that the presence of infinitely many conserved quantities severely constrains the dynamics of a model: in particular, once the scattering amplitudes are known, the form factors of a given operator can be obtained by imposing that some natural requirements on their structure are satisfied: this is the so-called form factor bootstrap program [77–79]. The form factor bootstrap program has been the object of intense research in the past forty years, leading to the classification of the operator content in several integrable QFTs, featuring diagonal and non-diagonal scattering, and even the presence of unstable bound states: see for instance [62, 80, 81] for the Ising model, [82] for the Federbush model, [83] for the sinh-Gordon model, [84] for the scaling Lee-Yang model, [85] for the sine-Gordon model, [86, 87] for the homogeneous sine-Gordon model, and [88] for the Bullough-Dodd model. In the case of operators which are semi-local with respect to the fundamental fields of the theory, the form-factor bootstrap equations need to be slightly modified. The first treatment of form factors of semi-local operators can be found in [62], for the order and disorder fields of the Ising IQFT, and in [89, 90] for the U(1) vertex operator of the sine-Gordon model. The form factor equations for the branch-point twist field were first derived in [20], and employed to obtain a large- ℓ expansion of the two-point function (1.36) in the Ising and sinh-Gordon model. The result of [20] was then generalised in [21] to the case of massive QFT, not necessarily integrable: for an interval A of length ℓ , the ground state

 $^{^{5}}$ In this thesis, we will only be concerned with form factors of free IQFT, so we refer to [74–76] for good reviews on one-dimensional IQFTs.

bipartite entropy is given by

$$S_A = -\frac{c}{3}\log(m_1\epsilon) + U - \frac{1}{8}\sum_{\alpha=1}^N K_0(2\ell m_\alpha) + \mathcal{O}(e^{-3\ell m_1}), \qquad (1.38)$$

where U is a constant, m_{α} are the masses of the N species of particles in the theory $(m_1 = \xi^{-1})$ being the lightest), and K_0 is a modified Bessel function [91]. In general, the inclusion of up to N-particle scattering processes in the spectral expansion of the BPTF two-point function yields a correction to the saturation law (1.34) which at leading order is $\mathcal{O}(e^{-Nm_1\ell})$.

In this thesis, we will perform expansions of the form (1.36) for the free massive Dirac fermion and complex boson (Chapter 2) and for the Ising field theory (Chapters (4) and (5)), both at finite and infinite size. We will provide the necessary details as needed throughout the work.

1.3 The role of internal symmetries

Investigating the relationship between the internal symmetries of a system and its entanglement properties has emerged as one of the most productive research directions in the field of many-body physics over the past five years. The first work on this subject was [92], in which the notion of spin-resolved entanglement was introduced to better characterise the spectrum of entanglement of the critical XXZ spin- $\frac{1}{2}$ chain. A more general notion of symmetry-resolved entanglement entropy (SREE) was then put forward in [93] and [19]. In [93], the SREE was employed to explain the mechanism behind many-body localisation in disordered systems whereas, crucially, [19] contains a field-theoretic characterisation of this quantity for U(1) and discrete symmetries within a replica picture. The equipartition of the symmetry-resolved entropy, i.e. the fact that different symmetry sectors yield the same leading-order contribution to the total entanglement, was proved in [94]. After these first works, the SREE for a generic non-abelian Lie group in a 1 + 1D CFT was obtained in [95], where the author focused on the ground state of Wess-Zumino-Witten models. Among the several other papers on this topic, we mention the generalisations to other entanglement measures, namely the charge-imbalance resolution of negativity [96] and symmetry-resolved quantum distances [97]. Following [19], we present the notion of symmetry-resolved entanglement entropy for a U(1) symmetry.

The key idea is that if a system possesses an internal symmetry, then the reduced density matrix has a block-diagonal structure. Let us consider a pure state ρ in a bipartite geometry, $\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_B$, and an internal U(1) symmetry of the system generated by the charge \hat{Q} :

$$\hat{Q} = \hat{Q}_A \oplus \hat{Q}_B := \hat{Q}_A \otimes \mathbb{1}_B + \mathbb{1}_A \otimes \hat{Q}_B.$$
(1.39)

Then the Hilbert space is decomposed into irreducible representations of the symmetry, $\mathcal{H} = \bigoplus_q \mathcal{H}_q$,

 $q \in \mathbb{Z}$. We further assume that the state ρ is symmetric, and so is the reduced density matrix⁶ ρ_A , so that

$$[\rho, \hat{Q}] = 0, \quad [\rho_A, \hat{Q}_A]. \tag{1.40}$$

The symmetry allows us to write ρ_A in a block-diagonal form:

$$\rho_A = \oplus_q p(q)\rho_A(q), \quad p(q) := \operatorname{Tr}\left(\Pi_q \rho_A\right), \quad \rho_A(q) = \frac{\Pi_q \rho_A}{\operatorname{Tr}\left(\Pi_q \rho_A\right)}, \tag{1.41}$$

where p(q) is the probability of measuring a charge eigenvalue q in the state ρ_A and we introduced the projector $\Pi_q : \mathcal{H}_A \to \mathcal{H}_{A,q}$ onto the charge eigenspace of \mathcal{H}_A labeled by q. In the case of U(1), Π_q can be written as

$$\Pi_q = \int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha \, e^{2\pi i \alpha (\hat{Q}_A - q)}, \quad q \in \mathbb{Z}.$$
(1.42)

It is straightforward to check from the above expression that Π_q is idempotent, Hermitian and it commutes with ρ_A .

We then define the charged replica partition function:

$$\mathcal{Z}_n(q) = \operatorname{Tr}\left(\Pi_q \rho_A^n\right), \quad n \in \mathbb{N}, \quad n > 1, \tag{1.43}$$

together with the *n*th symmetry-resolved Rényi entropy (SRRE):

$$S_n(q) = \frac{1}{1-n} \log \frac{\mathcal{Z}_n(q)}{[\mathcal{Z}_1(q)]^n} = \frac{1}{1-n} \log \operatorname{Tr} \left[\rho_A(q)\right]^n,$$
(1.44)

and the symmetry-resolved entanglement entropy (SREE):

$$S_A(q) = \lim_{n \to 1} S_n(q) = -\text{Tr} \left[\rho_A(q) \log \rho_A(q) \right].$$
(1.45)

Although the SREE, and particularly the charged partition function, has an interesting interpretation in terms of a modified replica manifold, which we describe below, in many situations one is ultimately interested in the total bipartite entropy S_A . This is related to the symmetry-resolved entropy as follows:

$$S_A = -\sum_{q} p(q) \operatorname{Tr} \left[\prod_{q} \rho_A \log \rho_A \right]$$
$$= \underbrace{-\sum_{q} p(q) \log p(q)}_{S_A^f} + \underbrace{\sum_{q} p(q) S_A(q)}_{S_A^c}.$$
(1.46)

⁶If the group acts unitarily on both \mathcal{H}_A and \mathcal{H}_B then it is easy to show that the symmetry of ρ implies that of ρ_A [95].

In the expression above, in which the second line follows from the definition (1.41) after some straightforward algebraic manipulation, we highlighted two contributions to the total entropy: the configurational entropy S_A^c and the fluctuation entropy S_A^f (also referred to as number entropy [22]). The quantity S_A^f measures the entropy related to fluctuations of the total charge in the subsystem A, which arise because of particle motion through the boundary ∂A . On the other hand, S_A^c is the averaged symmetry-resolved entanglement entropy of the two subsystems, where each contribution at a fixed charge q is weighted by the corresponding probability p(q). Although both quantities can be experimentally probed [22], it is only S_A^c that can be used as a resource in quantum information. Namely, S_A^c provides an upper bound to the operationally accessible entanglement, i.e. the entanglement that can be transferred to a quantum register by means of LOCC [25–27].

The charged partition function (1.43) is usually computed by means of its charged moments. For a fixed value of n, the charged moment $Z_n(\alpha)$ is the Fourier transform of $\mathcal{Z}_n(q)$ (or its Fourier series in the case of a discrete group). For a U(1), this is defined as

$$Z_n(\alpha) := \operatorname{Tr}\left(\rho_A^n e^{2\pi i \alpha}\right),\tag{1.47}$$

so that, using expression (1.42) one obtains:

$$\mathcal{Z}_n(q) = \int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha Z_n(\alpha) e^{-2\pi i \alpha q}.$$
 (1.48)

Expression (1.47) appeared in [19], although similar quantities were previously employed to describe Rényi entropies charged by a magnetic potential in holographic settings [98–100], theories with symmetry-protected topological phases [101], 2D CFTs and free-fermionic field theories in higher dimensions [102]. Physically, the computation of (1.47) amounts to evaluating the partition function over a replica manifold with the insertion of an Aharonov-Bohm flux $2\pi\alpha$ along the branch cut [103]. This is represented in Figure 1.2. Thus, a charged particle that encircled the branch point x_1 for ntimes, crossing all the replicas and returning to the initial position, would acquire a phase $e^{2\pi i\alpha}$. By fractionalising the total flux, so that the flux between consecutive sheets is $\frac{2\pi\alpha}{n}$, the boundary conditions (1.22) imposed on the (now charged) fields φ_i are modified to:

$$\mathcal{C}_{\alpha}(x_{1}, x_{2}): \begin{cases} \varphi_{i}(x, 0^{+}) = e^{\frac{2\pi i \alpha}{n}} \varphi_{i+1}(x, 0^{-}), & \forall x \in [x_{1}, x_{2}] \\ \varphi_{i}(x, 0^{+}) = \varphi_{i}(x, 0^{-}), & \forall x \notin [x_{1}, x_{2}] \end{cases}, \quad i = 1, \dots, n, \quad (1.49)$$

for a bipartition $A = [x_1, x_2], B = \overline{A}$. These boundary conditions are implemented in the path integral by *charged* or *composite* branch-point twist fields (CBPTF). If \mathcal{T} is the the branch-point twist field defined in equations (1.24), (1.25), and if \mathcal{V}_{α} is the operator that generates the U(1) flux, then the



Figure 1.2 Replica manifold \mathcal{R}_n for n = 3 with the insertion of a flux $2\pi\alpha$ along the branch cut in the region $A = [x_1, x_2]$. Picture taken from [19] and re-adapted.

CBPTF \mathcal{T}^{α} is obtained by "fusing" \mathcal{T} and \mathcal{V}_{α} . In particular, in a conformal field theory \mathcal{T}^{α} is defined as the lightest operator⁷ appearing in the operator product expansion (OPE) of those two fields, and the conjugate field $\tilde{\mathcal{T}}^{\alpha}$ is analogously defined starting from $\tilde{\mathcal{T}}$ and $\mathcal{V}_{-\alpha}$. The off-critical deformation of the CFT fields \mathcal{T}^{α} , $\tilde{\mathcal{T}}^{\alpha}$ define the composite branch-point twist fields in massive 1 + 1D QFTs with internal symmetries, and in the case of integrable theories they can be fully characterised by means of their form factors. We provide a more detailed characterisation of the composite fields associated to continuous and discrete (abelian) internal symmetries in Chapters 2 and 4 respectively. We remark that since the charge operator \hat{Q}_A commutes with the reduced density matrix ρ_A , different choices of the flux fractionalisation will lead to the same expectation values of physical observables. However, a homogeneous fractionalisation of the flux among copies keeps the replica symmetry manifest and it is essential to diagonalise the action of twist fields in free theories. In the following Chapters, we will specify every time which choice of flux fractionalisation we are adopting.

In the bipartite geometry considered above, the expression of the charged moment $Z_n(\alpha)$ in the ground state of a 1 + 1D QFT in terms of CBPTFs is the natural generalisation of equation (1.28):

$$Z_n(\alpha) = \zeta_{n,\alpha} \epsilon^{4\Delta_{\mathcal{T}^\alpha}} \langle \mathcal{T}^\alpha(x_1, 0) \tilde{\mathcal{T}}^\alpha(x_2, 0) \rangle_{\mathcal{L}^{(n)}:\mathbb{R}^2},$$
(1.50)

where $\zeta_{n,\alpha}$ is a normalisation constant and $\Delta_{\mathcal{T}^{\alpha}}$ is the conformal dimension of the CBPTFs (or of their critical counterpart). By means of charged moments, the bipartite SREE was computed in

⁷That is, the operator in the OPE with the lowest scaling dimension Δ above the identity operator (for which $\Delta = 0$).

the ground state of 1 + 1D CFTs [19, 94, 95], in the lowest-lying CFT excited states [104], in the ground state of massive 1 + 1D IQFTs with continuous [105, 106] and discrete [107, 108] symmetries. In Chapter 2, we provide a natural completion to this picture by computing charged moments and (excess of) SREEs of excited states in massive QFT, obtaining results which extend to non-integrable and higher-dimensional theories as well.

Before proceeding, we briefly comment on the notion of symmetry decomposition of logarithmic negativity, introduced in [96]. In a tripartite space $A \cup B \cup C$, the partial transpose $\rho_{A \cup B}^{T_B}$ admits a decomposition according to the irreducible representations of some internal charge of the theory. In this case, if $\hat{Q} = \hat{Q}_A \oplus \hat{Q}_B \oplus \hat{Q}_C$ is the charge operator on the total Hilbert space, then the operator that generates the symmetry of $\rho_{A \cup B}^{T_B}$ is the *charge imbalance*:

$$\hat{Q}_A - \hat{Q}_B^T := \hat{Q}_A \otimes \mathbb{1}_B - \mathbb{1}_A \otimes \hat{Q}_B^T.$$
(1.51)

In the replica theory, the charged moments of the partial transpose are naturally interpreted as partition functions on the Riemann surface, with the insertion of an Aharonov-Bohm flux only on region A. We defer a more detail discussion and a literature review on the subject to Chapter 3, where we obtain the symmetry-resolved logarithmic negativity and Rényi negativities of excited states in a QFT.

The idea of using entanglement measures to probe the *breaking* of an internal symmetry appeared about five years after the first works on symmetry-resolved entanglement in many-body systems. The notion of entanglement asymmetry was defined in [29] as a quantifier of the amount of symmetry breaking in a non-symmetric state ρ_A . A similar notion had already appeared in [109], where it was used as a measure of inseparability of a global state ρ with a conserved charge. In [29], the authors quenched an initial U(1) symmetry-breaking state to the XX Hamiltonian, which preserves the symmetry, and observed a dynamical restoration of the latter. Surprisingly, the symmetry was restored more quickly when it was initially more broken, a phenomenon that was dubbed quantum Mpemba effect. The reference is to the "classical" Mpemba effect, i.e. the counterintuitive phenomen for which, in a certain range of initial temperature and pressure, hot water freezes faster than cooler water [110]. A further investigation of the U(1) entanglement asymmetry in ground states of spin chains, together with a quasiparticle interpretation of the dynamical restoration of symmetry (or of its absence) was the object of [30, 31, 111]. The breaking and restoration of U(1) symmetry in random qubit states was also used to model the loss of information in black-hole radiation [34]. The case of the discrete group \mathbb{Z}_n was first addressed in [112], where the specific example of \mathbb{Z}_2 symmetry-breaking in the XY ground state was discussed. In [32], the authors computed the U(1) entanglement asymmetry in the ground state of a CFT by relating it to the presence of topological defects in the replica surface.

Finally, the symmetry breaking pattern $SU(2) \rightarrow U(1)$ in the XXZ critical chain was first investigated in [33].

To define the U(1) entanglement asymmetry, following [29, 30], we once again consider a pure state ρ of a bipartite Hilbert space $\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_B$, together with a global symmetry of the system generated by $\hat{Q} = \hat{Q}_A \oplus \hat{Q}_B$. We further assume that the RDM $\rho_A = \text{Tr}_{\mathcal{H}_B}\rho$ is not symmetric:

$$[\rho_A, \hat{Q}_A] \neq 0. \tag{1.52}$$

The symmetrised state $\rho_{A,Q}$ is built by retaining only the part of ρ_A which is block-diagonal in the eigenbasis of \hat{Q}_A , that is:

$$\rho_{A,Q} = \sum_{q} \Pi_q \rho_A \Pi_q, \tag{1.53}$$

where Π_q , for $q \in \mathbb{Z}$, are the projectors onto the U(1) charge eigenspaces. The state $\rho_{A,Q}$ satisfies $[\rho_{A,Q}, \hat{Q}_A] = 0$ by construction and it is immediate to check that $\rho_A = \rho_{A,Q}$ if and only if the state is symmetric. The entanglement asymmetry ΔS_A is defined as the difference of the von Neumann entropies of the states $\rho_{A,Q}$ and ρ_A :

$$\Delta S_A := S[\rho_{A,Q}] - S[\rho_A] = -\operatorname{Tr}\left(\rho_{A,Q}\log\rho_{A,Q}\right) + \operatorname{Tr}\left(\rho_A\log\rho_A\right), \qquad (1.54)$$

and it can be conveniently obtained by taking the limit $n \to 1$ of the *Rényi asymmetry*

$$\Delta S_A^{(n)} := \frac{1}{1-n} \left[\log \operatorname{Tr} \left(\rho_{A,Q}^n \right) - \log \operatorname{Tr} \left(\rho_A^n \right) \right].$$
(1.55)

Moreover, it is possible to express ΔS_A as the relative entropy between the states ρ_A and $\rho_{A,Q}$ [113]:

$$\Delta S_A = \operatorname{Tr}\left[\rho_A \left(\log \rho_A - \log \rho_{A,Q}\right)\right] =: S(\rho_A || \rho_{A,Q}).$$
(1.56)

The above equality follows from the definition (1.54) noting that

$$\operatorname{Tr}\left(\rho_{A,Q}\log\rho_{A,Q}\right) = \sum_{q} \operatorname{Tr}\left(\rho_{A}\Pi_{q}\log\rho_{A,Q}\Pi_{q}\right)$$
$$= \sum_{q} \operatorname{Tr}\left(\rho_{A}\log\rho_{A,Q}\Pi_{q}\right) = \operatorname{Tr}\left(\rho_{A}\log\rho_{A,Q}\right), \qquad (1.57)$$

where we used $[\Pi_q, \log \rho_{A,Q}] = 0$, together with the idempotence of the projector and completeness of the expansion in charge eigenstates. The fact that ΔS_A can be written as a relative entropy implies the following important properties:

• $\Delta S_A \ge 0$ for every state ρ_A ,

• $\Delta S_A = 0$ if and only if $\rho_A = \rho_{A,Q}$.

Like the SREE, the entanglement asymmetry can be computed by means of charged moments. Using the explicit expression (1.42), we can write

$$\rho_{A,Q} = \int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha \, e^{-2\pi i \alpha \hat{Q}_A} \rho_A e^{2\pi i \alpha \hat{Q}_A},\tag{1.58}$$

from which it follows that

$$\operatorname{Tr}\left(\rho_{A,Q}^{n}\right) = \int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha_{1} \dots \mathrm{d}\alpha_{n} Z_{n}(\boldsymbol{\alpha}), \quad \boldsymbol{\alpha} = (\alpha_{1}, \dots, \alpha_{n}), \quad (1.59)$$

where:

$$Z_n(\boldsymbol{\alpha}) = \operatorname{Tr}\left(\rho_A e^{2\pi i (\alpha_1 - \alpha_2)\hat{Q}_A} \rho_A \dots \rho_A e^{2\pi i (\alpha_n - \alpha_1)\hat{Q}_A}\right).$$
(1.60)

We stress that, because ρ_A and \hat{Q}_A do not commute, the order of the operators in the above expression is relevant.

In a replica picture, $Z_n(\alpha)$ is the partition function on a charged Riemann manifold similar to that depicted in Figure 1.2, with the important difference that the total Aharonov-Bohm flux, i.e. the sum of all the fluxes inserted between consecutive replicas, is zero. In Chapter 5, we propose a way to compute entanglement asymmetry in terms of generalisations of branch-point twist fields and we show that twist fields associated to a vanishing total flux are indeed unitarily equivalent to the standard (uncharged) BPTFs.

1.4 Structure of the thesis

This thesis is organised as follows. In Chapter 2, which is based on the works [1, 2], we compute the ratio of U(1) charged moments between zero-density excited states and the ground state in free 1+1D massive fermionic and bosonic QFTs by means of a form factor expansion of the twist field two-point function. The ratio of charged moments is then used to obtain the excess of SREE between the excited states and the ground state. This quantity displays the same universal features already observed for the total excess of entropy in [114, 115] and a very simple dependence on the U(1) charge. We then show how the QFT results can be derived within a much simpler framework, in which the excitations are represented by qubit states. Moreover, we prove that the validity of our formulae extends beyond the domain of free two-dimensional QFTs: this is done by explicitly computing the SREE of one-and two-magnon states of a spin chain and by developing the formalism of twist operators for generic algebras of observables in *D*-dimensional QFTs. In particular, the algebraic twist operator approach allows us to bypass the computation of form factors and establish the validity of the free-theory results

for local interacting theories, not necessarily integrable. We conclude the chapter by numerically checking our results on two one-dimensional discrete models with U(1) symmetry: a chain of spinless fermions and a complex harmonic chain.

We complete our investigation of symmetry resolution of entanglement for theories with a global U(1) symmetry in Chapter 3, based on [3], where we focus on symmetry-resolved negativity. Employing the qubit picture and the algebraic twist operator formalism introduced in Chapter 2, we obtain universal results for the difference of symmetry-resolved Rényi and logarithmic negativity between zero-density excited states and the ground state. Our results extend those found in [116] for free massive 1 + 1D QFTs with no internal symmetry, and at leading order in the large-volume limit they also apply to interacting and higher-dimensional theories. We provide distinct treatments of bosonic and fermionic excitations, as the correct definition of entanglement negativity for fermions requires a slightly different construction of the partial transpose of the RDM. We test our results numerically on a 1D chain of spinless fermions.

The first two chapters form part I of the thesis. Part II starts with Chapter 4, based on the work [4]. In this chapter we turn our attention to the massive Ising QFT in 1 + 1D, which features a global \mathbb{Z}_2 symmetry, and obtain an exact formula for the cumulant expansion of $\log \langle \mathcal{T}_{\mu}(x_1, 0) \tilde{\mathcal{T}}_{\mu}(x_2, 0) \rangle$ in the paramagnetic ground state, where \mathcal{T}_{μ} is the composite twist field obtained by fusing the BPTF \mathcal{T} and the disorder field μ . The cumulant expansion for the correlator of standard BPTFs is known for free 1 + 1D theories [63, 117], but in the case of composite twist fields there are convergence issues which we show how to solve. We conclude the chapter by performing the analytic continuation of the cumulant expansion to real values of the replica index.

In Chapter 5, based on [5], we propose a field-theoretic framework for the computation of entanglement asymmetry in the ordered phase of 1 + 1D, which display symmetry-breaking vacua. Our approach, valid for any discrete global symmetry G, is based on the algebraic construction of generalised twist operators and is applied to the Ising QFT in the ferromagnetic phase. We characterise the generalised twist operators by means of form factor bootstrap and obtain a formula for the \mathbb{Z}_2 entanglement asymmetry in the Ising QFT which is valid up to two-particle contributions. Moreover, we show as a byproduct that the total bipartite entropy in the ferromagnetic phase differs from the one in the dual point of the paramagnetic phase. We close the chapter by presenting a conjecture on the general form of the entanglement asymmetry in the case of partial symmetry breaking $G \to H$.

We conclude with a final summary of what we accomplished in this thesis and outline what we believe are the most promising research directions opened up by this work.

PART I : SYMMETRY-RESOLVED ENTANGLEMENT OF LOCALISED EXCITATIONS IN MASSIVE COMPLEX FREE THEORIES
CHAPTER

TWO

SYMMETRY-RESOLVED ENTROPY OF EXCITED STATES

The excess entanglement resulting from exciting a finite number of quasiparticles above the ground state of a free 1+1D QFT has been investigated quite extensively in the literature. It has been found that it takes a very simple form, depending only on the number of excitations and their statistics. There is now mounting evidence that such formulae also apply to interacting and even higher-dimensional quantum theories. In this chapter, based on [1, 2], we extend the known results by studying the symmetry-resolved entanglement entropy of such zero-density excited states in 1+1D QFTs that possess an internal symmetry. The ratio of charged moments between the excited and ground states, from which the symmetry-resolved entanglement entropy can be obtained, takes a very simple and universal form, which in addition to the number and statistics of the excitations, now depends also on the symmetry charge. Using form factor techniques, we obtain both the ratio of moments and the symmetry-resolved entanglement entropies in complex free theories which possess U(1) symmetry. The same formulae are found for simple multi-qubit states. We then generalise our results in two directions: by showing that they apply also to some excited states of quantum spin chains (one- and two-magnon states) and by developing a higher-dimensional generalisation of the branch-point twist field picture, leading to results in (interacting) higher-dimensional models. Finally, we provide numerical evidence for our formulae by computing functions of the charged moments in two free lattice theories: a 1D Fermi gas and a complex harmonic chain.

2.1 Introduction and summary of results

Symmetry-resolved entanglement and zero-density states. Starting from the basic ideas that we presented in Section 1.3, SREEs have been computed and discussed for many classes of models,

ranging from 1+1D CFTs [19, 92, 94, 95, 97, 104, 118–122], to free [105, 123, 124] and interacting integrable QFT [106–108], holographic settings [98, 102, 125–127], lattice models [26, 27, 92, 94, 118, 119, 123, 128–133], out of equilibrium systems [23, 118, 131, 134–137] and for systems with more exotic types of dynamics [138–143]. In this chapter, we compute the SREEs of zero-density excited states in 1+1D gapped systems in the scaling limit, focusing of excitations of free QFTs, the massive Dirac fermion and the complex boson. We then extend these results to generic zero-density excited states of interacting QFTs and spin chains.

The systems we consider are at zero temperature and finite volume with periodic boundary conditions. The two complementary regions A and \overline{A} have lengths ℓ and $L - \ell$. Eventually, we will consider the infinite-volume limit in which the relative size of the two subsystems is fixed:

$$\ell, L \to \infty$$
, with $r := \frac{\ell}{L}$ fixed, $r \in [0, 1]$. (2.1)

In this limit, a zero-density excited state is a state describing a finite number of excitations above the QFT vacuum whose momenta are fixed as the volume increases. In the series of papers [114-116,144], the excess of entanglement entropy and logarithmic negativity¹ of the zero-density states with respect to their ground state values was computed and found to take a remarkably universal and simple form: it depends only on r, on the number of excitations and on their statistics. The results were originally derived by employing the branch-point twist field approach in free fermion and free boson theories. However, it was argued in [114] (and illustrated with the example of one- and two-magnon states) that the formulae should hold much more generally, for interacting and even higher-dimensional theories², as long as a notion of localised excitations exists. These claims have been substantiated through additional recent results. In particular, a series of works by Rajabpour and collaborators [145–150] has expanded previous work in various directions: by obtaining finite-volume corrections, new formulae for systems where quasiparticles are not localised, and finally by establishing that the formulae indeed hold for generic magnon states, thus also in interacting theories. Similar formulae have also been found for interacting higher-dimensional theories in [151] and even in the presence of an external potential, arising from a semiclassical limit [152]. Indeed, the formulae found in [114] were not entirely unexpected as they can be derived for semiclassical systems [153], however their wide range of applicability, well beyond the semiclassical regime, as well as their derivation in the context of QFT were new.

In CFT, the excess of entanglement entropy of some excited states with respect to the ground state was studied in [154, 155]. The states considered therein are the lowest-lying excited states of the

¹In some of these works, more complex partitions were also considered, e.g. multiple disconnected regions.

²In [144] the same formulae were shown to hold for free bosons in any dimension if r is replaced by the ratio of generalised volumes.

theory in the same geometric setting described above. Namely, for states of the form:

$$|\Upsilon\rangle := \lim_{z,\bar{z} \to 0} \Upsilon(z,\bar{z})|0\rangle, \tag{2.2}$$

where $\Upsilon(z, \bar{z})$ is a primary field³ of conformal weights $(\Delta, \bar{\Delta})$, the excess of entropy with respect to the vacuum state in the limit of small subsystem size $r \ll 1$ is given by:

$$S_{\Upsilon} - S_0 \simeq \frac{2\pi^2}{3} (\Delta + \bar{\Delta}) r^2 + \mathcal{O}(r^{2d_{\Psi}}), \qquad (2.3)$$

with $d_{\Psi} = \Delta_{\Psi} + \bar{\Delta}_{\Psi}$ the scaling dimension of the field defined by the operator product expansion $\Upsilon \times \Upsilon^{\dagger} = 1 + \Psi + \dots$ This expression was then checked for the lowest-lying excited states of the complex compactified boson (c = 2), which has ($\Delta, \bar{\Delta}$) = (1,0). The computation of the excess of U(1) symmetry-resolved entropy for these low-lying states was performed in [104]. On the other hand, for zero-density excited states of the massive free boson and massive free fermion, the excess entropy is [114, 115]:

$$S_{\text{exc}} - S_0 = -r \log r - (1 - r) \log(1 - r).$$
(2.4)

By comparing these two expressions, it is clear how the entanglement content of zero-density excitations in massive QFT and that of low-lying states of critical systems are captured by different physical pictures.

In this chapter we combine these two topics, symmetry-resolved entropies and excited states, to investigate how the entropy of excited states may be seen as a sum over symmetry sectors in the presence of an internal symmetry. We will initially focus our attention on the complex free fermion and boson theories. The total excited state entanglement of (real) free fermions and bosons was obtained in [115, 116], while the SREE in the ground state of the 1 + 1D massive Dirac fermion and complex boson was studied in [105]. This chapter can be seen as a generalisation of these works. Our motivation to study these types of states from this viewpoint is, first and foremost, to provide exact formulae for the SREEs of at least a class of excited states in 1+1D QFT. This is interesting because the SREE of the ground state of 1+1D QFTs has generally a very complicated form, only accessible perturbatively in some parameter, as discussed in many papers [19, 92, 94, 95, 97, 104–108, 118–124]. On the other hand, it is possible to show that the SREEs of zero-density excited states are as complex as that of the ground state, i.e. knowing the symmetry-resolved entanglement content of the ground state is sufficient to reconstruct that of the excited states. Moreover, for special cases when the ground state is trivial, the SREEs can be obtained exactly. Further motivation is provided by the fact that, unlike the total entropy, the SREEs are entanglement measures that allow us to distinguish between

³The complex coordinates (z, \bar{z}) are introduced as customary in the radial quantisation scheme [74], in which $|z| \rightarrow 0$ corresponds to the infinite past in Euclidean time.

charged and neutral excitations, although in the present chapter we only focus on charged particles. An example where both charged and neutral excitations are present is the sine-Gordon model in the interacting regime, which was studied in [106].

Main results and outline. The main results can be summarised as follows. Let $Z_n^{\Psi}(L, \ell, \alpha)$ be the charged moments of the *n*th symmetry-resolved Rényi entropy (SRRE) of a connected region of length ℓ , in a pure state $|\Psi\rangle_L^n$ of an *n*-replica theory in finite volume *L*. Then, the ratio of moments

$$M_n^{\Psi}(r;\alpha) := \lim_{L \to \infty} \frac{Z_n^{\Psi}(L, rL;\alpha)}{Z_n^0(L, rL;\alpha)},$$
(2.5)

between the state $|\Psi\rangle_L^n$ and the ground state $|0\rangle_L^n$, in the infinite-volume limit with r fixed, is given by a universal formula, which depends very simply on r and α . We call this quantity a charged ratio. There are two particularly useful cases from which more general formulae can be constructed. When $|\Psi\rangle_L^n = |1^{\epsilon}\rangle_L^n$ is a state of a single particle excitation with U(1) charge $\epsilon = \pm 1$ we have that

$$M_n^{1^{\epsilon}}(r;\alpha) = e^{2\pi i \epsilon \alpha} r^n + (1-r)^n,$$
(2.6)

whereas for a state of k identical excitations, that is excitations with the same momenta and charges ϵ , we have

$$M_n^{k^{\epsilon}}(r;\alpha) = \sum_{j=0}^{k} [f_j^k(r)]^n e^{2\pi i j \epsilon \alpha}, \quad f_j^k(r) := \binom{k}{j} r^j (1-r)^{k-j}.$$
 (2.7)

Formula (2.7) is the building block for all other results (formula (2.6) is the k = 1 case of (2.7)). A generic state comprising s groups of $k_i^{\epsilon_i}$ identical particles of charge ϵ_i will have

$$M_{n}^{k_{1}^{\epsilon_{1}}\dots k_{s}^{\epsilon_{s}}}(r;\alpha) = \prod_{i=1}^{s} M_{n}^{k_{i}^{\epsilon_{i}}}(r;\alpha).$$
(2.8)

Note that the possibility of having identical excitations is excluded for fermionic theories. For $\alpha = 0$ these formulae reduce to those found in [114, 115], later generalised to entanglement measures of multiple disconnected regions [116] and to higher dimensions for free bosons in [144]. These results in turn have been extended in a series of works [145–150] to deal with finite-volume corrections and non-localised excitations. More recently, some of the $\alpha = 0$ results were recovered as a semiclassical limit in the presence of an interaction potential [152]. This semiclassical picture had already been invoked much earlier, see for instance [153]. However, it is worth emphasizing that our formulae are not merely semiclassical limits but hold for genuine quantum theories. The quantum nature of the model is encoded in the symmetry-resolved entanglement entropy of the ground state (and its associated moments), which is indeed highly non-trivial. In other words, it is only the ratios (2.5) that are simple, not the individual charged moments.

In order to obtain the SREE it is necessary to isolate the charged moments of the excited state. This can be easily done analytically, and it allows us to write the SREEs in terms of the ground state entropies. For instance, for the state $|\Psi\rangle_L^n = |1^{\epsilon}\rangle_L^n$ considered above, the SREEs (Rényi and von Neumann) are given by

$$S_n^{1^{\epsilon}}(r;q) = \frac{1}{1-n} \log \frac{\mathcal{Z}_n^{1^{\epsilon}}(r,q)}{(\mathcal{Z}_1^{1^{\epsilon}}(r,q))^n} = \frac{1}{1-n} \log \frac{\mathcal{Z}_n^0(q-\epsilon)r^n + \mathcal{Z}_n^0(q)(1-r)^n}{(\mathcal{Z}_1^0(q-\epsilon)r + \mathcal{Z}_1^0(q)(1-r))^n},$$
(2.9)

and

$$S_{1}^{1^{\epsilon}}(r;q) = -\frac{\mathcal{Z}_{1}^{0}(q-\epsilon)r\log r + \mathcal{Z}_{1}^{0}(q)(1-r)\log(1-r) + [r\partial_{n}\mathcal{Z}_{n}^{0}(q-\epsilon) + (1-r)\partial_{n}\mathcal{Z}_{n}^{0}(q)]_{n=1}}{r\mathcal{Z}_{1}^{0}(q-\epsilon) + (1-r)\mathcal{Z}_{1}^{0}(q)} + \log(\mathcal{Z}_{1}^{0}(q-\epsilon)r + \mathcal{Z}_{1}^{0}(q)(1-r)),$$
(2.10)

in terms of the ground state partition functions and their derivatives, which can be related back to the ground state entropies. Here $\mathcal{Z}_n^{\Psi}(r,q)$ are the symmetry-resolved partition functions in the state $|\Psi\rangle$ and $\mathcal{Z}_n^0(q)$ are those of the ground state, which are independent of r in the scaling limit considered here. The formulae for the SREEs of other states are rather cumbersome and we discuss more general cases in Section 2.2.

This chapter is organised as follows: In Section 2.2 we employ the U(1) composite branch-point twist fields to express the charged ratio of moments between zero-density states and the ground state. We show how the problem of computing this quantity can be simplified by diagonalising the fields in the replica space and present explicit form factor computations for zero-density excited states of the complex free boson and fermion. We then discuss how the symmetry-resolved entropies can be obtained from the ratios of charged moments. Some details of the calculations are left to Appendix 2.A and Appendix 2.B, while Appendix 2.C contains a form factor expansion of the two-point function of composite twist fields in the ground state. In Section 2.3 we show how the results obtained via the twist field approach can be derived in a simpler framework of qubit states. Multi-qubit states have coefficients that represent the probabilities of finding a certain number of excitations in a certain space region. In this case the symmetry-resolved entanglement of the excited states can be obtained explicitly. In Section 2.4 we generalise the results to interacting and higher dimensional theories. In the first part of the section we show that the formulae derived for free theories also hold in some magnonic states of spin chains in the presence of interactions. In the second part of the section we introduce the notion of twist operator, which extends that of twist field to theories (either free or interacting) in higher space-time dimensions. Twist operators will be used extensively in other chapters of this thesis. In Section 2.5 we present numerical results for two (free) discrete systems: a 1D lattice Fermi gas and a 1D complex harmonic chain. We find that in both cases the formulae presented above are reproduced with great precision, even if the scaling limit of the Fermi gas is a massless free fermion.

For the complex free boson, we show how the SREE can be computed by employing a wave-functional method, some details of which are left in Appendix 2.D. Conclusions and outlook are in Section 2.6.

2.2 Charged moments and SREE of free theories: form factor approach

In this section we express the charged moments of excited states of U(1) free theories as two-point functions of composite branch-point twist fields (CBPTF) in finite volume, and we show how their computation can be much simplified by diagonalising the action of the twist fields in the replica space. Once we have expressed the twist fields and the excited states in the diagonal basis, we compute the two-point functions via a form factor expansion, obtaining the ratio of charged moments for the complex boson and the Dirac fermion. Finally, from the ratio of charged moment and the knowledge of the charged moments in the ground state, we reconstruct the SREE of the excited states by performing simple Fourier transforms.

2.2.1 Composite branch-point twist field factorisation

Let us consider, as above, a system of total length L with periodic boundary conditions, and a connected subsystem A extending from x = 0 to $x = \ell$. The moments of the RDM of a pure state in the replica theory, that is, a tensor product $|\Psi\rangle_L^n := |\Psi\rangle_L \otimes \cdots \otimes |\Psi\rangle_L$ of n identical states, can be obtained from the equal-time correlator of two branch-point twist fields, thanks to the identification⁴:

$$\operatorname{Tr}\rho_{A}^{n} = \varepsilon^{4\Delta_{\mathcal{T}}}{}_{L}^{n} \langle \Psi | \mathcal{T}(0) \tilde{\mathcal{T}}(\ell) | \Psi \rangle_{L}^{n}, \qquad (2.11)$$

where ε is a short-distance cut-off and Δ_{τ} is the conformal dimension of the branch-point twist field, defined in (1.27). Because of the definitions (1.9), (1.11), differences of Rényi or von Neumann entropies are independent of ε . They depend only on the ratio

$$R_n^{\Psi}(\ell,L) := \frac{{}_L^n \langle \Psi | \mathcal{T}(0) \hat{\mathcal{T}}(\ell) | \Psi \rangle_L^n}{{}_L^n \langle 0 | \mathcal{T}(0) \tilde{\mathcal{T}}(\ell) | 0 \rangle_L^n},$$
(2.12)

where $|0\rangle_L^n := \otimes_{j=1}^n |0\rangle_{j,L}^n$ is the finite-volume replica ground state. In the limit (2.1) this becomes a function $R_n^{\Psi}(r)$ of $|\Psi\rangle^n$ and r only and, for the states considered in the previous section, it is given by the same equations (2.6), (2.7) and (2.8) if we set $\alpha = 0$. In theories possessing a U(1) symmetry the formulation is a very natural generalisation of the previous case, in which the branch-point twist field

⁴Note that here we do not include a non-universal normalisation factor to account for the norm of the state. The reason is that we are computing the correlator in finite volume and, as we will show, working in finite volume provides a natural way to regularise the divergences arising from the norm of the state when the large-volume limit is taken.

 \mathcal{T} is replaced by the CBPTF \mathcal{T}^{α} and the charged moment is given by the two-point function

$$\operatorname{Tr}_{A}(\rho_{A}^{n}e^{2\pi i\alpha\hat{Q}_{A}}) = Z_{n}^{\Psi}(L,\ell;\alpha) = \varepsilon^{4\Delta_{\mathcal{T}^{\alpha}}}{}_{L}^{n}\langle\Psi|\mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell)|\Psi\rangle_{L}^{n}.$$
(2.13)

The field \mathcal{T}^{α} and its conjugate $\tilde{\mathcal{T}}^{\alpha}$ can be understood as massive counterparts of the corresponding CFT field. This means that the field \mathcal{T}^{α} in the massive theory obtained as a certain relevant perturbation of a CFT flows to the field:

$$\mathcal{T}^{\alpha}(y) :=: \mathcal{T}\mathcal{V}_{\alpha}: (y) = n^{2\Delta_{\alpha}-1} \lim_{x \to y} |x-y|^{2\Delta_{\alpha}(1-\frac{1}{n})} \sum_{j=1}^{n} \mathcal{T}(y)\mathcal{V}^{j}_{\alpha}(x),$$
(2.14)

when the conformal limit is performed. Above, \mathcal{V}_{α} is the vertex operator associated with the U(1)symmetry of the theory, corresponding to the insertion of an Aharonov-Bohm phase $e^{2\pi i \alpha}$ on the Riemann surface, Δ_{α} is the conformal dimension of this field, and \mathcal{V}_{α}^{j} is a copy of this field living in copy j of the replica theory. That is, \mathcal{T}^{α} is the lightest field appearing in the operator product expansion (OPE) of \mathcal{T} and \mathcal{V}_{α} . As shown in [105, 124, 156–158], the conformal dimension of the fields $\mathcal{T}^{\alpha}, \tilde{\mathcal{T}}^{\alpha}$ is

$$\Delta_{\mathcal{T}^{\alpha}} := \Delta_{\mathcal{T}} + \frac{\Delta_{\alpha}}{n}, \qquad (2.15)$$

where, for $\alpha \in [-\frac{1}{2}, \frac{1}{2}]$:

$$\Delta_{\alpha} = \frac{\alpha^2}{2} \qquad \text{for Dirac fermions,} \qquad (2.16)$$

$$\Delta_{\alpha} = \frac{|\alpha| - \alpha^2}{2} \quad \text{for complex bosons.}$$
 (2.17)

The main result of this section is the finding that, similar to the quantity (2.12), also the ratio of the moments (2.13) between an excited state and the ground state:

$$M_n^{\Psi}(r;\alpha) = \lim_{L \to \infty} \frac{\frac{n}{L} \langle \Psi | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(rL) | \Psi \rangle_L^n}{\frac{n}{L} \langle 0 | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(rL) | 0 \rangle_L^n},$$
(2.18)

takes a simple, universal form which is a function of the ratio r, the charge α and the number and statistics of excitations in the state $|\Psi\rangle^n$.

Let us now discuss how these ratios may be computed in practice, employing a form factor approach. A key technical problem that was solved in [115] is the question of how to evaluate finite-volume matrix elements of the branch-point twist field. The same question arises for the composite field. Although a finite-volume form factor program for generic local fields exists [159, 160] this cannot be directly employed for twist fields (its extension to this case is still an open problem). In the absence of such a program, an alternative approach can be used for complex free theories, where the internal U(1)symmetry on each replica can be exploited to diagonalise the action of the CBPTF [20, 161, 162]. In fact, this diagonalisation procedure can also be employed in infinite volume to compute the form factors of \mathcal{T}^{α} , as done in [105]. The idea is the following. Let us denote by $\Phi = (\Phi_1, \ldots, \Phi_n)^T$ the multiplet of free complex fields the replica manifold, i.e. Φ_j is the local field in the *j*th replica (for now, we use the same symbol Φ to denote a multiplet of free complex bosons or free Dirac fermions). Since there is a U(1) symmetry in each copy, and the free replica action is quadratic, the theory is enhanced with a SU(n) symmetry. Thus, there is a global SU(n) transformation $\Phi \mapsto \tilde{\Phi} = (\tilde{\Phi}_1, \ldots, \tilde{\Phi}_n)^T$ such that the action of \mathcal{T}^{α} and $\tilde{\mathcal{T}}^{\alpha}$ is diagonal in the new basis of local fields. Assuming a homogeneous fractionalisation of the phase $e^{2\pi i \alpha}$ among all replicas, the fields that diagonalise the action of \mathcal{T}^{α} and $\tilde{\mathcal{T}}^{\alpha}$ are:

$$\tilde{\Phi}_p = \frac{1}{\sqrt{n}} \sum_{j=1}^n e^{-\frac{2\pi i j p}{n}} \Phi_j, \quad \begin{cases} p = 1, \dots, n & \text{for complex bosons} \\ p = -\frac{n-1}{2}, \dots, \frac{n-1}{2} & \text{for Dirac fermions} \end{cases},$$
(2.19)

and the eigenvalues of the transformation are given by

$$\lambda_p = e^{\frac{2\pi i(p+\alpha)}{n}}.$$
(2.20)

In free theories, correlation functions of the composite branch-point twist fields factorise in free theories, and therefore we can write $\mathcal{T}^{\alpha}(x,t)$, $\tilde{\mathcal{T}}^{\alpha}(x,t)$ as products of fields acting non-trivially only on one copy. Namely, the factorisation is:

$$\mathcal{T}^{\alpha}(x,t) = \prod_{p=1}^{n} \mathcal{T}_{p+\alpha}(x,t), \qquad \tilde{\mathcal{T}}^{\alpha}(x,t) = \prod_{p=1}^{n} \mathcal{T}_{-p-\alpha}(x,t), \qquad (2.21)$$

for complex free bosons and

$$\mathcal{T}^{\alpha}(x,t) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \mathcal{T}_{p+\alpha}(x,t), \qquad \tilde{\mathcal{T}}^{\alpha}(x,t) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \mathcal{T}_{-p-\alpha}(x,t),$$
(2.22)

for complex free fermion. The factors $\mathcal{T}_{p+\alpha}$ are all U(1) fields with charge $p + \alpha$, resulting from the fusion of two U(1) fields of charges p and α : namely, the fields \mathcal{T}_p employed in [115], in terms of which one can decompose the (standard) branch-point twist field (1.26) in a free theory, and the U(1) vertex operators \mathcal{V}_{α} . This factorisation was already employed in [105, 124], albeit with a different normalisation of the parameter α . There, it was shown that the conformal dimension of these fields is $\Delta_{\frac{\alpha+p}{n}}$ for the free fermion and $\Delta_{\frac{|\alpha|+p}{n}}$ for the free boson, with Δ_{α} given by (2.17), as indeed by summing these quantities over the allowed values of p one obtains (2.15) in the two cases. The fields $\mathcal{T}_{\pm(p+\alpha)}$ satisfy the usual equal-time exchange relations for U(1) fields, which involve what is termed a factor of local commutativity $\gamma_{p+\alpha}^{\pm} = \exp(\pm 2\pi i(p+\alpha)/n)$, that is, the phase that a field $\tilde{\Phi}_p(x)$ of

charge +1 accrues when taking a trip around the U(1) field:

$$\mathcal{T}_{\pm(p+\alpha)}(x,t)\tilde{\Phi}_q(y,t) = \begin{cases} (\gamma_{p+\alpha}^{\pm})^{\delta_{p,q}}\tilde{\Phi}_q(y,t)\mathcal{T}_{\pm(p+\alpha)}(x,t) & y > x\\ \tilde{\Phi}_q(y,t)\mathcal{T}_{\pm(p+\alpha)}(x,t) & y < x \end{cases}.$$
(2.23)

As we will review shortly, the factor of local commutativity is the key ingredient in determining the form factors of these fields.

The computation presented in [115] for the total entanglement entropy may be easily extended to the case of the ratio $M_n^{\Psi}(r; \alpha)$ in excited states. First, a word is due regarding the excited state $|\Psi\rangle_L^n$. In general, any state in the replica QFT can be characterised in terms of the rapidities and quantum numbers of the excitations above the ground state. Considering a free complex theory, we may define creation operators $(a_j^{\epsilon})^{\dagger}(\theta)$ where $\epsilon = \pm 1$ is the U(1) charge of the particle, θ is its rapidity and $j = 1, \ldots, n$ is the copy number. Unlike the works [114–116, 144], where complex theories were considered only in order to access results for real ones, here we are interested in obtaining results for complex models. The type of k-particle excited state that we are interested in consists of n identical copies of a standard k-particle state:

$$|\Psi\rangle_L^n = \prod_{j=1}^n \left(\prod_{i=1}^k (a_j^{\epsilon_i})^{\dagger}(\theta_i)\right) |0\rangle_L^n.$$
(2.24)

We will start by considering the complex boson and then move to the complex fermion.

2.2.2 Complex free boson

In order to represent the state, it is convenient to move to the basis (2.19) in which the action of the CBPTF is factorised and diagonal. In this basis, the state can be expressed in terms of creation operators $\mathfrak{a}_{j}^{\dagger}(\theta)$ and $\mathfrak{b}_{j}^{\dagger}(\theta)$ associated with bosons of charge +1 and -1 respectively. They are related to the creation operators in the standard basis as [115]

$$\mathfrak{a}_p^{\dagger}(\theta) = \frac{1}{\sqrt{n}} \sum_{j=1}^n e^{\frac{2\pi i j p}{n}} (a_j^{\dagger})^{\dagger}(\theta) \quad \text{and} \quad \mathfrak{b}_p^{\dagger}(\theta) = \frac{1}{\sqrt{n}} \sum_{j=1}^n e^{-\frac{2\pi i j p}{n}} (a_j^{-})^{\dagger}(\theta), \quad (2.25)$$

where p = 1, ..., n. That is, the sets of creation operators in the two basis are Fourier modes of each other. The annihilation operators $\mathfrak{a}_p(\theta)$, $\mathfrak{b}_p(\theta)$ are defined analogously and the only non-vanishing commutation relations are:

$$[\mathfrak{a}_p(\theta_1),\mathfrak{a}_p^{\dagger}(\theta_2)] = [\mathfrak{b}_p(\theta_1),\mathfrak{b}_p^{\dagger}(\theta_2)] = \delta(\theta_1 - \theta_2).$$
(2.26)

As an example, let us consider the case of one single excitation of charge ϵ , and rapidity θ , which

we write⁵ as $|1^{\epsilon}\rangle_{L}^{n}$. In the original basis, this would be the state $\prod_{j=1}^{n} (a_{j}^{\epsilon})^{\dagger}(\theta)|0\rangle_{L}^{n}$, that is a state where a single complex boson of rapidity θ and charge ϵ is present in each replica. In the diagonal basis, by inverting the relations (2.25), such a state takes the form

$$|1^{+}\rangle_{L}^{n} = \sum_{\{N^{+}\}} A_{n}(\{N^{+}\}) \prod_{p=1}^{n} [\mathfrak{a}_{p}^{\dagger}(\theta)]^{N_{p}^{+}} |0\rangle_{p,L}^{n}, \quad |1^{-}\rangle_{L}^{n} = \sum_{\{N^{-}\}} A_{n}(\{N^{-}\}) \prod_{p=1}^{n} [\mathfrak{b}_{p}^{\dagger}(\theta)]^{N_{p}^{-}} |0\rangle_{p,L}^{n}, \quad (2.27)$$

where the indices $\{N^{\pm}\} := \{N_1^{\pm}, \dots, N_n^{\pm}\}$ are boson occupation numbers in each sector and they are constrained by the condition that they must add up to n

$$\sum_{p=1}^{n} N_p^{\pm} = n.$$
 (2.28)

The coefficients $A(\{N^{\pm}\})$ can be obtained systematically from equations (2.25) and their inverses. Combining the factorisation of the CBPFT and equation (2.27), we can expand the two-point function in the excited state $|1^{\pm}\rangle_L^n$ as follows:

$$\sum_{L}^{n} \langle 1^{+} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{+} \rangle_{L}^{n} = \sum_{\{N^{+}\}} \sum_{\{M^{+}\}} A_{n}^{*}(\{N^{+}\}) A_{n}(\{M^{+}\})$$

$$\times \prod_{p=1}^{n} \sum_{p,L}^{n} \langle 0 | [\mathfrak{a}_{p}(\theta)]^{N_{p}^{+}} \mathcal{T}_{p+\alpha}(0) \mathcal{T}_{-p-\alpha}(\ell) [\mathfrak{a}_{p}^{\dagger}(\theta)]^{M_{p}^{+}} | 0 \rangle_{p,L}^{n} ,$$

$$\sum_{L}^{n} \langle 1^{-} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{-} \rangle_{L}^{n} = \sum_{\{N^{-}\}} \sum_{\{M^{-}\}} A_{n}^{*}(\{N^{-}\}) A_{n}(\{M^{-}\})$$

$$\times \prod_{p=1}^{n} \sum_{p,L}^{n} \langle 0 | [\mathfrak{b}_{p}(\theta)]^{N_{p}^{-}} \mathcal{T}_{p+\alpha}(0) \mathcal{T}_{-p-\alpha}(\ell) [\mathfrak{b}_{p}^{\dagger}(\theta)]^{M_{p}^{-}} | 0 \rangle_{p,L}^{n} .$$

$$(2.29)$$

This can be computed in the standard way by inserting a sum over a complete set of states between the two fields as detailed in Appendix 2.A. A particular subtlety of this kind of computation is that, because of finite volume, the momenta of the excitations are quantised and non-zero matrix elements correspond to particular quantisation conditions that take the monodromy of the fields into account. In particular we have:

$$P(\theta_i^{\pm}) = m \sinh \theta_i^{\pm} = 2\pi J_i^{\pm} \pm \frac{2\pi (p+\alpha)}{n}, \qquad J_i^{\pm} \in \mathbb{Z},$$
(2.31)

where θ_i^+ denotes rapidities of particles created by \mathfrak{a}_j^{\dagger} and θ_i^- denotes rapidities of particles created by \mathfrak{b}_j^{\dagger} . The reason is that these states, being inserted between the two composite twist fields, belong to a twisted sector of the Hilbert space (see [115] for more details), and thus the monodromy relation

⁵The fact that we do not label the state using the rapidity θ is justified *a posteriori*, as our results do not depend on the energy and momentum of the zero-density excitations.

is $\tilde{\Phi}_p(x+L) = e^{\pm \frac{2\pi i(p+\alpha)}{n}} \tilde{\Phi}_p(x)$, from which (2.31) follows. On the other hand, the rapidity θ of the external "untwisted" states is quantised through $P(\theta) = 2\pi I$ for $I \in \mathbb{Z}$, i.e. the usual Bethe-Yang quantisation condition for a free theory [163–165]. Note that the quantity $\frac{p+\alpha}{n}$ is never an integer for $\alpha \in [-\frac{1}{2}, \frac{1}{2}]$ and $p \neq n$ (p = n corresponds to the identity field). This guarantees that only non-diagonal form factors (that is matrix elements involving only distinct right and left states) will be involved in the computation of the leading large-volume contribution to (2.29).

Once a sum over a complete set of states is inserted in (2.29), the problem reduces to the computation of matrix elements of the U(1) fields $\mathcal{T}_{p+\alpha}$. Such matrix elements have been known for a long time but they were re-derived in [105, 115]. Because of the free nature of the theory, all matrix elements are given in terms of the two-particle form factor

$$F_n^{p+\alpha|+-}(\theta_{12}) := {}_p\langle 0|\mathcal{T}_{p+\alpha}(0)\mathfrak{a}_p^{\dagger}(\theta_1)\mathfrak{b}_p^{\dagger}(\theta_2)|0\rangle_p = -\tau_{p+\alpha}\sin\frac{\pi(p+\alpha)}{n}\frac{e^{\left(\frac{p+\alpha}{n}-\frac{1}{2}\right)\theta_{12}}}{\cosh\frac{\theta_{12}}{2}},\tag{2.32}$$

where $\tau_{p+\alpha}$ is the vacuum expectation value of $\mathcal{T}_{p+\alpha}$:

$$\tau_{p+\alpha} = {}_p \langle 0 | \mathcal{T}_{p+\alpha} | 0 \rangle_p, \tag{2.33}$$

and $\theta_{12} = \theta_1 - \theta_2$. Since the composite twist field preserves the total charge of the state, it follows that

$$F_n^{p+\alpha|++}(\theta_{12}) := {}_p\langle 0|\mathcal{T}_{p+\alpha}(0)\mathfrak{a}_p^{\dagger}(\theta_1)\mathfrak{a}_p^{\dagger}(\theta_2)|0\rangle_p = 0, \qquad (2.34)$$

$$F_n^{p+\alpha|--}(\theta_{12}) := {}_p\langle 0|\mathcal{T}_{p+\alpha}(0)\mathfrak{b}_p^{\dagger}(\theta_1)\mathfrak{b}_p^{\dagger}(\theta_2)|0\rangle_p = 0.$$

$$(2.35)$$

The two-point function (2.32) is the only solution of the Watson equations:

$$F_n^{p+\alpha|\pm\mp}(\theta) = F_n^{p+\alpha|\pm\pm}(-\theta), \quad F_n^{p+\alpha|\pm\mp}(\theta+2\pi i) = \gamma_{p+\alpha}^{\pm}F_n^{p+\alpha|\pm\mp}(\theta), \tag{2.36}$$

and of the kinematic residue equation:

$$\operatorname{Res}_{\theta=0} F_n^{p+\alpha|\pm\mp}(\theta+i\pi) = i(1-\gamma_{p+\alpha}^{\pm})\tau_{p+\alpha}.$$
(2.37)

Higher-particle form factors simply follow from Wick's Theorem, which for the free boson reads:

$$F_2^{p+\alpha,n}m(\theta_1,\ldots,\theta_m,\beta_1,\ldots,\beta_m) := {}_p\langle 0|\mathcal{T}_{p+\alpha}(0)\mathfrak{a}_p^{\dagger}(\theta_1)\ldots\mathfrak{a}_p^{\dagger}(\theta_m)\mathfrak{b}_p^{\dagger}(\beta_1)\ldots\mathfrak{b}_p^{\dagger}(\theta_m)|0\rangle_p$$
$$= \tau_{p+\alpha}\sum_{\sigma\in S_m} f_{p+\alpha}^n(\theta_{\sigma(1)}-\beta_1)\ldots f_{p+\alpha}^n(\theta_{\sigma(m)}-\beta_m), \qquad (2.38)$$

where S_m is the set of permutations of the first *m* positive integers. For free fermions, one has to

take into account also the parity of the permutations σ . In the formula above, we introduced the normalised two-particle form factor:

$$f_{p+\alpha}^{n}(\theta) := \frac{F_{n}^{p+\alpha|+-}(\theta)}{\tau_{p+\alpha}}.$$
(2.39)

The finite-volume matrix elements can then be obtained thanks to the results of [159, 160] (see also Appendix 2.A).

In summary, all results obtained in [115] follow through for the CBPTF with the replacement $p \rightarrow p + \alpha$ and the choice of an appropriate state. In particular, the ratio of charged moments for a state describing one excitation is given by

$$M_n^{1^{\pm}}(r;\alpha) = \sum_{\{N^{\pm}\}} |A_n(\{N^{\pm}\})|^2 \prod_{p=1}^n (N_p^{\pm}!) [g_{\pm(p+\alpha)}^n(r)]^{N_p^{\pm}} = e^{\pm 2\pi i \alpha} r^n + (1-r)^n,$$
(2.40)

which is, as anticipated, the formula (2.6) and where

$$g_p^n(r) := 1 - (1 - e^{\frac{2\pi i p}{n}})r.$$
(2.41)

For free bosons, this can be generalised to states containing k identical excitations to find (2.7). For states containing k different excitations (with different rapidities and any combination of charges ϵ_i) the result is

$$M_{n}^{1^{\epsilon_{1}}\dots1^{\epsilon_{k}}}(r;\alpha) = \prod_{s=1}^{k} \sum_{\{N^{\pm}\}} |C_{n}(\{N^{\pm}\})|^{2} \prod_{p=1}^{n} N_{p,s}^{+}! N_{p,s}^{-}! (g_{p+\alpha}^{n}(r))^{N_{p,s}^{+}} (g_{-p-\alpha}^{n}(r))^{N_{p,s}^{-}}$$
$$= \prod_{j=1}^{k} \left[e^{2\pi i \epsilon_{j} \alpha} r^{n} + (1-r)^{n} \right].$$
(2.42)

In these formulae, $C_n(\{N^{\pm}\})$ and $A_n(\{N^{\pm}\})$ are coefficients which are determined by the form of the state in the diagonal basis. Both results are special cases of (2.8).

2.2.3 Complex free fermion

For complex free fermions the computation is very similar, although states involving identical excitations are forbidden. The Fourier modes of the creation operators $(a_i^{\pm})^{\dagger}(\theta)$ are:

$$\mathfrak{a}_p^{\dagger}(\theta) = \frac{1}{\sqrt{n}} \sum_{j=1}^n e^{\frac{2\pi i j p}{n}} (a_j^{\dagger})^{\dagger}(\theta) \quad \text{and} \quad \mathfrak{b}_p^{\dagger}(\theta) = \frac{1}{\sqrt{n}} \sum_{j=1}^n e^{-\frac{2\pi i j p}{n}} (a_j^{-})^{\dagger}(\theta), \quad (2.43)$$

where now $p = -\frac{n-1}{2}, \dots, \frac{n-1}{2}$, and the only non-vanishing anticommutators are:

$$\{\mathfrak{a}_p(\theta_1),\mathfrak{a}_p^{\dagger}(\theta_2)\} = \{\mathfrak{b}_p(\theta_1),\mathfrak{b}_p^{\dagger}(\theta_2)\} = \delta(\theta_1 - \theta_2).$$
(2.44)

For a free fermion, the two-particle form factor of the factorised CBPTF is modified to [82, 89]

$$F_n^{p+\alpha|+-}(\theta_{12}) := {}_p\langle 0|\mathcal{T}_{p+\alpha}(0)\mathfrak{a}_p^{\dagger}(\theta_1)\mathfrak{b}_p^{\dagger}(\theta_2)|0\rangle_p = i\tau_{p+\alpha}\sin\frac{\pi(p+\alpha)}{n}\frac{e^{\left(\frac{p+\alpha}{n}\right)\theta_{12}}}{\cosh\frac{\theta_{12}}{2}}.$$
 (2.45)

The structure of a state consisting of a single particle excitation is as for the free boson, namely

$$|1^{+}\rangle_{L}^{n} = \prod_{j=1}^{n} (a_{j}^{+})^{\dagger}(\theta)|0\rangle_{L}^{n} = \prod_{j=1}^{n} \frac{1}{\sqrt{n}} \sum_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \omega^{jp} \mathfrak{a}_{p}^{\dagger}(\theta)|0\rangle_{L}^{n},$$
(2.46)

$$|1^{-}\rangle_{L}^{n} = \prod_{j=1}^{n} (a_{j}^{-})^{\dagger}(\theta)|0\rangle_{L}^{n} = \prod_{j=1}^{n} \frac{1}{\sqrt{n}} \sum_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \omega^{-jp} \mathfrak{b}_{p}^{\dagger}(\theta)|0\rangle_{L}^{n},$$
(2.47)

with $\omega = e^{-\frac{2\pi i}{n}}$. For instance, for n = 2 we have:

$$|1^{+}\rangle_{L}^{2} = \frac{1}{2}(i\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta) - i\mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta))(-\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta) - \mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta))|0\rangle_{L}^{2} = -i\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta)\mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta)|0\rangle_{L}^{2},$$
(2.48)

and

$$|1^{-}\rangle_{L}^{2} = \frac{1}{2}(-i\mathfrak{b}_{-\frac{1}{2}}^{\dagger}(\theta) + i\mathfrak{b}_{\frac{1}{2}}^{\dagger}(\theta))(-\mathfrak{b}_{-\frac{1}{2}}^{\dagger}(\theta) - \mathfrak{b}_{\frac{1}{2}}^{\dagger}(\theta))|0\rangle_{L}^{2} = i\mathfrak{b}_{-\frac{1}{2}}^{\dagger}(\theta)\mathfrak{b}_{\frac{1}{2}}^{\dagger}(\theta)|0\rangle_{L}^{2}.$$
 (2.49)

Similarly, for n = 3:

$$|1^{+}\rangle_{L}^{3} = i\mathfrak{a}_{-1}^{\dagger}(\theta)\mathfrak{a}_{0}^{\dagger}(\theta)\mathfrak{a}_{1}^{\dagger}(\theta)|0\rangle_{L}^{3} \quad , \quad |1^{-}\rangle_{L}^{3} = -i\mathfrak{b}_{-1}^{\dagger}(\theta)\mathfrak{b}_{0}^{\dagger}(\theta)\mathfrak{b}_{1}^{\dagger}(\theta)|0\rangle_{L}^{3}.$$
(2.50)

Due to the anticommutation relations, many contributions now cancel off and the states take extremely simple forms in the diagonal basis. It is easy to show that the general structure of the states (2.46) and (2.47) is:

$$|1^{+}\rangle_{L}^{n} = e^{i\kappa} \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \mathfrak{a}_{p}^{\dagger}(\theta)|0\rangle_{L}^{n}, \quad |1^{-}\rangle_{L}^{n} = e^{-i\kappa} \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \mathfrak{b}_{p}^{\dagger}(\theta)|0\rangle_{L}^{n}, \tag{2.51}$$

with κ a real parameter that can be computed for each specific state and will not play a role in our computation. Making use of the factorisation (2.22) we can expand the fermionic two-point function in terms of a sum over the form factors (2.45). The details are presented in Appendix 2.B. For a state

consisting of a single excitation the result is

$$M_n^{1^{\pm}}(r;\alpha) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{\pm p\pm\alpha}^n(r) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \left[1 - (1 - e^{\pm \frac{2\pi i(p+\alpha)}{n}})r \right].$$

Since the quantities $e^{\pm \frac{2\pi i p}{n}}$ are the *n*th roots of +1 for *n* odd, and the *n*th roots of -1 for *n* even, it follows that

$$\prod_{=-\frac{n-1}{2}}^{\frac{n-1}{2}} (x - e^{\pm \frac{2\pi i p}{n}} y) = x^n + (-y)^n,$$
(2.52)

which, after setting x = 1 - r, $y = -re^{\pm \frac{2\pi i \alpha}{n}}$ gives:

p

$$\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{\pm p\pm\alpha}^n(r) = e^{\pm 2\pi i\alpha} r^n + (1-r)^n, \qquad (2.53)$$

that is, the same result as (2.40) for free bosons, albeit resulting from a rather different product of g-functions. Similarly, all free boson formulae presented in the previous subsection are recovered for free fermions, as long as we consider only distinct excitations.

2.2.4 Symmetry-resolved entanglement entropies

Having obtained the ratios of charged moments we now proceed to computing the SREE of excited states. To this aim, we need to isolate the charged moments of the excited state and then compute their Fourier transform to obtain the charged partition functions (1.48). In other words, we need to multiply our results of the previous section by the ground state correlator in the infinite-volume limit considered here. Note that this ground state correlator will be different for different theories, even if formulae (2.6)-(2.8) are always satisfied.

For (local) 1+1D QFTs, such as complex free theories, the ground state correlator in our scaling limit reduces to its disconnected part, that is the square of the vacuum expectation value (VEV) of the field \mathcal{T}^{α} . This result follows simply from clustering of correlators in local QFT, but can also be demonstrated explicitly from the finite-volume expansion of the ground state two-point function. This expansion is presented in Appendix 2.C for complex free fermions⁶. In particular, looking at equation (2.218) we can see how, despite the complexity of the expansion, in infinite volume the only surviving term in the sum corresponds to the product of VEVs $|\tau_{p+\alpha}|^2$. The same statement holds for complex free bosons. As mentioned earlier, it is common to normalise the correlators by the inclusion of a UV

⁶A large-distance expansion of the two-point function of \mathcal{T}^{α} in free complex theories was performed in [105], although in that case the authors considered QFTs in infinite volume from the very beginning.

cut-off, so that the natural quantity to compute is

$$Z_n^{\Psi}(r;\alpha) = Z_n^0(\alpha) M_n^{\Psi}(r;\alpha) \qquad \text{with} \quad Z_n^0(\alpha) = \varepsilon^{4\Delta_\alpha} \langle \mathcal{T}^\alpha \rangle^2, \tag{2.54}$$

where $Z_n^{\Psi}(r; \alpha)$ are the charged moments of the excited state in our particular scaling limit, $Z_n^0(\alpha)$ are the moments of the ground state and $\langle \mathcal{T}^{\alpha} \rangle$ is the VEV of the composite twist field. From general dimensionality arguments, as can be found for instance in [166], the VEV has a very particular dependence on the mass scale and the conformal dimension of the CBPTF. In fact, we have

$$\langle \mathcal{T}^{\alpha} \rangle = v_n^{\alpha} m^{2\Delta_{\mathcal{T}^{\alpha}}},\tag{2.55}$$

where v_n^{α} is a function that depends on the model and can be determined by requiring CFT normalisation of the composite twist field (that is, that the CFT two-point function has numerical coefficient of 1) and $\Delta_{\mathcal{T}^{\alpha}}$ is given by (2.15). The Fourier transform of the ground state moments has been studied in detail for free QFTs in [105, 124], thus we will not revisit its computation here. Instead, we show that, assuming $Z_n^0(\alpha)$ to be known, it is possible to express the symmetry-resolved partition functions and entropies of excited states fully in terms of those of the ground state. The reason for this is that the functions $M_n^{\Psi}(r; \alpha)$ depend on α in an extremely simple manner, namely through factors of the form $e^{\pm 2\pi i j \alpha}$ only. Thus, in order to compute the SREE of an excited state, the only non-trivial integrals that we need to consider are of the form

$$\int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha \, Z_n^0(\alpha) e^{-2\pi i \alpha (q \pm j)} = \mathcal{Z}_n^0(q \pm j).$$
(2.56)

For instance, using (2.6), the simple example of a single excitation of charge ϵ gives the following relationship amongst partition functions

$$\mathcal{Z}_{n}^{1^{\epsilon}}(r;q) = \mathcal{Z}_{n}^{0}(q-\epsilon)r^{n} + \mathcal{Z}_{n}^{0}(q)(1-r)^{n}.$$
(2.57)

Therefore, the symmetry-resolved Rényi and von Neumann entropies of such a state are given by formulae (2.9) and (2.10), respectively. They can in turn be written in terms of the SREE and partition function of the ground state (i.e. eliminating derivative terms) by recalling that

$$\partial_n \mathcal{Z}_n^0(q) \big|_{n=1} = -\mathcal{Z}_1^0(q) [S_1^0(q) - \log Z_1^0(q)].$$
(2.58)

Similar relations are found for more complicated cases, such as (2.7), that is an excited state of k

identical excitations of charge ϵ . In this case we find instead

$$S_{n}^{k^{\epsilon}}(r;q) = \frac{1}{1-n} \log \frac{\sum_{j=0}^{k} \left[f_{j}^{k}(r)\right]^{n} \mathcal{Z}_{n}^{0}(q-\epsilon j)}{\left[\sum_{j=0}^{k} f_{j}^{k}(r) \mathcal{Z}_{1}^{0}(q-\epsilon j)\right]^{n}},$$
(2.59)

and the symmetry-resolved von Neumann entropy

$$S_{1}^{k^{\epsilon}}(r;q) = -\frac{\sum_{j=0}^{k} \left[\mathcal{Z}_{1}^{0}(q-\epsilon j) f_{j}^{k}(r) \log f_{j}^{k}(r) + f_{j}^{k}(r) \partial_{n} \mathcal{Z}_{n}^{0}(q-\epsilon j) \right]_{n=1}}{\sum_{j=0}^{k} f_{j}^{k}(r) \mathcal{Z}_{1}^{0}(q-\epsilon j)} + \log \sum_{j=0}^{k} f_{j}^{k}(r) \mathcal{Z}_{1}^{0}(q-\epsilon j).$$
(2.60)

Unlike for the charged moments, the entropies of other states are not expressed by particularly simple formulae. However, the kind of integrals involved are of the same type so that the computation can be performed in a similar manner for any excited state. As a last example, let us consider the ratio of charged moments for an excited state of two particles of opposite charges. We have that

$$\mathcal{Z}_{n}^{1^{+}1^{-}}(r;q) = \int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha \, Z_{n}^{0}(\alpha) (r^{n} + e^{2\pi i\alpha} (1-r)^{n}) (r^{n} + e^{-2\pi i\alpha} (1-r)^{n}) e^{-2\pi i\alpha q}$$

$$= \mathcal{Z}_{n}^{0}(q) (r^{2n} + (1-r)^{2n}) + (\mathcal{Z}_{n}^{0}(q-1) + \mathcal{Z}_{n}^{0}(q+1)) r^{n} (1-r)^{n}, \qquad (2.61)$$

so that the Rényi entropy is

$$S_n^{1^+1^-}(r;q) = \frac{1}{1-n} \log \frac{\mathcal{Z}_n^0(q)(r^{2n} + (1-r)^{2n}) + (\mathcal{Z}_n^0(q-1) + \mathcal{Z}_n^0(q+1))r^n(1-r)^n}{[\mathcal{Z}_1^0(q)(r^2 + (1-r)^2) + (\mathcal{Z}_1^0(q-1) + \mathcal{Z}_1^0(q+1))r(1-r)]^n},$$
(2.62)

from which the von Neumann entropy follows as above.

In conclusion, the SREE of the kind of excited states considered here can be expressed in terms of the SREE and partition function of the ground state. This statement holds for any system where formulae (2.6)-(2.8) apply and where the ground state contribution is well-defined. As we shall see below, this includes a wide range of models, well beyond free QFTs. As a final remark, we recall that a key property of the SREEs of the ground state both in QFT [19] and interacting quantum spin chains [94] is the property of equipartition at leading order. That is, within a certain range of parameters⁷ the SREEs of all charge sectors are charge independent. It is clear from the formulae above that this property also holds for the SREEs of excited states, as their charge dependence is solely encoded in the symmetry-resolved partition function and entropies of the ground state. Thus, if the entropy is equipartite in the ground state it will also be so in excited states.

⁷For massive QFT this range typically corresponds to the double limit of large subsystem size $\ell \gg 1$ and $|\log(m\epsilon)| \ll 1$ where *m* is a typical mass scale and ϵ a UV cut-off (see e.g. [106, 108]).

2.3 The qubit picture

Besides the QFT approach based on twist fields that we have presented so far, there are alternative ways in which the entanglement of excited states may be studied. In the works [114–116] several models and approaches were considered, including the study of the entanglement of certain qubit states. In the present context, such states are also useful as they provide a simpler way of obtaining our formulae for the ratios of charged moments, even if their associated SREEs will be different, in fact much simpler than those of QFT states.

2.3.1 Charged moments of multi-qubit states

The main idea behind the qubit picture is that of representing a localised excitation in $A \cup \overline{A}$ via a superposition of two two-qubit states:

$$|\Psi_{\rm qb}^{(1)}\rangle = \sqrt{r}|1\rangle_A \otimes |0\rangle_{\bar{A}} + \sqrt{1-r}|0\rangle_A \otimes |1\rangle_{\bar{A}}, \qquad (2.63)$$

where the qubit state 1(0) represents the presence (absence) of the excitation in the corresponding spacial region. The coefficients are chosen in such a way to reproduce a uniform probability of finding the excitation in $A \cup \overline{A}$, a choice dictated by the fact that the SREEs computed via the form factor approach are independent on the energy of the excitation. A state consisting of k indistinguishable excitations (same rapidity and same charge) is described by defining a Hilbert space $\mathcal{H}_A \otimes \mathcal{H}_{\overline{A}}$, with:

$$\mathcal{H}_A = \operatorname{span}\{|q\rangle_A, q = 0, \dots, k\} \simeq \mathcal{H}_{\bar{A}} \simeq \mathbb{C}^{k+1},$$
(2.64)

and a subspace

$$\mathcal{H}^{(k)} = \operatorname{span}\{|q\rangle_A \otimes |k-q\rangle_{\bar{A}}, q = 0, \dots, k\} \subset \mathcal{H}_A \otimes \mathcal{H}_{\bar{A}}.$$
(2.65)

Then:

$$|\Psi_{\rm qb}^{(k)}\rangle := \sum_{q=0}^{k} \sqrt{f_q^k(r)} |q\rangle_A \otimes |k-q\rangle_{\bar{A}} \in \mathcal{H}^{(k)}, \tag{2.66}$$

where the function $f_q^k(r)$ (see the definition in (2.7)) represents the probability of finding q out of k indistinguishable particles in the region A. The RDM can be computed in the orthonormal multi-qubit basis of $\mathcal{H}_A \otimes \mathcal{H}_{\bar{A}}$, yielding:

$$\rho_A^{(k)} = \operatorname{Tr}_{\bar{A}} |\Psi_{qb}^{(k)}\rangle \langle \Psi_{qb}^{(k)}|
= \operatorname{Tr}_{\bar{A}} \sum_{q,q'} \sqrt{f_q^k(r) f_{q'}^k(r)} |q\rangle_{AA} \langle q'| \otimes |k-q\rangle_{\bar{A}\bar{A}} \langle k-q'|
= \sum_{q=0}^k f_q^k(r) |q\rangle_{AA} \langle q|,$$
(2.67)

that is, the reduced density matrix is diagonal. Assuming that the charge operator associated with the internal symmetry is $\hat{Q} = \hat{Q}_A \oplus \hat{Q}_{\bar{A}}$ and that, without loss of generality, all the particles in the state above have charge $\epsilon = +1$, then $e^{2\pi i \alpha \hat{Q}_A} |q\rangle_A = e^{2\pi i \alpha q} |q\rangle_A$ and the charged moments can be easily computed as follows:

$$\operatorname{Tr}\left[\left(\rho_{A}^{(k)}\right)^{n}e^{2\pi i\alpha\hat{Q}_{A}}\right] = \sum_{q_{1},\dots,q_{n}=0,\dots,k} {}_{A}\langle q_{1}|\rho_{A}^{(k)}e^{\frac{2\pi i\alpha\hat{Q}_{A}}{n}}|q_{2}\rangle_{AA}\langle q_{2}|\rho_{A}^{(k)}e^{\frac{2\pi i\alpha\hat{Q}_{A}}{n}}|q_{3}\rangle_{A}\dots_{A}\langle q_{n}|\rho_{A}^{(k)}e^{\frac{2\pi i\alpha\hat{Q}_{A}}{n}}|q_{1}\rangle_{A} = \sum_{\{q_{i}\}}\prod_{i=1}^{n}f_{q_{i}}^{k}(r)e^{\frac{2\pi i\alpha q_{i}}{n}}\delta_{q_{i},q_{i+1}} = \sum_{q=0}^{k}[f_{q}^{k}(r)]^{n}e^{2\pi i\alpha q}.$$

$$(2.68)$$

Thus, the result (2.7) is reproduced. The more general situation one can consider is that of a state containing N sets of indistinguishable excitations, with the set j formed by k_j particles of the same rapidity and charge ϵ_j , for j = 1, ..., N. By keeping fixed the total number of particles $k = \sum_j k_j$, the Hilbert space of the multi-qubit states is now $\mathcal{H}_A \otimes \mathcal{H}_{\bar{A}}$, with:

$$\mathcal{H}_A = \operatorname{span}\{|q_1^{\epsilon_1}, \dots, q_N^{\epsilon_N}\rangle_A, q_j = 0, \dots, k_j, \epsilon_j = \pm 1, j = 1, \dots, N\} \simeq \mathcal{H}_{\bar{A}} \simeq \bigotimes_{j=1}^N \mathbb{C}^{k_j+1}.$$
 (2.69)

A multi-qubit state with k particles uniformly distributed over $A \cup \overline{A}$ is:

$$|\Psi_{qb}^{(k_1,\epsilon_1;\ldots;k_N,\epsilon_N)}\rangle := \sum_{q_1,\ldots,q_N} \sqrt{\prod_{j=1}^N f_{q_j}^{k_j}(r)} |q_1^{\epsilon_1},\ldots,q_N^{\epsilon_N}\rangle_A \otimes |\bar{q}_1^{\epsilon_1},\ldots,\bar{q}_N^{\epsilon_N}\rangle_{\bar{A}}, \quad \bar{q}_j := k_j - q_j.$$
(2.70)

The state is normalised:

$$\langle \Psi_{\rm qb}^{(k_1',\epsilon_1';...;k_N',\epsilon_N')} | \Psi_{\rm qb}^{(k_1,\epsilon_1;...;k_N,\epsilon_N)} \rangle = \prod_{j=1}^N \delta_{k_j,k_j'} \delta_{\epsilon_j,\epsilon_j'}, \qquad (2.71)$$

and factorises into a product of multi-qubit states (2.66):

$$|\Psi_{\rm qb}^{(k_1,\epsilon_1;\ldots;k_N,\epsilon_N)}\rangle = \bigotimes_{j=1}^N |\Psi_{\rm qb}^{(k_j,\epsilon_j)}\rangle = \bigotimes_{j=1}^N \left(\sum_{q_j=0}^{k_j} \sqrt{f_{q_j}^{k_j}(r)} |q_j^{\epsilon_j}\rangle_A \otimes |\bar{q}_j^{\epsilon_j}\rangle_{\bar{A}}\right).$$
(2.72)

Because of this factorisation, the charged moments are obtained in a straightforward way from those of the state (2.66). Indeed, the RDM matrix is:

$$\rho_A^{(k_1,\epsilon_1;\dots;k_N,\epsilon_N)} = \bigotimes_{j=1}^N \rho_A^{(k_j,\epsilon_j)} = \bigotimes_{j=1}^N \sum_{q_j=0}^{k_j} f_{q_j}^{k_j}(r) |q_j^{\epsilon_j}\rangle_{A\bar{A}} \langle q_j^{\epsilon_j}|, \qquad (2.73)$$

and, decomposing $\hat{Q}_A = \bigoplus_{j=1}^N \hat{Q}_A^{(j)}$, we obtain the expression:

$$\operatorname{Tr}\left[\left(\rho_{A}^{(k_{1},\epsilon_{1};\ldots;k_{N},\epsilon_{N})}\right)^{n}e^{2\pi i\alpha\hat{Q}_{A}}\right] = \prod_{j=1}^{N}\operatorname{Tr}\left[\left(\rho_{A}^{(k_{j},\epsilon_{j})}\right)^{n}e^{2\pi i\alpha\hat{Q}_{A}^{(j)}}\right] = \prod_{j=1}^{N}\sum_{q_{j}=0}^{k_{j}}[f_{q_{j}}^{k_{j}}(r)]^{n}e^{2\pi i\alpha q_{j}\epsilon_{j}}, \quad (2.74)$$

which indeed reproduces the most general result (2.8) (with s = N).

2.3.2 SREE of multi-qubit states

We close this section by noting that for multi-qubit states the results obtained are directly the charged moments of the state rather than ratios of moments. This is so because the qubit ground state $|0\rangle_A \otimes |0\rangle_{\bar{A}}$ is trivial from the point of view of entanglement. This means that the formulae (2.6)-(2.8) are the quantities we need to Fourier-transform in order to obtain the SREEs. The simplicity of the formulae allows us to obtain the SREEs exactly, something that is typically beyond reach for QFT. By using

$$\int_{-\frac{1}{2}}^{\frac{1}{2}} \mathrm{d}\alpha \, e^{-2\pi i \alpha x} = \delta_{x,0}, \qquad \text{for} \quad x \in \mathbb{Z},$$
(2.75)

it immediately follows from (2.68) that

$$S_{n}^{k^{\epsilon}}(r;q) = \frac{1}{1-n} \log \frac{\sum_{j=0}^{k} \left[f_{j}^{k}(r) \right]^{n} \delta_{q,\epsilon j}}{\left[\sum_{j=0}^{k} f_{j}^{k}(r) \delta_{q,\epsilon j} \right]^{n}},$$
(2.76)

and in particular

$$S_n^{1^{\epsilon}}(r;q) = \frac{1}{1-n} \log \left[\frac{\delta_{q,\epsilon} r^n + \delta_{q,0} (1-r)^n}{(\delta_{q,\epsilon} r + \delta_{q,0} (1-r))^n} \right].$$
 (2.77)

The von Neumann entropies easily follow from these expressions. Due to the simplicity of the states, however, we see that all the entropies above are identically zero whenever any of the delta-functions is 1. This can be interpreted as the statement that the symmetry-resolution of the entropy does not give any additional information about these states. Another way to put this is to say that the only property that matters in establishing formulae (2.77)-(2.76) is whether particles are distinguishable or not and in both formulae particles are identical by construction, so that specifying the charge does not add any relevant information.

The situation is different if we consider states containing distinct excitations. For instance, for a state of k distinct excitations of the same charge ϵ the charged moments are given by

$$(r^{n} + e^{2\pi i\epsilon\alpha}(1-r)^{n})^{k} = \sum_{j=0}^{k} \binom{k}{j} (1-r)^{nj} e^{2\pi i\epsilon\alpha j} r^{n(k-j)},$$
(2.78)



Figure 2.1 Symmetry-resolved entropies of various qubit states in the zero charge sector as functions of r. Left: The symmetry-resolved von Neumann entropy of the charge zero sector for states of equal numbers k of identical positively and negatively charged particles. In the figure k = 1, 2, 3, 4 giving larger entropy for higher k. The maxima at r = 1/2 are $\log 2, \log 6, \log 20$ and $\log 70$, that is $\log(2k)! - 2\log(k!)$ which counts the number of distinct arrangements of two groups of k identical particles. Right: The symmetry-resolved Rényi entropy of the charge zero sector of a state consisting of four identical positively and four identical negatively charged excitations for n = 2, 4, 8, 20. The larger n is, the more sharply peaked at r = 1/2 the functions become. The value at r = 1/2 is $\log 70$, independent of n.

so performing the Fourier transform we get

$$S_n^{1^{\epsilon_1 \epsilon_{\dots 1} \epsilon}}(r;q) = \frac{1}{1-n} \log \frac{\sum_{j=0}^k {\binom{k}{j}} (1-r)^{nj} \delta_{q,\epsilon_j} r^{n(k-j)}}{\left[\sum_{j=0}^k {\binom{k}{j}} (1-r)^j \delta_{q,\epsilon_j} r^{(k-j)}\right]^n},$$
(2.79)

thus for $q = \epsilon j, j = 0, \ldots, k$, we have

$$S_n^{1^{\epsilon_1\epsilon}\dots 1^{\epsilon}}(r;\epsilon j) = \frac{1}{1-n} \log \frac{\binom{k}{j} (1-r)^{nj} r^{n(k-j)}}{\left[\binom{k}{j} (1-r)^{j} r^{(k-j)}\right]^n} = \log \binom{k}{j}.$$
 (2.80)

In this case the symmetry-resolved Rényi entropy tells us about the number of equally likely configurations which produce a charge ϵj in region A, and it is independent of n. Many other configurations can be considered, all of which produce different results, with similar interpretations. For instance, for a state with one positively and one negatively charged particle, the Fourier transform of the function

$$(r^{n} + e^{2\pi i\alpha}(1-r)^{n})(r^{n} + e^{-2\pi i\alpha}(1-r)^{n}), \qquad (2.81)$$

yields the simple formula

$$S_n^{1^+1^-}(r;q) = \frac{1}{1-n} \log \frac{(r^{2n} + (1-r)^{2n})\delta_{q,0} + r^n (1-r)^n (\delta_{q,1} + \delta_{q,-1})}{\left[(r^2 + (1-r)^2)\delta_{q,0} + r(1-r)(\delta_{q,1} + \delta_{q,-1})\right]^n},$$
(2.82)

and

$$S_n^{1^+1^-}(r;0) = \frac{1}{1-n} \log \frac{r^{2n} + (1-r)^{2n}}{(r^2 + (1-r)^2)^n}, \quad S_n^{1^+1^-}(r;\pm 1) = 0.$$
(2.83)

In this case the q = 0 result is *n*-dependent and gives a non-trivial symmetry-resolved von Neumann entropy:

$$S_1^{1^{+1^{-}}}(r;0) = \log(r^2 + (1-r)^2) - \frac{r^2 \log r^2 + (1-r)^2 \log(1-r)^2}{r^2 + (1-r)^2}, \quad S_1^{1^{+1^{-}}}(r;\pm 1) = 0.$$
(2.84)

In this example the SREE of the q = 0 sector is non-trivial as there are now two possible configurations that we can associate with such a charge, namely both particles being in region A and no particle being in region A. Thus there is a difference in the SREEs of states involving two particles with the same or distinct charges, even for the simple states considered here. Additional examples are presented in Fig. 2.1.

It is worth noting that all formulae in this section are in agreement with those in Section 2.2.4 if one identifies the ground state partition function $\mathcal{Z}_n^0(q-\epsilon j)$ with the quantity $\delta_{q,\epsilon j}$. Therefore, the study of multi-qubit states provides a neat application of the general results of the previous section to the case of a trivial, unentangled, ground state.

Because of the simplicity and explicit nature of the formulae for the multi-qubit states, it is possible to compute precisely the two contributions to the total von Neumann entropy, that is the configurational entropy and the number entropy [19, 93, 94]. Calling $S_1^{\Psi}(r)$ the total von Neumann entropy of the state $|\Psi\rangle$, we can write

$$S_1^{\Psi}(r) = \sum_q (p(q)S_1^{\Psi}(q;r) - p(q)\log p(q)), \qquad (2.85)$$

where $p(q) := \mathcal{Z}_1^{\Psi}(r;q)$, that is the symmetry-resolved partition function of the state for n = 1, and the term $\sum_q p(q) \log p(q)$ is the number entropy. The quantity p(q) represents the probability of obtaining the value q when measuring the charge. It is easy to work out an explicit example and see the features of these two contributions. For the state consisting of two distinct excitations with different charges the SREEs are given by (2.84). Thus, from

$$p(0) = r^2 + (1-r)^2, \qquad p(\pm 1) = r(1-r),$$
 (2.86)

it follows that the number entropy is simply

$$(r^{2} + (1 - r)^{2})\log(r^{2} + (1 - r)^{2}) + 2r(1 - r)\log(r(1 - r)),$$
(2.87)

while the configuration entropy is

$$(r^{2} + (1 - r)^{2})S_{1}^{1^{+}1^{-}}(r; 0), (2.88)$$

By summing these contributions we recover the known formula for the total von Neumann entropy of a state of two distinct excitations $-2r \log r - 2(1-r) \log(1-r)$ as found in [114, 115]. We note also that the number entropy takes its maximum (absolute) value $3/2 \log 2$ at r = 1/2, and that it can itself be considered a measure of entanglement, as discussed for other examples in [93, 109].

We close this section by noting that the entropy formulae for qubit states considered here do not have the property of equipartition, that is, they depend explicitly on the charge sector as we see for instance from Eq. (2.83). This is no contradiction, since the property of equipartition [94] is typically a leading order property (for instance in [94] it holds for small magnetisation), whereas in the case of multi-qubit states we have exact formulae rather than leading order expressions. Indeed, these states provide probably the simplest example where such a dependence on the charge sector can be fully computed.

2.4 Generalisations to interacting theories and higher dimensions

2.4.1 Magnon states

The agreement of the results obtained via form factors of CBPTFs in Section 2.2 and via multi-qubit states in Section 2.3 provide evidence that the formulae we obtained are correct. This is further substantiated by the numerical results we will present in the next Section. We now consider how our results might be applicable in a broader context. A natural starting point are magnon states. Such states describe the eigenstates of a variety of spin chain Hamiltonians, with or without interactions. They admit a simple explicit form in the spin basis so that entanglement computations are easy to perform. We also know from [115, 150] that the total entanglement entropy of magnon states is described by our formulae with $\alpha = 0$. As we see below, even in the presence of non-trivial scattering, the agreement extends to $\alpha \neq 0$.

The main idea behind this construction is somewhat similar in spirit to the qubit picture [1, 115], namely, that the entanglement content of quasiparticles can be easily understood if one factors out the zero-point fluctuations. In other words, instead of considering the full quantum theory where the quasiparticles are constructed on top of a nontrivial ground state, which in general has its own entanglement content, we consider a simpler theory in which particles are constructed above a trivial ground state. It turns out that the entanglement of this simpler model keeps track of the exact entanglement of the quasiparticle and explicitly discards explicitly the entanglement of the true ground state. Our magnon states belong to a Fock space generated by multi-particle configurations, endowed with an internal symmetry that acts as a phase on each multi-particle state. This phase is directly related to the quantum numbers of each particle.

2.4.1.1 One-magnon states

We firstly focus on a single magnon state on the lattice, belonging to the one-particle sector of a quantum spin- $\frac{1}{2}$ chain of length L. A one-magnon state can be written as a superposition of localised excitations with momentum p:

$$|\Psi_1\rangle = \frac{1}{\sqrt{L}} \sum_{j=1}^{L} e^{ipj} |j\rangle, \quad |j\rangle = |\uparrow\rangle_1 \otimes \dots |\downarrow\rangle_j \otimes \dots |\uparrow\rangle_L.$$
(2.89)

If one imposes boundary conditions on the chain, the momentum p is quantised as follows

$$p \in \frac{2\pi}{L}\mathbb{Z}.$$
(2.90)

We introduce the action of the symmetry operator $e^{2\pi i \alpha \hat{Q}}$, where \hat{Q} is associated with an internal symmetry. For our purposes we just need to specify its action on the vacuum state $|0\rangle = \bigotimes_{j=1}^{L} |\uparrow\rangle_{j}$ and on the one-particle sector. In addition we assume that the magnon is charged with respect to \hat{Q} , and it has charge +1.

We are interested in the entanglement of the one-magnon state with respect to the partition of spin sites into $A = \{1, 2, \dots, \ell\}$ and $\bar{A} = \{\ell + 1, \dots, L\}$. We associate to the region A a restricted symmetry generator $e^{2\pi i \alpha \hat{Q}_A}$, which acts as

$$e^{i2\pi\alpha\hat{Q}_A}|0\rangle = |0\rangle \qquad e^{2\pi i\alpha\hat{Q}_A}|j\rangle = e^{2\pi i\alpha\delta_{j\in A}}|j\rangle$$
(2.91)

where $\delta_{j\in A} = 1$ if $j \in A$ and $\delta_{j\in A} = 0$ otherwise. The reduced density matrix of the region A is

$$\rho_A := \operatorname{Tr}_{\bar{A}}(|\Psi_1\rangle\langle\Psi_1|) = \frac{1}{L} \sum_{j,j'\in A} e^{ip(j-j')} |j\rangle_{A\,A} \langle j'| + (1-r)|0\rangle_{A\,A} \langle 0|, \qquad (2.92)$$

where the states $|0\rangle_A$, $|j\rangle_A$ are defined by restricting the tensor products to the sites in A and $r = \ell/L$. The two terms appearing in the formula above are interpreted as the contributions associated to the presence/absence of the magnon in subsystem A. It is easy to show that

$$e^{2\pi i\alpha \hat{Q}_A} \rho_A^n = \left(\frac{1}{L} \sum_{j,j' \in A} e^{ip(j-j')} |j\rangle_{AA} \langle j'|\right)^n e^{2\pi i\alpha} + (1-r)^n |0\rangle_{AA} \langle 0|,$$
(2.93)

and that $[\rho_A, e^{2\pi i \alpha \hat{Q}_A}] = 0$. After a straightforward calculation one gets

$$\operatorname{Tr}_{A}\left(\frac{1}{L}\sum_{j,j'\in A}e^{ip(j-j')}|j\rangle_{AA}\langle j'|\right)^{n} = r^{n}, \quad \operatorname{Tr}_{A}\left((1-r)^{n}|0\rangle_{AA}\langle 0|\right) = (1-r)^{n}.$$
 (2.94)

Putting the pieces together, we arrive at the expected final result

$$\operatorname{Tr}\left(\rho_{A}^{n}e^{2\pi i\alpha\hat{Q}_{A}}\right) = r^{n}e^{2\pi i\alpha} + (1-r)^{n},$$
(2.95)

which provides the exact charged moments of a single magnon state.

2.4.1.2 Two-magnon states

In the following we consider a state of two magnons with the same symmetry charge. This example is more interesting because it allows us to test whether the presence of non-trivial interaction changes our results. Given a pair of momenta p and p', we parametrise the two-magnon state in the following way

$$|\Psi_{2}\rangle = \frac{1}{\sqrt{L}} \sum_{j,j'}^{L} S_{j,j'} e^{ipj + ip'j'} |jj'\rangle,$$
 (2.96)

where S is a scattering matrix and $|jj'\rangle$ is the state with two localised magnons in sites j and j'. The choice of the S-matrix is not really relevant for our purpose, but for the sake of concreteness we set

$$S_{jj'} = \begin{cases} e^{i\varphi} \text{ for } j > j', \\ 1 \text{ for } j < j', \\ 0 \text{ for } j = j', \end{cases}$$
(2.97)

using the same conventions as in [114]. The action of the restricted symmetry operator $e^{2\pi i \alpha \hat{Q}_A}$ on the two-particle sector of the Hilbert space is

$$e^{2\pi i \alpha \hat{Q}_A} |jj'\rangle = e^{2\pi i \alpha (\delta_{j \in A} + \delta_{j' \in A})} |jj'\rangle.$$
(2.98)

It is possible to decompose $\rho_A = \text{Tr}_{\bar{A}}(|\Psi_2\rangle\langle\Psi_2|)$ in a block-diagonal way as follows

$$\rho_A = \frac{1}{L} \left(\rho_A^{(1)} + \rho_A^{(2)} + \rho_A^{(3)} \right), \qquad (2.99)$$

where $\rho_A^{(1)}$ is the two-particle contribution (both particles in A), $\rho_A^{(2)}$ is the vacuum contribution (no particles in A) and $\rho_A^{(3)}$ is the one-particle contribution (one particle in A and one in \bar{A}). The introduction of the flux gives rise to the following relation

$$\rho_A^n e^{2\pi i \alpha \hat{Q}_A} = \frac{1}{L^n} \left((\rho_A^{(1)})^n e^{4\pi i \alpha} + \left(\rho_A^{(2)} \right)^n + \left(\rho_A^{(3)} \right)^n e^{2\pi i \alpha} \right).$$
(2.100)

No approximation was made up to this point, but the explicit expressions of $\rho_A^{(j)}$, given in [114], are cumbersome and not particularly enlightening for our purpose. However, one can show that in the limit (2.1) and with $p \neq p'$ kept fixed, $\operatorname{Tr}_A\left((\rho_A^{(j)})^n\right)$ simplifies drastically, and the leading contributions are:

$$\operatorname{Tr}_A\left((\rho_A^{(1)})^n\right) \simeq L^n r^{2n}, \quad \operatorname{Tr}_A\left((\rho_A^{(2)})^n\right) \simeq L^n (1-r)^{2n}, \quad \operatorname{Tr}_A\left((\rho_A^{(3)})^n\right) \simeq 2L^n r^n (1-r)^n.$$
 (2.101)

Putting all the pieces together, one finally gets

$$\operatorname{Tr}_{A}\left(\rho_{A}^{n}e^{2\pi i\alpha\hat{Q}_{A}}\right) \simeq r^{2n}e^{4\pi i\alpha} + 2r^{n}(1-r)^{n}e^{2\pi i\alpha} + (1-r)^{2n} = (r^{n}e^{2\pi i\alpha} + (1-r)^{n})^{2}.$$
 (2.102)

This computation shows that in this particular scaling limit the interaction between particles has no effect on the final result, and the total charged moment is just a product of two single-particle charged moments. A different result is obtained if p = p' and fixed. In that case, the magnons are indistinguishable and one can prove that

$$\operatorname{Tr}_{A}\left((\rho_{A}^{(1)})^{n}\right) \simeq L^{n}r^{2n}, \quad \operatorname{Tr}_{A}\left((\rho_{A}^{(2)})^{n}\right) \simeq L^{n}(1-r)^{2n}, \quad \operatorname{Tr}_{A}\left((\rho_{A}^{(3)})^{n}\right) \simeq 2^{n}L^{n}r^{n}(1-r)^{n}, \quad (2.103)$$

so that

$$\operatorname{Tr}_{A}\left(\rho_{A}^{n}e^{2\pi i\alpha\hat{Q}_{A}}\right) \simeq r^{2n}e^{4\pi i\alpha} + 2^{n}r^{n}(1-r)^{n}e^{2\pi i\alpha} + (1-r)^{2n},$$
(2.104)

which no longer factorises into one-magnon contributions. Both results are special cases of (2.6) and (2.7).

The results of this section generalise previous work for the excess entanglement entropy of excited states [115] and are also related to the results of [150] where the entanglement of magnon states was considered more generally. In particular, it was shown that for states consisting of several magnons, entanglement will factorise into the contributions of groups of magnons which are well-separated from each other in momentum space (that is, their momentum difference is of order $\mathcal{O}(1)$ rather than $\mathcal{O}(1/L)$, as the length of the system grows). Such results also apply to the present case up to the introduction of the appropriate phases.

2.4.2 Twist operator approach

So far we have derived the behaviour of the charged moments of the SREE of quasiparticle excited states making use of two different types of formalism: the form factor expansion in 1+1D (free) IQFTs [1] and the analysis of qubit/magnon states on the lattice. Unfortunately, these techniques are suited for a limited range of situations. Indeed, on the one hand, the description of the Rényi entropy as a correlation function of branch-point twist fields inserted at different points is special of 1+1D QFT [12, 20]. On the other hand, while the description of excited states as magnons or qubit states can be generalised to higher dimensions, it has the disadvantage of not taking into account the zero-point fluctuations. In this Section, we want to consider instead a generic QFT in higher dimensions. To this aim we introduce a slightly different approach.

Despite the technical limitations outlined above we expect that, in the particular scaling limit we are considering, the universal entanglement content of the symmetry-resolved Rényi entropy should not depend on dimensionality, on the presence of interactions and even on the integrability of the theory. The computation performed on interacting two-magnon states in the previous Subsection partially supports this claim. Concerning theories in higher dimensions, at least one precedent for this generalisation already exists. In a previous work [144], the excitations of the free massive boson in D := d + 1 dimensions were analysed and their Rényi entropy was computed in terms of graph partition functions. The results obtained therein agree with the formulae in [115], with r replaced by the ratio of generalised volumes.

In this section we slightly generalise the formalism of [144] to take into account possible interactions and provide, as a proof of concept, a simple calculation of symmetry-resolved entanglement of a single-particle excited state. The key ingredients we need are the description of the excited states as local operators acting on a vacuum state and a semi-local twist operator, which generalises the composite branch-point twist field to higher dimensional settings. The only strong assumption we make in our derivation is the presence of a finite mass gap m, with correlation length $\xi = m^{-1}$ much smaller than the typical lengths of the system.⁸

We anticipate here that our formulae (2.6)-(2.8) are unchanged in higher dimensional theories, up to the identification

$$r = \frac{V_A}{V},\tag{2.105}$$

which is the ratio between the (generalised) volumes of subsystem A and of the total system. It may seem surprising that the results should only depend on r and not on other features of the entanglement region, such as the connectivity and smoothness of its boundary. Indeed, the charged moments and

⁸The emergence of the universal entanglement content is also expected for some high-energy states in massless theories (see [148, 150] for the analysis of the gapless XY chain). However, here we keep the assumption of a finite mass gap to avoid technical complications.

symmetry-resolved entropies of both the ground state and excited states depend on such properties, as would finite-volume corrections to our results. However, our computations deliver results for the ratio of charged moments in the infinite-volume limit, and it is this ratio in this limit which is universal and independent of boundary features. This independence of boundary features has been analytically shown for the ratio of (uncharged) moments in the case of one-dimensional disconnected regions, where the same formulae as for one connected region were found to apply, with r the sum of the lengths of all disconnected parts [116].

2.4.2.1 Excited states and operator algebra

Let us consider the vacuum state $|0\rangle$ of a Hilbert space \mathcal{H} , together with an algebra \mathcal{A} of observables⁹ acting on \mathcal{H} which has $|0\rangle$ as a cyclic vector (see [167] for a modern review of this algebraic viewpoint in QFT). This allows us to represent any state $|\Psi\rangle$ of the Hilbert space as

$$|\Psi\rangle = \mathcal{O}|0\rangle$$
 with $\mathcal{O} \in \mathcal{A}$. (2.106)

We would like to assume that the vacuum state is translationally invariant, namely that it is invariant under a faithful representation of the translation group in d dimensions. However, since we consider a finite-size system, we modify this requirement by defining the system on a d-dimensional torus \mathcal{M} of volume V and requiring that $|0\rangle$ is invariant under the isometries of the torus. Other boundary conditions can be considered too, but they do not change the picture in the scaling limit we are interested in. We also require locality of the observables, asking that at any point $\mathbf{x} \in \mathcal{M}$, \mathcal{A} is generated by a set of fields $\{\mathcal{O}(\mathbf{x})\}$.

We define the Fourier transform of the field $\mathcal{O}(\mathbf{x})$ as

$$\mathcal{O}(\mathbf{p}) = \int_{\mathcal{M}} \mathrm{d}^d x \, e^{-i\mathbf{p}\mathbf{x}} \mathcal{O}(\mathbf{x}), \qquad (2.107)$$

where we adopt the same symbol \mathcal{O} in real and momentum space for notational convenience. The Fourier-transformed fields are building blocks for the following set of translationally invariant states

$$\mathcal{O}_1(\mathbf{p}_1)\dots\mathcal{O}_k(\mathbf{p}_k)|0\rangle,$$
 (2.108)

which correspond to k particles distributed over \mathcal{M} with momenta $\mathbf{p}_1, \ldots, \mathbf{p}_k$, and the choice of the fields $\{\mathcal{O}_j\}$ depend on the particle species and quantum numbers. This construction is similar to the usual way of generating particle states in free theories acting with creation operators on the vacuum of a Fock space. However, the advantage of our formulation is that it is directly related to local

⁹In the case of a single real boson, \mathcal{A} is just the algebra of operators generated by the field $\Phi(x)$ and its conjugated momentum $\Pi(x)$.

observables, a property which is fundamental to correctly define entanglement measures.

Let us take a set of orthogonal fields $\{\mathcal{O}_j\}$, so that the correlation function $\langle 0|\mathcal{O}_i^{\dagger}(\mathbf{x})\mathcal{O}_j(\mathbf{x}')|0\rangle$ vanishes for $i \neq j$. In other words, the fusion rule

$$[\mathcal{O}_i^{\dagger}] \times [\mathcal{O}_j] \to [1], \tag{2.109}$$

is present only if i = j and the Operator Product Expansion (OPE) can be expressed formally as

$$\mathcal{O}_{i}^{\dagger}(\mathbf{x})\mathcal{O}_{j}(\mathbf{x}') \simeq \delta_{ij} \langle 0|\mathcal{O}_{i}^{\dagger}(\mathbf{x}-\mathbf{x}')\mathcal{O}_{i}(0)|0\rangle + \dots, \qquad (2.110)$$

where in the right-hand side we neglected contributions coming from operators less relevant than the identity. The exact evaluation of the correlation function above can be hard, but the assumption of a finite gap m ensures that the latter is exponentially damped for $|\mathbf{x} - \mathbf{x}'| \gg m^{-1}$. This is the only property we really need in the following discussion.

We now consider a restriction of the modes $\mathcal{O}(\mathbf{p})$ with support in a subsystem only, that is a spacial region $A \subseteq \mathcal{M}$:

$$\mathcal{O}_A(\mathbf{p}) = \int_A \mathrm{d}^d x \, e^{-i\mathbf{p}\mathbf{x}} \mathcal{O}(\mathbf{x}). \tag{2.111}$$

Given any two regions $A, A' \subseteq \mathcal{M}$, we can compute¹⁰ $\mathcal{O}_{A}^{\dagger}(-\mathbf{p})\mathcal{O}_{A'}(\mathbf{p}')$ by making use of some approximations. First, we consider only the most relevant term in the fields OPE

$$\mathcal{O}_{A}^{\dagger}(-\mathbf{p})\mathcal{O}_{A'}(\mathbf{p}') = \int_{A} \mathrm{d}^{d}x \int_{A'} \mathrm{d}^{d}x' e^{i\mathbf{p}\mathbf{x}-i\mathbf{p}'\mathbf{x}'} \mathcal{O}^{\dagger}(\mathbf{x})\mathcal{O}(\mathbf{x}')$$
$$\simeq \int_{A} \mathrm{d}^{d}x \int_{A'} \mathrm{d}^{d}x' e^{i\mathbf{p}\mathbf{x}-i\mathbf{p}'\mathbf{x}'} \langle 0|\mathcal{O}^{\dagger}(\mathbf{x})\mathcal{O}(\mathbf{x}')|0\rangle.$$
(2.112)

Second, since we are working in the limit of small correlation length (compared to the geometry), the leading contribution comes from the insertion of the fields at small distances, which is present if $\mathbf{x}, \mathbf{x}' \in A \cap A'$; this observation motivates the change of variable $\mathbf{x}'' = \mathbf{x}' - \mathbf{x}$, and the subsequent approximation¹¹

$$\mathcal{O}_{A}^{\dagger}(-\mathbf{p})\mathcal{O}_{A'}(\mathbf{p}') \simeq \int_{A \cap A'} \mathrm{d}x \, e^{i(\mathbf{p}-\mathbf{p}')\mathbf{x}} \, \int_{\mathcal{M}} \mathrm{d}x'' \, e^{-i\mathbf{p}'\mathbf{x}''} \langle 0|\mathcal{O}^{\dagger}(0)\mathcal{O}(\mathbf{x}'')|0\rangle.$$
(2.113)

The second integral may be difficult to compute and in principle it could require a UV regularisation

¹⁰One should note that Hermitian conjugation and Fourier transform do not commute. Indeed, with our notation

 $[\]mathcal{O}_{A}^{\dagger}(-\mathbf{p}) = (\mathcal{O}_{A}(\mathbf{p}))^{\dagger}.$ ¹¹To be more rigorous: in approximating the integration domain $\mathbf{x} \in A$, $\mathbf{x}' \in A'$ with $\mathbf{x}, \mathbf{x}' \in A \cap A'$ we are neglecting terms where at least one of the two variables, say **x**, lies outside of $A \cap A'$. If the distance between **x** and $A \cap A'$ is smaller than the correlation length there is no exponential suppression, however, the integration over \mathbf{x} produces a term which grows with the area of the boundary $\partial(A \cap A')$ and hence is subleading in the large-volume limit. On the other hand, the integration domain of the variable \mathbf{x}'' depends in principle on the set $A \cap A'$, and its approximation with \mathcal{M} is valid up to terms exponentially damped in the correlation length.

for $|\mathbf{x}''| < \epsilon \ll m^{-1}$. However, because the integration is over \mathcal{M} , the result does not depend on the regions A, A' and in our computation this integral appears only as a multiplicative constant. In conclusion, we end up with

$$\mathcal{O}_{A}^{\dagger}(-\mathbf{p})\mathcal{O}_{A'}(\mathbf{p}') \propto V_{A \cap A'}\delta_{\mathbf{p},\mathbf{p}'}, \qquad (2.114)$$

where $V_{A \cap A'}$ is the volume of $A \cap A'$. Equation (2.114) is the main result of this Subsection. Since the volume in (2.114) emerges from the integrals (2.113), which involve a Fourier transform, we require that subsystem $A \cap A'$ consists of a finite number of disconnected regions, whose boundaries are piecewise smooth.

It is natural to ask how the discussion above would be modified for a vanishing gap m = 0. The main change is in the scaling of correlations functions: exponential localisation of the correlation function in a region of typical length m^{-1} does not hold any longer, due to the long algebraic tails of the correlation functions. We conjecture that, as long as the momenta are fixed in the infinite-volume limit, the main conclusion (2.114) is unchanged. A qualitative argument is that in this case the inverse momentum, say the De Broglie length, plays the role of typical length scale. In order to make this consideration more precise, let us analyse Eq. (2.113) for a 1+1D CFT, where \mathcal{O} is a field of conformal dimension $\Delta_{\mathcal{O}}$. We focus on the following integral

$$\int_{\mathcal{M}} \mathrm{d}x \, e^{-i\mathbf{p}\mathbf{x}} \langle 0|\mathcal{O}^{\dagger}(0)\mathcal{O}(\mathbf{x})|0\rangle, \qquad (2.115)$$

which we regulate both in the UV, with a cutoff ϵ , and in the IR, with a cutoff L, as follows

$$\int_{\epsilon}^{L} \mathrm{d}x \, e^{-ipx} \frac{1}{x^{4\Delta \mathcal{O}}} + (\mathrm{c.c.}). \tag{2.116}$$

This integral can be explicitly computed. However, the important feature is that for $\Delta_{\mathcal{O}} > 0$, p > 0and $\epsilon > 0$ all fixed, the integral converges to a finite value when $L \to +\infty$. This is no longer the case if $p \sim 1/L$ in the infinite-volume limit. In practice, this means that for small momentum and scaling dimension $0 < \Delta_{\mathcal{O}} \leq 1$ the considerations we made so far regarding the scaling at large sizes cannot be applied. As a matter of fact, for free CFTs the scaling dimensions of the fundamental fields are smaller than 1: the fermionic field Ψ has dimension 1/2 while the derivative of a compact boson $\partial_x \Phi$ has dimension 1. While these considerations are not mathematically rigorous in establishing convergence of the OPE in the large-volume limit, they are sufficient to explain why low-energy states of gapless theories, or multi-particle states with small momenta difference, are not well captured by our predictions. Indeed, for such states the excess entanglement was computed in [154, 155], and the results obtained therein are different from those of [114, 115].

2.4.2.2 Replica construction for symmetry-resolved entanglement

Consider now a replica version of the theory, consisting of n copies of the latter. For any state $|\Psi\rangle$ we denote its replicated version by $|\Psi\rangle^n$. Our goal is to define a U(1) composite twist operator which generalises to higher dimensions the composite branch-point twist field defined in [19]. We construct this operator in such a way that its expectation value over $|\Psi\rangle^n$ gives exactly the charged moments $Z_n(\alpha)$, allowing for the computation of symmetry-resolved Rényi entropies in higher dimensional QFTs. This type of operator was already considered in the literature, mostly without the flux insertion (see for example [144, 168–170]). However, the novelty of our approach consists in establishing the relation between the twist operator and the algebra of local operators.

The first point we have to clarify regards the symmetry and its action on the space of fields. Starting from $e^{2\pi i \alpha Q}$, the global generator of U(1) symmetry in the non-replicated theory, we say that $\mathcal{O}(\mathbf{x})$ has charge $\kappa_{\mathcal{O}}$ if

$$e^{2\pi i\alpha \bar{Q}}\mathcal{O}(\mathbf{x})e^{-2\pi i\alpha \bar{Q}} = e^{2\pi i\alpha\kappa_{\mathcal{O}}}\mathcal{O}(\mathbf{x}).$$
(2.117)

Since one can decompose the space of fields in irreducible representations of U(1), we restrict our analysis to charged fields. Going back to the replicated theory, we define the algebra of replicated observables \mathcal{A}^n as the algebra generated by the tensor product of n observables in \mathcal{A} . Thus, to any field $\mathcal{O}(\mathbf{x}) \in \mathcal{A}$, we associate $\mathcal{O}^j(\mathbf{x}) \in \mathcal{A}^n$ defined as

$$\mathcal{O}^{j}(\mathbf{x}) = 1 \otimes \cdots \otimes 1 \otimes \mathcal{O}(\mathbf{x}) \otimes 1 \cdots 1, \qquad (2.118)$$

where $\mathcal{O}(\mathbf{x})$ is inserted in the *j*th replica. By requiring orthogonality of local fields in different copies, the momentum space OPE (2.114) generalises to:

$$\mathcal{O}_{A}^{i,\dagger}(-\mathbf{p})\mathcal{O}_{A'}^{j}(\mathbf{p}') \propto V_{A \cap A'} \delta_{\mathbf{p},\mathbf{p}'} \delta_{i,j}.$$
(2.119)

Consider now a spacial region $A \subset \mathcal{M}$ and its complement \overline{A} . We define a composite twist operator T_A^{α} which implements the structure of the *n*-sheeted, cyclically connected, Riemann surface where the replica theory is defined. That is, T_A^{α} implements the "gluing" of the replicas along A with an additional flux insertion due to the action of the U(1) symmetry. The commutation relations of T_A^{α} with any charged field $\mathcal{O}^j(\mathbf{x})$ generalise in an obvious way those of a standard CBPTF:

$$T_{A}^{\alpha}\mathcal{O}^{j}(\mathbf{x}) = \begin{cases} e^{2\pi i\kappa_{\mathcal{O}}\alpha\delta_{j,n}}\mathcal{O}^{j+1}(\mathbf{x})T_{A}^{\alpha} & \mathbf{x} \in A, \\ \mathcal{O}^{j}(\mathbf{x})T_{A}^{\alpha} & \mathbf{x} \in \bar{A}. \end{cases}$$
(2.120)

Here, the flux is inserted only between the nth and the first replica, but other choices are possible. We emphasize that a similar definition has already appeared in the context of 1+1D integrable QFTs (see [105–108]). In particular, for $A = [0, \ell]$ one can identify

$$T_A^{\alpha} = \mathcal{T}_n^{\alpha}(0)\tilde{\mathcal{T}}_n^{\alpha}(\ell), \qquad (2.121)$$

and the commutation relations for T_A^{α} can be obtained from those of the CBPTF and of its Hermitian conjugate. However, such a representation in terms of the usual CBPTFs of one-dimensional theories is in general not possible in higher dimensions. Regarding the definition of the twist operator T_A^{α} , it is worth mentioning also that in QFT it is known[171, 172] that the twist operators are not local observables of the algebra \mathcal{A}^n . Rather, they are observables of the orbifolded algebra $\mathcal{A}^n/\mathbb{Z}_n$, which is obtained by taking the coset of \mathcal{A}^n over the cyclic symmetry of replicas.

We are now ready to relate the twist operator to the symmetry-resolved entanglement. The charged moments of $|\Psi\rangle$ are given by

$$Z_n^{\Psi}(\alpha) = \frac{{}^n \langle \Psi | T_A^{\alpha} | \Psi \rangle^n}{{}^n \langle \Psi | \Psi \rangle^n}.$$
(2.122)

The above definition, together with the commutation relations (2.120) and the momentum space OPE (2.119), is enough to obtain the explicit expression for the ratio of charged moments between the state $|\Psi\rangle$ and the ground state, as defined in (2.5). We now show how these ideas come together with a simple example.

2.4.2.3 Single-particle state

In this Section, we consider an excited state $|\Psi\rangle$ consisting of a single quasiparticle with momentum **p** generated by a charged field \mathcal{O} . Its explicit expression is given by

$$|\Psi\rangle = \mathcal{O}(\mathbf{p})|0\rangle, \qquad (2.123)$$

and the replicated version is just

$$|\Psi\rangle^n = \mathcal{O}^1(\mathbf{p})\dots\mathcal{O}^n(\mathbf{p})|0\rangle^n.$$
(2.124)

For the sake of convenience, we split the observable

$$\mathcal{O}^{j}(\mathbf{p}) = \mathcal{O}^{j}_{A}(\mathbf{p}) + \mathcal{O}^{j}_{\bar{A}}(\mathbf{p}), \qquad (2.125)$$

so that its commutation relations with T_A^{α} become more transparent. Indeed, using (2.120) one can write

$$T_{A}^{\alpha}|\Psi\rangle^{n} = T_{A}^{\alpha}(\mathcal{O}_{A}^{1}(\mathbf{p}) + \mathcal{O}_{\bar{A}}^{1}(\mathbf{p}))\dots(\mathcal{O}_{A}^{n}(\mathbf{p}) + \mathcal{O}_{\bar{A}}^{n}(\mathbf{p}))|0\rangle^{n} =$$

$$(\mathcal{O}_{A}^{2}(\mathbf{p}) + \mathcal{O}_{\bar{A}}^{1}(\mathbf{p}))\dots(\mathcal{O}_{A}^{1}(\mathbf{p})e^{2\pi i\alpha\kappa_{\mathcal{O}}} + \mathcal{O}_{\bar{A}}^{n}(\mathbf{p}))T_{A}^{\alpha}|0\rangle^{n}.$$
(2.126)

Up to now, everything is exact. However, to compute the expectation value ${}^{n}\langle\Psi|T_{A}^{\alpha}|\Psi\rangle^{n}$ we make use of the OPE contraction (2.119). Among all the terms which are generated, all but two are vanishing and they give

$${}^{n}\langle\Psi|T_{A}^{\alpha}|\Psi\rangle^{n} \simeq e^{2\pi i\alpha\kappa_{\mathcal{O}}} {}^{n}\langle0|(\mathcal{O}^{\dagger})_{A}^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})_{A}^{1}(-\mathbf{p})\mathcal{O}_{A}^{2}(\mathbf{p})\dots\mathcal{O}_{A}^{n}(\mathbf{p})\mathcal{O}_{A}^{1}(\mathbf{p})T_{A}^{\alpha}|0\rangle^{n}$$

$$+{}^{n}\langle0|(\mathcal{O}^{\dagger})_{\bar{A}}^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})_{\bar{A}}^{1}(-\mathbf{p})\mathcal{O}_{\bar{A}}^{1}(\mathbf{p})\mathcal{O}_{\bar{A}}^{2}(\mathbf{p})\dots\mathcal{O}_{\bar{A}}^{n}(\mathbf{p})T_{A}^{\alpha}|0\rangle^{n}$$

$$\propto \left(e^{2\pi i\alpha\kappa_{\mathcal{O}}}V_{A}^{n}+(V-V_{A})^{n}\right)\frac{{}^{n}\langle0|T_{A}^{\alpha}|0\rangle^{n}}{{}^{n}\langle0|0\rangle^{n}}.$$

$$(2.127)$$

To evaluate the first of the two terms we needed to commute the local observable before applying the OPE contraction, but the commutators always produce terms which are subleading in the volume, as it is obvious e.g. from the inspection of the free theory. Similarly, we can evaluate the norm ${}^{n}\langle\Psi|\Psi\rangle^{n}$ which does not require the splitting of $\mathcal{O}^{j}(\mathbf{p})$

$${}^{n}\langle\Psi|\Psi\rangle^{n} = {}^{n}\langle0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\mathcal{O}^{1}(\mathbf{p})\dots\mathcal{O}^{n}(\mathbf{p})|0\rangle^{n} \propto V^{n}.$$
(2.128)

In the evaluation of the ratio

$$\frac{{}^{n}\langle\Psi|T_{A}^{\alpha}|\Psi\rangle^{n}}{{}^{n}\langle\Psi|\Psi\rangle^{n}},$$
(2.129)

the proportionality constant (which is non-universal and could be absorbed in a redefinition of the field) cancels out, and one can write

$$\frac{{}^{n}\langle\Psi|T_{A}^{\alpha}|\Psi\rangle^{n}}{{}^{n}\langle\Psi|\Psi\rangle^{n}} \simeq \left(e^{2\pi i\alpha\kappa_{\mathcal{O}}}r^{n} + (1-r)^{n}\right)\frac{{}^{n}\langle0|T_{A}^{\alpha}|0\rangle^{n}}{{}^{n}\langle0|0\rangle^{n}}$$
(2.130)

with $r = \frac{V_A}{V}$. In the expression above the term in brackets is universal, while the term factored out is not universal and it is the *n*th charged moment of the ground state. Therefore, as anticipated, the ratio of charged moments is given by:

$$M_n^{\Psi}(r,\alpha) = \frac{n\langle \Psi | T_A^{\alpha} | \Psi \rangle^n}{n\langle \Psi | \Psi \rangle^n} \frac{n\langle 0 | 0 \rangle^n}{n\langle 0 | T_A^{\alpha} | 0 \rangle^n} \simeq e^{2\pi i \alpha \kappa_{\mathcal{O}}} r^n + (1-r)^n.$$
(2.131)

Results for multi-particle states can be obtained in a similar fashion.

In conclusion, the striking simplicity of these results relies on the truncation of the OPE in (2.110), which is expected to become exact in the limit $m^d V \gg 1$. We expect that for finite $m^d V$ further contributions in the OPE can be recast as a (possibly non-integer) power series in $(m^d V)^{-1}$, which generalises the explicit $(mL)^{-1}$ power expansion that is obtained for 1+1D free theories using form factor techniques (see Appendix 2.C). In massless theories, we instead expect corrections as a power series in $(|\mathbf{p}|^d V)^{-1}$. The explicit evaluation of these non-universal corrections, as well as the treatment of possible divergences in the power series, are all beyond the purpose of this Thesis.

2.5 Numerical results

In this section we present numerical results for two very different discrete models. First we consider a 1D lattice Fermi gas, which has a gapless phase but it also possesses highly excited states whose entanglement is well described by our formulae, and then we look at the harmonic chain, which becomes a U(1) massive free boson in the scaling limit. While for the first model we can only consider distinct excitations, for the second we consider also identical excitations. The treatment of the complex harmonic chain is based on a generalisation of the wave-functional method as presented in [115] which we derive in detail. The good agreement with the theoretical predictions confirms the picture already put forward for the total entropy in [114], i.e. that our formulae hold under the broad assumption of localised excitations.

2.5.1 1D lattice Fermi gas

In this subsection we analyse a particle-hole excited state of a 1D lattice Fermi gas, comparing our analytical predictions with the numerical data. Even though the model is critical, it was realised in [147] that certain highly energetic quasiparticle excitations still have a universal entanglement content. More precisely, if one considers a set of quasiparticles with small enough De Broglie wavelengths (compared to the typical geometric lengths) and sufficiently separated momenta, then these quasiparticles will be essentially uncorrelated with each other and with respect to zero-point fluctuations. We refer to [145–150] for further details about the universal entanglement content of quasiparticles in critical systems.

Our goal here is to briefly review the numerical techniques involved in the characterisation of fermionic Gaussian states [173] and their application to the computation of symmetry-resolved measures. We start by considering the Hamiltonian of free spinless fermions on a circle of length L

$$H = -\frac{1}{2} \sum_{j} f_{j+1}^{\dagger} f_{j} + f_{j}^{\dagger} f_{j+1} + \mu \sum_{j} f_{j}^{\dagger} f_{j}, \qquad (2.132)$$

where μ is the chemical potential and $\{f_j\}_{j=1,\dots,L}$, $\{f_j^{\dagger}\}_{j=1,\dots,L}$ are the ladder operators obeying the standard anticommutation relations

$$\{f_j, f_{j'}\} = \{f_j^{\dagger}, f_{j'}^{\dagger}\} = 0, \qquad \{f_j, f_{j'}^{\dagger}\} = \delta_{jj'}.$$
(2.133)

When $|\mu| < 1$ the theory is gapless, as can be seen by moving to momentum space, where the Hamiltonian is diagonal:

$$H = \sum_{q=0}^{L-1} \left[\mu - \cos\left(\frac{2\pi q}{L}\right) \right] \tilde{f}_q^{\dagger} \tilde{f}_q, \qquad (2.134)$$

and the ground state is a Fermi sea with Fermi momentum $k_F = \arccos(\mu)$. The two-point function evaluated in the ground state at Fermi momentum k_F takes the following form [174, 175]:

$$C_0(j,j') := \langle f_j^{\dagger} f_{j'} \rangle_0 = \frac{\sin k_F(j-j')}{L \sin \frac{\pi(j-j')}{L}}.$$
(2.135)

Here, we analyse the quasiparticle excited state described by the two-point function

$$C(j,j') = C_0(j,j') + \frac{1}{L}e^{-i(k_F + \frac{\pi}{4} - \frac{\pi}{L})(j-j')} - \frac{1}{L}e^{-i(k_F - \frac{\pi}{4} + \frac{\pi}{L})(j-j')},$$
(2.136)

which corresponds to the insertion of a fermion of momentum $k = k_F + \frac{\pi}{4} - \frac{\pi}{L}$ above the ground state and the removal of another fermion (or equivalently, the insertion of a hole) at $k = k_F - \frac{\pi}{4} + \frac{\pi}{L}$. The choice of the momentum shift is not important in the continuum limit, where the only necessary condition is that $|k - k_F|$ remains finite when $L \to \infty$ ¹². We now have to specify the symmetry of the model. The Hamiltonian (2.132) is invariant under an internal U(1) symmetry associated to the number of fermions, generated by the operator

$$\hat{Q} = \sum_{j} f_j^{\dagger} f_j, \qquad (2.137)$$

which satisfies the locality condition $\hat{Q} = \hat{Q}_A + \hat{Q}_{\bar{A}}$, with

$$\hat{Q}_A = \sum_{j \in A} f_j^{\dagger} f_j, \quad \hat{Q}_{\bar{A}} = \sum_{j \in \bar{A}} f_j^{\dagger} f_j.$$
 (2.138)

As done in Section 2.4.1, the subsystems A and \overline{A} are defined by the sets of fermion sites $A = \{1, \ldots, \ell\}$, $\overline{A} = \{\ell + 1, \ldots, L\}$. We denote by C_0^A and C^A the $\ell \times \ell$ matrices resulting from projection of the matrices C_0 and C (defined by equations (2.135) and (2.136) respectively) onto subsystem A, so that the indices $j, j' \in A$. Following [104] we express the charged moments of the particle-hole state and the ground state by means of the determinants

$$\operatorname{Tr}_{A}(\rho_{A}^{n}e^{2\pi i\alpha \hat{Q}_{A}}) = \det\left((C^{A})^{n}e^{2\pi i\alpha} + (1 - C^{A})^{n}\right), \qquad (2.139)$$

$$\operatorname{Tr}_{A}(\rho_{A,0}^{n}e^{2\pi i\alpha\hat{Q}_{A}}) = \det\left((C_{0}^{A})^{n}e^{2\pi i\alpha} + (1 - C_{0}^{A})^{n}\right),$$
(2.140)

with ρ_A and $\rho_{A,0}$ the RDMs of the two states. According to our analytical predictions, because the fermion and the hole are distinguishable quasiparticles with opposite charges ± 1 , we expect the ratio

¹²In the work [104] a particle-hole state satisfying $|k - k_F| \sim 1/L$ was considered. Unlike the present case, the entanglement measures of such a low-lying state is captured by CFT predictions, due to the strong correlation effects between the particle/hole and the zero-point fluctuations.

of charged moment to take the universal form

$$\frac{\operatorname{Tr}_{A}(\rho_{A}^{n}e^{2\pi i\alpha\hat{Q}_{A}})}{\operatorname{Tr}_{A}(\rho_{A,0}^{n}e^{2\pi i\alpha\hat{Q}_{A}})} \simeq (r^{n}e^{2\pi i\alpha} + (1-r)^{n})(r^{n}e^{-2\pi i\alpha} + (1-r)^{n}), \qquad (2.141)$$

where the equality only holds in the scaling limit of the lattice model, with $r = \ell/L$ fixed.

To test the validity of Eq. (2.141) we consider two entanglement measures, namely the excess of (total) Rényi entropy and the so-called (following the terminology of [104]) "excess of variance". The excess of entropy is recovered from our formulae for $\alpha = 0$, and for two distinct excitations it takes the simple form

$$\Delta S_n = \frac{1}{1-n} \log \frac{\text{Tr}_A(\rho_A^n)}{\text{Tr}_A(\rho_{A,0}^n)} \simeq \frac{\log \left(r^n + (1-r)^n\right)^2}{1-n}.$$
(2.142)

We define the variance¹³ associated to ρ_A as

$$\langle \Delta \hat{Q}_A^2 \rangle_n := \frac{\operatorname{Tr}_A(\rho_A^n \hat{Q}_A^2)}{\operatorname{Tr}_A(\rho_A^n)} - \left(\frac{\operatorname{Tr}_A(\rho_A^n \hat{Q}_A)}{\operatorname{Tr}_A(\rho_A^n)} \right)^2 = \frac{1}{(2\pi i)^2} \frac{d^2}{d\alpha^2} \log \frac{\operatorname{Tr}_A(\rho_A^n e^{2\pi i\alpha \hat{Q}_A})}{\operatorname{Tr}_A(\rho_A^n)} \bigg|_{\alpha=0}.$$
 (2.143)

Similarly, we denote by $\langle \Delta \hat{Q}_A^2 \rangle_{n,0}$ the variance of the ground state $\rho_{A,0}$. From (2.141) it follows that the excess of variance is given by

$$\langle \Delta \hat{Q}_A^2 \rangle_n - \langle \Delta \hat{Q}_A^2 \rangle_{n,0} \simeq \frac{2r^n (1-r)^n}{(r^n + (1-r)^n)^2}.$$
 (2.144)

A way to physically interpret the result (2.144) is to regard this excess of variance as twice the contribution associated to a single quasiparticle, since particles and antiparticles contribute in the same way. The latter is just the variance of a Bernoulli random variable with success probability given by

$$p = \frac{r^n}{r^n + (1 - r)^n},\tag{2.145}$$

namely the probability one associates to the presence of a quasiparticle in A computed with the density matrix ρ_A^n . Since the variance of a Bernoulli variable with probability p is just p(1-p), one gets Eq. (2.144).

In Fig. 2.2 we report our analytical predictions and the numerical values of ΔS_n and $\langle \Delta \hat{Q}_A^2 \rangle_n - \langle \Delta \hat{Q}_A^2 \rangle_{n,0}$, computed from (2.139) and (2.140) using exact diagonalisation of the correlation matrices $C_A, C_{A,0}$. We keep L fixed, analysing different values of $r = \ell/L$. Our choice is motivated by the expectation that these plots should be "universal" at large L, meaning that data obtained with different values L should collapse to the same universal prediction (independent of lattice details such as the value of k_F) when $L \to \infty$. As we see from the plots of both measures, the match between numerics

¹³The choice of this terminology comes from the fact that for n = 1 this measure reproduces the physical variance of the charge in the state ρ_A .



Figure 2.2 Numerical data versus analytical prediction for the particle-hole excited state described by the correlation function (2.136). Here, $k_F = \pi/2$, L = 200 and we considered the first values of nfor $r = \ell/L \in [0, 1]$. Left: Excess Rényi entropy checked against Eq. (2.142). Right: Excess variance, checked against Eq. (2.144). The numerical results are in very good agreement with the analytical formulae.

and analytics is really good.

2.5.2 Complex free boson and 1D harmonic chain

In this section we consider a complex massive free boson. Unlike the 1D Fermi gas, this model and its lattice version allow us to test formulae for states containing two or more identical excitations. Our numerical computation is based on the wave-functional method introduced in [115] (see Appendix A of that paper). Here we need to extend the technique to a complex theory and to the symmetry-resolved moments. These extensions are not entirely trivial and for that reason we review the wave-functional method in detail.

Let us consider a 1D complex massive boson on the line [0, L] with Hamiltonian:

$$H = \int_0^L \mathrm{d}x \left(\Pi^{\dagger} \Pi + (\partial_x \Phi)^{\dagger} (\partial_x \Phi) + m^2 \Phi^{\dagger} \Phi \right), \qquad (2.146)$$

where

$$\Pi(x) = \dot{\Phi}^{\dagger}(x), \quad \Pi^{\dagger}(x) = \dot{\Phi}(x). \tag{2.147}$$

Alternatively, one can introduce a pair of real bosons Φ_1, Φ_2 , and express Φ and Π as

$$\Phi = \frac{\Phi_1 + i\Phi_2}{\sqrt{2}}, \quad \Pi = \frac{\Pi_1 + i\Pi_2}{\sqrt{2}}, \quad (2.148)$$

so that the Hamiltonian becomes that of two real bosons. The only non-vanishing equal-time commutators are:

$$[\Phi(x), \Pi(y)] = [\Phi^{\dagger}(x), \Pi^{\dagger}(y)] = i\delta(x - y).$$
(2.149)
Since space is compact, there are discrete energy levels with dispersion relation:

$$E_p = \sqrt{m^2 + p^2}, \quad p \in \frac{2\pi}{L} \mathbb{Z}, \tag{2.150}$$

and the Hamiltonian is diagonalised via the introduction of two sets of creation/annihilation operators. The annihilation operators are:

$$A_p = \frac{1}{\sqrt{2LE_p}} \int_0^L \mathrm{d}x e^{-ipx} (E_p \Phi(x) + i\Pi^{\dagger}(x)), \qquad (2.151)$$

$$B_p = \frac{1}{\sqrt{2LE_p}} \int_0^L \mathrm{d}x e^{-ipx} (E_p \Phi^{\dagger}(x) + i\Pi(x)).$$
(2.152)

and as usual A_p^{\dagger} (B_p^{\dagger} resp.) creates from the a positively (negatively) U(1)-charged particle with momentum p from the vacuum. These operators satisfy:

$$[A_p, A_{p'}^{\dagger}] = [B_p, B_{p'}^{\dagger}] = \delta_{p,p'}.$$
(2.153)

 $\Phi(x)$, $\Pi(x)$ then admit the usual Fourier decomposition

$$\Phi(x) = \sum_{p} \frac{1}{\sqrt{2LE_p}} \left(A_p e^{ipx} + B_p^{\dagger} e^{-ipx} \right), \qquad (2.154)$$

$$\Pi(x) = -i\sum_{p} \sqrt{\frac{E_p}{2L}} \left(B_p e^{ipx} - A_p^{\dagger} e^{-ipx} \right).$$
(2.155)

Finally, the charge operator corresponding to the U(1) symmetry of the theory is:

$$Q = i \int_0^L \mathrm{d}x \, : \left(\Phi^{\dagger}(x)\Pi^{\dagger}(x) - \Phi(x)\Pi(x)\right) := \sum_p (A_p^{\dagger}A_p - B_p^{\dagger}B_p), \tag{2.156}$$

where $: \mathcal{O} :$ denotes the usual normal ordering of \mathcal{O} .

2.5.2.1 The wave-functional method

A very useful way to represent the states in the theory is provided by the wave-functional formalism. In this approach, we associate to every state a functional Ψ acting on the space of classical field configurations $\phi, \phi^{\dagger} : [0, L] \to \mathbb{C}$ and formally defined by:

$$\Psi[\phi, \phi^{\dagger}] = \langle \phi, \phi^{\dagger} | \Psi \rangle.$$
(2.157)

This definition resembles that of the wave function $\psi(x) = \langle x | \psi \rangle$ in non-relativistic quantum mechanics, and the action of the operators Φ , Π on the state $|\phi, \phi^{\dagger}\rangle$ mimics that of the position and

momentum operators on the state $|x\rangle$:

$$\Phi(x)\Psi[\phi,\phi^{\dagger}] = \phi(x)\Psi[\phi,\phi^{\dagger}], \quad i\Pi(x)\Psi[\phi,\phi^{\dagger}] = \frac{\delta\Psi[\phi,\phi^{\dagger}]}{\delta\phi(x)},$$
(2.158)

and analogously

$$\Phi^{\dagger}(x)\Psi[\phi,\phi^{\dagger}] = \phi^{\dagger}(x)\Psi[\phi,\phi^{\dagger}], \quad i\Pi^{\dagger}(x)\Psi[\phi,\phi^{\dagger}] = \frac{\delta\Psi[\phi,\phi^{\dagger}]}{\delta\phi^{\dagger}(x)}.$$
(2.159)

Notice that, for consistency, when $i\Pi$, $i\Pi^{\dagger}$ act on a ket $|\phi, \phi^{\dagger}\rangle$ there is a minus sign in front of the functional derivative. The vacuum state functional is defined by:

$$A_p \Psi_{\text{vac}} = B_p \Psi_{\text{vac}} = 0 \quad \forall \, p \in \frac{2\pi}{L} \mathbb{Z},$$
(2.160)

and the only solution to these functional differential equations up to normalisation is the Gaussian functional:

$$\Psi_{\text{vac}}[\phi, \phi^{\dagger}] = \exp\left[-\int_{0}^{L} \mathrm{d}x \int_{0}^{L} \mathrm{d}y \,\phi^{\dagger}(x) K(x-y)\phi(y)\right], \quad K(x-y) = \frac{1}{L} \sum_{p} E_{p} e^{ip(x-y)}. \quad (2.161)$$

Notice that K(x) is a real and even function of x. The functionals of the positively and negatively charged one-particle states are obtained through the action of A_p^{\dagger} and B_p^{\dagger} :

$$A_p^{\dagger} \Psi_{\text{vac}} = \alpha_p[\phi^{\dagger}] \Psi_{\text{vac}}, \quad \alpha_p[\phi^{\dagger}] = \sqrt{\frac{2E_p}{L}} \int_0^L \mathrm{d}x \, e^{ipx} \phi^{\dagger}(x), \tag{2.162}$$

$$B_p^{\dagger} \Psi_{\text{vac}} = \beta_p[\phi] \Psi_{\text{vac}}, \quad \beta_p[\phi] = \sqrt{\frac{2E_p}{L}} \int_0^L dx \, e^{ipx} \phi(x), \qquad (2.163)$$

and the functional of a state with k^+ positive excitations and k^- negative excitations (all with different momenta) is:

$$\Psi_{\{p_i,q_j\}}^{k^+,k^-}[\phi,\phi^{\dagger}] = \prod_{i=1}^{k^+} A_{p_i}^{\dagger} \prod_{j=1}^{k^-} B_{q_j}^{\dagger} \Psi_{\text{vac}}[\phi,\phi^{\dagger}] = \prod_{i=1}^{k^+} \alpha_{p_i}[\phi^{\dagger}] \prod_{j=1}^{k^-} \beta_{q_j}[\phi] \Psi_{\text{vac}}[\phi,\phi^{\dagger}].$$
(2.164)

A correct choice of the normalisation in (2.161) ensures that the functional above has unit norm with respect to the bra-ket product. It is then immediate to construct the functionals of multi-particle states with identical excitations. If there are k_i^+ (k_i^-) positively (negatively) charged particles with momentum p_i (q_i), for $i = 1, ..., m^+$ (m^-) we define:

$$\Psi_{\{p_i,q_j\}}^{\{k_i^+,k_j^-\}}[\phi,\phi^{\dagger}] = \prod_{i=1}^{m^+} \frac{(A_{p_i}^{\dagger})^{k_i^+}}{\sqrt{k_i^+!}} \prod_{j=1}^{m^-} \frac{(B_{q_j}^{\dagger})^{k_j^-}}{\sqrt{k_j^-!}} \Psi_{\text{vac}}[\phi,\phi^{\dagger}].$$
(2.165)

The action of the charge operator on a wave-functional immediately follows from that of the fields:

$$Q\Psi[\phi,\phi^{\dagger}] = \int_{0}^{L} \mathrm{d}x \, \left(\phi^{\dagger}(x)\frac{\delta}{\delta\phi^{\dagger}(x)} - \phi(x)\frac{\delta}{\delta\phi(x)}\right)\Psi[\phi,\phi^{\dagger}],\tag{2.166}$$

and in particular one finds that the vacuum functional and the functionals of multi-particle states are charge eigenstates:

$$Q\Psi_{\text{vac}} = 0 \quad , \quad Q\Psi_{\{p_i,q_j\}}^{k^+,k^-} = (k^+ - k^-)\Psi_{\{p_i,q_j\}}^{k^+,k^-}.$$
(2.167)

However, because $|\phi, \phi^{\dagger}\rangle$ is not associated to any charged state in the Fock space, it is not a charge eigenstate. The exponential of the charge operator acts on $|\phi, \phi^{\dagger}\rangle$ by introducing phases (notice the minus sign in front of the integral):

$$e^{2\pi i\alpha Q} |\phi, \phi^{\dagger}\rangle = \exp\left[-2\pi i\alpha \int_{0}^{L} \mathrm{d}x \left(\phi^{\dagger}(x) \frac{\delta}{\delta \phi^{\dagger}(x)} - \phi(x) \frac{\delta}{\delta \phi(x)}\right)\right] |\phi, \phi^{\dagger}\rangle = |e^{2\pi i\alpha} \phi, e^{-2\pi i\alpha} \phi^{\dagger}\rangle.$$
(2.168)

Employing these results it is possible to show (see Appendix 2.D for the derivation) that

$$\operatorname{Tr}_{A}(\rho_{0,A}^{n}e^{2\pi i\alpha Q_{A}}) = \int \mathcal{D}\phi_{1}\mathcal{D}\phi_{1}^{\dagger}\dots\mathcal{D}\phi_{n}\mathcal{D}\phi_{n}^{\dagger}\exp\left[-\mathcal{G}_{\alpha}\right], \qquad (2.169)$$

where \mathcal{G}_{α} is a known Gaussian functional of the fields $\phi(x), \phi^{\dagger}(x)$ given in (2.248). Results for the harmonic chain can then be obtained by discretisation, as we see in the next Subsection.

2.5.2.2 The harmonic chain

Since the Hamiltonian (2.146) reduces to the sum of two Hamiltonians for the real bosons Φ_1 , Φ_2 with prefactors $\frac{1}{2}$, the discretisation proceeds exactly as for the real boson [115]. We divide the interval [0, L] in N parts by introducing a spacing:

$$\Delta x = \frac{L}{N},\tag{2.170}$$

and we define $x = \frac{L}{N}\bar{x}, \bar{x} \in \{0, 1, \dots, N-1\}$, so that we can replace every integral with a sum:

$$\int_{A\cup\bar{A}} \mathrm{d}x \to \frac{L}{N} \sum_{x=0}^{L-\Delta x}, \quad \int_{A} \mathrm{d}x \to \frac{L}{N} \sum_{x=0}^{\ell-\Delta x}, \quad \int_{\bar{A}} \mathrm{d}x \to \frac{L}{N} \sum_{x=\ell}^{L-\Delta x}.$$
 (2.171)

If we discretise the Laplace operator as:

$$\partial_x^2 \Phi(x) \rightarrow \frac{\Phi(x + \Delta x) + \Phi(x - \Delta x) - 2\Phi(x)}{(\Delta x)^2},$$
(2.172)

and impose periodic boundary conditions $\Phi(0) = \Phi(L)$, $\Phi^{\dagger}(0) = \Phi^{\dagger}(L)$, the Hamiltonian (2.146) reduces (after integration by parts) to two independent harmonic chains for real fields. The set of

momenta is now restricted to the first Brillouin zone, $p = \frac{2\pi}{L}\bar{p}, \bar{p} \in \{0, 1, \dots, N-1\}$, and the dispersion relation becomes:

$$E_p = \sqrt{m^2 + \left(\frac{2N}{L}\sin\frac{pL}{2N}\right)^2},\tag{2.173}$$

from which the relativistic relation $E_p^2 = m^2 + p^2$ is obtained when $\frac{pL}{2N} \ll 1$. Notice that since we restrict the set of momenta, the function K(x) defined in (2.161) becomes a finite sum:

$$K(x) = \frac{1}{L} \sum_{p=0}^{2\pi(N-1)/L} E_p e^{ipx},$$
(2.174)

thus it is no longer an even function of x, though the property $K^*(x) = K(-x)$ still holds.

For the sake of simplicity we will take Φ and Φ^{\dagger} to be the fundamental degrees of freedom in the following, while keeping in mind that the real degrees of freedom are recovered using (2.148). In the formula (2.251) the functions U_i^{\pm} , V_i^{\pm} are modified by simply replacing the integrals with sums following the prescription (2.171). On the other hand, discretisation of the measure \mathcal{G}_{α} leads to a finite-dimensional $(nN) \times (nN)$ matrix G which couples the fields $\phi_i^{\dagger}(x)$ and $\phi_j(y)$:

$$\mathcal{G}_{\alpha} = \left(\frac{L}{N}\right)^{2} \sum_{i=1}^{n} \left[2 \left(\sum_{x \in A, y \in A} + \sum_{x \in \bar{A}, y \in \bar{A}} \right) \phi_{i}^{\dagger}(x) K(x-y) \phi_{i}(y) \right. \\ \left. + \sum_{x \in A, y \in \bar{A}} \left(\phi_{i}^{\dagger}(x) + \phi_{i+1}^{\dagger}(x) e^{-2\pi i \alpha \delta_{i,n}} \right) K(x-y) \phi_{i}(y) \right. \\ \left. + \sum_{x \in \bar{A}, y \in A} \phi_{i}^{\dagger}(x) K(x-y) \left(\phi_{i}(y) + \phi_{i+1}(y) e^{2\pi i \alpha \delta_{i,n}} \right) \right] =: \sum_{i,j=1}^{n} \sum_{x,y=0}^{L} \phi_{i}^{\dagger}(x) G_{ix,jy} \phi_{j}(y), \quad (2.175)$$

where we explicitly wrote the complex conjugate in (2.248) before discretising the integrals. Wick's theorem ensures that the Gaussian average in (2.251) can be computed from the contractions of pairs of fields, which are in turn obtained via the inversion of the matrix G:

$$\phi_i(x)^{\dagger}\phi_j(y) = (G^{-1})_{ix,jy}.$$
 (2.176)

The matrix G has a block structure, consisting of n^2 blocks $G_{i,j}$, each of which is an $N \times N$ matrix. From the above expression we see that the only non-vanishing blocks are either in the diagonal $G_{i,i}$ or just off the diagonal, $G_{i,(i\pm 1)}$. Each block G admits a sub-block structure in terms of the matrices $K_{Q_1Q_2}, Q_1, Q_2 \in \{A, \bar{A}\}$, whose elements are:

$$(K_{Q_1Q_2})_{xy} = \left(\frac{L}{N}\right)^2 K(x-y), \quad x \in Q_1, y \in Q_2.$$
 (2.177)

Notice that K_{AA} is a square matrix with dimensions $\frac{\ell}{L}N \times \frac{\ell}{L}N$, $K_{A\bar{A}}$ has dimensions $\frac{\ell}{L}N \times \frac{L-\ell}{L}N$

and so on. From (2.175) we obtain the following basic structures:

• Diagonal blocks $G_{i \cdot , i}$.

$$\begin{split} &\left(\sum_{x \in A, y \in A} + \sum_{x \in \bar{A}, y \in \bar{A}} + \sum_{x \in \bar{A}, y \in \bar{A}} + \sum_{x \in A, y \in \bar{A}} + \sum_{x \in \bar{A}, y \in A}\right) \phi_i^{\dagger}(x) G_{ix,iy} \phi_i(y) \\ &= \left(\frac{L}{N}\right)^2 \left[2 \sum_{\substack{x \in A \\ y \in A}} \phi_i^{\dagger}(x) K(x-y) \phi_i(y) + 2 \sum_{\substack{x \in \bar{A} \\ y \in \bar{A}}} \phi_i^{\dagger}(x) K(x-y) \phi_i(y) \right] \\ &+ \sum_{x \in A, y \in \bar{A}} \phi_i^{\dagger}(x) K(x-y) \phi_i(y) + \sum_{x \in \bar{A}, y \in A} \phi_i^{\dagger}(x) K(x-y) \phi_i(y) \right] \\ &\Rightarrow G_{i\cdot,i\cdot} = \begin{pmatrix}2K_{AA} & K_{A\bar{A}} \\ K_{\bar{A}A} & 2K_{\bar{A}\bar{A}}\end{pmatrix}. \end{split}$$

• Off-diagonal blocks $G_{i,(i+1)}$.

$$\sum_{x \in \bar{A}, y \in A} \phi_i^{\dagger}(x) G_{ix,(i+1)y} \phi_{i+1}(y) = \left(\frac{L}{N}\right)^2 \sum_{x \in \bar{A}, y \in A} \phi_i^{\dagger}(x) K(x-y) \phi_{i+1}(y) e^{2\pi i \alpha \delta_{i,n}}$$
$$\Rightarrow G_{i \cdot,(i+1) \cdot} = \begin{pmatrix} 0 & 0 \\ K_{\bar{A}A} e^{2\pi i \alpha \delta_{i,n}} & 0 \end{pmatrix}.$$

• Off-diagonal blocks $G_{(i+1)\cdot,i}$.

$$\begin{split} &\sum_{x \in A, y \in \bar{A}} \phi_{i+1}^{\dagger}(x) G_{(i+1)x, iy} \phi_i(y) = \left(\frac{L}{N}\right)^2 \sum_{x \in A, y \in \bar{A}} \phi_{i+1}^{\dagger}(x) K(x-y) \phi_i(y) e^{-2\pi i \alpha \delta_{i,n}} \\ &\Rightarrow G_{(i+1)\cdot, i\cdot} = \begin{pmatrix} 0 & K_{A\bar{A}} e^{-2\pi i \alpha \delta_{i,n}} \\ 0 & 0 \end{pmatrix}. \end{split}$$

Note that the block structure is different from that in [115] because the roles of regions A and \overline{A} are now exchanged. Although this exchange has no effect on the form of the entanglement entropy, for the SREE it makes a difference as the symmetry between A, \overline{A} is broken when we choose to place the charge in subsystem A.

In terms of the $N \times N$ blocks above, we can schematically write the matrix G as follows:

		1		2			<i>n</i> -1		n	
		A	Ā	A	Ā		A	Ā	A	Ā
		$\overbrace{}$	$\overbrace{}$	$\overline{}$	\sim	I I				$\overbrace{}$
1 · 2 ·	$\int A \Big\{$	$2K_{AA}$	$K_{A\bar{A}}$	0			0		0	$\lambda_{\alpha}^* K_{A\bar{A}}$
	$\left(\bar{A} \right)$	K _{ĀA}	$2K_{ar{A}ar{A}}$	$K_{\bar{A}A}$	0		0	0	0	0
	$\int A \bigg\{$	0	$K_{A\bar{A}}$	$2K_{AA}$	$K_{A\bar{A}}$		0		0	0
	$\Big] \bar{A} \Big\{$	0	0	$K_{\bar{A}A}$	$2K_{ar{A}ar{A}}$		0	0	0	0
		:			:	·		:		
n-1	$\int A \left\{ \right.$	0	0	0	0		$2K_{AA}$	$K_{A\bar{A}}$	0	0
	$\left \bar{A} \right $	0	0	0	0	•••	$K_{\bar{A}A}$	$2K_{ar{A}ar{A}}$	$K_{\bar{A}A}$	0
	$\int A \bigg\{$	0	0	0	0		0	$K_{A\bar{A}}$	$2K_{AA}$	$K_{Aar{A}}$
	\bar{A}	$\lambda_{\alpha}K_{\bar{A}A}$	0	0	0		0		$K_{\bar{A}A}$	$2K_{ar{A}ar{A}}$

where we introduced $\lambda_{\alpha} = e^{2\pi i \alpha}$.

2.5.2.3 Numerical results

In Appendix 2.D we have explicitly derived the ratio of charged moments for excited states. We thus have all the ingredients needed to obtain numerical results. Let us take a bi-partition where A is a segment made of $N_A \leq N$ consecutive sites, with $N_A/N = r$ and analyse the behaviour of $M_n^{\Psi}(r; \alpha)$.

In Fig. 2.3 we compare results for two kinds of two-particle excited states: those of particles with identical charges and either distinct or equal momenta p_1 and p_2 . Our analytical predictions for $M_n(r; \alpha)$ are

$$M_n^{1^{+1^{+}}}(r;\alpha) = (r^n e^{2i\pi\alpha} + (1-r)^n)^2, \quad p_1 \neq p_2,$$

$$M_n^{2^{+}}(r;\alpha) = r^{2n} e^{4\pi i\alpha} + 2^n (1-r)^n r^n e^{2\pi i\alpha} + (1-r)^{2n}, \quad p_1 = p_2.$$
(2.178)

In our numerics we have chosen L = N = 30, so that the lattice spacing L/N = 1. We also fix the mass scale to m = 0.1, which corresponds to a typical correlation length of $\xi = m^{-1} = 10$ sites. Finally we choose either $p_1 = p_2 = \pi$ or $p_1 = \pi$ and $p_2 = \frac{2\pi}{5}$, both in units of the lattice spacing.

Similarly, Fig. 2.4, we consider the following three-particle excited states: a state of three equal momenta, that is $p_1 = p_2 = p_3$, a state of two equal momenta among the three, that is $p_1 = p_2 \neq p_3$



Figure 2.3 Charged moments of two-particle excited states in the 1D harmonic chain Numerical data (triangles) versus analytical predictions (dashed lines) for $M_2^{1^+1^+}(r;\alpha)$ (top row) and $M_2^{2^+}(r;\alpha)$ (bottom row). We consider n = 2, system size L = 30 with m = 0.1. The left/right panels in each row show the real/imaginary part of the function. In both rows we take values of the flux $\alpha = 0, 0.1, \ldots 0.5$. For the numerics of the top row figures we use momenta $p_1 = \pi, p_2 = 2\pi/5$ whereas for the bottom row we take equal momenta $p_1 = p_2 = \pi$.

 p_3 , and a state with three distinct momenta, that is p_1, p_2, p_3 distinct. In this case the analytical predictions are

$$M_n^{3^+}(r;\alpha) = r^{3n}e^{6i\pi\alpha} + 3^n r^{2n}(1-r)^n e^{4i\pi\alpha} + 3^n r^n(1-r)^{2n}e^{2i\pi\alpha} + (1-r)^{3n}, \quad p_1 = p_2 = p_3, \quad (2.179)$$

and

$$M_n^{2^{+1^+}}(r;\alpha) = M_n^{2^+}(r;\alpha)(r^n e^{2i\pi\alpha} + (1-r)^n), \quad p_1 = p_2 \neq p_3, \tag{2.180}$$

$$M_n^{1^+1^+}(r;\alpha) = (r^n e^{2i\pi\alpha} + (1-r)^n)^3, \quad p_1 \neq p_2 \neq p_3.$$
(2.181)

The set of momenta is $p_1 = p_2 = p_3 = \pi$ for the first excited state, $p_1 = p_2 = \pi, p_3 = \pi/3$ for the second state, and $p_1 = \pi, p_2 = \pi/3, p_3 = \pi/5$ for the third one.

In all our figures we chose non-negative values of α . Given the formulae above, taking $\alpha < 0$ is equivalent to complex conjugation with α positive, so the figures for negative α are identical except for a change of sign in the imaginary part of all functions. We have also considered the value $\alpha = 0$ (in



Figure 2.4 Numerical data (triangles) versus analytical predictions (dashed lines) for $M_2^{3^+}(r; \alpha)$ (top row), $M_2^{2^+1+}(r; \alpha)$ (central row), and $M_2^{1^+1+1^+}(r; \alpha)$ (bottom row). We consider n = 2, system size L = 30 with m = 0.1. The left/right panels in each row show the real/imaginary part of the function. In each rows we take values of the flux $\alpha = 0, 0.1, \ldots 0.5$. For the numerics of the top row figures we use momenta $p_1 = p_2 = p_3 = \pi$, for the central row $p_1 = p_2 = \pi, p_3 = \pi/3$, whereas for the bottom row we take $p_1 = \pi, p_2 = \pi/3, p_3 = \pi/5$.

green) which is the limit where there is no flux. As expected, in this case our formulae recover those for the excess Rényi Entropies in [114, 115], which are symmetric in r and have vanishing imaginary part. Despite the fact that the correlation length is not particularly small with respect to the system size L ($\xi = m^{-1} = \frac{L}{3}$), we took highly energetic states (momenta being fixed in the large-volume limit) and we thus expect our predictions to remain valid.

In both Fig. 2.3 and 2.4, we plot the numerical data (triangles) against analytical predictions

((2.178) and (2.181)) as functions of r fixing n = 2 for several values of α between 0 and $\frac{1}{2}$, which correspond to flux ± 1 , respectively. At these two points, the ratio becomes purely real. The figures show excellent agreement between numerical data and analytical predictions.

2.6 Concluding remarks

In this chapter we have computed the symmetry-resolved entanglement entropy and its moments for zero-density excited states. These are excited states consisting of a finite number of excitations above the ground state in a scaling limit where both the volume of the system and the volume of each subsystem are taken to infinity, keeping their ratio constant.

It is known from previous work [114–116, 144–150] that the difference between the entanglement entropy of the excited state and that of the ground state, also known as excess entropy, takes an extremely simple and universal form for non-interacting 1+1D QFTs and also for certain highly excited states of CFT [104]. Since this excess of entanglement represents the extra contribution to entanglement of an excited state above a non-trivially entangled ground state, the same extra contribution is obtained when the ground state is trivial. For this reason both a free QFT and a qubit picture lead to the same results, even if the underlying theories are extremely different. Finally, it has also been shown that the results extend to free bosons in any dimension [116] and more generally they are expected to hold in any situations where excitations are localised. Here, we extended the work summarised above to the computation of excess symmetry-resolved Rényi entropies and entanglement entropies in excited states of theories possessing a U(1) symmetry. The ratio of charged moments between the excited and ground states takes the same universal form in the 1+1Dmassive Dirac fermion and 1+1D complex free boson. By employing the form factor program for composite branch-point twist fields, we computed the ratio of charged moments and from the latter we obtained exact expressions for the SREE of the excited states. Interestingly, these expressions can be written solely in terms of the SREE and symmetry-resolved partition function of the ground state. As we expected, the results obtained in free 1+1D QFTs are recovered using the qubit picture also in the symmetry-resolved case.

The greatest novelty of the work presented in this chapter consists in showing that our results, i.e. the formulae for the excess of SREE, apply much more broadly than the form factor computation would suggest: namely, they apply to any localised excitations of interacting and higher dimensional theories with U(1) symmetry as long as the correlation length is much smaller than the typical subsystem size. As a first, simple example we considered one- and two-magnon states of quantum spin chains, with and without interaction. Their entanglement entropies can be computed analytically via free fermion techniques and we showed that, irrespective of interaction, the formulae presented in Section 2.1 still apply, in line with observations made in [115, 150]. Furthermore, we proposed a general method to compute the charged moments of zero-density excited states of QFTs in higher dimensions with or without interactions. Our method is based on the notion of algebra of local observables of a theory, and on the commutation relations between composite twist operators -generalisations of CBPTFs to higher dimensions- and the local observables. The formulae already obtained for free 1+1D theories can then be derived in full generality by making few natural assumptions on the OPE of observables in the theory and on the correlation length of the latter.

Finally, we performed numerical tests of our results. For this, we considered a 1D Fermi gas and a complex harmonic chain. Although both the models are discrete and amenable to numerical computations, their microscopic features are quite different. Whereas the 1D Fermi gas is a fermionic theory which possesses a gapless phase, the complex harmonic chain is a gapped theory which becomes the complex massive free boson in continuous limit. It is therefore quite remarkable that the same set of formulae for the ratios of charged moments apply for both theories. This is nonetheless the case. Whereas for the 1D Fermi gas our formulae hold for highly excited states containing excitations of large momenta, thus small De Broglie wavelengths, for the complex harmonic charge the formulae hold as long as the correlation length is small compared to subsystem size.

In the following chapter, we will extend our investigation of zero-density excited states to another symmetry-resolved measure of entanglement, the logarithmic negativity [23, 116, 136]. In the conclusion of that chapter, we will also discuss some research directions that naturally emerge from our work.

2.A Complex free boson computation

In this appendix we present the form factor computation of the ratio of charged moments in detail, focusing on the complex free boson theory.

2.A.1 Single-particle excited states

The two-point function in (2.29) can be computed by inserting a sum over a complete set of states between the U(1) fields as follows:

$${}^{n}_{L}\langle 1^{+}|\mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell)|1^{+}\rangle_{L}^{n} = \sum_{\{N^{+}\}}\sum_{\{M^{+}\}}A_{n}^{*}(\{N^{+}\})A_{n}(\{M^{+}\})\prod_{p=1}^{n}\sum_{m^{\pm}=0}^{\infty}\sum_{\{J^{\pm}\}}\prod_{j=1}^{m^{+}}\prod_{r=1}^{m^{-}}\frac{1}{m^{+}!m^{-}!}$$

$$\times_{p,L}^{n}\langle 0|[\mathfrak{a}_{p}(\theta)]^{N_{p}^{+}}\mathcal{T}_{p+\alpha}(0)\mathfrak{a}_{p}^{\dagger}(\theta_{j}^{+})\mathfrak{b}_{p}^{\dagger}(\theta_{r}^{-})|0\rangle_{p,L}^{n}\times \sum_{p,L}^{n}\langle 0|\mathfrak{a}_{p}(\theta_{j}^{+})\mathfrak{b}_{p}(\theta_{r}^{-})\mathcal{T}_{-p-\alpha}(\ell)[\mathfrak{a}_{p}^{\dagger}(\theta)]^{M_{p}^{+}}|0\rangle_{p,L}^{n},$$

$$(2.182)$$

and similarly for the $|1^-\rangle_L^n$ case. As shown in [115] following [159, 160], the finite-volume matrix elements involved in the expression above differ from the corresponding infinite-volume form factors for a term $\mathcal{O}(e^{-\mu L})$, with μ a characteristic mass scale, in this case $\mu = m$. Thus, we can rewrite the previous expression up to exponentially decaying corrections as

$${}^{n}_{L} \langle 1^{+} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{+} \rangle_{L}^{n} = \sum_{\{N^{+}\}} \sum_{\{M^{+}\}} A_{n}^{*}(\{N^{+}\}) A_{n}(\{M^{+}\}) \prod_{p=1}^{n} \sum_{m^{\pm}=0}^{\infty} \sum_{\{J^{\pm}\}} \frac{1}{m^{+}! m^{-}!}$$
(2.183)

$$\times \frac{e^{i\ell \left(\sum_{j=1}^{m^{+}} P(\theta_{j}^{+}) + \sum_{r=1}^{m^{-}} P(\theta_{r}^{-}) - M_{p}^{+} P(\theta)\right)}}{\sqrt{LE(\theta)}^{N_{p}^{+} + M_{p}^{+}} \prod_{j=1}^{m^{+}} LE(\theta_{j}^{+}) \prod_{r=1}^{m^{-}} LE(\theta_{r}^{-})} F_{N_{p}^{+} + m^{+} + m^{-}}^{p+\alpha,n}(\theta_{1}^{+} \dots \theta_{m^{+}}^{+}, \hat{\theta}, \dots \hat{\theta}, \theta_{1}^{-} \dots \theta_{m^{-}}^{-})$$
$$\times F_{M_{p}^{+} + m^{+} + m^{-}}^{n-\rho-\alpha,n}(\theta \dots \theta, \hat{\theta}_{1}^{-} \dots \hat{\theta}_{m^{-}}^{-}, \hat{\theta}_{1}^{+} \dots \hat{\theta}_{m^{+}}^{+}),$$

being $\hat{\theta}_j^{\pm} = \theta_j^{\pm} + i\pi$, and $E(\theta)$, $P(\theta)$ are the one-particle energy and momentum. The momenta are quantised according to the usual Bethe-Yang condition or to equation (2.31) for the intermediate rapidities θ_j^{\pm} . The complete formula for the form factors above was given in [115] and they can be fully expressed as sums of products of two-particle form factors. They are non-vanishing for $N_p^+ = M_p^+ = m^+ - m^-$ and zero otherwise.

If the same intermediate rapidity θ_j^+ is paired up (in the Wick-contraction sense) with the rapidity of the excited state θ from the in- an out-states, the dominant contribution in the form factor product will come from kinematic poles. In other words, if $\theta_j^+ \simeq \theta$ two-particle form factors will appear as follows:

$$\begin{split} F^{p+\alpha,n}_{N^+_p+m^++m^-}(\theta^+_1\dots\theta^+_{m^+},\hat\theta\dots\hat\theta,\dots) &\simeq N^+_p f^n_{p+\alpha}(\theta^+_j-\hat\theta) \\ &\times F^{p+\alpha,n}_{N^+_p+m^++m^--2}(\theta^+_1\dots\theta^+_{j-1}\theta^+_{j+1}\dots\theta^+_{m^+},\hat\theta\dots\hat\theta\dots) \\ F^{n-p-\alpha,n}_{M^+_p+m^++m^-}(\theta\dots\theta,\dots\hat\theta^+_1\dots\hat\theta^+_{m^+}) &\simeq M^+_p f^n_{n-(p+\alpha)}(\hat\theta-\theta^+_j) \\ &\times F^{n-p-\alpha,n}_{M^+_p+m^++m^--2}(\theta\dots\theta,\dots\hat\theta^+_1\dots\hat\theta^+_{j-1}\hat\theta^+_{j+1}\dots\hat\theta^+_{m^+}), \end{split}$$

where the number of rapidities $\hat{\theta}$ (θ) in the arguments of the form factors in the right-hand side are now $N_p^+ - 1$ ($M_p^+ - 1$). The main property of the matrix elements in (2.182) in determining the final formula for (2.29) is the infinite-volume limit of the terms such as

$$\sum_{J^{+}\in\mathbb{Z}} \frac{f_{p+\alpha}^{n}(\theta^{+}-\hat{\theta})f_{n-p-\alpha}^{n}(\theta-\hat{\theta}^{+})e^{i\ell(P(\theta^{+})-P(\theta))}}{\cosh\theta\cosh\theta^{+}} \simeq (mL)^{2}\sum_{J_{i}^{+}\in\mathbb{Z}} \frac{\sin^{2}\frac{\pi(p+\alpha)}{n}}{\pi^{2}} \frac{e^{2\pi ir(J^{+}-I+\frac{p+\alpha}{n})}}{(J^{+}-I+\frac{p+\alpha}{n})^{2}} = (mL)^{2}g_{p+\alpha}^{n}(r),$$
(2.184)

with $g_{p+\alpha}^n(r)$ the functions defined in (2.41) and the indices J^+ , I are integers resulting from the quantisation conditions of the rapidities of intermediate states (2.31) and of the rapidity of the

physical one-particle state respectively. We can proceed in an analogous way for (2.30) obtaining $(mL)^2 g^n_{-(p+\alpha)}(r)$ as the leading contribution.

Once all possible contractions with a rapidity of the excited in- and out- state have been carried out, the leading large-volume contribution from the summation over the quantum number J^+ is of order $\mathcal{O}(L^0)$ and comes from terms with N = M, as shown in Appendix B of [115]. This contribution reads

$$\sum_{L}^{n} \langle 1^{+} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{+} \rangle_{L}^{n} = \sum_{\{N^{+}\}} |A_{n}(\{N^{+}\})|^{2} \prod_{p=1}^{n} N_{p}^{+} ! \left[g_{p+\alpha}^{n}(r)\right]^{N_{p}^{+}} \prod_{q^{+}=0}^{\infty} \prod_{m^{-}=0}^{\infty} \frac{1}{q^{+}! m^{-}!} \sum_{\{J^{\pm}\} \in \mathbb{Z}} \frac{e^{i\ell \left(\sum_{j=1}^{q^{+}} P(\theta_{j}^{+}) + \sum_{r=1}^{m^{-}} P(\theta_{r}^{-})\right)}}{\prod_{j=1}^{q^{+}} L^{2} E(\theta_{j}^{+}) \prod_{r=1}^{m^{-}} L^{2} E(\theta_{r}^{-})} F_{q^{+}+m^{-}}^{p,n}(\theta_{1}^{+} \dots \theta_{q^{+}}^{+}, \theta_{1}^{-} \dots \theta_{m^{-}}^{-}) F_{q^{+}+m^{-}}^{n-p,n}(\hat{\theta}_{1}^{+} \dots \hat{\theta}_{q^{+}}^{+}, \hat{\theta}_{1}^{-} \dots \hat{\theta}_{m^{-}}^{-})}$$

with $q^+ = m^+ - N_p^+$. Dividing by the finite-volume vacuum two-point function:

$$\prod_{p=1}^{n} \prod_{p,L}^{n} \langle 0 | \mathcal{T}_{p+\alpha}(0) \mathcal{T}_{-p-\alpha}(\ell) | 0 \rangle_{p,L}^{n}$$

we obtain the formula (2.40) for the ratio of moments of the SREE for a one excitation state. We stress that the approximations made in considering the leading contributions become exact if the large-volume limit (2.1) is taken while keeping the mass m fixed.

2.A.2 Free boson (k = 1, n = 2)

In this Section we work out an example in detail. Consider a single particle excited state consisting of a complex boson excitation above the ground state. The relevant state is

$$(a_{1}^{+})^{\dagger}(\theta)(a_{2}^{+})^{\dagger}(\theta)|0\rangle_{L}^{2} = \frac{1}{2}(-\mathfrak{a}_{1}^{\dagger}(\theta) + \mathfrak{a}_{2}^{\dagger}(\theta))(\mathfrak{a}_{1}^{\dagger}(\theta) + \mathfrak{a}_{2}^{\dagger}(\theta))|0\rangle_{L}^{2},$$
(2.185)

and comparing this expression to (2.27) we obtain $A_2(2,0) = -A_2(0,2) = -\frac{1}{2}$. Thus

$$M_n^{1^+}(r;\alpha) = \frac{2!}{4} (g_{1+\alpha}^2(r)^2 + g_{2+\alpha}^2(r)^2)$$

$$= \frac{1}{2} \left(1 - r + re^{\pi i(1+\alpha)} \right)^2 + \frac{1}{2} \left(1 - r + re^{\pi i(2+\alpha)} \right)^2$$

$$= \frac{1}{2} ((1-r)^2 + r^2 e^{2\pi i\alpha} - 2r(1-r)e^{i\pi\alpha} + (1-r)^2 + r^2 e^{2\pi i\alpha} + 2r(1-r)e^{i\pi\alpha})$$

$$= (1-r)^2 + e^{2\pi i\alpha} r^2.$$
(2.186)

2.A.3 Multi-particle excited states

Below, we describe in detail the computation of the ratio of moments of the SREE for a state consisting of k particle excitations with equal rapidities and charge signs. These states have the form:

$$|k^{\pm}\rangle_{L}^{n} = \prod_{j=1}^{n} (a_{j}^{\pm})^{\dagger}(\theta)|0\rangle_{L}^{n} = \frac{1}{(\sqrt{k!})^{n}} \sum_{\{N^{\pm}\}} D_{n}(\{N^{\pm}\}) \prod_{p=1}^{n} \left[\left(\tilde{a}_{p}^{\pm}\right)^{\dagger}(\theta) \right]^{N_{p}^{\pm}} |0\rangle_{L}^{n},$$
(2.187)

where in the second equality we used the expression of the creation operators in the diagonal basis (2.25), with the convenient notation $(\tilde{a}_p^+)^{\dagger} = \mathfrak{a}_p^{\dagger}$, $(\tilde{a}_p^-)^{\dagger} = \mathfrak{b}_p^{\dagger}$. After inserting a the resolution of the identity between the twist fields, the two-point function reads:

$${}^{n}_{L}\langle k^{\pm}|\mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell)|k^{\pm}\rangle_{L}^{n} = \frac{1}{(k!)^{n}}\sum_{\{N^{\pm}\}}\sum_{\{M^{\pm}\}}D_{n}(\{N^{+}\}) D_{n}^{*}(\{M^{+}\})\prod_{p=1}^{n}\sum_{m^{\pm}=0}^{\infty}\sum_{\{J^{\pm}\}}\prod_{j=1}^{m^{+}}\prod_{r=1}^{m^{-}}\frac{1}{m^{+}!m^{-}!}$$

$$\times{}^{n}_{p,L}\langle 0|[\tilde{a}_{p}^{\pm}(\theta)]^{N_{p}^{\pm}}\mathcal{T}_{p+\alpha}(0)\mathfrak{a}_{p}^{\dagger}(\theta_{j}^{+})\mathfrak{b}_{p}^{\dagger}(\theta_{r}^{-})|0\rangle_{p,L}^{n}}\sum_{p,L}\langle 0|\mathfrak{a}_{p}(\theta_{j}^{+})\mathfrak{b}_{p}(\theta_{r}^{-})\mathcal{T}_{-p-\alpha}(\ell)[(\tilde{a}_{p}^{\pm})^{\dagger}(\theta)]^{M_{p}^{\pm}}|0\rangle_{p,L}^{n},$$

$$(2.188)$$

Employing the relation between these matrix elements and the finite-volume form factors, together with the action of the translation operator, we get:

$$\begin{split} & {}_{L}^{n} \langle k^{\pm} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | k^{\pm} \rangle_{L}^{n} = \frac{1}{(k!)^{n}} \sum_{\{N^{+}\}} \sum_{\{M^{+}\}} D_{n}(\{N^{+}\}) D_{n}^{*}(\{M^{+}\}) \prod_{p=1}^{n} \sum_{m^{\pm}=0}^{\infty} \sum_{\{J^{\pm}\}} \frac{1}{m^{+}! m^{-}!} \\ & \times \frac{e^{i\ell \left(\sum_{j=1}^{m^{+}} P(\theta_{j}^{+}) + \sum_{r=1}^{m^{-}} P(\theta_{r}^{-}) - M_{p}^{\pm} P(\theta)\right)}}{\sqrt{LE(\theta)^{N_{p}^{\pm} + M_{p}^{\pm}} \prod_{j=1}^{m^{+}} LE(\theta_{j}^{+}) \prod_{r=1}^{m^{-}} LE(\theta_{r}^{-})}} F_{N_{p}^{\pm} + m^{+} + m^{-}}^{n,p}(\theta_{1}^{+} \dots \theta_{m^{+}}^{+}, \hat{\theta}, \dots \hat{\theta}, \theta_{1}^{-} \dots \theta_{m^{-}}^{-})} \\ & \times F_{M_{p}^{+} + m^{+} + m^{-}}^{n-p,n}(\theta \dots \theta, \hat{\theta}_{1}^{-} \dots \hat{\theta}_{m^{-}}^{-}, \hat{\theta}_{1}^{+} \dots \hat{\theta}_{m^{+}}^{+}). \end{split}$$

Once all possible intermediate rapidities have been paired up with the same rapidity of the excited state in both form factors and the contribution of the ground state has been factored, the leading large-volume contribution of the ratio of moments can be written as:

$$M_n^{k^{\pm}}(r;\alpha) = \frac{1}{(k!)^n} \sum_{\{N^{\pm}\}} |D_n(\{N^{\pm}\})|^2 \prod_{p=1}^n N_p^{\pm}! \left(g_{\pm(p+\alpha)}^n(r)\right)^{N_p^{\pm}}$$
$$= \sum_{p=0}^k \left[\binom{k}{p} r^p (1-r)^{k-p}\right]^n e^{\pm 2\pi i \alpha p}.$$
(2.189)

On the other hand, if the k-particle excitations have distinct rapidities there can be two different cases:

- k-particle excitations with distinct rapidities but equal charge sign
- k-particle excitations with distinct rapidities and charge sign

We can summarize the computations for both cases if we consider the following ansatz for the excited state:

$$|1^{\epsilon_1} 1^{\epsilon_2} \dots 1^{\epsilon_k}\rangle_L^n = \prod_{i=1}^k \prod_{j=1}^n (a_j^{\epsilon_i})^{\dagger}(\theta_i)|0\rangle_L^n = \sum_{\{N^{\pm}\}} C_n(\{N^{\pm}\}) \prod_{s=1}^k \prod_{p=1}^n [\mathfrak{a}_p^{\dagger}(\theta_s)]^{N_{p,s}^+} [\mathfrak{b}_p^{\dagger}(\theta_s)]^{N_{p,s}^-} |0\rangle_L^n, \quad (2.190)$$

where each ϵ_i with i = 1, ..., k can be either + or -. If we consider k-particle excitations with distinct rapidities but equal charge sign + (-) then all the ϵ_i are the same sign and $N_{p,s}^-$ ($N_{p,s}^+$) vanish. When pairing up the intermediate rapidities with the same rapidity of the excited state in both form factors, the conditions for the matrix elements to be non vanishing are:

$$N_{p,s}^{-} + m^{+} = N_{p,s}^{+} + m^{-},$$

$$m^{-} + M_{p,s}^{+} = m^{+} + M_{p,s}^{-}.$$

Subtracting the contribution of the ground state, the leading large-volume contribution to the ratio of moments can be written as:

$$M_n^{1^{\epsilon_1}1^{\epsilon_2}\dots1^{\epsilon_k}}(r;\alpha) = \prod_{s=1}^k \left[\sum_{\{N^{\pm}\}} |C_n(\{N^{\pm}\})|^2 \prod_{p=1}^n N_{p,s}^{+}! \ N_{p,s}^{-}! \ [g_{p+\alpha}^n(r)]^{N_{p,s}^+} \ [g_{-p-\alpha}^n(r)]^{N_{p,s}^-} \right].$$
(2.191)

Notice that if we study k-particle excitations with distinct rapidities but equal charge sign, the above expression reduces to:

$$M_n^{1^{\pm}1^{\pm}\dots1^{\pm}}(r;\alpha) = \prod_{s=1}^k \left[\sum_{\{N^{\pm}\}} |C_n(\{N^{\pm}\})|^2 \prod_{p=1}^n N_{p,s}^{\pm}! \; [g_{\pm(p+\alpha)}^n(r)]^{N_{p,s}^{\pm}} \right],\tag{2.192}$$

with the conditions $m^{\mp} + M_{p,s}^{\pm} = m^{\pm} = N_{p,s}^{\pm} + m^{\mp}$ for the \pm sign state.

2.A.4 Free boson (k = 2, n = 2)

As an example, consider the following two-particle excited states with distinct rapidities:

$$|1^{+}1^{+}\rangle_{L}^{2} = (a_{1}^{+})^{\dagger}(\theta_{1})(a_{1}^{+})^{\dagger}(\theta_{2})(a_{2}^{+})^{\dagger}(\theta_{1})(a_{2}^{+})^{\dagger}(\theta_{2})|0\rangle_{L}^{2} = \frac{1}{4} \left([\mathfrak{a}_{1}^{\dagger}(\theta_{1})]^{2} [\mathfrak{a}_{1}^{\dagger}(\theta_{2})]^{2} + [\mathfrak{a}_{2}^{\dagger}(\theta_{1})]^{2} [\mathfrak{a}_{2}^{\dagger}(\theta_{2})]^{2} - [\mathfrak{a}_{1}^{\dagger}(\theta_{2})]^{2} [\mathfrak{a}_{2}^{\dagger}(\theta_{1})]^{2} - [\mathfrak{a}_{1}^{\dagger}(\theta_{1})]^{2} [\mathfrak{a}_{2}^{\dagger}(\theta_{2})]^{2} \right) |0\rangle_{L}^{2}, \qquad (2.193)$$

$$|1^{+}1^{-}\rangle_{L}^{2} = (a_{1}^{+})^{\dagger}(\theta_{1})(a_{1}^{-})^{\dagger}(\theta_{2})(a_{2}^{+})^{\dagger}(\theta_{1})(a_{2}^{-})^{\dagger}(\theta_{2})|0\rangle_{L}^{2} = \frac{1}{4} \left([\mathfrak{a}_{1}^{\dagger}(\theta_{1})]^{2} \, [\mathfrak{b}_{1}^{\dagger}(\theta_{2})]^{2} + [\mathfrak{a}_{2}^{\dagger}(\theta_{1})]^{2} \, [\mathfrak{b}_{2}^{\dagger}(\theta_{2})]^{2} - [\mathfrak{a}_{2}^{\dagger}(\theta_{1})]^{2} \, [\mathfrak{b}_{1}^{\dagger}(\theta_{2})]^{2} - [\mathfrak{a}_{1}^{\dagger}(\theta_{1})]^{2} \, [\mathfrak{b}_{2}^{\dagger}(\theta_{2})]^{2} \right) |0\rangle_{L}^{2}.$$

$$(2.194)$$

The charged ratios for these excited states are given by

$$M_{2}^{1+1+}(r;\alpha) = \frac{1}{4} \left([g_{1+\alpha}^{2}(r)]^{4} + [g_{2+\alpha}^{2}(r)]^{4} + 2 [g_{1+\alpha}^{2}(r)]^{2} [g_{2+\alpha}^{2}(r)]^{2}) \right)$$

$$= \left((1-r)^{2} + r^{2} e^{2\pi i \alpha} \right)^{2}, \qquad (2.195)$$

$$M_{2}^{1+1-}(r;\alpha) = \frac{1}{4} \left([g_{1+\alpha}^{2}(r)]^{2} [g_{-1-\alpha}^{2}(r)]^{2} + [g_{2+\alpha}^{2}(r)]^{2} [g_{-2-\alpha}^{2}(r)]^{2} + [g_{1+\alpha}^{2}(r)]^{2} [g_{-2-\alpha}^{2}(r)]^{2} \right)$$

$$+ \left[g_{-1-\alpha}^{2}(r) \right]^{2} [g_{2+\alpha}^{2}(r)]^{2} \right) = \left((1-r)^{2} + r^{2} e^{2\pi i \alpha} \right) \left((1-r)^{2} + r^{2} e^{-2\pi i \alpha} \right). \qquad (2.196)$$

On the other hand, if we consider ta two-particle excited state with coinciding rapidities:

$$|2^{-}\rangle_{L}^{2} = [(a_{1}^{-})^{\dagger}(\theta)]^{2} [(a_{2}^{-})^{\dagger}(\theta)]^{2} |0\rangle_{L}^{2}$$

$$= \frac{1}{8} \left([\mathfrak{b}_{1}^{\dagger}(\theta)]^{4} + [\mathfrak{b}_{2}^{\dagger}(\theta)]^{4} - 2[\mathfrak{b}_{1}^{\dagger}(\theta)]^{2} [\mathfrak{b}_{2}^{\dagger}(\theta)]^{2} \right) |0\rangle_{L}^{2}, \qquad (2.197)$$

the charged moment is

$$M_2^{2^-}(r;\alpha) = \frac{1}{8^2} (4! [g_{-1-\alpha}^2(r)]^4 + 4! [g_{-2-\alpha}^2(r)]^4 + 4^2 [g_{-1-\alpha}^2(r)]^2 [g_{-2-\alpha}^2(r)]^2)$$

= $(1-r)^4 + 4r^2(1-r)^2 e^{-2\pi i\alpha} + e^{-4\pi i\alpha} r^4.$ (2.198)

2.B Complex free fermion computation

In this appendix we present the form factor computation of the ratio of charged moments in detail, focusing on the complex free fermion theory.

2.B.1 Single-particle excited states

Below, we present the explicit computation of the fermionic two-point function in an excited state consisting of a single positively-charged particle. Thanks to the factorisation (2.22), the latter can be cast as:

$${}_{L}^{n} \langle 1^{+} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{+} \rangle_{L}^{n}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} {}_{p,L}^{n} \langle 0 | \mathfrak{a}_{p} \left(\theta \right) \mathcal{T}_{p+\alpha}(0) \mathcal{T}_{-p-\alpha}(\ell) \mathfrak{a}_{p}^{\dagger}(\theta) | 0 \rangle_{p,L}^{n}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} {}_{s=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{1}{s!(s+1)!} {}_{p,L}^{n} \langle 0 | \mathfrak{a}_{p} \left(\theta \right) \mathcal{T}_{p+\alpha}(0) \mathfrak{a}_{p}^{\dagger}(\theta_{1}) \dots \mathfrak{a}_{p}^{\dagger}(\theta_{s+1}) \mathfrak{b}_{p}^{\dagger}(\theta_{s+2}) \dots \mathfrak{b}_{p}^{\dagger}(\theta_{2s+1}) | 0 \rangle_{p,L}^{n}$$

$$\times {}_{p,L}^{n} \langle 0 | \mathfrak{a}_{p} \left(\theta_{1} \right) \dots \mathfrak{a}_{p}(\theta_{s+1}) \mathfrak{b}_{p}(\theta_{s+2}) \dots \mathfrak{b}_{p}(\theta_{2s+1}) \mathcal{T}_{-p-\alpha}(0) \mathfrak{a}_{p}^{\dagger}(\theta) | 0 \rangle_{p,L}^{n} e^{i\ell(\sum_{i=1}^{2s+1} P(\theta_{i}) - P(\theta))}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} {}_{s=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{|F_{2s+2}^{p+\alpha,n}(\theta_{1}, \dots, \theta_{s+1}; \theta + i\pi, \theta_{s+2}, \dots, \theta_{2s+1})|^{2}}{s!(s+1)!LE(\theta) \prod_{i=1}^{2s+1}(\theta_{i})LE(\theta_{i})} e^{i\ell(\sum_{i=1}^{2s+1} P(\theta_{i}) - P(\theta))}, \quad (2.199)$$

where the resolution of the identity is inserted in such a way as to preserve the total charge of the one-particle state and the Bethe quantum numbers $\{J_i^{\pm}\}$ are defined as in (2.31). Notice that since the excitations are fermionic, one could either have $J_i^{\pm} \in \mathbb{Z}$ or $J_i^{\pm} \in \mathbb{Z} + \frac{1}{2}$: for the sake of simplicity, we consider the case where these numbers are integer, as this assumption does not make a difference in the final result.

The non-vanishing contributions in the large-volume limit come from the terms in the previous expression in which the rapidity of the excited state is contracted with θ_i , $i = 1, \ldots, s + 1$ in both form factors. The s + 1 possible contractions in $F_{2s+2}^{p+\alpha,n}$ give rise to:

$$F_{2s+2}^{p+\alpha,n}(\theta_1, \dots, \theta_{s+1}; \hat{\theta}, \theta_{s+2}, \dots, \theta_{2s+1}) \simeq f_{p+\alpha}^n(\theta_i - \hat{\theta}) F_{2s}^{p+\alpha,n}(\theta_1, \dots, \check{\theta}_i, \dots, \theta_{s+1}; \theta_{s+2}, \dots, \theta_{2s+1}),$$
(2.200)

where around the pole:

$$f_{p+\alpha}^{n}(\theta_{i}-\hat{\theta}) \underset{\theta\simeq\theta_{i}}{\simeq} \frac{mL\sin\frac{\pi(p+\alpha)}{n}\cosh\theta \, e^{\frac{i\pi(p+\alpha)}{n}}}{\pi(J_{i}^{+}-I+\frac{p+\alpha}{n})}.$$
(2.201)

Thus, considering also the contraction coming from the form factor of $\mathcal{T}_{-p-\alpha}$, we can separately perform the s + 1 summations over the quantum numbers J_i^+ as

$$\sum_{\substack{J_i^+ \in \mathbb{Z} \\ \theta \simeq \theta_i}} \frac{|f_{p+\alpha}^n(\theta_i - \hat{\theta})|^2 e^{i\ell(P(\theta_i) - P(\theta))}}{Lm \cosh \theta Lm \cosh \theta_i}$$
$$\underset{\theta \simeq \theta_i}{\simeq} \sum_{\substack{J_i^+ \in \mathbb{Z} \\ \pi^2(J_i^+ - I + \frac{p+\alpha}{n})^2}} \frac{\sin^2 \frac{\pi(p+\alpha)}{n} e^{2\pi i r \left(J_i^+ - I + \frac{p+\alpha}{n}\right)}}{\pi^2 (J_i^+ - I + \frac{p+\alpha}{n})^2} = g_{p+\alpha}^n(r).$$
(2.202)

We therefore obtain, in the limit (2.1) and for fixed m:

$${}^{n}_{L}\langle 1^{+} | \mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{+} \rangle_{L}^{n}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{p+\alpha}^{n}(r) \sum_{s=0}^{\infty} \frac{1}{(s!)^{2}} \sum_{\{J_{i}^{\pm}\}} |F_{2s}^{p+\alpha,n}(\theta_{1},\ldots,\theta_{s};\beta_{1},\ldots,\beta_{s})|^{2} \frac{e^{i\ell \sum_{i=1}^{s} (P(\theta_{i})+P(\beta_{i}))}}{\prod_{i=1}^{s} LE(\theta_{i})LE(\beta_{i})}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{p+\alpha}^{n}(r) \times_{p,L}^{n} \langle 0 | \mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell) | 0 \rangle_{p,L}^{n},$$
(2.203)

where we re-labeled the rapidities of the negatively charged intermediate states as $\beta_i = \theta_{s+i+1}$ for i = 1, ..., s. We can now make use of (2.53) in the evaluation of the ratio, so that we finally obtain

for the free fermionic one-particle states:

$$M_n^{1^+}(r;\alpha) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{p+\alpha}^n(r) = (1-r)^n + e^{2\pi i \alpha} r^n.$$
(2.204)

An analogous result can be obtained for a negatively charged particle, where the phase above picks up an extra minus sign.

2.B.2 Multi-particle excited states

The anticommuting nature of the creation/annihilation operators allows us to obtain an exact expression for the ratio of the charged moments in the fermionic case, which (unlike for the free boson) does not require a case-by-case calculation. This is because in the free fermion case, the structure of the states in the transformed base is extremely simple, as we shall see. We have non-vanishing two-point functions only with two kind of states:

- k-particle excitations with distinct rapidities, irrespective of the charge signs;
- 2-particle excitations with equal rapidities and different charge signs: $|1^+1^-\rangle_L^n$.

Below, we consider in detail the case of k-particle states with distinct rapidities. Such states are written exactly as in the bosonic case:

$$|1^{\epsilon_1} 1^{\epsilon_2} \dots 1^{\epsilon_k}\rangle_L^n = \prod_{i=1}^k \prod_{j=1}^n (a_j^{\epsilon_i})^{\dagger}(\theta_i) |0\rangle_L^n,$$
(2.205)

where $\epsilon_i = \pm 1$, $\theta_i \neq \theta_{i'}$ if $i \neq i'$. Unlike the bosonic case, however, all the operators anticommute, so that we can make the ansatz:

$$|1^{\epsilon_1} 1^{\epsilon_2} \dots 1^{\epsilon_k}\rangle_L^n = e^{i\kappa} \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \prod_{i=1}^k (\mathfrak{a}_p^{\epsilon_i})^{\dagger}(\theta_i) |0\rangle_L^n, \qquad (2.206)$$

with the identification $(\mathfrak{a}_p^+)^{\dagger}(\theta_i) = \mathfrak{a}_p^{\dagger}(\theta_i)$, $(\mathfrak{a}_p^-)^{\dagger}(\theta_i) = \mathfrak{b}_p^{\dagger}(\theta_i)$ and the only unspecified parameter is the phase $\kappa = \kappa(k, n; \{\epsilon_i\})$. Notice that the order of the operators in the double product can be arbitrarily altered, resulting only in a change in the phase. Without giving a full proof of the validity of this formula, let us consider a few simple cases and introduce the notations k^{\pm} to indicate the number of positively/negatively charged excitations in the state, with $k = k^+ + k^-$:

• $n = 2, k^+ = 2$:

$$\begin{split} |1^{+}1^{+}\rangle_{L}^{2} &= \prod_{i=1}^{2} \prod_{j=1}^{2} \frac{1}{\sqrt{2}} \sum_{p=-\frac{1}{2}}^{\frac{1}{2}} e^{-\frac{2\pi i j p}{2}} \mathfrak{a}_{p}^{\dagger}(\theta_{i}) |0\rangle_{L}^{2} \\ &= \frac{1}{4} \prod_{i=1}^{2} (i\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta_{i}) - i\mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta_{i})) (-\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta_{i}) - \mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta_{i})) |0\rangle_{L}^{2} \\ &= \frac{1}{4} \prod_{i=1}^{2} (-2i\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta_{i})\mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta_{i})) |0\rangle_{L}^{2} = -\prod_{p=-\frac{1}{2}}^{\frac{1}{2}} \prod_{i=1}^{2} \mathfrak{a}_{p}^{\dagger}(\theta_{i}) |0\rangle_{L}^{2} \end{split}$$

•
$$n = 2, k^+ = 2, k^- = 1$$
:

$$\begin{split} |1^{+}1^{+}1^{-}\rangle_{L}^{2} &= \left(\prod_{i=1}^{2}\prod_{j=1}^{2}\frac{1}{\sqrt{2}}\sum_{p=-\frac{1}{2}}^{\frac{1}{2}}e^{-\frac{2\pi i j p}{2}}\mathfrak{a}_{p}^{\dagger}(\theta_{i})\right)\prod_{j=1}^{2}\frac{1}{\sqrt{2}}\sum_{p=-\frac{1}{2}}^{\frac{1}{2}}e^{\frac{2\pi i j p}{2}}\mathfrak{b}_{p}^{\dagger}(\theta_{3})|0\rangle_{L}^{2} \\ &= \left(\frac{1}{4}\prod_{i=1}^{2}(-2i\mathfrak{a}_{-\frac{1}{2}}^{\dagger}(\theta_{i})\mathfrak{a}_{\frac{1}{2}}^{\dagger}(\theta_{i}))\right)\frac{1}{2}(2i\mathfrak{b}_{-\frac{1}{2}}^{\dagger}(\theta_{3})\mathfrak{b}_{\frac{1}{2}}^{\dagger}(\theta_{3}))|0\rangle_{L}^{2} \\ &= -i\prod_{p=-\frac{1}{2}}^{\frac{1}{2}}\left(\prod_{i=1}^{2}\mathfrak{a}_{p}^{\dagger}(\theta_{i})\right)\mathfrak{b}_{p}^{\dagger}(\theta_{3})|0\rangle_{L}^{2} \end{split}$$

• $n = 3, k^+ = 2$:

$$\begin{split} |1^{+}1^{+}\rangle_{L}^{3} &= \prod_{i=1}^{2} \prod_{j=1}^{3} \frac{1}{\sqrt{3}} \sum_{p=-1}^{1} e^{-\frac{2\pi i j p}{3}} \mathfrak{a}_{p}^{\dagger}(\theta_{i}) |0\rangle_{L}^{3} \\ &= \prod_{i=1}^{2} \frac{1}{3^{\frac{3}{2}}} \left(e^{\frac{2\pi i}{3}} \mathfrak{a}_{-1}^{\dagger}(\theta_{i}) + \mathfrak{a}_{0}^{\dagger}(\theta_{i}) + e^{-\frac{2\pi i}{3}} \mathfrak{a}_{1}^{\dagger}(\theta_{i}) \right) \\ &\times \left(e^{\frac{4\pi i}{3}} \mathfrak{a}_{-1}^{\dagger}(\theta_{i}) + \mathfrak{a}_{0}^{\dagger}(\theta_{i}) + e^{-\frac{4\pi i}{3}} \mathfrak{a}_{1}^{\dagger}(\theta_{i}) \right) \left(\mathfrak{a}_{-1}^{\dagger}(\theta_{i}) + \mathfrak{a}_{0}^{\dagger}(\theta_{i}) + \mathfrak{a}_{1}^{\dagger}(\theta_{i}) \right) |0\rangle_{L}^{3} \\ &= -\prod_{p=-1}^{1} \prod_{i=1}^{2} \mathfrak{a}_{p}^{\dagger}(\theta_{i}) |0\rangle_{L}^{3}. \end{split}$$

For a fixed value of n, the structure of more complicated states can be easily worked out following these simple examples. Using equation (2.206) and the twist field factorisation (2.22), the two-point

function reads:

$${}^{n}_{L} \langle 1^{\epsilon_{1}} 1^{\epsilon_{2}} \dots 1^{\epsilon_{k}} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{\epsilon_{1}} 1^{\epsilon_{2}} \dots 1^{\epsilon_{k}} \rangle_{L}^{n}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \prod_{p,L}^{n} \langle 0 | \mathfrak{a}_{p}(\theta_{1}) \dots \mathfrak{a}_{p}(\theta_{k}+) \mathfrak{b}_{p}(\beta_{1}) \dots \mathfrak{b}_{p}(\beta_{k}-) \mathcal{T}_{p+\alpha}(0) \mathcal{T}_{-p-\alpha}(\ell) \mathfrak{a}_{p}^{\dagger}(\theta_{1}) \dots \mathfrak{a}_{p}^{\dagger}(\theta_{k}+) \mathfrak{b}_{p}^{\dagger}(\beta_{1}) \dots \mathfrak{b}_{p}^{\dagger}(\beta_{k}-) | 0 \rangle_{p,L}^{n}$$

$$= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \sum_{s=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{1}{s!(s+q)!} e^{i\ell \left(\sum_{i=1}^{q+s} P(\tilde{\theta}_{i}) + \sum_{i=1}^{s} P(\tilde{\beta}_{i}) - \sum_{i=1}^{k+1} P(\theta_{i}) - \sum_{i=1}^{k-1} P(\beta_{i})\right)}$$

$$\times \prod_{p,L}^{n} \langle 0 | \mathfrak{a}_{p}(\theta_{1}) \dots \mathfrak{a}_{p}(\theta_{k}^{+}) \mathfrak{b}_{p}(\beta_{1}) \dots \mathfrak{b}_{p}(\beta_{k}-) \mathcal{T}_{p+\alpha}(0) \mathfrak{a}_{p}^{\dagger}(\tilde{\theta}_{1}) \dots \mathfrak{a}_{p}^{\dagger}(\tilde{\theta}_{q+s}) \mathfrak{b}_{p}^{\dagger}(\tilde{\beta}_{1}) \dots \mathfrak{a}_{p}^{\dagger}(\tilde{\beta}_{s}) | 0 \rangle_{p,L}^{n}$$

$$\times \prod_{p,L}^{n} \langle 0 | \mathfrak{a}_{p}(\tilde{\theta}_{1}) \dots \mathfrak{a}_{p}(\tilde{\theta}_{q+s}) \mathfrak{b}_{p}(\tilde{\beta}_{1}) \dots \mathfrak{b}_{p}(\tilde{\beta}_{s}) \mathcal{T}_{-p-\alpha}(0) \mathfrak{a}_{p}^{\dagger}(\theta_{1}) \dots \mathfrak{a}_{p}^{\dagger}(\theta_{k}+) \mathfrak{b}_{p}^{\dagger}(\beta_{1}) \dots \mathfrak{b}_{p}^{\dagger}(\beta_{k-}) | 0 \rangle_{p,L}^{n}.$$

$$(2.207)$$

In the expansion above we assumed the total charge of the excited state to be positive, $q := k^+ - k^- > 0$. However, the computation steps are unchanged if one assumes q < 0, the only difference being in the structure of the resolution of the identity. Denoting $\hat{x} := x + i\pi$, the infinite-volume form factor corresponding to the first matrix element reads:

$$F_{q+2s+k}^{p+\alpha,n}(\hat{\theta}_1,\ldots,\hat{\theta}_{q+s},\hat{\beta}_1,\ldots,\hat{\beta}_{k-};\hat{\beta}_1,\ldots,\hat{\beta}_s,\hat{\theta}_1,\ldots,\hat{\theta}_{k+}),$$
(2.208)

where the total charge conservation is ensured by the equality $q + s + k^- = s + k^+$. When turning to the finite-volume, one needs to divide the previous infinite-volume form factor by a quantity:

$$\left[\prod_{i=1}^{q+s} LE(\tilde{\theta}_i) \prod_{i=1}^{k^-} LE(\beta_i) \prod_{i=1}^{s} LE(\tilde{\beta}_i) \prod_{i=1}^{k^+} LE(\theta_i)\right]^{\frac{1}{2}}.$$
 (2.209)

Taking into account also the contribution coming from the other form factor, this results into a factor of order $\mathcal{O}(L^{-q-2s-k})$ for every term in the expansion (2.207), and the latter reads:

$$\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \sum_{s=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{1}{s!(s+q)!} e^{i\ell \left(\sum_{i=1}^{q+s} P(\tilde{\theta}_{i}) + \sum_{i=1}^{s} P(\tilde{\beta}_{i}) - \sum_{i=1}^{k^{+}} P(\theta_{i}) - \sum_{i=1}^{k^{-}} P(\beta_{i})\right)} \times \frac{|F_{q+2s+k}^{p+\alpha,n}(\tilde{\theta}_{1},\ldots,\tilde{\theta}_{q+s},\hat{\beta}_{1},\ldots,\hat{\beta}_{k-};\tilde{\beta}_{1},\ldots,\tilde{\beta}_{s},\hat{\theta}_{1},\ldots,\hat{\theta}_{k+})|^{2}}{\prod_{i=1}^{q+s} LE(\tilde{\theta}_{i})\prod_{i=1}^{k^{-}} LE(\beta_{i})\prod_{i=1}^{s} LE(\tilde{\beta}_{i})\prod_{i=1}^{k^{+}} LE(\theta_{i})}.$$
(2.210)

In the large-volume limit, the leading contributions are those coming from simultaneous contractions in both form factors. In turn, in each form factor the simple poles arise from the pairings of the rapidities $\hat{\theta}_i$ with $\tilde{\theta}_i$ (these are at most k^+ contractions) and from the pairings of the rapidities $\hat{\beta}_i$ with $\tilde{\beta}_i$ (these are at most k^- contractions). Again, these pairings have to be made simultaneously. The terms with $s < k^-$ (or equivalently $q + s < k^+$) do not contribute in this limit, as they contain some extra factors of L in the denominator. On the other hand, the terms with $s > k^-$ contain a sum of $k^+!k^-!$ products of the form:

$$|f_{p+\alpha}^n(\tilde{\theta}_i - \hat{\theta}_j)f_{p+\alpha}^n(\hat{\beta}_{i'} - \tilde{\beta}_{j'})|^2 \times \text{residual form factors.}$$

By making use of (2.201), the simultaneous expansion around the poles leads to the following leading contribution:

$$\sum_{J_{i}^{+},J_{j'}^{-}} \frac{|f_{p+\alpha}^{n}(\tilde{\theta}_{i}-\hat{\theta}_{j})f_{p+\alpha}^{n}(\hat{\beta}_{i'}-\tilde{\beta}_{j'})|^{2}}{\cosh\tilde{\theta}_{i}\cosh\theta_{i}\cosh\theta_{i}\cosh\beta_{i'}\cosh\tilde{\beta}_{j'}} e^{i\ell[P(\tilde{\theta}_{i})-P(\theta_{j})+P(\tilde{\beta}_{j'})-P(\beta_{i'})]}$$

$$\simeq (mL)^{4} \sum_{J_{i}^{+}\in\mathbb{Z}} \frac{\sin^{2}\frac{\pi(p+\alpha)}{n}}{\pi^{2}} \frac{e^{2\pi i r(J_{i}^{+}-I_{j}+\frac{p+\alpha}{n})}}{(J_{i}^{+}-I_{j}+\frac{p+\alpha}{n})^{2}} \sum_{J_{j'}^{-}\in\mathbb{Z}} \frac{\sin^{2}\frac{\pi(p+\alpha)}{n}}{\pi^{2}} \frac{e^{2\pi i r(J_{j'}^{-}-I_{i'}-\frac{p+\alpha}{n})}}{(J_{j'}^{-}-I_{i'}-\frac{p+\alpha}{n})^{2}}$$

$$= (mL)^{4} g_{p+\alpha}^{n}(r) g_{-p-\alpha}^{n}(r).$$
(2.211)

Since each of the $k^+!k^-!$ terms in the sum contains exactly k^+ functions $g_{p+\alpha}^n$ and k^- functions $g_{-p-\alpha}^n$ we have, relabelling $m = s - k^-$:

$$\frac{{}_{L}^{n} \langle 1^{\epsilon_{1}} 1^{\epsilon_{2}} \dots 1^{\epsilon_{k}} | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 1^{\epsilon_{1}} 1^{\epsilon_{2}} \dots 1^{\epsilon_{k}} \rangle_{L}^{n}}{= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \sum_{m=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{1}{(m!)^{2}} (g_{p+\alpha}^{n}(r))^{k^{+}} (g_{-p-\alpha}^{n}(r))^{k^{-}}}{\times e^{i\ell \sum_{i=1}^{m} (P(\tilde{\theta}_{i}) + P(\tilde{\beta}_{i}))} \frac{|F_{2m}^{p+\alpha,n}(\tilde{\theta}_{1}, \dots, \tilde{\theta}_{m}; \tilde{\beta}_{1}, \dots, \tilde{\beta}_{m})|^{2}}{\prod_{i=1}^{m} LE(\tilde{\theta}_{i}) LE(\tilde{\beta}_{i})}} = \left(\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{p+\alpha}^{n}(r)\right)^{k^{+}} \left(\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} g_{-p-\alpha}^{n}(r)\right)^{k^{-}} \frac{n}{L} \langle 0|\mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell)|0\rangle_{L}^{n}, \quad (2.212)$$

and therefore, thanks to (2.22), the ratio of charged moments is:

$$M_n^{1^{\pm}1^{\pm}\dots1^{\pm}}(r;\alpha) = ((1-r)^n + e^{2\pi i\alpha}r^n)^{k^+}((1-r)^n + e^{-2\pi i\alpha}r^n)^{k^-}.$$
 (2.213)

2.C Finite-volume two-point function in the ground state

In this appendix we investigate the large-volume expansion of the correlator

$${}_{L}^{n}\langle 0|\mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(r\,L)|0\rangle_{L}^{n},\tag{2.214}$$

which is the denominator of the ratio of charged moments $M_n^{\Psi}(r; \alpha)$ in (2.18). We show that, as expected, the leading contribution in the limit (2.1) is given by the squared modulus of the vacuum expectation value of the composite twist field \mathcal{T}^{α} , and we compute the first finite-volume corrections to this quantity. The calculations are carried out in the fermionic case, but they apply to the free boson case with few changes.

The first step in the evaluation of (2.214) is as usual the insertion of a projection onto asymptotic states:

$$\frac{n}{L} \langle 0 | \mathcal{T}^{\alpha}(0) \tilde{\mathcal{T}}^{\alpha}(\ell) | 0 \rangle_{L}^{n} = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \sum_{p,L}^{n} \langle 0 | \mathcal{T}_{p+\alpha}(0) \mathcal{T}_{-p-\alpha}(\ell) | 0 \rangle_{p,L}^{n} \\
= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \sum_{s=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{1}{(s!)^{2}} \sum_{p,L}^{n} \langle 0 | \mathcal{T}_{p+\alpha}(0) \mathfrak{a}_{p}^{\dagger}(\theta_{1}) \dots \mathfrak{a}_{p}^{\dagger}(\theta_{s}) \mathfrak{b}_{p}^{\dagger}(\beta_{1}) \dots \mathfrak{b}_{p}^{\dagger}(\beta_{s}) | 0 \rangle_{p,L}^{n} \\
\times \sum_{p,L}^{n} \langle 0 | \mathfrak{a}_{p}(\theta_{1}) \dots \mathfrak{a}_{p}(\theta_{s}) \mathfrak{b}_{p}(\beta_{1}) \dots \mathfrak{b}_{p}(\beta_{s}) \mathcal{T}_{-p-\alpha}(0) | 0 \rangle_{p,L}^{n} e^{i\ell \sum_{i=1}^{s} (P(\theta_{i}) + P(\beta_{i}))} \\
= \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \sum_{s=0}^{\infty} \sum_{\{J_{i}^{\pm}\}} \frac{|F_{2s}^{p+\alpha,n}(\theta_{1},\dots,\theta_{s};\beta_{1},\dots,\beta_{s})|^{2}}{(s!)^{2} \prod_{i=1}^{s} L^{2} E(\theta_{i}) E(\beta_{i})} e^{i\ell \sum_{i=1}^{s} (P(\theta_{i}) + P(\beta_{i}))}. \quad (2.215)$$

Notice that the equal number of particles of the two types is dictated by the fact that twist fields preserve the total charge of the state. In a free fermion theory, the infinite-volume form factor of an even number of particle is given by Wick's theorem:

$$F_{2s}^{p+\alpha,n}(\theta_1,\ldots,\theta_s;\beta_1,\ldots,\beta_s) = \sum_{\sigma\in\mathcal{P}_s} \tau_{p+\alpha} \operatorname{sgn} \sigma \prod_{i=1}^s f_{p+\alpha}^n(\theta_{\sigma(i)} - \beta_i), \qquad (2.216)$$

where the normalised fermionic two-particle form factor is obtained diving (2.45) by the VEV $\tau_{p+\alpha}$. Thus, the squared modulus of the 2*s*-particle form factor is a sum of $(s!)^2$ terms, each of which is a product of *s* terms of the type

$$f_{p+\alpha}^{n}(\theta_{i}-\beta_{j})f_{p+\alpha}^{n}(\theta_{k}-\beta_{j})^{*} = \sin^{2}\frac{\pi(p+\alpha)}{n}\frac{e^{\frac{p+\alpha}{n}(\theta_{i}+\theta_{k}-2\beta_{j})}}{\cosh\frac{\theta_{i}-\beta_{j}}{2}\cosh\frac{\theta_{k}-\beta_{j}}{2}}.$$
(2.217)

We can therefore explicitly rewrite (2.215) as:

$${}^{n}_{L}\langle 0|\mathcal{T}^{\alpha}(0)\tilde{\mathcal{T}}^{\alpha}(\ell)|0\rangle_{L}^{n} = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} |\tau_{p+\alpha}|^{2} \sum_{s=0}^{\infty} \frac{\sin^{2s} \frac{\pi(p+\alpha)}{n}}{(s!)^{2} (mL)^{2s}} \sum_{\{J_{i}^{\pm}\}} \sum_{\sigma,\omega\in\mathcal{P}_{s}} \operatorname{sgn}\sigma \operatorname{sgn}\omega$$
$$\times \prod_{i=1}^{s} \frac{e^{i\ell(P(\theta_{i})+P(\beta_{i}))}}{\cosh\theta_{i}\cosh\beta_{i}} \frac{e^{\frac{p+\alpha}{n}(\theta_{\sigma(i)}-\theta_{i})-2\beta_{i}}}{\cosh\frac{\theta_{\sigma(i)}-\beta_{i}}{2}\cosh\frac{\theta_{\omega(i)}-\beta_{i}}{2}}.$$
(2.218)

From this expression we easily see that the vacuum expectation value is corrected by contributions of multi-particle states, and that in general every 2s-particle state (containing s particles with positive charge and s particles with negative charge) contributes with a leading large-volume term $\mathcal{O}((mL)^{-2s})$. Further corrections can be obtained by working out how the product in the second line of (2.218) depends on L. This is done by solving the Bethe equations for $\sinh \theta_i$ and $\sinh \beta_i$:

$$\sinh \theta_i = \frac{2\pi \left(J_i^+ + \frac{p+\alpha}{n}\right)}{mL} := \frac{c_i^+}{mL},\tag{2.219}$$

$$\sinh \beta_i = \frac{2\pi \left(J_i^- - \frac{p+\alpha}{n}\right)}{mL} := \frac{c_i^-}{mL}.$$
(2.220)

Where either $J_i^+, J_i^- \in \mathbb{Z}$ or $J_i^+, J_i^- \in \mathbb{Z} + \frac{1}{2}$. Let us consider in detail the expansion up to s = 1 terms, assuming that $J_1^+, J_1^- \in \mathbb{Z}$. Some elementary algebra shows that:

$$\frac{e^{2(\theta_1 - \beta_1)\frac{p + \alpha}{n}}}{\cosh^2 \frac{\theta_1 - \beta_1}{2}} = 1 + 2\left(\frac{p + \alpha}{n}\right)\frac{c_1^+ - c_1^-}{mL} + \mathcal{O}\left(\frac{1}{(mL)^2}\right)$$
(2.221)

and:

$$\frac{e^{i\ell(P(\theta_1)+P(\beta_1))}}{\cosh\theta_1\cosh\beta_1} = e^{2\pi i r (J_1^+ + J_1^-)} \left(1 - \frac{(c_1^+)^2 + (c_1^-)^2}{2(mL)^2} + \mathcal{O}\left(\frac{1}{(mL)^4}\right)\right).$$
(2.222)

Thus we have:

$$\prod_{p=-\frac{n-1}{2}}^{n} |\tau_{p+\alpha}|^2 \left[1 + \frac{\sin^2 \frac{\pi(p+\alpha)}{n}}{(mL)^2} \sum_{J_1^+, J_1^- \in \mathbb{Z}} e^{2\pi i r (J_1^+ + J_1^-)} \left(1 + 2\left(\frac{p+\alpha}{n}\right) \frac{c_1^+ - c_1^-}{mL} + \mathcal{O}\left(\frac{1}{(mL)^2}\right) \right) \right].$$
(2.223)

We immediately notice that there is no contribution of order $\mathcal{O}((mL)^{-2})$, as we can regularise the non convergent double sum by introducing two small real parameters ε , δ :

$$\sum_{\substack{J_1^+, J_1^- \in \mathbb{Z} \\ \delta \to 0}} e^{2\pi i r (J_1^+ + J_1^-)} = \lim_{\substack{\varepsilon \to 0 \\ \delta \to 0}} \left(-1 + \sum_{J \ge 0} e^{2\pi i J (r+i\varepsilon)} + \sum_{J \ge 0} e^{-2\pi i J (r-i\varepsilon)} \right) \left(-1 + \sum_{J \ge 0} e^{2\pi i J (r+i\delta)} + \sum_{J \ge 0} e^{-2\pi i J (r-i\delta)} \right) = 0.$$
(2.224)

The term of order $\mathcal{O}((mL)^{-3})$ can similarly be regularised to zero. Therefore, the first finite-volume correction to the ratio of moments of the SREE is (at least) of order $\mathcal{O}((mL)^{-4})$. However, at that order the computation becomes more involved as there are contributions coming from terms with s > 1 in (2.218).

2.D Trace calculations via wave-functional method

In this appendix we present explicit calculations of the ratio of charged moments for excited states in a complex free boson theory using the wave-functional method introduced in Section 2.5.2. The discretisation of these results leads to the formulae for the complex harmonic chain presented in Subsection 2.5.2.2.

2.D.1 Zero flux

We wish to compute the ratio of charged moments

$$\frac{\operatorname{Tr}_{A}\left(\rho_{A}^{n} e^{2\pi i \alpha Q_{A}}\right)}{\operatorname{Tr}_{A}\left(\rho_{0,A}^{n} e^{2\pi i \alpha Q_{A}}\right)},\tag{2.225}$$

where ρ_A and $\rho_{0,A}$ are the reduced density matrices of the excited and ground states, respectively.

We define restricted wave-functionals which take as arguments (complex) fields that either have support on region $A := [0, \ell)$ or on $\bar{A} := [\ell, L]$. Leaving the dependence on the conjugate fields ϕ^{\dagger} implicit, we write:

$$\Phi(x)\Psi[\phi_A,\phi'_{\bar{A}}] = \left(\delta_{x \in A}\phi(x) + \delta_{x \in \bar{A}}\phi'(x)\right)\Psi[\phi_A,\phi'_{\bar{A}}], \qquad (2.226)$$

$$i\Pi(x)\Psi[\phi_A,\phi'_{\bar{A}}] = \left(\delta_{x \in A}\frac{\delta}{\delta\phi(x)} + \delta_{x \in \bar{A}}\frac{\delta}{\delta\phi'(x)}\right)\Psi[\phi_A,\phi'_{\bar{A}}],\tag{2.227}$$

and similarly for the action of $\Phi^{\dagger}(x)$, $i\Pi^{\dagger}(x)$. The action of the charge operator \hat{Q}_A on the wave-functional is simply:

$$\hat{Q}_A \Psi[\phi_A, \phi'_{\bar{A}}] = \int_A \mathrm{d}x \, \left(\phi^\dagger(x) \frac{\delta}{\delta \phi^\dagger(x)} - \phi(x) \frac{\delta}{\delta \phi(x)}\right) \Psi[\phi_A, \phi'_{\bar{A}}], \tag{2.228}$$

while from (2.168) it follows that

$$e^{2\pi i\alpha \hat{Q}_A} |\phi_A, \phi'_{\bar{A}}\rangle = |e^{2\pi i\alpha} \phi_A, \phi'_{\bar{A}}\rangle.$$
(2.229)

The reduced density matrix then admits the functional representation

$$\rho_A = \operatorname{Tr}_{\bar{A}}\rho = \int \mathcal{D}\phi_{\bar{A}}\langle\phi_{\bar{A}}|\rho|\phi_{\bar{A}}\rangle \Rightarrow \langle\phi_A'|\rho_A|\phi_A''\rangle = \int \mathcal{D}\phi_{\bar{A}}\Psi[\phi_A',\phi_{\bar{A}}]\Psi[\phi_A'',\phi_{\bar{A}}]^*,$$
(2.230)

where $\mathcal{D}\phi$ is shorthand for $\mathcal{D}\phi\mathcal{D}\phi^{\dagger}$.

We start with the simple case where there is no symmetry resolution, that is, we compute (2.225)

for $\alpha = 0$, following [115]. For n integer, we insert the resolution of the identity n times and obtain:

$$\operatorname{Tr}_{A}(\rho_{0,A}^{n}) = \int \mathcal{D}\phi_{1A} \dots \mathcal{D}\phi_{nA} \langle \phi_{1A} | \rho_{\operatorname{vac},A} | \phi_{2A} \rangle \dots \langle \phi_{nA} | \rho_{\operatorname{vac},A} | \phi_{1A} \rangle$$
$$= \int \mathcal{D}\phi_{1} \dots \mathcal{D}\phi_{n} \prod_{i=1}^{n} \Psi_{\operatorname{vac}}[\phi_{i,A}, \phi_{i,\bar{A}}] \prod_{i=1}^{n} \Psi_{\operatorname{vac}}[\phi_{i+1,A}, \phi_{i,\bar{A}}]^{*}.$$
(2.231)

The product of the diagonal terms, i.e. those in which the fields in subsystems A and \overline{A} are in the same copy, is

$$\prod_{i=1}^{n} \Psi_{\text{vac}}[\phi_{i,A}, \phi_{i,\bar{A}}] = \exp\left[-\sum_{i=1}^{n} \int_{x,y \in A \cup \bar{A}} \mathrm{d}x \mathrm{d}y \,\phi_{i}^{\dagger}(x) K(x-y)\phi_{i}(y)\right],\tag{2.232}$$

with K(x) defined in (2.161). For the non-diagonal terms we notice that since $K(x)^* = K(-x)$ we have $\Psi_{\text{vac}}[\phi_{i+1,A}, \phi_{i,\bar{A}}]^* = \Psi_{\text{vac}}[\phi_{i+1,A}, \phi_{i,\bar{A}}]$ and:

$$\prod_{i=1}^{n} \Psi_{\text{vac}}[\phi_{i+1,A}, \phi_{i,\bar{A}}] = \exp\left\{-\sum_{i=1}^{n} \left[\int_{\substack{x \in A \\ y \in A}} \mathrm{d}x \mathrm{d}y \,\phi_{i+1}^{\dagger}(x) K(x-y) \phi_{i+1}(y)\right]$$
(2.233)

$$+ \int_{\substack{x \in \bar{A} \\ y \in \bar{A}}} \mathrm{d}x \mathrm{d}y \,\phi_i^{\dagger}(x) K(x-y)\phi_i(y) + \left(\int_{\substack{x \in A \\ y \in \bar{A}}} \mathrm{d}x \mathrm{d}y \,\phi_{i+1}^{\dagger}(x) K(x-y)\phi_i(y) + \mathrm{c.c.} \right) \right] \right\}.$$
(2.234)

Putting all the terms together, we end up with the Gaussian measure:

$$\operatorname{Tr}_{A}(\rho_{0,A}^{n}) = \int \mathcal{D}\phi_{1}\dots\mathcal{D}\phi_{n} \exp\left[-\mathcal{G}\right], \qquad (2.235)$$

where

$$\mathcal{G} = \sum_{i=1}^{n} 2 \left(\int_{\substack{x \in A \\ y \in \overline{A}}} + \int_{\substack{x \in \overline{A} \\ y \in \overline{A}}} \right) dx dy \, \phi_i^{\dagger}(x) K(x-y) \phi_i(y)$$

+
$$\sum_{i=1}^{n} \left(\int_{\substack{x \in A \\ y \in \overline{A}}} dx dy \, (\phi_i(x) + \phi_{i+1}(x))^{\dagger} K(x-y) \phi_i(y) + \text{c.c.} \right).$$
(2.236)

Let us now compute the numerator of (2.225). For a state of the form (2.164) we have:

$$\operatorname{Tr}_{A}(\rho_{A}^{n}) = \int \mathcal{D}\phi_{1} \dots \mathcal{D}\phi_{n} \prod_{i=1}^{n} \Psi_{\{p_{j+}, p_{j-}\}}[\phi_{i,A}, \phi_{i,\bar{A}}] \prod_{i=1}^{n} \Psi_{\{p_{j+}, p_{j-}\}}[\phi_{i+1,A}, \phi_{i,\bar{A}}]^{*},$$
(2.237)

with:

$$\prod_{i=1}^{n} \Psi_{\{p_{j^{+}}, p_{j^{-}}\}}[\phi_{i,A}, \phi_{i,\bar{A}}] = \left(\prod_{j^{+}} \frac{2E_{p_{j^{+}}}}{L} \prod_{j^{-}} \frac{2E_{p_{j^{-}}}}{L}\right)^{\frac{n}{2}} \prod_{i=1}^{n} \prod_{j^{+}} \int_{A \cup \bar{A}} \mathrm{d}x e^{ip_{j^{+}}x} \phi_{i}^{\dagger}(x) \\
\times \prod_{j^{-}} \int_{A \cup \bar{A}} \mathrm{d}x e^{ip_{j^{-}}x} \phi_{i}(x) \prod_{i=1}^{n} \Psi_{\mathrm{vac}}[\phi_{i,A}, \phi_{i,\bar{A}}],$$
(2.238)

$$\prod_{i=1}^{n} \Psi_{\{p_{j+},p_{j-}\}} [\phi_{i+1,A}, \phi_{i,\bar{A}}]^{*} = \left(\prod_{j+1} \frac{2E_{p_{j+1}}}{L} \prod_{j-1} \frac{2E_{p_{j-1}}}{L} \right)^{\frac{n}{2}} \times \prod_{i=1}^{n} \prod_{j+1} \left(\int_{A} \mathrm{d}x e^{-ip_{j+1}x} \phi_{i+1}(x) + \int_{\bar{A}} \mathrm{d}x e^{-ip_{j+1}x} \phi_{i}(x) \right) \times \prod_{j-1} \left(\int_{A} \mathrm{d}x e^{-ip_{j-1}x} \phi_{i+1}^{\dagger}(x) + \int_{\bar{A}} \mathrm{d}x e^{-ip_{j-1}x} \phi_{i}^{\dagger}(x) \right) \prod_{i=1}^{n} \Psi_{\mathrm{vac}} [\phi_{i+1,A}, \phi_{i,\bar{A}}].$$
(2.239)

Putting everything together, we obtain

$$\frac{\operatorname{Tr}_{A}(\rho_{A}^{n})}{\operatorname{Tr}_{A}(\rho_{0,A}^{n})} = \left(\prod_{j^{+}} \frac{2E_{p_{j^{+}}}}{L} \prod_{j^{-}} \frac{2E_{p_{j^{-}}}}{L}\right)^{n} \langle \prod_{i=1}^{n} \prod_{j^{+}} U_{i}^{+}(p_{j^{+}})V_{i}^{+}(p_{j^{+}}) \prod_{j^{-}} U_{i}^{-}(p_{j^{-}})V_{i}^{-}(p_{j^{-}})\rangle$$
(2.240)

where the correlation function is defined with respect to the Gaussian measure:

$$\langle \mathcal{O}[\phi_1, \phi_1^{\dagger} \dots, \phi_n, \phi_n^{\dagger}] \rangle := \frac{\int \mathcal{D}\phi_1 \mathcal{D}\phi_1^{\dagger} \dots \mathcal{D}\phi_n \mathcal{D}\phi_n^{\dagger} \mathcal{O}[\phi_1, \phi_1^{\dagger} \dots, \phi_n, \phi_n^{\dagger}] \exp\left[-\mathcal{G}\right]}{\int \mathcal{D}\phi_1 \mathcal{D}\phi_1^{\dagger} \dots \mathcal{D}\phi_n \mathcal{D}\phi_n^{\dagger} \exp\left[-\mathcal{G}\right]},$$
(2.241)

and the operators are

$$U_{i}^{+}(p) := \int_{A \cup \bar{A}} \mathrm{d}x \, e^{ipx} \phi_{i}^{\dagger}(x), \quad U_{i}^{-}(p) := \int_{A \cup \bar{A}} \mathrm{d}x \, e^{ipx} \phi_{i}(x) \tag{2.242}$$

$$V_i^+(p) := \int_A \mathrm{d}x \, e^{-ipx} \phi_{i+1}(x) + \int_{\bar{A}} \mathrm{d}x \, e^{-ipx} \phi_i(x), \tag{2.243}$$

$$V_i^{-}(p) := \int_A \mathrm{d}x \, e^{-ipx} \phi_{i+1}^{\dagger}(x) + \int_{\bar{A}} \mathrm{d}x \, e^{-ipx} \phi_i^{\dagger}(x). \tag{2.244}$$

If the excited state is of the form (2.165), the result (2.240) is minimally modified. The terms inside the correlator are exactly the same and the prefactor is modified to:

$$\left(\prod_{j^{+}=1}^{m^{+}} \frac{1}{k_{j^{+}}^{+}!} \left(\frac{2E_{p_{j^{+}}}}{L}\right)^{k_{j^{+}}^{+}} \prod_{j^{-}=1}^{m^{-}} \frac{1}{k_{j^{-}}^{-}!} \left(\frac{2E_{p_{j^{-}}}}{L}\right)^{k_{j^{-}}^{-}}\right)^{n}.$$
(2.245)

2.D.2 Non-trivial flux insertion

We now come to the quantity (2.225) with $\alpha \neq 0$, assuming the excited state to be of the form (2.164). Let us consider the denominator first. Because of the flux insertion to the right of the *n*th operator $\rho_{0,A}$, everything is the same as in (2.231) except for the last resolution of the identity, which produces a term:

$$\langle \Psi | e^{2\pi i \alpha \hat{Q}_A} | \phi_{1A}, \phi_{n\bar{A}} \rangle = \langle \Psi | e^{2\pi i \alpha} \phi_{1A}, \phi_{n\bar{A}} \rangle = \Psi [\phi_{1A} e^{2\pi i \alpha}, \phi_{n\bar{A}}]^*.$$
(2.246)

Thus, the Gaussian measure in the presence of the charge is modified as follows:

$$\operatorname{Tr}_{A}(\rho_{0,A}^{n}e^{2\pi i\alpha \hat{Q}_{A}}) = \int \mathcal{D}\phi_{1}\dots\mathcal{D}\phi_{n} \exp\left[-\mathcal{G}_{\alpha}\right], \qquad (2.247)$$

with

$$\mathcal{G}_{\alpha} = \sum_{i=1}^{n} \left[2 \left(\int_{\substack{x \in A \\ y \in A}} + \int_{\substack{x \in \bar{A} \\ y \in \bar{A}}} \right) dx dy \, \phi_{i}^{\dagger}(x) K(x-y) \phi_{i}(y) + \left(\int_{\substack{x \in A \\ y \in \bar{A}}} dx dy \, \left(\phi_{i}^{\dagger}(x) + \phi_{i+1}^{\dagger}(x) e^{-2\pi i \alpha \delta_{i,n}} \right) K(x-y) \phi_{i}(y) + \text{c.c.} \right) \right].$$
(2.248)

As for the numerator of (2.225), we have similarly

$$\operatorname{Tr}_{A}(\rho_{A}^{n}) = \int \mathcal{D}\phi_{1} \dots \mathcal{D}\phi_{n} \prod_{i=1}^{n-1} \Psi_{\{p_{j+},p_{j-}\}}[\phi_{iA},\phi_{i\bar{A}}]\Psi_{\{p_{j+},p_{j-}\}}[\phi_{i+1,A},\phi_{i\bar{A}}]^{*} \\ \times \Psi_{\{p_{j+},p_{j-}\}}[\phi_{nA},\phi_{n\bar{A}}]\Psi_{\{p_{j+},p_{j-}\}}[\phi_{1,A}e^{2\pi i\alpha},\phi_{n\bar{A}}]^{*}.$$
(2.249)

The product in the first line yields the same quantity as in the $\alpha = 0$ case, with the replica index taking values up to n - 1 only. The product of the last two functionals gives:

$$\left(\prod_{j^{+}} \alpha_{p_{j^{+}}} [\phi_{n}^{\dagger}] \alpha_{p_{j^{+}}} [\phi_{1A}^{\dagger} e^{-2\pi i \alpha}, \phi_{n\bar{A}}^{\dagger}]^{*} \prod_{j^{-}} \beta_{p_{j^{-}}} [\phi_{n}] \beta_{p_{j^{-}}} [\phi_{1A} e^{2\pi i \alpha}, \phi_{n\bar{A}}]^{*}\right) \times \Psi_{\text{vac}} [\phi_{nA}, \phi_{n\bar{A}}] \Psi_{\text{vac}} [\phi_{1,A} e^{2\pi i \alpha}, \phi_{n\bar{A}}]^{*}.$$

$$(2.250)$$

and from the definitions (2.162) we obtain our final result:

$$\frac{\operatorname{Tr}_{A}\left(\rho_{A}^{n} e^{2\pi i \alpha \hat{Q}_{A}}\right)}{\operatorname{Tr}_{A}\left(\rho_{0,A}^{n} e^{2\pi i \alpha \hat{Q}_{A}}\right)} = \left(\prod_{j^{+}} \frac{2E_{p_{j^{+}}}}{L} \prod_{j^{-}} \frac{2E_{p_{j^{-}}}}{L}\right)^{n} \times \left\langle\prod_{i=1}^{n} \prod_{j^{+}} U_{i}^{+}(p_{j^{+}})V_{i}^{+}(p_{j^{+}}) \prod_{j^{-}} U_{i}^{-}(p_{j^{-}})V_{i}^{-}(p_{j^{-}})\right\rangle_{\alpha}, \quad (2.251)$$

where $\langle \ldots \rangle_{\alpha}$ is the average with Gaussian measure (2.248). The quantities $U_i^{\pm}(p)$ are defined exactly as in (2.242), while the functions $V_i^{\pm}(p)$ are now modified by the presence of the charge as follows:

$$V_i^+(p) = e^{2\pi i \alpha \delta_{i,n}} \int_A \mathrm{d}x \, e^{-ipx} \phi_{i+1}(x) + \int_{\bar{A}} \mathrm{d}x \, e^{-ipx} \phi_i(x), \qquad (2.252)$$

$$V_i^{-}(p) = e^{-2\pi i \alpha \delta_{i,n}} \int_A \mathrm{d}x \, e^{-ipx} \phi_{i+1}^{\dagger}(x) + \int_{\bar{A}} \mathrm{d}x \, e^{-ipx} \phi_i^{\dagger}(x).$$
(2.253)

If the excited state is given by (2.165), the right-hand side of (2.251) is changed exactly as in the case $\alpha = 0$.

CHAPTER

THREE

SYMMETRY-RESOLVED NEGATIVITY OF EXCITED STATES

In the previous chapter, we studied the symmetry-resolved Rényi entropies of quasi-particle excited states in QFTs with U(1) symmetry. We found that the entropies display some model-independent features which we characterised using different approaches. In this chapter, based on [3], we extend this line of investigation by providing analytical and numerical evidence that a similar universal behavior arises for the symmetry-resolved negativity. In particular, we compute the ratio of charged moments of the partially transposed reduced density matrix. These charged ratios are given by expectation values of the composite twist operators introduced in the previous chapter: their use allows us to perform the computation in an arbitrary number of spacial dimensions. We show that, in the large-volume limit, only the commutation relations between the twist operators and local fields matter, and computations reduce to a purely combinatorial problem. We address some specific issues regarding fermionic excitations, whose treatment requires the notion of partial time-reversal transformation, and we discuss the differences with their bosonic counterpart. We find that although the operation of partial transposition requires a redefinition for fermionic theories, the ratio of the negativity moments between an excited state and the ground state is universal and identical for fermions and bosons, as it is found by performing computations on QFT states as well as simple qubit states. Our predictions are tested numerically on a 1D Fermi chain.

3.1 Introduction

Over the past two decades, entanglement measures have been widely studied in the context of low-dimensional QFT, starting with several seminal works [11-16, 48] which focused on one measure (the entanglement entropy [17]) and on one type of theory, largely 1+1D CFT and its discrete

counterpart, critical spin chains. From these papers sprang several important ideas and techniques which have been extensively exploited thereafter. Notable among them is the numerical and analytical observation that the entanglement entropy exhibits universal properties, i.e. properties that depend only on the theory's universality class characterised by the central charge c. As already discussed in the previous chapters, from a computational point of view, an important idea to emerge from [12, 48] and later reinterpreted and generalised to non-critical theories in [20] is that entanglement measures can be written in terms of correlation functions of semi-local fields of a replica version of the QFT under study. One particular development of these ideas has been the proposal and study of new measures of entanglement, each tailored to capturing particular features of entanglement and/or of the state whose entanglement is being measured. One such new measure is the (logarithmic) negativity [56, 57, 176–180] which we defined in (1.16) for a tripartite system consisting of subsystems A, B and $C := \overline{A \cup B}$ and Hilbert space given by (1.14).

An interesting issue that is specific to the logarithmic negativity is the fact that the definition of \mathcal{E} in terms of the partial transpose $\rho_{A\cup B}^{T_B}$, whose matrix elements are given in (1.15), directly apply to spin chains or bosonic systems, but it is ill-suited for fermionic systems. The reason for this is rather technical and can be explained in different ways. Unlike for bosons, the partial transpose of the Gaussian density matrix of a free fermion state is not Gaussian, which makes the computation of the negativity spectrum particularly difficult. There have been several proposals as to how to modify the definition of \mathcal{E} in a way that is better adapted to deal with fermionic degrees of freedom. The first definition of partial transposition specifically modified for fermionic states was introduced in [181]. However, in [182] it was proved that, because of the anticommuting nature of the fermionic degrees of freedom, two of the standard requirements of a partial transposition operation, namely that if $\rho \equiv \rho_{A\cup B}$ then

$$(\rho^{T_A})^{T_B} = \rho^T \quad \text{and} \quad \rho_1^{T_A} \otimes \dots \otimes \rho_n^{T_A} = (\rho_1 \otimes \dots \otimes \rho_n)^{T_A},$$
(3.1)

with T representing transposition over the total space (here $T = T_{A\cup B}$), may not hold with the definition given in [181]. On the other hand, these properties are satisfied with the definition introduced in [182]: this is the *time-reversal* (or fermionic) negativity, which accounts for the locality properties of fermions [182, 183] and which we present below. Following [183], we now take $\mathcal{H}_A \otimes \mathcal{H}_B$ to be a Fock space associated to fermionic degrees of freedom in $\ell_A + \ell_B$ sites and choose an occupation number basis:

$$|\{n_{j}\}_{A},\{n_{j}\}_{B}\rangle := |n_{1},\ldots,n_{\ell_{A}},n_{\ell_{A}+1},\ldots,n_{\ell_{A}+\ell_{B}}\rangle$$
$$= (f_{1}^{\dagger})^{n_{1}}\ldots(f_{n_{\ell_{A}}}^{\dagger})^{n_{\ell_{A}}}(f_{n_{\ell_{A}+1}}^{\dagger})^{n_{\ell_{A}+1}}\ldots(f_{n_{\ell_{A}+\ell_{B}}}^{\dagger})^{n_{\ell_{A}+\ell_{B}}}|0\rangle,$$
(3.2)

such that all the $n_j \in \{0,1\}$ and f_j^{\dagger} is a fermionic creation operator at site j. Then, given the RDM

$$\rho_{A\cup B} = \sum_{\substack{\{n_j\}_A, \{n_j\}_B\\\{n'_j\}_A, \{n'_j\}_B}} |\{n_j\}_A, \{n_j\}_B\rangle \langle \{n_j\}_A, \{n_j\}_B | \rho_{A\cup B} | \{n'_j\}_A, \{n'_j\}_B\rangle \langle \{n'_j\}_A, \{n'_j\}_B |,$$
(3.3)

we define the fermionic partial transposition as

$$\rho_{A\cup B}^{R_B} := \sum_{\substack{\{n_j\}_A, \{n_j\}_B\\\{n'_j\}_A, \{n'_j\}_B}} i^{\phi(\{n_j\}, \{n'_j\})} |\{n_j\}_A, \{n'_j\}_B \rangle \langle \{n_j\}_A, \{n_j\}_B | \rho_{A\cup B} | \{n'_j\}_A, \{n'_j\}_B \rangle \langle \{n'_j\}_A, \{n_j\}_B | ,$$

$$(3.4)$$

where $\phi(\{n_j\}, \{n'_j\})$ is given by

$$\phi(\{n_j\},\{n'_j\}) = (\tau_B + \tau'_B) (\text{mod } 2) + 2(\tau_A + \tau'_A)(\tau_B + \tau'_B), \qquad (3.5)$$

and $\tau_{A/B} = \sum_{j \in A/B} n_j$, $\tau'_{A/B} = \sum_{j \in A/B} n'_j$ are the numbers of occupied states in each subsystem. Thus, the novelty of the fermionic partial transposition (3.4), as compared to (1.15), is the presence of an additional phase shift which depends on the number of fermions. The operation R_B defined in (3.4) is also called partial time reversal, as in a path integral formalism for spinless fermions it implements time reversal only in the subsystem B [184]. While in general $\rho_{A\cup B}^{R_B}$ is no longer Hermitian for fermionic systems, one can still use the definition (1.16) for fermionic logarithmic negativity, but where now

$$|\rho_{A\cup B}^{R_B}| = \sqrt{(\rho_{A\cup B}^{R_B})^{\dagger} \rho_{A\cup B}^{R_B}}, \qquad (3.6)$$

which is a positive semi-definite matrix.

Let us now add the final layer of definitions by introducing symmetry-resolved negativities. As seen in the previous chapters, symmetry-resolved entanglement measures have become very popular in the past few years and extend the standard definitions by exploiting the presence of internal symmetries. While the earliest studies focused on the entanglement entropies (see [19, 94] for the CFT/QFT and quantum spin chain constructions), more recently also the logarithmic negativity has been generalised in a similar fashion [96, 120, 185, 186]. Let us consider a theory with a global U(1) symmetry (i.e. a complex free boson/fermion). In that case, a global U(1) charge $\hat{Q}_{A\cup B} = \hat{Q}_A \oplus \hat{Q}_B$ commutes with the state $\rho_{A\cup B}$:

$$[\rho_{A\cup B}, \hat{Q}_A \oplus \hat{Q}_B] = 0.$$
(3.7)

Then, it has been shown in [96] that the charge imbalance operator $\hat{Q}_A - \hat{Q}_B^T$, defined in (1.51), commutes with $\rho_{A\cup B}^{T_B}$

$$[\rho_{A\cup B}^{T_B}, \hat{Q}_A - \hat{Q}_B^T] = 0, \qquad (3.8)$$

and it generates a U(1) symmetry for the (bosonic or fermionic) partial transpose. The matrix representation of the charge imbalance operator is basis-dependent, and in the occupation number basis $\hat{Q}_B^T = \hat{Q}_B$ so we will drop the transposition on \hat{Q}_B from now on. At this point, it is natural to consider the charged moments of the partial transpose. These are defined as

$$\operatorname{Tr}\left((\rho_{A\cup B}^{T_B})^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right), \quad \alpha \in [-1/2, 1/2],$$
(3.9)

and, generalising the standard moments (1.30), when integrated over α they provide the symmetry-resolved Rényi negativities. For fermions, the definition above is changed to

$$\operatorname{Tr}\left(|\rho_{A\cup B}^{R_B}|^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right), \quad \alpha \in [-1/2, 1/2].$$
(3.10)

The computation of charged moments of the partial transpose was performed in [120] in the ground and thermal state of massless free fermions in 1+1 dimensions, where the universal UV divergences were captured by the underlying CFT. They have also been measured in an experimental setup in [23]. In this chapter, we are interested in zero-density quasi-particle states of (massive) QFT, obtained as excitations of the ground state with finite number of particles at given momenta. These are the same states we considered in the previous chapter, and that were already used in [114–116, 144] for the standard (uncharged) entanglement measures. We aim to compute the contribution to the charged moments given by the quasi-particles, which arises in addition to the zero-point fluctuations.

We will now briefly state the main results of this chapter. Let us consider a QFT in D = d + 1dimensions carrying a global U(1) symmetry and two non-complementary spacial regions A and B. Let $|0\rangle$ be the vacuum state in the Hilbert space (1.14), and let $|\Psi^{(k)}\rangle$ be an excited state containing kidentical excitations with unit charge. We construct the associated reduced density matrices (RDM) over $A \cup B$ as

$$\rho_{A\cup B,0} := \operatorname{Tr}_C |0\rangle \langle 0|, \quad \rho_{A\cup B} := \operatorname{Tr}_C |\Psi^{(k)}\rangle \langle \Psi^{(k)}|, \quad \text{with} \quad C := \overline{A \cup B}.$$
(3.11)

Then, generalising what we did in the previous chapter, we consider the limit in which the generalised volume V_A, V_B, V_C of each region goes to infinity while the ratios

$$r_A := \frac{V_A}{V}, \quad r_B := \frac{V_B}{V} \quad \text{and} \quad r := \frac{V_C}{V} = 1 - r_A - r_B,$$
 (3.12)

are finite and $V = V_A + V_B + V_C$. We then define the ratio of charged moments:

$$\mathcal{R}_{k}^{n}(r_{A}, r_{B}, r; \alpha) := \frac{\operatorname{Tr}\left((\rho_{A\cup B}^{T_{B}})^{n} e^{2\pi i \alpha (\hat{Q}_{A} - \hat{Q}_{B})}\right)}{\operatorname{Tr}\left((\rho_{A\cup B, 0}^{T_{B}})^{n} e^{2\pi i \alpha (\hat{Q}_{A} - \hat{Q}_{B})}\right)}.$$
(3.13)

We find this ratio to be universal in the large-volume limit, and for a single particle excitation (k = 1) it is given by

$$\mathcal{R}_{1}^{n}(r_{A}, r_{B}, r; \alpha) = e^{2\pi i \alpha} r_{A}^{n} + e^{-2\pi i \alpha} r_{B}^{n} + \left(\frac{r + \sqrt{r^{2} + 4r_{A}r_{B}}}{2}\right)^{n} + \left(\frac{r - \sqrt{r^{2} + 4r_{A}r_{B}}}{2}\right)^{n} . \quad (3.14)$$

We notice that, because $r_A, r_B, r \in (0, 1)$, the last term in (3.14) is positive (negative) when n is an even (odd) integer and therefore one needs to consider two distinct analytic continuations over the even and odd integers. In particular, the analytic continuation from n even to n = 1 gives

$$\lim_{n \to \frac{1}{2}} \mathcal{R}_1^{2n}(r_A, r_B, r; \alpha) = e^{2\pi i \alpha} r_A + e^{-2\pi i \alpha} r_B + \sqrt{r^2 + 4r_A r_B} \,. \tag{3.15}$$

Comparing these expression to the $\alpha = 0$ results found in [116, 144], we see that the phases $e^{\pm 2\pi i \alpha}$ appear only in the first two terms of (3.14), whereas the other terms are unchanged. This provides a useful hint as to how more complicated formulae for multi-particle states will generalise to the symmetry-resolved measure, namely by the substitutions $r_A \mapsto e^{\frac{2\pi i \alpha}{n}} r_A$ and $r_B \mapsto e^{-\frac{2\pi i \alpha}{n}} r_B$. Therefore, we can write:

$$\mathcal{R}_1^n(r_A, r_B, r; \alpha) = \mathcal{R}_1^n\left(e^{\frac{2\pi i\alpha}{n}}r_A, e^{-\frac{2\pi i\alpha}{n}}r_B, r; 0\right).$$
(3.16)

The generalisation to states of many distinct quasiparticles is straightforward, and each particle contributes independently (multiplicatively) to the ratio of charged moments, similar to the structure found in the previous chapter for the charged Rényi entropies. We highlight that the result (3.14) holds also for fermions with the definitions (3.6) and (3.10) when n is even. This is the case we will consider in the following when treating the fermionic case. For bosonic systems, states of multiple identical excitations can also be considered. The total (uncharged) negativity of these states was obtained in [116, 144], and for the ratio of charged moments in this case we find:

$$\mathcal{R}_{k}^{n}(r_{A}, r_{B}, r; \alpha) = \sum_{p=-k}^{k} \sum_{q=\max(0, -np)}^{\left[\frac{n}{2}(k-p)\right]} \mathcal{A}_{p,q} r_{A}^{np+q} r^{n(k-p)-2q} r_{B}^{q} e^{2\pi i \alpha p},$$
(3.17)

where $[\cdot]$ represents the integer part and

$$\mathcal{A}_{p,q} := \sum_{\{k_1,\dots,k_n\} \in P_n(q)} \prod_{j=1}^n \frac{k!}{(p+k_j)!(k-p-k_j-k_{j+1})!k_j!},$$
(3.18)

are combinatorial factors, with the sum running over the set $P_n(q)$ of partitions of $q \in \mathbb{N}_0$ into n non-negative integers.

We structure this chapter as follows. In Section 3.2 we analyse in detail a simplified model consisting

of a state of few qubits. In spite of the simplicity of these states, their symmetry-resolved negativity moments capture the main universal features found in QFT. In Section 3.3 we give a field-theoretic formulation of the charged moments of the partially transposed RDM, employing the notion of twist operators. The difference between fermionic and bosonic particles is thoroughly discussed in this context, and the evaluation of the moments is shown to reduce to a combinatorial problem which we solve exactly for single and multiple distinct excitations. Some details of our combinatorial arguments are left to Appendix 3.A, while in Appendix 3.B we prove some useful identities that we use in Section 3.3. We check numerically our predictions on a 1D Fermi chain in Section 3.4 and find good agreement. We conclude in Section 3.5.

3.2 Qubit computation

In this section we derive the main formulae for the charged replica negativities in a multi-qubit system. This toy model was already employed in [114–116], as well as in the previous chapter, following the observation that even if multi-qubit states are much simpler than the excited states of a QFT, they both produce the same universal contribution to entanglement entropies and negativities. The advantage of this picture is that the ground state of a multi-qubit system is trivial from the point of view of the entanglement content, which means that the excess of entropy, or the excess of negativity, of an excited state with respect to the ground state effectively reduces to the entropy or negativity of the excited state. Analogously, the ratio of charged moments of the partial transpose coincides with the charged moment of the partial transpose of the excited state. The notion of charge imbalance in a qubit setup was introduced in [96] and the notion of fermionic partial transposition in the same setup was later used in [120].

The main result of this section is to show that equation (3.14) for a state consisting of a single excitation can be derived employing either the bosonic or the fermionic notion of partial transposition. Interestingly, even if the intermediate steps of the computation are different in the two cases, the final result is still the same. For bosonic theories, the result can be generalised to multiple identical excitations to give (3.17).

3.2.1 Single bosonic excitation

Proceeding exactly as in Section 2.3, let us assume that a single bosonic excitation of charge¹ +1 is localised in space according to a uniform probability distribution, so that r_A , r_B and r can be regarded as the probabilities for the excitation to be found in regions A, B, C respectively. Then the state in

¹Without loss of generality, in this section we take all the quasiparticle excitations to be positively charged. In this way the action of the U(1) charge \hat{Q} on the multi-qubit states is identical to that of the number operator \hat{N} .

 $\mathcal{H}_A \otimes \mathcal{H}_B \otimes \mathcal{H}_C$ representing a single excitation can be written as

$$|\Psi_{\rm qb}^{(1)}\rangle = \sqrt{r_A}|100\rangle + \sqrt{r_B}|010\rangle + \sqrt{r}|001\rangle, \qquad (3.19)$$

where the values 0 (1) represent the absence (presence) of the excitation and the coefficients can be interpreted as probabilities of finding the excitation in a particular region. Here and in the following we omit the tensor products in writing qubit states, identifying $|k_1k_2k_3\rangle := |k_1\rangle_A \otimes |k_2\rangle_B \otimes |k_3\rangle_C$. The RDM $\rho_{A\cup B}$ is obtained taking the trace over \mathcal{H}_C :

$$\rho_{A\cup B} = \text{Tr}_{C} |\Psi_{qb}^{(1)}\rangle \langle \Psi_{qb}^{(1)}|$$

= $r_{A} |10\rangle \langle 10| + r_{B} |01\rangle \langle 01| + \sqrt{r_{A}r_{B}} (|01\rangle \langle 10| + |10\rangle \langle 01|) + r|00\rangle \langle 00|,$ (3.20)

or in matrix form

$$\rho_{A\cup B} = \begin{pmatrix}
00 & 01 & 10 & 11 \\
00 & r & 0 & 0 & 0 \\
01 & 0 & r_B & \sqrt{r_A r_B} & 0 \\
10 & 0 & \sqrt{r_A r_B} & r_A & 0 \\
11 & 0 & 0 & 0 & 0
\end{pmatrix}.$$
(3.21)

This matrix has a block-diagonal structure with respect to the charge operator $\hat{Q}_{A\cup B} = \hat{Q}_A \oplus \hat{Q}_B$:

$$(\hat{Q}_A \oplus \hat{Q}_B) | k_A k_B \rangle = (k_A + k_B) | k_A k_B \rangle, \quad k_A, k_B \in \{0, 1\},$$
 (3.22)

as indeed we can decompose it according to the eigenspaces of $\hat{Q}_{A\cup B}$, with eigenvalues Q = 0, 1, 2:

$$\rho_{A\cup B} = (r)_{Q=0} \oplus \begin{pmatrix} r_B & \sqrt{r_A r_B} \\ \sqrt{r_A r_B} & r_A \end{pmatrix}_{Q=1} \oplus (0)_{Q=2}.$$
(3.23)

Let us now come to the partially transposed matrix $\rho_{A\cup B}^{T_B}$. From the definition (1.15), it follows:

$$\rho_{A\cup B}^{T_B} = r_A |10\rangle \langle 10| + r_B |01\rangle \langle 01| + \sqrt{r_A r_B} (|00\rangle \langle 11| + |11\rangle \langle 00|) + r|00\rangle \langle 00|
= \begin{pmatrix} 00 & 01 & 10 & 11 \\ \hline 00 & r & 0 & 0 & \sqrt{r_A r_B} \\ 01 & 0 & r_B & 0 & 0 \\ 10 & 0 & 0 & r_A & 0 \\ 11 & \sqrt{r_A r_B} & 0 & 0 & 0 \end{pmatrix},$$
(3.24)

which is a block-diagonal matrix with respect to the charge imbalance operator $\hat{Q}_A - \hat{Q}_B$:

$$(\hat{Q}_A - \hat{Q}_B)|k_A k_B\rangle = (k_A - k_B)|k_A k_B\rangle, \quad k_A, k_B \in \{0, 1\},$$
(3.25)

and denoting the eigenvalues of the charge imbalance by $\Delta Q = 1, 0, -1$ we can write:

$$\rho_{A\cup B}^{T_B} = (r_A)_{\Delta Q=1} \oplus \begin{pmatrix} r & \sqrt{r_A r_B} \\ \sqrt{r_A r_B} & 0 \end{pmatrix}_{\Delta Q=0} \oplus (r_B)_{\Delta Q=-1}.$$
(3.26)

The partial transpose $\rho_{A\cup B}^{T_B}$ has four distinct eigenvalues, one of which is negative:

$$\sigma\left(\rho_{A\cup B}^{T_B}\right) = \left\{r_A, r_B, \frac{r + \sqrt{r^2 + 4r_A r_B}}{2}, \frac{r - \sqrt{r^2 + 4r_A r_B}}{2}\right\},\tag{3.27}$$

and therefore this system has non-vanishing negativity. From the block-diagonal structure of $\rho_{A\cup B}^{T_B}$ it is immediate to construct the matrix $(\rho_{A\cup B}^{T_B})^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}$ and we finally obtain the expected result:

$$\operatorname{Tr}\left((\rho_{A\cup B}^{T_B})^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right) = \mathcal{R}_1^n(r_A, r_B, r; \alpha), \qquad (3.28)$$

with $\mathcal{R}_1^n(r_A, r_B, r; \alpha)$ given by (3.14). We highlight that the knowledge of the spectrum (3.27) allows for a direct evaluation of the logarithmic negativity (1.16) without the need to obtain the moments first.

3.2.2 Single fermionic excitation

In this section we show that the result (3.14) holds for n even also if we adopt the fermionic prescription (3.4) for the partial transposition. With that definition, the matrix $\rho_{A\cup B}^{R_B}$ differs from $\rho_{A\cup B}^{T_B}$ as given in (3.26) only because there is now an extra phase in the off-diagonal elements:

$$(|10\rangle\langle 01|)^{R_B} = -i|11\rangle\langle 00|, \quad (|01\rangle\langle 10|)^{R_B} = -i|00\rangle\langle 11|, \qquad (3.29)$$

while the diagonal elements are not modified. It follows that $\rho_{A\cup B}^{R_B}$ is still block-diagonal with respect to the imbalance operator:

$$\rho_{A\cup B}^{R_B} = (r_A)_{\Delta Q=1} \oplus \begin{pmatrix} r & -i\sqrt{r_A r_B} \\ -i\sqrt{r_A r_B} & 0 \end{pmatrix}_{\Delta Q=0} \oplus (r_B)_{\Delta Q=-1}.$$
(3.30)

The two eigenvalues in the sector $\Delta Q = 0$ can now be imaginary, depending on the values of r_A and r_B . However, we are eventually interested in the evaluation of the charged moments $\operatorname{Tr}\left(|\rho_{A\cup B}^{R_B}|^n e^{i2\pi\alpha(\hat{Q}_A-\hat{Q}_B)}\right)$ for n even, which requires only the knowledge of the eigenvalues of

 $|\rho_{A\cup B}^{R_B}|^2$. From the definition (3.6), and making use of the above block diagonal decomposition we can write:

$$(\rho_{A\cup B}^{R_B})^{\dagger}\rho_{A\cup B}^{R_B} = (r_A^2)_{\Delta Q=1} \oplus \begin{pmatrix} r^2 + r_A r_B & ir\sqrt{r_A r_B} \\ -ir\sqrt{r_A r_B} & r_A r_B , \end{pmatrix}_{\Delta Q=0} \oplus (r_B^2)_{\Delta Q=-1}, \quad (3.31)$$

and

$$\sigma\left((\rho_{A\cup B}^{R_B})^{\dagger}\rho_{A\cup B}^{R_B}\right) = \left\{r_A^2, r_B^2, \left(\frac{r+\sqrt{r^2+4r_Ar_B}}{2}\right)^2, \left(\frac{r-\sqrt{r^2+4r_Ar_B}}{2}\right)^2\right\}.$$
 (3.32)

Since these eigenvalues are the squares of the eigenvalues of $\rho_{A\cup B}^{T_B}$, the result (3.14) is recovered here for *n* even.

3.2.3 Multiple distinct excitations

We now consider states containing k distinct excitations. Among these states, let us focus on those which are tensor products of the linear combination (3.19), that is, on states of the form:

$$|\Psi_{\rm qb}^{(1,\dots,1)}\rangle := |\Psi_{\rm qb}^{(1)}\rangle^{\otimes k}.$$
 (3.33)

For k = 2 we have for instance

$$\begin{split} \Psi_{\rm qb}^{(1,1)} \rangle &= r_A |100\rangle \otimes |100\rangle + r_B |010\rangle \otimes |010\rangle + r |001\rangle \otimes |001\rangle \\ &+ \sqrt{r_A r_B} (|100\rangle \otimes |010\rangle + |010\rangle \otimes |100\rangle) \\ &+ \sqrt{r_A r} (|100\rangle \otimes |001\rangle + |001\rangle \otimes |100\rangle) \\ &+ \sqrt{r_B r} (|010\rangle \otimes |001\rangle + |001\rangle \otimes |010\rangle) \,. \end{split}$$
(3.34)

For a tensor product state the density matrix is a tensor product of single-particle matrices, and this applies to the RDM and to its partial transpose as well:

$$\rho^{(1,\dots,1)} = \rho^{\otimes k}, \quad \rho^{(1,\dots,1)}_{A\cup B} = (\rho_{A\cup B})^{\otimes k}, \quad \rho^{(1,\dots,1),T_B}_{A\cup B} = (\rho^{T_B}_{A\cup B})^{\otimes k}, \tag{3.35}$$

with $\rho_{A\cup B}$ and $\rho_{A\cup B}^{T_B}$ given by (3.21) and (3.24) respectively. Decomposing the charge imbalance as $\hat{Q}_A - \hat{Q}_B = \bigoplus_{j=1}^k (\hat{Q}_A^{(j)} - \hat{Q}_B^{(j)})$, we can compute the charged replica negativities using the bosonic partial transposition (1.15):

$$\operatorname{Tr}\left(\left(\rho_{A\cup B}^{(1,\dots,1),T_B}\right)^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right) = \prod_{j=1}^k \operatorname{Tr}\left(\left(\rho_{A\cup B}^{T_B}\right)^n e^{2\pi i \alpha (\hat{Q}_A^{(j)} - \hat{Q}_B^{(j)})}\right) = \left(\mathcal{R}_1^n(r_A, r_B, r; \alpha)\right)^k.$$
 (3.36)
The result above is not surprising and it is a consequence of the choice of the state: if the multi-particle state is a tensor product then there is no correlation between different particles and the total charged moment is the product of the single-particle ones. As we shall see below, this is not the case when the particles are indistinguishable.

3.2.4 Multiple identical excitations

In this section we consider a state consisting of k positively charged indistinguishable excitations. Its associated qubit state can be written as:

$$|\Psi_{\rm qb}^{(k)}\rangle = \sum_{k_A, k_B, k_C \in \mathbb{N}_0} c_{k_A, k_B, k_C} |k_A k_B k_C\rangle, \qquad (3.37)$$

where the coefficient

$$c_{k_A,k_B,k_C} := \sqrt{\frac{k! r_A^{k_A} r_B^{k_B} r^{k_C}}{k_A! k_B! k_C!}} \delta_{k_A + k_B + k_C,k}, \qquad (3.38)$$

is the square root of the probability of finding k_A identical particles in region A, k_B identical particles in region B and the remaining k_C particles in region C. From $r_A + r_B + r = 1$ it is easy to show that if the vectors $|k_A k_B k_C\rangle$ form an orthonormal set, then also $\langle \Psi_{qb}^{(k)} | \Psi_{qb}^{(k)} \rangle = 1$.

From this expression it is then possible to explicitly construct the matrix elements of the (bosonic) partially transposed density matrix as:

$$\langle k_A^1 k_B^1 | \rho_{A \cup B}^{T_B} | k_A^2 k_B^2 \rangle = \sum_{k_C \in \mathbb{N}_0} c_{k_A^1 k_B^2 k_C} c_{k_A^2 k_B^1 k_C}, \qquad (3.39)$$

where the sum represents taking the trace over the degrees of freedom in C and the partial transposition exchanges the indices k_B^1 and k_B^2 in the coefficients. The matrix elements of the *n*th power can then be computed inserting *n* resolutions of the identity in $\mathcal{H}_{A\cup B}$:

$$\langle k_A^1 k_B^1 | \left(\rho_{A \cup B}^{T_B} \right)^n | k_A^{n+1} k_B^{n+1} \rangle = \sum_{\substack{k_A^s, k_B^s \in \mathbb{N}_0 ; s=2,\dots,n \\ k_C^r \in \mathbb{N}_0 ; r=1,\dots,n}} \prod_{j=1}^n c_{k_A^j k_B^{j+1} k_C^j} c_{k_A^{j+1} k_B^j k_C^j}$$

$$= \sum_{\substack{k_A^s, k_B^s \in \mathbb{N}_0 ; s=2,\dots,n \\ k_C^r \in \mathbb{N}_0 ; r=1,\dots,n}} \prod_{j=1}^n \frac{k! r_A^{k_A^j r k_C^j} r_B^{k_B^j}}{k_A^j ! k_C^j ! k_B^j !} \delta_{k_A^j + k_B^{j+1} + k_C^j, k} \delta_{k_A^{j+1} + k_B^j + k_C^j, k}, \quad (3.40)$$

While expression (3.40) was already presented in [116], its symmetry-resolved version is new and can

be easily written by introducing phase factors in the sum above. We have

$$\langle k_A^1 k_B^1 | \left(\rho_{A \cup B}^{T_B} \right)^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)} | k_A^{n+1} k_B^{n+1} \rangle$$

$$= \sum_{\substack{k_A^s, k_B^s \in \mathbb{N}_0 ; s=2,\dots,n \\ k_C^r \in \mathbb{N}_0 ; r=1,\dots,n}} \prod_{j=1}^n \frac{k! r_A^{k_j^j} r_C^{k_j^j} r_B^{k_j^j}}{k_A^j ! k_C^j ! k_B^j !} e^{\frac{2\pi i \alpha}{n} (k_A^j - k_B^j)} \delta_{k_A^j + k_B^{j+1} + k_C^j, k} \, \delta_{k_A^{j+1} + k_B^j + k_C^j, k}.$$
(3.41)

Starting with this result, the derivation of equations (3.17) and (3.18) is identical to that presented in [116, 144]. The idea is to employ all the delta-function constraints on the values of k_A^j , k_B^j and k_C^j in order to reduce the number of terms in the sum. After all the constraints have been implemented, equation 3.41 reproduces the result (3.17) with (3.18).

Before concluding the section let us analyse the simplest case, k = 1. For a single excitation, the right-hand side of (3.17) is given by:

$$\mathcal{A}_{-1,n} r_B^n e^{-2\pi i \alpha} + \mathcal{A}_{1,0} r_A^n e^{2\pi i \alpha} + \sum_{q=0}^{[n/2]} \mathcal{A}_{0,q} (r_A r_B)^q r^{n-2q}.$$
(3.42)

By looking at the definition (3.18) one immediately gets $\mathcal{A}_{-1,n} = \mathcal{A}_{1,0} = 1$. On the other hand, $\mathcal{A}_{0,q} = \sum_{\{k_1,\ldots,k_n\} \in P_n(q)} 1$, where each $k_j \in \{0,1\}$ and whenever $k_j = 1$ then $k_{j+1} = 0$. Counting the number of sequences (k_1, \ldots, k_n) that satisfy these constraints is a combinatorial problem identical to the one we solve in the next section. The number of these sequences is $\frac{n}{n-q} \binom{n-q}{n}$. As we explain in the next section, this implies that expression (3.42) exactly reproduces (3.14). More generally, looking at the coefficients (3.18) it is clear that even for k > 1 the computation has an underlying combinatorial interpretation. For the $\alpha = 0$ case this has been established by reinterpreting the sum (3.17) as a partition function for a certain class of graphs [144]. A combinatorial picture will emerge again in the next section in a related context: the computation using twist operators.

3.3 Twist operator approach

In this section, we provide a field-theoretic description of the charged moments of the partially transposed RDM, valid in principle for any local QFT and in any spacetime dimensions. To do so, we employ the replica construction and the formalism of twist operators that we introduced in Section 2.4. There, we showed that it is possible to compute entanglement entropies (or at least charged ratios) of quasi-particle states just by relying on a few algebraic properties: the exchange relations between local fields and twist operators and the operator product expansion (OPE) of the local fields. On the other hand, most of the theory-dependent features are hidden in the zero-point fluctuations (ground state entanglement), which factors out. In this section we show that the same applies to the logarithmic negativity.

Proceeding as we did in the previous section, here we compute the charged moments of the partial transpose in a bosonic and a fermionic theory, starting with single-particle states and generalising the result to excitations containing many quasiparticles. In the fermionic case, we discuss how to modify the algebra of twist operators and fermionic fields and, as a byproduct, we perform a simpler derivation valid for free fermions in any dimension.

3.3.1 Single bosonic excitation

We consider now a bosonic QFT, described by its algebra of local observables \mathcal{A} , acting on the Hilbert space \mathcal{H} , and $|0\rangle \in \mathcal{H}$ is the ground state of the theory. We then consider the replica version of this theory, consisting of n non-interacting copies of the same model. The algebra of observables is now denoted by \mathcal{A}^n , so that \mathbb{Z}_n becomes an internal (global) symmetry, which includes cyclic permutation symmetry among copies [58, 59, 187, 188]. We also assume that the QFT carries an additional global U(1) symmetry. This procedure allows us to introduce a set of twist operators, supported on extended spacial regions, which mix cyclic permutation and internal U(1) symmetries, and generalise the notion of composite twist fields of 1+1D QFT [19, 107, 117, 156–158].

Let $\mathcal{O}^{j}(\mathbf{x}) \in \mathcal{A}^{n}$ be a local bosonic field of the *j*-th replica (j = 1, ..., n) with U(1) charge $\kappa_{\mathcal{O}}$. We consider a region A, and we associate to the twist operator T_{A}^{α} which satisfies equations (2.120). That is, the action of T_{A}^{α} is non-trivial only in the region A, where it consists of a replica shift $j \to j + 1$ followed by the insertion of a U(1) flux between the *n*-th and the first copy. The operator T_{A}^{α} is sufficient for the computation of moments of the RDM in case of a spacial bipartition. However, here we need to define a conjugate twist operator \tilde{T}_{A}^{α} so that

$$\tilde{T}_{A}^{\alpha}\mathcal{O}^{j}(\mathbf{x}) = \begin{cases} e^{-2\pi i\kappa_{\mathcal{O}}\alpha\delta_{j,1}}\mathcal{O}^{j-1}(\mathbf{x})\tilde{T}_{A}^{\alpha} & \mathbf{x} \in A, \\ \mathcal{O}^{j}(\mathbf{x})\tilde{T}_{A}^{\alpha} & \mathbf{x} \notin A, \end{cases}$$
(3.43)

and we can identify $\tilde{T}_A^{\alpha} = (T_A^{\alpha})^{\dagger}$. These operators will now allow us to develop a field-theoretic formulation of the symmetry-resolved negativity and its moments. Let A and B be two non-complementary connected regions which share at most a boundary, $|\Psi\rangle \in \mathcal{H}$ a zero-density state and $\rho_{A\cup B}$ its RDM over $A \cup B$. In 1+1D QFT one can interpret the moments of the partial transpose as functions over a *n*-sheeted Riemann surface with two branch-cuts in A and B, connecting the replicas in two opposite directions [182]. A similar construction in the presence of fluxes has been proposed in [96, 120]. In analogy with the works above, we establish the following relation between the charged moments and the twist operators:

$$\operatorname{Tr}\left((\rho_{A\cup B}^{T_B})^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right) = \frac{{}^n \langle \Psi | T_A^{\alpha} T_B^{\alpha} | \Psi \rangle^n}{{}^n \langle \Psi | \Psi \rangle^n}, \qquad (3.44)$$

where $|\Psi\rangle^n$ represents the replicated version of the state $|\Psi\rangle$.

Following the construction of the previous chapter, we consider a one-particle excitation of fixed momentum \mathbf{p} and charge +1, which is created by a field \mathcal{O} acting on the ground state as

$$|\Psi^{(1)}\rangle = \mathcal{O}(\mathbf{p})|0\rangle. \tag{3.45}$$

Here $\mathcal{O}(\mathbf{p})$ is the Fourier transform of $\mathcal{O}(\mathbf{x})$

$$\mathcal{O}(\mathbf{p}) = \int_{\mathcal{M}} \mathrm{d}^d x e^{-i\mathbf{p}\cdot\mathbf{x}} \,\mathcal{O}(\mathbf{x})\,,\tag{3.46}$$

and \mathcal{M} is the whole space, which, for simplicity, we take to be a *d*-dimensional torus. We note that $\mathcal{O}^{\dagger}(-\mathbf{p}) = [\mathcal{O}(\mathbf{p})]^{\dagger}$ which will be important for later computations. Our aim, as in the previous section, is to compute the ratio of charged moments, which up to a normalisation constant is given by

$$\frac{{}^{n}\langle \Psi^{(1)}|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|\Psi^{(1)}\rangle^{n}}{{}^{n}\langle 0|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|0\rangle^{n}}.$$
(3.47)

We define the projection of $\mathcal{O}(\mathbf{p})$ over a generic region A as the restricted integral

$$\mathcal{O}_A(\mathbf{p}) = \int_A \mathrm{d}^d x e^{-i\mathbf{p}\cdot\mathbf{x}} \mathcal{O}(\mathbf{x}) \,, \tag{3.48}$$

then we can write

$$\frac{{}^{n}\langle\Psi^{(1)}|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|\Psi^{(1)}\rangle^{n}}{{}^{n}\langle\Psi^{(1)}|\Psi^{(1)}\rangle^{n}} = \frac{{}^{n}\langle0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}\mathcal{O}^{1}(\mathbf{p})\dots\mathcal{O}^{n}(\mathbf{p})|0\rangle^{n}}{{}^{n}\langle0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\mathcal{O}^{1}(\mathbf{p})\dots\mathcal{O}^{n}(\mathbf{p})|0\rangle^{n}}.$$
(3.49)

We point out that bosonic creation operators, whether on the same or on different copies, commute with each other, therefore the order of a string of operators $\mathcal{O}^{j}(\boldsymbol{p})$ is irrelevant. We now observe that

$$\mathcal{O}^{j}(\mathbf{p}) = \mathcal{O}^{j}_{A}(\mathbf{p}) + \mathcal{O}^{j}_{B}(\mathbf{p}) + \mathcal{O}^{j}_{C}(\mathbf{p}), \qquad (3.50)$$

which, when inserted into (3.49), leads to a large numbers of terms both in the numerator and the denominator. The key idea is that in the infinite volume limit many of these terms are subleading and the leading contribution is the one which reproduces the expected result. We now present the details of the calculation.

Employing the exchange relations (2.120) and (3.43), we can bring all the bosonic fields $\mathcal{O}^{j}(\mathbf{p})$ to

the left of $T^{\alpha}_{A}\tilde{T}^{\alpha}_{B}$ in (3.49). This gives

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}\mathcal{O}^{1}(\mathbf{p})\dots\mathcal{O}^{n}(\mathbf{p})|0\rangle^{n} =$$

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})(\mathcal{O}_{A}^{2}(\mathbf{p})+\mathcal{O}_{B}^{n}(\mathbf{p})e^{-2\pi i\alpha}+\mathcal{O}_{C}^{1}(\mathbf{p}))\dots\times$$

$$(\mathcal{O}_{A}^{1}(\mathbf{p})e^{2\pi i\alpha}+\mathcal{O}_{B}^{n-1}(\mathbf{p})+\mathcal{O}_{C}^{n}(\mathbf{p}))T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|0\rangle^{n},$$
(3.51)

where the phases are attached only to the fields \mathcal{O}_A^1 and \mathcal{O}_B^n . To proceed with the evaluation of the expectation value, we focus on the large-volume behavior. In the previous chapter, we argued that the leading terms come from the contractions of fields belonging to the same replica, which amount to the following formal replacement inside the correlation function:

$$(\mathcal{O}^{\dagger})^{j}_{A'}(-\mathbf{p})\mathcal{O}^{j}_{A}(\mathbf{p}) \to \langle 0|(\mathcal{O}^{\dagger})^{j}_{A'}(-\mathbf{p})\mathcal{O}^{j}_{A}(\mathbf{p})|0\rangle \propto V_{A\cap A'}, \qquad (3.52)$$

with $A, A' \subseteq \mathcal{M}$ generic spacial regions. Proportionality to the volume is valid in the large-volume limit and the proportionality constant can in principle be absorbed in the normalisation of the field \mathcal{O} and does not affect the final result. We remarked in Section 2.4 that the accuracy of the approximation (3.52) is supported when the correlation length or the de Broglie length (represented by $|\mathbf{p}|^{-1}$) are significantly smaller than the size of the subsystem. This condition is an essential requirement for the excited states we aim to describe. In this regime, the vacuum expectation value of the twist operators also factors out:

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})(\mathcal{O}^{2}_{A}(\mathbf{p}) + \mathcal{O}^{n}_{B}(\mathbf{p})e^{-i2\pi\alpha} + \mathcal{O}^{1}_{C}(\mathbf{p}))\dots \times$$

$$(\mathcal{O}^{1}_{A}(\mathbf{p})e^{i2\pi\alpha} + \mathcal{O}^{n-1}_{B} + \mathcal{O}^{n}_{C}(\mathbf{p}))T^{\alpha}_{A}\tilde{T}^{\alpha}_{B}|0\rangle^{n} \simeq$$

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})(\mathcal{O}^{2}_{A}(\mathbf{p}) + \mathcal{O}^{n}_{B}(\mathbf{p})e^{-2\pi i\alpha} + \mathcal{O}^{1}_{C}(\mathbf{p}))\dots \times$$

$$(\mathcal{O}^{1}_{A}(\mathbf{p})e^{2\pi i\alpha} + \mathcal{O}^{n-1}_{B}(\mathbf{p}) + \mathcal{O}^{n}_{C}(\mathbf{p}))|0\rangle^{n} \times {}^{n}\langle 0|T^{\alpha}_{A}\tilde{T}^{\alpha}_{B}|0\rangle^{n},$$

$$(3.53)$$

which means that the charge moments of the ground state factor out and will subsequently be cancelled in the ratio (3.47). While the formula above is already a large-volume approximation, when the sums are expanded and each individual term considered, many terms that are subleading for large volume still appear. To make things clear, consider the terms

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\mathcal{O}_{A}^{2}(\mathbf{p})\dots\mathcal{O}_{A}^{n}(\mathbf{p})\mathcal{O}_{A}^{1}(\mathbf{p})|0\rangle^{n}e^{2\pi i\alpha}\simeq V_{A}^{n}e^{2\pi i\alpha},\qquad(3.54)$$

and

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\mathcal{O}_{B}^{n}(\mathbf{p})\mathcal{O}_{B}^{1}(\mathbf{p})\dots\mathcal{O}_{B}^{n-1}(\mathbf{p})|0\rangle^{n}e^{-2\pi i\alpha}\simeq V_{B}^{n}e^{-2\pi i\alpha}.$$
(3.55)

These are the terms that generate the highest powers in the volume of regions A and B respectively. Among the other terms that are generated, the leading ones at large volume are those containing a string of operators $\mathcal{O}^{j}(\mathbf{p})$ and their daggered versions all inserted at different replicas, just as in the examples above. If that is not the case, there is at least a pair of operators that can not be contracted as in (3.52) and the term is subleading.

We now proceed with the systematic evaluation and counting of all the leading terms generated in the expansion (3.53). We introduce the following notation to identify each term

$$(A_1 \dots A_n) := {}^n \langle 0 | (\mathcal{O}^{\dagger})^n (-\mathbf{p}) \dots (\mathcal{O}^{\dagger})^1 (-\mathbf{p}) \mathcal{O}_{A_1}^{j_1}(\mathbf{p}) \dots \mathcal{O}_{A_n}^{j_n}(\mathbf{p}) | 0 \rangle^n,$$
(3.56)

where $A_i \in \{A, B, C\}$ and $j_i \in \{1, \ldots, n\}, \forall i = 1, \ldots, n$. We observe that, once the sequence of regions $(A_1 \ldots A_n)$ is identified, the sequence of replica indices $(j_1 \ldots j_n)$ is fixed unambiguously, as in fact

$$j_{i} = \begin{cases} i+1, & \text{if } A_{i} = A \\ i, & \text{if } A_{i} = C \\ i-1, & \text{if } A_{i} = B \end{cases}$$
(3.57)

hence the choice of notation above. Moreover, due to the contraction rules discussed above, only the terms for which $(j_1 \dots j_n) = (\sigma(1), \dots, \sigma(n))$, with σ a permutation of the indices $\{1, \dots, n\}$ are non-vanishing. As a consequence, one can show (See Appendix 3.A.1) that the only possible non-vanishing terms fit into one of these two categories:

- Either $(A_1 \dots A_n) = (A \dots A)$ or $(A_1 \dots A_n) = (B \dots B)$, the two cases which have been already discussed in equations (3.54), (3.55),
- Or, whenever A appears in $(A_1 \ldots A_n)$, it has to be followed by B. Similarly, if B appears in $(A_1 \ldots A_n)$, then it has to be preceded by A.

We focus on the second set of terms. It is convenient to split this into two additional subsets, which we call type-I and type-II

- Type-I: $(A_1 \dots A_n) = (B \ A_2 \dots A_{n-1} \ A),$
- Type-II: $(A_1 \dots A_n)$ with $A_1 \neq B, A_n \neq A$.

Thus, each string in both these subsets contains a number $k \ge 0$ of pairs AB and n - 2k C's, and according to (3.52) each of them will yield a term proportional to $(V_A V_B)^k V_C^{n-2k}$. Due to the balance of A's and B's there is no phase present in these terms (no α dependence). We now just need to count how many strings of each type we have.

Among the strings of type-I, there is always at least one pair AB and there are at most [n/2] - 1additional pairs AB that can be present. The number of such strings is precisely (a proof is given in 3.A.2):

$$\binom{n-k-1}{k-1}$$
 for $k = 0, \dots [n/2].$ (3.58)

Similarly, one can show that the number of type-II strings consisting of k pairs of consecutive A and B is:

$$\binom{n-k}{k} \quad \text{for} \quad k = 0, \dots [n/2]. \tag{3.59}$$

In summary,

$${}^{n}\langle 0|(\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\dots(\mathcal{O}^{\dagger})^{1}(-\mathbf{p})(\mathcal{O}^{2}_{A}(\mathbf{p}) + \mathcal{O}^{n}_{B}(\mathbf{p})e^{-i2\pi\alpha} + \mathcal{O}^{1}_{C}(\mathbf{p}))\dots \times$$
$$(\mathcal{O}^{1}_{A}(\mathbf{p})e^{i2\pi\alpha} + \mathcal{O}^{n-1}_{B}(\mathbf{p})e^{i2\pi\alpha} + \mathcal{O}^{n}_{C}(\mathbf{p}))|0\rangle^{n} \simeq$$
$$V^{n}_{A}e^{2\pi i\alpha} + V^{n}_{B}e^{-2\pi i\alpha} + \sum_{k=0}^{\left[\frac{n}{2}\right]}\frac{n}{n-k}\binom{n-k}{k}(V_{A}V_{B})^{k}(V_{C})^{n-2k},$$
(3.60)

where we used the identity

$$\binom{n-k}{k} + \binom{n-k-1}{k-1} = \frac{n}{n-k} \binom{n-k}{k}.$$
(3.61)

The denominator in (3.49) can be fully contracted and yields V^n (up to a non-universal normalisation constant). Therefore, when properly normalised, the ratio (3.47) becomes a function of the variables r_A, r_B and r defined in (3.12) and we we obtain

$$\mathcal{R}_{1}^{n}(r_{A}, r_{B}, r; \alpha) = \frac{n \langle \Psi^{(1)} | T_{A}^{\alpha} \tilde{T}_{B}^{\alpha} | \Psi^{(1)} \rangle^{n}}{n \langle \Psi^{(1)} | \Psi^{(1)} \rangle^{n}} \frac{n \langle 0 | 0 \rangle^{n}}{n \langle 0 | T_{A}^{\alpha} \tilde{T}_{B}^{\alpha} | 0 \rangle^{n}}$$
$$= r_{A}^{n} e^{2\pi i \alpha} + r_{B}^{n} e^{-2\pi i \alpha} + \sum_{k=0}^{\left[\frac{n}{2}\right]} \frac{n}{n-k} \binom{n-k}{k} (r_{A} r_{B})^{k} r^{n-2k}, \qquad (3.62)$$

which is the main result of this section. Note that although this formula looks different from (3.14), they are in fact equivalent. That is

$$\left(\frac{r+\sqrt{r^2+4r_Ar_B}}{2}\right)^n + \left(\frac{r-\sqrt{r^2+4r_Ar_B}}{2}\right)^n = \sum_{k=0}^{\left[\frac{n}{2}\right]} \frac{n}{n-k} \binom{n-k}{k} (r_Ar_B)^k r^{n-2k}.$$
 (3.63)

This relation was used in [116, 144] without a proof. The proof is indeed quite involved, and can be performed using properties of the generalised Lucas' polynomials. This is presented in Appendix 3.B.1, where we also derive two interesting corollaries. The equality (3.63) is particularly interesting because it shows that the result is always a polynomial in integer powers of r_A, r_B, r for n positive, even or odd. However, its analytic continuation from n even to n = 1 does contain a square root, as seen in (3.15).

3.3.2 Single fermionic excitation

Let us consider a theory for which the algebra \mathcal{A} contains fermionic observables. In other words, we assume that \mathcal{A} is a \mathbb{Z}_2 -graded algebra (superalgebra) generated by bosonic/fermionic fields, which are even/odd with respect to the \mathbb{Z}_2 fermionic parity. Here, to generalise properly the twist operator construction, one must take care of the fermionic nature of the fields. Indeed, two such fields sitting at distinct points, say $\Psi(\mathbf{x})$ and $\Psi(\mathbf{x}')$ will now anticommute

$$\Psi(\mathbf{x})\Psi(\mathbf{x}') = -\Psi(\mathbf{x}')\Psi(\mathbf{x}). \tag{3.64}$$

Moreover, when the replica construction is performed, we require that fermionic fields on distinct replicas also anticommute

$$\Psi^{j}(\mathbf{x})\Psi^{j'}(\mathbf{x}') = -\Psi^{j'}(\mathbf{x}')\Psi^{j}(\mathbf{x}).$$
(3.65)

As a result, the algebra of the replica theory \mathcal{A}^n is not a conventional tensor product.

As before, we assume that an additional U(1) symmetry is present in the theory. Let A be a spacial region, and we associate to it a twist operator T_A^{α} which shifts the replica index and appends a U(1)flux to the field. The natural generalisation of (2.120) for a fermionic field Ψ of charge κ_{Ψ} is:

$$T_{A}^{\alpha}\Psi^{j}(\mathbf{x}) = \begin{cases} (-1)^{(n-1)\delta_{j,n}} e^{2\pi i\kappa_{\Psi}\alpha\delta_{j,n}}\Psi^{j+1}(\mathbf{x})T_{A}^{\alpha} & \mathbf{x} \in A, \\ \Psi^{j}(\mathbf{x})T_{A}^{\alpha} & \mathbf{x} \notin A. \end{cases}$$
(3.66)

We point out that the only difference with respect to (2.120) is the presence of an additional flux $(-1)^{n-1}$ between the *n*-th and the first replica, a factor that was already introduced in [162] and employed for instance in [20] in the calculation of the vacuum expectation value (VEV) of the Ising twist field. A derivation of (3.66) for $\alpha = 0$ in 1+1D using the coherent state representation of fermionic density matrices can be found in [182].

For fermionic theories we need to define another twist operator which implements explicitly the fermionic partial transposition, and from now on we only consider n even. It has been shown in [182] that the effect of the partial transposition on the fermions gives rise to an additional insertion of a flux (-1) among any pair of consecutive replicas, in addition to the usual replica shift. To implement this construction, we define a twist operator \tilde{T}^{α}_{A} satisfying

$$\tilde{T}_{A}^{\alpha}\Psi^{j}(\mathbf{x}) = \begin{cases} -(-1)^{(n-1)\delta_{j,1}}e^{-2\pi i\kappa_{\Psi}\alpha\delta_{j,1}}\Psi^{j-1}(\mathbf{x})\tilde{T}_{A}^{\alpha} & \mathbf{x} \in A, \\ \Psi^{j}(\mathbf{x})\tilde{T}_{A}^{\alpha} & \mathbf{x} \notin A. \end{cases}$$
(3.67)

We are now ready to compute the ratio of charged moments, along the same lines of the previous

computation. Namely, given a fermionic field $\Psi(\mathbf{x})$ with U(1) charge +1, we consider the state

$$|\Psi^{(1)}\rangle = \Psi(\mathbf{p})|0\rangle, \qquad (3.68)$$

and its replicated version

$$|\Psi^{(1)}\rangle^n = \Psi^1(\mathbf{p})\dots\Psi^n(\mathbf{p})|0\rangle.$$
(3.69)

Given two regions A and B defined as in the previous section, we express the ratio of charged moments of the partial transpose also in this case as (up to a normalisation constant)

$$\frac{{}^{n}\langle \Psi^{(1)}|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|\Psi^{(1)}\rangle^{n}}{{}^{n}\langle 0|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|0\rangle^{n}}.$$
(3.70)

We then expand the expectation value of the twist operators as follows:

$${}^{n}\langle 0|(\Psi^{\dagger})^{n}(-\mathbf{p})\dots(\Psi^{\dagger})^{1}(-\mathbf{p})(\Psi^{2}_{A}(\mathbf{p})+\Psi^{n}_{B}(\mathbf{p})e^{-2\pi i\alpha}+\Psi^{1}_{C}(\mathbf{p}))\cdots\times$$

$$(-\Psi^{1}_{A}(\mathbf{p})e^{2\pi i\alpha}-\Psi^{n-1}_{B}(\mathbf{p})e^{2\pi i\alpha}+\Psi^{n}_{C}(\mathbf{p}))T^{\alpha}_{A}\tilde{T}^{\alpha}_{B}|0\rangle^{n}\simeq$$

$${}^{n}\langle 0|(\Psi^{\dagger})^{n}(-\mathbf{p})\dots(\Psi^{\dagger})^{1}(-\mathbf{p})(\Psi^{2}_{A}(\mathbf{p})+\Psi^{n}_{B}(\mathbf{p})e^{-2\pi i\alpha}+\Psi^{1}_{C}(\mathbf{p}))\cdots\times$$

$$(-\Psi^{1}_{A}(\mathbf{p})e^{2\pi i\alpha}-\Psi^{n-1}_{B}(\mathbf{p})+\Psi^{n}_{C}(\mathbf{p}))|0\rangle^{n}\times{}^{n}\langle 0|T^{\alpha}_{A}\tilde{T}^{\alpha}_{B}|0\rangle^{n}.$$

$$(3.71)$$

As in the bosonic case, many terms are generated from the expansion of the above expression, and we can apply similar considerations as to which of these terms are leading and which are sub-leading in the large-volume limit. However, since the fermionic fields anticommute, we need to pay attention to the order of the fields. For example, following the notation (3.56), the term $(A \ A \dots A)$ can be evaluated to

$${}^{n}\langle 0|(\Psi^{\dagger})^{n}(-\mathbf{p})\dots(\Psi^{\dagger})^{1}(-\mathbf{p})\Psi_{A}^{2}(\mathbf{p})\dots\Psi_{A}^{1}(\mathbf{p})|0\rangle^{n}(-e^{2\pi i\alpha}) =$$

$${}^{n}\langle 0|(\Psi^{\dagger})^{n}(-\mathbf{p})\dots(\Psi^{\dagger})^{1}(-\mathbf{p})\Psi_{A}^{1}(\mathbf{p})\Psi_{A}^{2}(\mathbf{p})\dots\Psi_{A}^{n}(\mathbf{p})|0\rangle^{n}e^{2\pi i\alpha} \simeq e^{2\pi i\alpha}V_{A}^{n},$$
(3.72)

where $\Psi_A^1(\mathbf{p})$ has been recast in the first position after crossing n-1 (odd) fermions, thus acquiring an additional phase -1. Similarly, it is easy to show that

$${}^{n}\langle 0|(\Psi^{\dagger})^{n}(-\mathbf{p})\dots(\Psi^{\dagger})^{1}(-\mathbf{p})\Psi_{B}^{n}(\mathbf{p})\dots\Psi_{B}^{n-1}(\mathbf{p})|0\rangle^{n}(-e^{-2\pi i\alpha})\simeq V_{B}^{n}e^{-2\pi i\alpha}.$$
(3.73)

In general, each term of the expansion is weighted with a phase which arises from the commutation relations between twist operators and fermions and from those between fermions and fermions. This is the crucial difference with respect to the calculation presented for the boson. We can summarise the total contribution to the phase for a generic $term^2$

$$(A_1...A_n) := {}^n \langle 0 | (\Psi^{\dagger})^n (-\mathbf{p}) \dots (\Psi^{\dagger})^1 (-\mathbf{p}) \Psi_{A_1}^{j_1}(\mathbf{p}) \dots \Psi_{A_n}^{j_n}(\mathbf{p}) | 0 \rangle^n,$$
(3.74)

as follows:

- If $j_n = 1$ and $A_n = A$ there is a $-e^{2\pi i \alpha}$ phase. Similarly, if $j_1 = 1$, and $A_1 = B$, there is a contribution of $-e^{-2\pi i \alpha}$.
- In addition to the previous phase, an additional -1 is present for each B which appears in the string $(A_2 \ldots A_n)$. This is due to the fermionic partial transposition over B.
- Once the contraction is performed, there is a sign coming from the order of the fields. Given $(j_1 \ldots j_n) = (\sigma(1) \ldots \sigma(n))$, with σ a permutation of the replica indices, one can show that the sign appearing after the contraction is sign (σ) .

Having suitably modified the definition of the twist operators for the fermion, the phase of each term appearing in (3.71) is the same as the one for the corresponding term in (3.53), leading to the same result for fermions and bosons. To see this, let us first analyse a term of type-I:

$$(A_1, \dots, A_n) = (B, A_2, \dots, A_{n-1}, A).$$
 (3.75)

The phases coming from the last A and the first B cancel each other. Let $A_{i+1} = B$: then this must be preceded by $A_i = A$. The resulting replica indices at the corresponding positions are $j_{i+1} = i$ and $j_i = i + 1$. In other words, there is an exchange of the replica indices i and i + 1, which changes the sign of the permutation σ and it contributes as -1 after the contraction, but it is compensated by the -1 due to the presence of a B. Similar considerations apply straightforwardly to the strings of type-II. Putting everything together, the same formula (3.14) is obtained again, which is the final result. We emphasise that this derivation relies on the assumption that n is even, something that was not necessary for a single bosonic excitation.

3.3.2.1 Single fermionic excitation via replica diagonalisation

In this section, we show that the result just derived for fermions in full generality can also be obtained via replica diagonalisation in the free case in 1+1D. This is essentially the procedure employed in Section 2 of the previous chapter, but the twist operator approach, together with the simple assumptions on the large-volume behaviour of the OPE, allows us to bypass the lengthy form factor calculations. The key observation is that, as show in Ref. [182], the fermionic partial transpose of a Gaussian state is still Gaussian. This allows to simplify the analysis of the replica theory, reducing it

²given a sequence $(A_1 \ldots A_n)$, the indices j_i are fixed by (3.57) also for the fermion.

to a single-replica model in the presence of proper fluxes. For instance, given a Gaussian state and its RDM, and taking again n even, one can show that the following factorisation [182] holds:

$$\operatorname{Tr}\left(|\rho_{A\cup B}^{R_B}|^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \operatorname{Tr}\left(\rho_{A\cup B} e^{\frac{2\pi i (\alpha+p)}{n} (\hat{Q}_A - \hat{Q}_B) + i\pi \hat{Q}_B}\right).$$
(3.76)

Each term appearing inside the product is a single-copy charged partition function with fluxes along A and B. Thus, we can evaluate the ratio of charged moments of the partial transpose as a product of ratios of single-copy correlators of twist operators, the only difference being that now we need to shift the flux α in T_A^{α} :

$$\mathcal{R}_{1}^{n}(r_{A}, r_{B}, r; \alpha) = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \frac{\operatorname{Tr}\left(\rho_{A\cup B}e^{\frac{2\pi i(\alpha+p)}{n}(\hat{Q}_{A}-\hat{Q}_{B})+i\pi\hat{Q}_{B}}\right)}{\operatorname{Tr}\left(\rho_{A\cup B,0}e^{\frac{2\pi i(\alpha+p)}{n}(\hat{Q}_{A}-\hat{Q}_{B})+i\pi\hat{Q}_{B}}\right)} \propto \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \frac{\langle\Psi|T_{A}^{\alpha+p}T_{B}^{-\alpha-p+\frac{1}{2}}|\Psi\rangle}{\langle0|T_{A}^{\alpha+p}T_{B}^{-\alpha-p+\frac{1}{2}}|0\rangle},$$
(3.77)

where the twist operators are defined by (3.66) with n = 1 and the proportionality constant is fixed by normalising the state. A similar calculation as in the previous sections for a one-particle states gives

$$\langle \Psi^{(1)} | T_A^{\alpha} T_B^{-\alpha + \frac{1}{2}} | \Psi^{(1)} \rangle = \langle 0 | \Psi^{\dagger}(\mathbf{p}) T_A^{\alpha} T_B^{-\alpha + \frac{1}{2}} \Psi(\mathbf{p}) | 0 \rangle =$$

$$\langle 0 | \Psi^{\dagger}(\mathbf{p}) (\Psi_A(\mathbf{p}) e^{2\pi i \alpha} - \Psi_B(\mathbf{p}) e^{-2\pi i \alpha} + \Psi_C(\mathbf{p})) T_A^{\alpha} T_B^{-\alpha + \frac{1}{2}} | 0 \rangle \simeq$$

$$\langle 0 | T_A^{\alpha} T_B^{-\alpha + \frac{1}{2}} | 0 \rangle (V_A e^{2\pi i \alpha} - V_B e^{-2\pi i \alpha} + V_C),$$

(3.78)

so that:

$$\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \frac{\langle \Psi^{(1)} | T_A^{\alpha+p} T_B^{-\alpha-p+\frac{1}{2}} | \Psi^{(1)} \rangle}{\langle 0 | T_A^{\alpha+p} T_B^{-\alpha-p+\frac{1}{2}} | 0 \rangle} \frac{\langle 0 | 0 \rangle}{\langle \Psi^{(1)} | \Psi^{(1)} \rangle} = \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} (r_A e^{\frac{2\pi i (\alpha+p)}{n}} - r_B e^{-\frac{2\pi i (\alpha+p)}{n}} + r).$$
(3.79)

This product can be shown yet again to be equal to (3.14), although the proof requires some mathematical identities that we present in Appendix (3.B.2).

3.3.3 Bosonic state with multiple distinct excitations

Consider now a k-particle bosonic state where all particles have the same (unitary) charge and momentum **p**. In order to ensure the presence of U(1) symmetry we consider the complex free boson, described by a field \mathcal{O} which satisfies Wick's theorem in the vacuum state. We thus describe the excited state as

$$|\Psi^{(k)}\rangle = (\mathcal{O}(\mathbf{p}))^k |0\rangle.$$
(3.80)

Before entering the core of the computation in the replica model, it is convenient to slightly modify the definition (2.120) of the twist operators as follows

$$T_{A}^{\alpha}\mathcal{O}^{j}(\mathbf{x}) = \begin{cases} e^{\frac{2\pi i\alpha}{n}}\mathcal{O}^{j+1}(\mathbf{x})T_{A}^{\alpha} & \mathbf{x} \in A, \\ \mathcal{O}^{j}(\mathbf{x})T_{A}^{\alpha} & \mathbf{x} \notin A. \end{cases}$$
(3.81)

This amounts to distributing the total flux $e^{2\pi i\alpha}$ in equal part among all copies, rather than inserting it between the *n*-th and the first replica only. The expectation values are not affected by this choice, which is however computationally useful. An analogous fractionalisation will be considered for \tilde{T}_A^{α} . We can now evaluate

$$\frac{{}^{n}\langle\Psi^{(k)}|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|\Psi^{(k)}\rangle^{n}}{{}^{n}\langle\Psi^{(k)}|\Psi^{(k)}\rangle^{n}} = \frac{{}^{n}\langle0|\left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\right)^{k}\dots\left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\right)^{k}T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}\left(\mathcal{O}^{1}(\mathbf{p})\right)^{k}\dots\left(\mathcal{O}^{n}(\mathbf{p})\right)^{k}|0\rangle^{n}}{{}^{n}\langle0|\left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\right)^{k}\dots\left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\right)^{k}\left(\mathcal{O}^{1}(\mathbf{p})\right)^{k}\dots\left(\mathcal{O}^{n}(\mathbf{p})\right)^{k}|0\rangle^{n}}.$$
 (3.82)

The denominator, required to ensure normalisation, can be computed using Wick's theorem, which gives the expectation value as a sum over the possible contractions:

$${}^{n}\langle 0|\left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\right)^{k}\dots\left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\right)^{k}\left(\mathcal{O}^{1}(\mathbf{p})\right)^{k}\dots\left(\mathcal{O}^{n}(\mathbf{p})\right)^{k}|0\rangle^{n} = \left[\langle 0|\left((\mathcal{O}^{\dagger})(-\mathbf{p})\right)^{k}\left(\mathcal{O}(\mathbf{p})\right)^{k}|0\rangle\right]^{n} \simeq V^{nk}(k!)^{n}.$$
(3.83)

The numerator in (3.82) can be manipulated in a way analogous to the single-particle state, now using the exchange relation (3.81):

$${}^{n}\langle 0|\left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\right)^{k}\ldots\left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\right)^{k}T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}\left(\mathcal{O}^{1}(\mathbf{p})\right)^{k}\ldots\left(\mathcal{O}^{n}(\mathbf{p})\right)^{k}|0\rangle^{n} =$$

$${}^{n}\langle 0|\left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\right)^{k}\ldots\left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\right)^{k}\left(\mathcal{O}_{A}^{2}(\mathbf{p})e^{\frac{2\pi i\alpha}{n}}+\mathcal{O}_{B}^{n}(\mathbf{p})e^{-\frac{2\pi i\alpha}{n}}+\mathcal{O}_{C}^{1}(\mathbf{p})\right)^{k}\cdots\times\left(\mathcal{O}_{A}^{1}(\mathbf{p})e^{\frac{2\pi i\alpha}{n}}+\mathcal{O}_{B}^{n-1}(\mathbf{p})e^{-\frac{2\pi i\alpha}{n}}+\mathcal{O}_{C}^{n}(\mathbf{p})\right)^{k}T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|0\rangle^{n}\simeq$$

$${}^{n}\langle 0|\left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p})\right)^{k}\ldots\left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p})\right)^{k}\left(\mathcal{O}_{A}^{2}(\mathbf{p})e^{\frac{2\pi \alpha}{n}}+\mathcal{O}_{B}^{n}(\mathbf{p})e^{-\frac{2\pi i\alpha}{n}}+\mathcal{O}_{C}^{1}(\mathbf{p})\right)^{k}\cdots\times\left(\mathcal{O}_{A}^{1}(\mathbf{p})e^{\frac{2\pi i\alpha}{n}}+\mathcal{O}_{B}^{n-1}(\mathbf{p})e^{-\frac{2\pi i\alpha}{n}}+\mathcal{O}_{C}^{n}(\mathbf{p})\right)^{k}|0\rangle^{n}\times{}^{n}\langle 0|T_{A}^{\alpha}\tilde{T}_{B}^{\alpha}|0\rangle^{n}.$$

$$(3.84)$$

The evaluation of the previous expression in the general case is a hard combinatorial task, but many crucial features are already apparent. Namely, it is clear that 3^{nk} terms are generated simply from the expansion of all the products. Each of these terms is a string containing nk daggered operators $(\mathcal{O}^{\dagger})^{j}(-\mathbf{p})$, where every $j \in \{1, \ldots, n\}$ appears exactly k times, followed by nk operators $\mathcal{O}_{A_{i}}^{j}(\mathbf{p})$: to have a non vanishing expectation value, also in these latter operators every $j \in \{1, \ldots, n\}$ must appear exactly k times. Each of the non vanishing terms can then be evaluated via Wick's theorem and will give rise to precisely $(k!)^{n}$ identical contractions, thus canceling the combinatorial factor in Eq. (3.83). Moreover, whenever the restriction of $\mathcal{O}_{A}^{j}(\mathbf{p})$ over the region A appears, a factor $V_{A}e^{\frac{2\pi i \alpha}{n}}$ is present after the Wick contraction; similarly, a factor $V_B e^{-\frac{2\pi i \alpha}{n}}$ appears with every B and a factor V_C for every C. Putting everything together, we can infer the general structure

$${}^{n}\langle 0| \left((\mathcal{O}^{\dagger})^{n}(-\mathbf{p}) \right)^{k} \dots \left((\mathcal{O}^{\dagger})^{1}(-\mathbf{p}) \right)^{k} \left(\mathcal{O}_{A}^{2}(\mathbf{p}) e^{2\pi i \alpha/n} + \mathcal{O}_{B}^{n}(\mathbf{p}) e^{-2\pi i \alpha/n} + \mathcal{O}_{C}^{1}(\mathbf{p}) \right)^{k} \dots \times \\ \left(\mathcal{O}_{A}^{1}(\mathbf{p}) e^{2\pi i \alpha/n} + \mathcal{O}_{B}^{n-1}(\mathbf{p}) e^{-2\pi i \alpha/n} + \mathcal{O}_{C}^{n}(\mathbf{p}) \right)^{k} |0\rangle^{n} \simeq \\ (k!)^{n} \sum_{k_{A},k_{B}} C_{k_{A},k_{B}} V_{A}^{k_{A}} V_{B}^{k_{B}}(V_{C})^{nk-k_{A}-k_{B}} e^{\frac{2\pi i \alpha(k_{A}-k_{B})}{n}},$$

$$(3.85)$$

so that the expectation value is a homogeneous polynomial of degree nk in V_A, V_B, V_C , and C_{k_A,k_B} is a combinatorial coefficient. Thus, generalising (3.62), the ratio of charged moments is:

$$\mathcal{R}_{k}^{n}(r_{A}, r_{B}, r; \alpha) = \frac{n \langle \Psi^{(1)} | T_{A}^{\alpha} \tilde{T}_{B}^{\alpha} | \Psi^{(1)} \rangle^{n}}{n \langle \Psi^{(1)} | \Psi^{(1)} \rangle^{n}} \frac{n \langle 0 | 0 \rangle^{n}}{n \langle 0 | T_{A}^{\alpha} \tilde{T}_{B}^{\alpha} | 0 \rangle^{n}} = \sum_{k_{A}, k_{B}} C_{k_{A}, k_{B}} r_{A}^{k_{A}} r_{B}^{k_{B}} r^{nk - k_{A} - k_{B}} e^{\frac{2\pi i \alpha (k_{A} - k_{B})}{n}}, \qquad (3.86)$$

valid as usual in the infinite volume limit. The closed formula for the combinatorial coefficient C_{k_A,k_B} at any k and n is difficult to obtain by this method, but it has been obtained earlier for simpler qubit states. Indeed, The numbers C_{k_A,k_B} are nothing but the coefficients $\mathcal{A}_{p,q}$ in (3.18). Rather then giving a complete proof of this claim, in the next section we study in detail a simple example.

3.3.3.1 An example: k = n = 2

Let us consider the example n = 2 and k = 2 to get an idea of how the combinatorics of (3.85) works in this case. First, we define the symbol

$$\begin{pmatrix} A_1 \dots A_{nk} \\ j_1 \dots j_{nk} \end{pmatrix} :=^n \langle 0 | \left((\mathcal{O}^{\dagger})^n (-\mathbf{p}) \right)^k \dots \left((\mathcal{O}^{\dagger})^1 (-\mathbf{p}) \right)^k \mathcal{O}_{A_1}^{j_1}(\mathbf{p}) \dots \mathcal{O}_{A_{kn}}^{j_{kn}}(\mathbf{p}) | 0 \rangle^n.$$
(3.87)

Among the strings which are generated, we only keep those for which any replica index j appears exactly k times among $(j_1 \dots j_{nk})$, as all others will vanish after Wick contractions. For n = k = 2 the length of the strings above is nk = 4. We notice that, unlike the case of a single excitation, the indices j_i in (3.87) are not uniquely fixed by the corresponding A_i 's: in particular, there may be different permutations of the A_i 's corresponding to the same sequence of j_i 's. For any given string, there are others that can be obtained via the following permutations of the A_i indices:

$$(A_1 \ A_2 \ A_3 \ A_4) \to (A_2 \ A_1 \ A_3 \ A_4), \quad (A_1 \ A_2 \ A_3 \ A_4) \to (A_1 \ A_2 \ A_4 \ A_3), (A_1 \ A_2 \ A_3 \ A_4) \to (A_3 \ A_4 \ A_1 \ A_2),$$
(3.88)

corresponding to a swap of the two operators in the first copy, a swap of the two operators in the second copy and to a cyclic permutation of the copies respectively. All these terms contribute equally.

Furthermore, as noticed in the previous section, each non-vanishing string yields a combinatorial factor $(k!)^n = 4$. These considerations allow us to slightly simplify the combinatorial counting, and we only list the terms coming from strings up to the transformations generated by Eq. (3.88), taking care of the degeneracy for each representative distinct string. Up to the combinatorial factors, the contributing terms are:

- $(A \ A \ A \ A)$: it yields $(V_A e^{\frac{2\pi i \alpha}{2}})^4$.
- $(B \ B \ B \ B)$: it yields $(V_B e^{-\frac{2\pi i \alpha}{2}})^4$.
- (A A B B) and (A B A B): they yield $6V_A^2 V_B^2$.
- $(B \ A \ A \ A)$: it yields $4(V_A e^{\frac{2\pi i \alpha}{2}})^3(V_B e^{-\frac{2\pi i \alpha}{2}})$.
- (A B B B): it yields $4(V_B e^{-2\pi i \alpha/2})^3 (V_A e^{2\pi i \alpha/2})$.
- $(A \ C \ B \ C)$: it yields $8V_A V_B V_C^2$.
- $(C \ C \ C \ C)$: it yields V_c^4 .
- $(A \ C \ A \ C)$: it yields $4V_c^2(V_A e^{\frac{2\pi i \alpha}{2}})^2$.
- (B C B C): it yields $4V_C^2(V_B e^{-\frac{2\pi i\alpha}{2}})^2$.

Putting all these pieces together and diving the result by (3.83), we obtain

$$\mathcal{R}_{2}^{2}(r_{A}, r_{B}, r; \alpha) = r_{A}^{4} e^{4\pi i \alpha} + r_{B}^{4} e^{-4\pi i \alpha} + 6r_{A}^{2} r_{B}^{2} + 4r_{A}^{3} r_{B} e^{2\pi i \alpha} + 4r_{B}^{3} r_{A} e^{-2\pi i \alpha} + 8r_{A} r_{B} r_{A} e^{-2\pi i \alpha} + 8r_{A} r_{B} r_{A} e^{-2\pi i \alpha} + 4r_{B}^{2} r_{A}^{2} e^{-2\pi i \alpha} + 4r_{B}^{2} r_{A}^{2} e^{-2\pi i \alpha} + 6r_{A}^{2} r_{B}^{2} e^{-2\pi i \alpha} + 4r_{B}^{2} r_{A}^{2} e^{-2\pi i \alpha} + 6r_{A}^{2} r_{A}^{2} e^{-2\pi$$

This result is consistent with the one obtained from a direct evaluation of the right hand side of (3.17) for n = 2, k = 2.

3.4 Numerics

In this section, we present numerical results for a 1D lattice Fermi gas, the same model we already considered in Section 2.5.1. In particular, we consider the ground state at vanishing chemical potential, which is a Fermi sea and has critical features. This Fermi sea is then excited through the insertion of an additional particle above the Fermi energy at large momentum. We aim to compute the ratio of charged moments for a state of a single excitation and show the validity of result (3.14) numerically. Agreement with the latter confirms the claim made earlier in this chapter and in the previous one, namely that while the ground state exhibits theory-dependent, highly non-trivial behavior (in this case, captured by a free fermion CFT [120]), the contribution given by the excitation is universal.

3.4.1 The method

Let us consider a Fermi chain of length L described by the fermionic operators $\{f_j, f_j^{\dagger}\}_{j=1,...,L}$ satisfying the standard anticommutation relations

$$\{f_j, f_{j'}\} = \{f_j^{\dagger}, f_{j'}^{\dagger}\} = 0, \qquad \{f_j, f_{j'}^{\dagger}\} = \delta_{jj'}.$$
(3.90)

We choose a Gaussian state with a given number of particles and consider its correlation matrix, denoted by

$$C(j,j') = \langle f_j^{\dagger} f_{j'} \rangle, \quad j,j' = 1,\dots,L.$$

$$(3.91)$$

Let us further define the $L \times L$ covariance matrix

$$\Gamma = 1 - 2C. \tag{3.92}$$

Given any two disjoint spacial subsystems A and B, of length ℓ_A, ℓ_B respectively, the restriction of Γ over $A \cup B$ is a $(\ell_A + \ell_B) \times (\ell_A + \ell_B)$ matrix defined by

$$\Gamma_{A\cup B} = \begin{pmatrix} \Gamma_{AA} & \Gamma_{AB} \\ \Gamma_{BA} & \Gamma_{BB} \end{pmatrix}, \qquad (3.93)$$

Following [189] one can show that, if $\rho_{A\cup B}$ is the RDM for this system, the fermionic partial transposition $\rho_{A\cup B}^{R_B}$ is Gaussian. Moreover, since in general $\rho_{A\cup B}^{R_B}$ is not Hermitian, it is convenient to introduce a matrix ρ^{\times} defined as

$$\rho^{\times} = \frac{(\rho_{A\cup B}^{R_B})(\rho_{A\cup B}^{R_B})^{\dagger}}{\operatorname{Tr}(\rho_{A\cup B}^2)}.$$
(3.94)

From the definition of partial time-reversal transposition, it is possible to show [190] that ρ^{\times} has unit trace. If one interprets ρ^{\times} as an unphysical mixed state of $A \cup B$, its associated covariance matrix is³ (See [189, 191])

$$\Gamma_{A\cup B}^{\times} \equiv \frac{2}{1+\Gamma_{A\cup B}^2} \begin{pmatrix} \Gamma_{AA} & 0\\ 0 & -\Gamma_{BB} \end{pmatrix}.$$
(3.95)

One can then express the even charged moments of the partially transposed RDM in terms of the eigenvalues of $\Gamma_{A\cup B}$ and $\Gamma_{A\cup B}^{\times}$ as (See also [120])

$$\log \operatorname{Tr}\left(|\rho_{A\cup B}^{R_B}|^n e^{2\pi i \alpha (\hat{Q}_A - \hat{Q}_B)}\right) = \operatorname{Tr}\log\left(\left(\frac{1 - \Gamma_{A\cup B}^{\times}}{2}\right)^{\frac{n}{2}} e^{2\pi i \alpha} + \left(\frac{1 + \Gamma_{A\cup B}^{\times}}{2}\right)^{\frac{n}{2}}\right) + \frac{n}{2} \operatorname{Tr}\log\left(\left(\frac{1 + \Gamma_{A\cup B}}{2}\right)^2 + \left(\frac{1 - \Gamma_{A\cup B}}{2}\right)^2\right).$$
(3.96)

³Strictly speaking, the correlation matrix is unitarily equivalent to the one in Eq. (3.95), as shown in [190]. However, this is not important for our purpose, as we are interested on its spectrum only.

We stress that Eq. (3.96) makes sense also if n is not an even integer, and it naturally provides the analytic continuation over n for the even charged moments.

3.4.2 Lattice Fermi gas

For our numerics we take the Hamiltonian of a lattice free Fermi gas on a ring of length L

$$H = -\frac{1}{2} \sum_{j} f_{j+1}^{\dagger} f_{j} + f_{j}^{\dagger} f_{j+1} \,. \tag{3.97}$$

Its ground state is a Fermi sea, with Fermi momentum $k_F = \pi/2$, and its correlation matrix is

$$C_0(j,j') := \langle f_j^{\dagger} f_{j'} \rangle_0 = \frac{\sin k_F(j-j')}{L \sin \frac{\pi(j-j')}{L}}.$$
(3.98)

We then consider the excited state obtained via the insertion of a particle at momentum

$$k = k_F + \frac{\pi}{2} - \frac{\pi}{L},\tag{3.99}$$

above the Fermi sea, whose correlation matrix is

$$C(j,j') = C_0(j,j') + \frac{1}{L}e^{-i(k_F + \frac{\pi}{2} - \frac{\pi}{L})(j-j')}.$$
(3.100)

Notice that this state differs from the particle-hole excitation considered in 2.5.1. While the specific choice of k is irrelevant for our purpose, it is important to require that $k - k_F$ is finite in the thermodynamic limit. We then consider the subsystems

$$A = \{1, \dots, \ell_A\}, \quad B = \{\ell_A + 1, \dots, \ell_A + \ell_B\},$$
(3.101)

and we fix the size of $A \cup B$ to be half the subsystem size:

$$\frac{\ell_A + \ell_B}{L} = \frac{1}{2}.$$
 (3.102)

Finally, by using (3.96) we evaluate numerically the difference of charged Rényi negativities

$$\mathcal{E}_{n}(\alpha) - \mathcal{E}_{n,0}(\alpha) := \log \frac{\operatorname{Tr}\left(|\rho_{A\cup B}^{R_{B}}|^{n}e^{2\pi i\alpha(\hat{Q}_{A}-\hat{Q}_{B})}\right)}{\operatorname{Tr}\left(|\rho_{A\cup B,0}^{R_{B}}|^{n}e^{2\pi i\alpha(\hat{Q}_{A}-\hat{Q}_{B})}\right)},$$
(3.103)

for some values of the flux α as a function of $r_A = \ell_A/L$, and we compare it with the prediction (3.14) (with $r_B = \ell_B/L = 1/2 - r_A$). In figures 3.1 and 3.2 we show the results for a chain of length L = 400and given values of n and α , while varying the value of r_A from 0 to 1/2. We consider also non-even



Figure 3.1 Difference of (uncharged, $\alpha = 0$) Rényi negativities for the one-particle state at n = 0.5, 1, 2. Note that although we derived equation (3.14) for n even via the replica approach, we can analytically continue the latter to any value of n. The theoretical results are the dashed lines, and the numerical values are the dots.

values of n, and we compare the numerics with the analytic continuation (over the even integers) of our predictions. The agreement between the numerical results and equation (3.14) is good, even if there are small discrepancies for small values of ℓ_A or ℓ_B (corresponding to $r_A \simeq 0$ and $r_A \simeq 0.5$) due to finite-size effects which vanish in the large-volume limit.

3.5 Concluding remarks

In this chapter we concluded the investigation of U(1) symmetry-resolved entanglement measures in zero-density excited states. Zero-density here means that volume is taken to infinity, while the number of excitations above the ground state (which may be trivial, as for a qubit state, or highly non-trivial as in QFT) is kept fixed and finite. The results presented in this chapter and in the previous one extend work on entanglement measures for zero-density excited states carried out in [114–116, 144]. Other important contributions to this research field are [146, 148–150, 152, 192, 193]. In line with the results of the previous chapter for the Rényi entropies, we expected and indeed found that the contribution of a finite number of excitations to the symmetry-resolved (logarithmic) negativity is given by a simple formula, a polynomial on the variables r_A, r_B and $r = 1 - r_A - r_B$, which represent the relative sizes of two subsystems A and B and their complement, respectively. For the symmetry-resolved moments of the negativity, this polynomial depends also on a parameter α related to the internal U(1) symmetry of the theory. The formulae that we obtained generalise the results for the uncharged moments in a simple way and are consistent with numerical results. However, some of the methods that we have employed to obtain these results are quite new and have potential for further use.

The method of twist operators, suggested by the computation of entanglement measures in d-dimensional free bosonic theories [144], was introduced in the previous chapter and here we provided a very non-trivial check of its validity. From this method alone, we can claim that our



Figure 3.2 Difference of charged Rényi negativities for the one-particle state at flux $\alpha = 0.1, 0.2, 0.3, 0.4$ and n = 1, 2 evaluated numerically (dots) versus the analytical predictions (3.14) (dashed lines). The left (right) panels show the real (imaginary) part of $\mathcal{E}_n(\alpha) - \mathcal{E}_{n,0}(\alpha)$. The size of the chain is L = 400, and we plot the results as functions of $r_A \in (0, 1/2)$.

formulae should be valid in any dimensionality and in the presence of short-range interactions as well. Compared to a computation based on branch-point twist fields for free QFT, as performed for the negativity in [116], the use of twist operators captures the same universal result through a significantly simpler computation. A further application of twist operators and one of the most interesting and novel results of this chapter is the fact these operators can be easily adapted to treat particles with both fermionic and bosonic statistics. In particular, it has been known for some time that the negativity of fermionic theories requires a redefinition of the operation of partial transposition [181, 182]. Here we find that, first, this redefinition is easy to implement in the context of twist operators and, second, that once implemented it leads to a result which is the same as for bosons. This ties in well with the idea that the universal part of the entanglement associated with these types of excitation has a semiclassical interpretation (as recently explored in [152]), thus the statistics of excitations plays no role at leading order in the large-volume expansion.

Looking ahead, there are many directions to explore in relation to the role of quasiparticle excitations in the context of symmetry-resolved entanglement measures. Via twist operators, any new measures of charged entanglement should be computable by a suitable redefinition of the operators. In particular, a quantum distance such as the relative entropy [113] can be computed to detect a difference in the entanglement content of zero-density states which have the same symmetry-resolved entropy and/or negativity. The symmetry-resolved relative entropy of excited states in 1+1D CFT was studied in [97].

Equally interesting would be the investigation of the crossover from low- to high-energy states in CFT, from the model-dependent predictions of [154] to the sort of universal results obtained in [1, 2, 114–116, 144] and here. The results of [154] apply to low-lying excited states of CFT, whereas the universal formulae obtained for zero-density excited states apply for large momentum/energy. This suggests that there must be a crossover between these two behaviours, which could be understood using CFT arguments.

Finally, twist operators seem to be a promising approach to computing entanglement measures in limiting cases where many details of the interaction can be neglected, i.e. the semiclassical limit. A field where these ideas can be applied is that of out-of-equilibrium protocols [136, 194]. In this context, characterising the entanglement growth for free or interacting theories in any dimension via the twist operator approach constitutes an interesting challenge. For some protocols, such as a global quench, we expect that the linear growth of entanglement may be captured by a semiclassical approximation of correlation functions, similar to what we considered in this chapter and in the previous one.

3.A Combinatorics

3.A.1 Non-vanishing strings

Here, we show which terms in (3.53) give rise to non-vanishing contractions in the large-volume limit, following the notation introduced in (3.56). As noted after equation (3.56), the strings that produce leading powers of the volume are those where the sequence of indices (j_1, \ldots, j_n) is a permutation $(\sigma(1), \ldots, \sigma(n))$, and the j_i are further constrained by (3.57).

Now suppose that $A_i = A$ for one value of *i*. this means that $j_i = i + 1$ and thus it can be either $j_{i+1} = i + 2$, which fixes $A_{i+1} = A$ or $j_{i+1} = i$, which fixes $A_{i+1} = B$. In the first case the fact that $A_i = A_{i+1} = A$ selects all the other $A_i = A$, while in the second case we simply have a pair AB. Analogously, we can start with $A_i = B$, $j_i = i - 1$. Now the two possibilities are $j_{i-1} = i$, that is $A_{i-1} = A$ and thus again we have a pair AB, or $j_{i-1} = i - 2$, that is $A_{i-1} = B$ and thus $A_i = B$ for every *i*. To summarise, there are only three possibilities: $A_i = A$ for every *i*, $A_i = B$ for every *i*, and finally whenever $A_i = A$, $A_{i+1} = B$. This is the claim presented after equation (3.57).

3.A.2 Combinatorial counting of strings

In this appendix we count the number of non-vanishing strings

$$(A_1 \dots A_n), \tag{3.104}$$

containing k pairs of consecutive A's and B's. We first focus on the type-I strings

$$(B \ A_2 \dots A_{n-1} \ A).$$
 (3.105)

The number of strings that satisfy the constraints derived in Section (3.3.1) is given by all the possible ways one can insert sequences of C's among any pair of A's and B's. In other words, the generic string will look like

$$(B C C \dots C A B C C \dots C A), \tag{3.106}$$

where k sequences of C's of length $\{x_i\}_{i=1,\dots,k}$ are present, and $x_i \ge 0$ are integer numbers. As the length of the total string is n, the $\{x_i\}_{i=1,\dots,k}$ satisfy the following constraint

$$x_1 + \dots + x_k = n - 2k. \tag{3.107}$$

We now make use of a remarkable mathematical result, namely that the number of non-negative integer solutions of $x_1 + \cdots + x_k = n$, that is the number of non-negative integer partitions of n into k parts is $\binom{n+k-1}{n}$ [195]. As a consequence, the number of type-I strings satisfying the previous constraints is

$$\binom{n-k-1}{k-1}.$$
(3.108)

Similarly, we consider now the type-II strings, having the following structure

$$(C C \dots C A B C C \dots C). \tag{3.109}$$

In this case, when there are k pairs of consecutive A's and B's, there are k+1 sequences of consecutive C's. Thus, we now have to count the number of non-negative integer solutions of

$$x_1 + \dots + x_{k+1} = n - 2k, \tag{3.110}$$

which is

$$\binom{n-k}{k}.$$
(3.111)

Summing up the contribution of both type of strings, we get precisely

$$\binom{n-k-1}{k-1} + \binom{n-k}{k} \tag{3.112}$$

as the total number of strings containing k pairs of consecutive A's and B's, and this proves the result (3.60). We notice that in type-I strings there is always at least one pair of consecutive A's and B's, given by $A_n = A$ and $A_1 = B$. Then, if k = 0 in (3.108) there are no strings satisfying the constraints, which is compatible with the convention $\binom{n-1}{-1} = 0$.

As a final remark, we stress that the problem of counting the number of type-I and type-II strings containing k pairs of consecutive A's and B's is identical to the problem mentioned after equation (3.42), namely counting the number of sequences (k_1, \ldots, k_n) with $k_i \in \{0, 1\}$ and $k_{i+1} = 0$ whenever $k_i = 0$.

3.B Mathematical identities

3.B.1 Generalised Lucas polynomials and a proof of equation (3.63)

In this appendix we prove the identity (3.63) by taking advantage of some properties of generalised Lucas polynomials. These are polynomials V(x, y) defined by the recurrence relation [196, 197]:

$$V_{n+2}(x,y) = xV_{n+1}(x,y) + yV_n(x,y), \quad n \in \mathbb{N}_0,$$
(3.113)

the first two polynomials being $V_0(x, y) = 2$, $V_1(x, y) = x$. The proof of (3.63) is based on the fact that the two sides of the equation are precisely two equivalent closed formulae for the *n*-th Lucas polynomial, with x = r, $y = r_A r_B$. In fact, we will now prove the following two statements:

1. For all integers $n \ge 0$, one has a generalised Binet formula

$$V_n(x,y) = \alpha^n + \beta^n, \quad \alpha = \frac{x + \sqrt{x^2 + 4y}}{2}, \quad \beta = \frac{x - \sqrt{x^2 + 4y}}{2}.$$
 (3.114)

This is immediate to prove, as (3.113) holds by inspection for n = 0, n = 1 and furthermore $\alpha^2 = x\alpha + y$, $\beta^2 = x\beta + y$, which implies that $\alpha^{n+2} = x\alpha^{n+1} + y\alpha^n$ and $\beta^{n+2} = x\beta^{n+1} + y\beta^n$. This means that $\alpha^n + \beta^n$ satisfies the relation (3.113) for all $n \ge 0$.

2. For all integers $n \ge 1$ another explicit formula for $V_n(x, y)$ is given by

$$V_n(x,y) = \sum_{k=0}^{\lfloor n/2 \rfloor} \frac{n}{n-k} \binom{n-k}{k} x^{n-2k} y^k .$$
(3.115)

We prove this by showing again that the recurrence relation is satisfied. For n = 1, n = 2 it is immediate to see that this reproduces the correct polynomials. For $n \ge 3$, we can make use of the identity

$$\frac{n}{n-k}\binom{n-k}{k} = \frac{n-1}{n-k-1}\binom{n-k-1}{k} + \frac{n-2}{n-k-1}\binom{n-k-1}{k-1},$$
 (3.116)

and we adopt the convention that $\binom{n}{k} = 0$ if k > n or k < 0. Let us now consider n = 2m, the case of n odd being completely analogous. If n = 2m, [n/2] = m, [(n-1)/2] = [(n-2)/2] = m-1. From (3.115) and (3.116) we have

$$V_{n}(x,y) = \sum_{k=0}^{m} \frac{2m}{2m-k} {\binom{2m-k}{k}} x^{2m-2k} y^{k}$$

$$= x \sum_{k=0}^{m} \frac{2m-1}{2m-1-k} {\binom{2m-1-k}{k}} x^{2m-1-2k} y^{k}$$

$$+ y \sum_{k=0}^{m} \frac{2m-2}{2m-1-k} {\binom{2m-1-k}{k-1}} x^{2m-2k} y^{k-1}.$$
 (3.117)

The first sum in the right-hand side vanishes if k = m, so that this term is $xV_{n-1}(x,y)$. The second sum on the other hand vanishes if k = 0, so we can shift the summation variable and we see that this term reproduces $yV_{n-2}(x,y)$. Hence the recurrence relation is proved.

Equation (3.63) follows from the identity of expressions (3.114) and (3.115), and it has two interesting implications. The first one comes from a direct expansion of the Binet formula using the binomial theorem:

$$\left(\frac{x-\sqrt{x^{2}+4y}}{2}\right)^{n} + \left(\frac{x+\sqrt{x^{2}+4y}}{2}\right)^{n} \\
= \frac{1}{2^{n}} \sum_{j=0}^{n} \left[(-1)^{k} \binom{n}{j} x^{n-j} (x^{2}+4y)^{j/2} + \binom{n}{j} x^{n-j} (x^{2}+4y)^{j/2} \right] \\
= 2^{1-n} \sum_{j=0}^{[n/2]} \left[\binom{n}{2j} x^{n-2j} (x^{2}+4y)^{j} \right] \\
= 2^{1-n} \sum_{j=0}^{[n/2]} \binom{n}{2j} x^{n-2j} \sum_{k=0}^{j} \binom{j}{k} x^{2j-2k} 2^{2k} y^{k} = 2^{1-n} \sum_{j=0}^{[n/2]} \sum_{k=0}^{j} \binom{n}{2j} \binom{j}{k} x^{n-2k} 2^{2k} y^{k} \\
= 2^{1-n} \sum_{k=0}^{[n/2]} x^{n-2k} 2^{2k} y^{k} \sum_{j=k}^{[n/2]} \binom{n}{2j} \binom{j}{k} = \sum_{k=0}^{[n/2]} \left(2^{1-n+2k} \sum_{j=k}^{[n/2]} \binom{n}{2j} \binom{j}{k} x^{n-2k} y^{k}, \quad (3.118)$$

where in the last line we rearranged the sums over j and k. This quantity equals (3.115), which implies the non-trivial combinatorial identity:

$$\frac{n}{n-k} \binom{n-k}{k} = 2^{1-n+2k} \sum_{j=k}^{\lfloor n/2 \rfloor} \binom{n}{2j} \binom{j}{k}.$$
(3.119)

To the best of our knowledge, this identity was only proved for n odd in [196].

The other interesting implication is obtained for x = y = 1. In this case, the Lucas polynomials (3.113) reduce to the Lucas numbers:

$$L_n = L_{n-1} + L_{n-2}, \quad n \ge 2, \tag{3.120}$$

with $L_0 = 2$, $L_1 = 1$. The recurrence formula is the same defining the Fibonacci sequence, except for the different initial values. Equation (3.114) with x = y = 1 gives a closed formula for the Lucas numbers, and thus we have, for $n \ge 1$:

$$\sum_{k=0}^{[n/2]} \frac{n}{n-k} \binom{n-k}{k} = \left(\frac{1+\sqrt{5}}{2}\right)^n + \left(\frac{1-\sqrt{5}}{2}\right)^n.$$
(3.121)

The quantity on the right-hand side is $\phi^n + (1 - \phi)^n$, with ϕ the golden ratio, and it is always a positive integer. On the other hand, the quantity on the left hand side is the number of non-vanishing strings of type-I and type-II (out of a total of 3^n possible strings) obtained via the contraction methods discussed in Section 3.3.

3.B.2 Product formulae

Here we point out two useful identities which are employed to obtain the free fermion result in Subsection 3.3.2.1. The first relation, which is immediate to check, is:

$$\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} (x+e^{\frac{2\pi i p}{n}}y) = x^n + y^n, \quad n \in \mathbb{N},$$
(3.122)

where the product is performed over p integer (resp. semi-integer) when n is odd (resp. even). A consequence of this identity is:

$$\begin{split} &\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \left(xe^{\frac{2\pi ip}{n}} + ye^{-\frac{2\pi ip}{n}} + z\right) \\ &= x^n \prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} \left[1 + \frac{z - \sqrt{z^2 - 4yx}}{2x}e^{-\frac{2\pi ip}{n}}\right] \left[1 + \frac{z + \sqrt{z^2 - 4yx}}{2x}e^{-\frac{2\pi ip}{n}}\right] \\ &= x^n \left[1 + \left(\frac{z - \sqrt{z^2 - 4yx}}{2x}\right)^n\right] \left[1 + \left(\frac{z + \sqrt{z^2 - 4yx}}{2x}\right)^n\right] \\ &= x^n + y^n + \left(\frac{z - \sqrt{z^2 - 4yx}}{2}\right)^n + \left(\frac{z + \sqrt{z^2 - 4yx}}{2}\right)^n. \end{split}$$

Finally, the equation above can be employed to evaluate

$$\prod_{p=-\frac{n-1}{2}}^{\frac{n-1}{2}} (r_A e^{\frac{2\pi i}{n}(\alpha+p)} - r_B e^{-\frac{2\pi i}{n}(\alpha+p)} + r)$$

= $e^{2\pi i \alpha} r_A^n + e^{-2\pi i \alpha} r_B^n + \left(\frac{r + \sqrt{r^2 + 4r_A r_B}}{2}\right)^n + \left(\frac{-r + \sqrt{r^2 + 4r_A r_B}}{2}\right)^n$, (3.123)

valid if n is an even integer ⁴.

⁴It is important here that *n* is even, as $(-r_B)^n = r_B^n$.

PART II : TWIST OPERATORS AND ENTANGLEMENT MEASURES IN THE ISING FIELD THEORY

CHAPTER

FOUR

TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

All standard measures of bipartite entanglement in one-dimensional QFT can be expressed in terms of correlators of the branch-point twist fields (BPTF) \mathcal{T} and $\tilde{\mathcal{T}}$. These are symmetry fields associated to cyclic permutation symmetry in a replica theory and having the smallest conformal dimension at the critical point. As seen in the previous chapters, the composite twist fields (CTF), typically of higher dimension, play a role in the study of symmetry-resolved measures of entanglement. In this chapter, based on [4], we give an exact expression for the two-point function of a CTF that arises in the Ising field theory. In doing so we extend the techniques originally developed for the standard BPTF in free theories as well as an existing computation [107] of the same two-point function which focussed on the leading large-distance contribution. We study the ground state two-point function of the composite twist field \mathcal{T}_{μ} and its conjugate $\tilde{\mathcal{T}}_{\mu}$. At criticality, this field can be defined as the leading field in the operator product expansion of \mathcal{T} and the disorder field μ . We find a general formula for the logarithm of $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell) \rangle$ and for (the derivative of) its analytic continuation to positive real replica numbers greater than 1. We check our formula for consistency by showing that at short distances it exactly reproduces the expected conformal dimension

4.1 Introduction

It is well known that the 1+1D Ising field theory, obtained as the continuum limit of the quantum Ising chain, is described near the critical point by the action of a free Majorana fermion [60, 61, 74, 198]:

$$\mathcal{S} = \int \mathrm{d}z \mathrm{d}\bar{z} \left(\psi(z) \bar{\partial} \psi(z) + \bar{\psi}(\bar{z}) \partial \bar{\psi}(\bar{z}) + m \bar{\psi}(\bar{z}) \psi(z) \right), \tag{4.1}$$

CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

where ψ , $\bar{\psi}$ are the two components of the Majorana field Ψ and the sign of the mass term indicates whether the theory is in the ordered or in the disordered phase (the critical point being at m = 0). This theory has an internal \mathbb{Z}_2 symmetry, as the action is invariant under $\Psi \to -\Psi$. This symmetry is associated to the order operator σ (the spin field) and the disorder operator μ (disorder operator), which are semi-local with respect to the fermion field Ψ , so that the three fields can be characterised by their mutual equal-time exchange relations [60, 61, 81]:

$$\Psi(x)\sigma(y) = \begin{cases} \sigma(y)\Psi(x) & y > x \\ \sigma(y)\Psi(x) & y < x \end{cases} \quad \text{and} \quad \Psi(x)\mu(y) = \begin{cases} -\mu(y)\Psi(x) & y > x \\ \mu(y)\Psi(x) & y < x \end{cases} \quad (4.2)$$

In the replica version of the theory, the fields above acquire an index $\{\mu_j, \sigma_j, \Psi_j\}$ with j = 1, ..., n, running over the copy numbers. The resulting model possesses a larger amount of symmetry, as the \mathbb{Z}_2 symmetry on each copy is now enhanced with symmetry under the exchange of any copies. As discussed in Section 1.2, the latter symmetry plays a fundamental role in computations of the entanglement entropy and other measures of entanglement [11, 12, 20].

In [20] the branch-point twist fields \mathcal{T} and its conjugate $\tilde{\mathcal{T}}$ were defined as the symmetry fields associated with cyclic permutation symmetry of copies in a replica theory. These fields too are characterised by their exchange relations with respect to the fermions, which are identical to those in (1.26):

$$\Psi_{j}(x)\mathcal{T}(y) = \begin{cases} \mathcal{T}(y)\Psi_{j+1}(x) & y > x \\ \mathcal{T}(y)\Psi_{j}(x) & y < x \end{cases} \quad \text{and} \quad \Psi_{j}(x)\tilde{\mathcal{T}}(y) = \begin{cases} \tilde{\mathcal{T}}(y)\Psi_{j-1}(x) & y > x \\ \tilde{\mathcal{T}}(y)\Psi_{j}(x) & y < x \end{cases}$$
(4.3)

for j = 1, ..., n and $n + 1 \equiv 1$. These relations can be written for any 1+1D QFT, however, in the context of massive integrable theories they provide -together with the two-body scattering matrixall the information needed to compute correlation functions and matrix elements of \mathcal{T} . These computations have now been carried out for many theories and entanglement measures (see e.g. [63, 117, 199, 200]) revealing many new insights into the universal properties of entanglement at near-critical points.

In recent years, it has been shown that also the fields resulting from the conformal OPE of \mathcal{T} with other fields of the Ising field theory can be of interest in the context of entanglement [19, 94, 156–158, 201]. In particular, the correlation functions of the leading field in the OPE of \mathcal{T} and $\sum_{j} \mu_{j}$, denoted by \mathcal{T}_{μ} , are related to the \mathbb{Z}_2 -resolved entanglement entropy [19, 94, 107]. \mathcal{T}_{μ} satisfies exchange relations which combine those for \mathcal{T} and μ as seen above:

$$\Psi_{j}(x)\mathcal{T}_{\mu}(y) = \begin{cases} -\mathcal{T}_{\mu}(y)\Psi_{j+1}(x) & y > x \\ \mathcal{T}_{\mu}(y)\Psi_{j}(x) & y < x \end{cases} \quad \text{and} \quad \Psi_{j}(x)\tilde{\mathcal{T}}_{\mu}(y) = \begin{cases} -\tilde{\mathcal{T}}_{\mu}(y)\Psi_{j-1}(x) & y > x \\ \tilde{\mathcal{T}}_{\mu}(y)\Psi_{j}(x) & y < x \end{cases} \quad (4.4)$$

The computation of the symmetry-resolved entanglement provides strong motivation to study correlators of \mathcal{T}_{μ} and this is the focus of the present chapter. Using techniques of integrable QFT, we find an exact analytic expression for the (logarithm of the) two-point function $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle$ in the disordered phase of the model. The applications of such a result in the context of entanglement will not be discussed here, but they follow quite straightforwardly from existing literature. In particular, the form factors of \mathcal{T}_{μ} and the leading contribution to its two-point function were computed in [107]. The present work is an extension of those results to include higher particle contributions and to show how non-trivial resummation identities allow for relatively simple closed formulae for all correlation function cumulants.

Correlation functions of composite twist fields have been studied in a number of works both for the Ising field theory and other, interacting models. Most of these results build upon the form factor program for the matrix elements of \mathcal{T} [20] and its extension to composite twist fields [107]. In [105, 123, 124] free theories were studied, whereas interacting IQFTs such as the Ising and sinh-Gordon models (both with discrete \mathbb{Z}_2 symmetry), the sine-Gordon model (with continuous U(1) symmetry) and the 3-state Potts model (with discrete \mathbb{Z}_3 symmetry) were studied in [106, 107] and [108], respectively. It is also possible to study composite twist fields where \mathcal{T} is composed with a local field *not* associated with an internal symmetry. Such composite fields are associated with cyclic permutation symmetry too and have a conformal dimension which is distinct from that of \mathcal{T} . In particular, for theories whose UV fixed point is described by a non-unitary CFT, it is possible to construct composite twist fields whose dimension is lower than that of \mathcal{T} and they play a critical role in describing the usual measures of entanglement [158]. This happens for instance for the Lee-Yang theory both at and away from criticality. The form factors and two-point functions of the BPTF and CTF for this theory were studied in [201]. The expectation values of composite twist fields involving the energy field in the Ising field theory were studied in [156, 157].

The structure of this chapter is as follows: In Section 4.2 we review form factor results for the order and disorder fields in the Ising field theory as well as for \mathcal{T} and \mathcal{T}_{μ} . We then present the cumulant expansion of two-point functions and introduce an example of the type of convergence issues that arise in the cumulant expansion of $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle/\langle \mathcal{T}_{\mu}\rangle^2$. In Section 4.3 we find closed formulae for all higher cumulants, leading to a close-form expression for the two-point function. In Section 4.4 we test this expression by obtaining the exact conformal dimension of \mathcal{T}_{μ} from resummation of leading terms in the short-distance expansion of the cumulants. We show that the normalised two-point function $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle/\langle \mathcal{T}_{\mu}\rangle^2$ is in fact proportional to the normalised two-point function $\langle \mu(0)\mu(\ell)\rangle/\langle \mu\rangle^2$, thus it factorises into *n*-dependent and *n*-independent parts. In Section 4.5 we show how to analytically continue the cumulant expansion from *n* integer and greater than 1 to *n* real. This allows us to write a formula for the *n*-derivative of the two-point function at n = 1, a quantity that typically plays a

CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

role in entanglement measures. We conclude in Section 4.6.

4.2 Field content of the Ising model and form factors

The correlation functions and form factors of the fields σ, μ in the disordered phase can be obtained via form factor bootstrap [77, 78] and were studied in the seminal papers [62, 81]. Form factors of descendent fields (in the CFT sense) of the energy field ε were studied in [80] and shown to match in number and spin the field content of the corresponding Verma module in the underlying Ising CFT. Starting from the relations (4.2) the form factor equations can be written and solved for matrix elements of σ, μ and these were found to take an extremely simple form [62], namely (the factor i^k is needed to satisfy the kinematic residue equation):

$$F_{2k}^{\mu}(\theta_1,\dots,\theta_{2k}) = i^k \langle \mu \rangle \prod_{1 \le i < j \le 2k} \tanh \frac{\theta_{ij}}{2}, \qquad (4.5)$$

$$F_{2k+1}^{\sigma}(\theta_1, \dots, \theta_{2k+1}) = i^k F_1^{\sigma} \prod_{1 \le i < j \le 2k+1} \tanh \frac{\theta_{ij}}{2},$$
(4.6)

with $\theta_{ij} := \theta_i - \theta_j$ and $\langle \mu \rangle$ and F_1^{σ} normalisation constants which can be identified with the vacuum expectation value of μ and the one-particle form factor of σ , respectively. In general, as discussed in Section 1.2, form factors are characterised by a set of quantum numbers specifying the particle types in the asymptotic state, but in the Ising model there is a single particle type so these do not need to be specified. For the field μ the products above can be rewritten as a Pfaffian of a $2k \times 2k$ antisymmetric matrix \mathcal{A} with entries $\mathcal{A}_{ij} = \tanh \frac{\theta_{ij}}{2}$. In particular this means that the vacuum expectation value of μ is non-vanishing, whereas it is vanishing for σ . This is a consequence of the fact that in the disordered phase the field σ (μ) is odd (even) with respect to the \mathbb{Z}_2 action on the the fermion field, which creates the asymptotic states, and hence σ (μ) will have a non vanishing correlator only with an odd (even) number of particles.

The form factors of the BPTFs \mathcal{T} and $\tilde{\mathcal{T}}$ in the (replica) Ising model have been known for some time [20, 63] and due to the free nature of the model they can also be expressed in terms of a Pfaffian

$$F_{2k}^{\mathcal{T}|11\dots 1}(\theta_1,\dots,\theta_{2k};n) = \langle \mathcal{T} \rangle \mathrm{Pf}(K), \qquad \mathrm{Pf}(K) = \sqrt{\det K}, \tag{4.7}$$

where n labels the number of replicas,

$$K_{ij} := k(\theta_{ij}) = \frac{\sin\frac{\pi}{n}}{2n\sinh\left(\frac{i\pi-\theta_{ij}}{2n}\right)\sinh\left(\frac{i\pi+\theta_{ij}}{2n}\right)} \frac{\sinh\frac{\theta_{ij}}{2n}}{\sinh\frac{i\pi}{2n}}, \quad \text{with} \quad i, j = 1, \dots, 2k,$$
(4.8)

and the superindices 11...1 indicate that all particles are in the same copy 1. From this representation

we also see that all form factors are functions of rapidity differences only, a property that holds for all spinless fields in relativistic QFT. The two-particle form factor is simply $F_2^{\mathcal{T}|11}(\theta_1, \theta_2; n) = \langle \mathcal{T} \rangle k(\theta_{12})$. Form factors for particles in copies $j_1 \dots j_{2k}$ can be obtained from the above using the standard form factor equations presented in [20]

$$F_{2k}^{\mathcal{T}|j_1\dots j_{2k}}(\theta_1,\dots,\theta_{2k};n) = F_{2k}^{\mathcal{T}|1\dots 1}(\theta_1^{j_1-1},\dots,\theta_{2k}^{j_k-1};n), \quad \text{for} \quad j_1 \ge j_2 \dots \ge j_{2k}, \quad (4.9)$$

with

$$\theta^j := \theta + 2\pi i j. \tag{4.10}$$

The form factors of the composite twist field \mathcal{T}_{μ} where first obtained in [107] and have again the Pfaffian structure typical of the Ising model, that is

$$F_{2k}^{\mathcal{T}_{\mu}|11\dots 1}(\theta_1,\dots,\theta_{2k};n) = \langle \mathcal{T}_{\mu} \rangle \mathrm{Pf}(W), \qquad (4.11)$$

with

$$W_{ij} := w(\theta_{ij}) = \frac{\sin\frac{\pi}{n}}{2n\sinh\left(\frac{i\pi-\theta_{ij}}{2n}\right)\sinh\left(\frac{i\pi+\theta_{ij}}{2n}\right)} \frac{\sinh\frac{\theta_{ij}}{n}}{\sinh\frac{i\pi}{n}}.$$
(4.12)

As we can see, this function differs from $k(\theta)$ above only because n is replaced by n/2 in the minimal part of the form factor (i.e. the part that does not contain kinematic poles). However, this small change leads to some important differences, the main one being the asymptotic properties

$$\lim_{\theta \to \infty} k(\theta) = 0, \quad \text{and} \quad \lim_{\theta \to \pm \infty} w(\theta) = \pm \frac{i}{n}, \quad (4.13)$$

as well as

$$\lim_{n \to 1} k(\theta) = 0, \quad \text{and} \quad \lim_{n \to 1} w(\theta) = i \tanh \frac{\theta}{2}.$$
(4.14)

Note that the last equality simply shows that the two-particle form factor of \mathcal{T}_{μ} reduces to that of μ for n = 1, as expected. This extends to higher-particle form factors too. It is known from the study of many models and arguments such as those presented in [202] that the asymptotics of two particle form factors should be related to the value of a one-particle form factor. This is a consequence of so-called cluster decomposition in momentum space. In simple theories, as assumed in [202], this one-particle form factor would be that of the same field. However, in the Ising model, due to \mathbb{Z}_2 symmetry there is a mixing between form factors of μ and σ and also those of \mathcal{T}_{σ} (defined as the composition of \mathcal{T} and $\sum_{j} \sigma_{j}$) and \mathcal{T}_{μ} , in such a way that:

$$\lim_{\theta \to \pm \infty} w(\theta) = \pm \tau^2, \tag{4.15}$$

where $\tau := F_1^{\mathcal{T}_{\sigma}|1}$ is the one-particle form factor of \mathcal{T}_{σ} , which by relativistic invariance does not depend

CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

on θ . Combining (4.15) with (4.13) we have that

$$|F_1^{\mathcal{T}_\sigma|1}|^2 = |\tau|^2 = \frac{1}{n}.$$
(4.16)

Higher form factors of \mathcal{T}_{σ} can also be related to Pfaffians by employing a more general version of the cluster decomposition property. Namely

$$\lim_{\theta_{2k+2} \to \infty} \langle \mathcal{T}_{\mu} \rangle^{-1} F_{2k+2}^{\mathcal{T}_{\mu}}(\theta_1, \dots, \theta_{2k+2}; n) = \tau F_{2k+1}^{\mathcal{T}_{\sigma}}(\theta_1, \dots, \theta_{2k+1}; n).$$
(4.17)

Note that the prefactor $\langle \mathcal{T}_{\mu} \rangle^{-1}$ ensures that when k = 0 both sides of the equation become τ^2 . In this way, the form factors $F_{2k+1}^{\mathcal{T}_{\sigma}}(\theta_1, \ldots, \theta_{2k+1}; n)$ can be computed systematically and it is easy to show that they can be written as sums of Pfaffians involving 2k variables. In fact, we can show that

$$F_{2k+1}^{\mathcal{T}_{\sigma}}(\theta_1,\dots,\theta_{2k+1};n) = \frac{\tau}{\langle \mathcal{T}_{\mu} \rangle} \sum_{j=1}^{2k+1} (-1)^{j+1} F_{2k}^{\mathcal{T}_{\mu}}(\theta_1,\dots,\bar{\theta}_j,\dots,\theta_{2k+1};n),$$
(4.18)

where the sign depends on the position of the variable θ_j and can be worked out by counting Wick contractions. Similarly, the symbol $\bar{\theta}_j$ means that this variable is removed, hence this is a sum over 2k-particle form factors depending on a subset of the variables $\{\theta_1, \ldots, \theta_{2k+1}\}$. For instance

$$F_{3}^{\mathcal{T}_{\sigma}}(\theta_{1},\theta_{2},\theta_{3};n) = \tau(w(\theta_{12}) - w(\theta_{13}) + w(\theta_{23}))$$

$$= \frac{\tau}{\langle \mathcal{T}_{\mu} \rangle} (F_{2}^{\mathcal{T}_{\mu}}(\theta_{1},\theta_{2};n) - F_{2}^{\mathcal{T}_{\mu}}(\theta_{1},\theta_{3};n) + F_{2}^{\mathcal{T}_{\mu}}(\theta_{2},\theta_{3};n)).$$
(4.19)

The formula (4.18) is, to the best of our knowledge, new and first presented here. However, this structure is the same relating the form factors (4.5) and (4.6) of the fields μ and σ which are obtained in the limit n = 1, and ultimately relies on the factorisation properties of the tanh function.

In [107] it was also shown that the form factor (4.12) gives the correct conformal dimension of \mathcal{T}_{μ} via the Δ -sum rule [202]. This dimension is [12, 19, 58, 59, 156]

$$\Delta_{\mathcal{T}_{\mu}} = \Delta_{\mathcal{T}} + \frac{\Delta_{\mu}}{n} = \frac{n}{48} + \frac{1}{24n}, \quad \text{with} \quad \Delta_{\mathcal{T}} = \frac{1}{48} \left(n - \frac{1}{n} \right), \quad \Delta_{\mu} = \frac{1}{16}. \quad (4.20)$$

We observe that, since $\Delta_{\mu} = \Delta_{\sigma}$, it also holds $\Delta_{\mathcal{T}_{\mu}} = \Delta_{\mathcal{T}_{\sigma}}$.

4.2.1 Two-point function and cumulant expansion

In this chapter, we are interested in the two-point function of the field \mathcal{T}_{μ} in the ground state of the disordered phase. Our goal is to write down an expansion of the form

$$\log\left(\frac{\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle}{\langle \mathcal{T}_{\mu}\rangle^{2}}\right) = \sum_{k=1}^{\infty} c_{k}^{\mathcal{T}_{\mu}}(\ell;n) \stackrel{m\ell\ll 1}{\simeq} -4\Delta_{\mathcal{T}_{\mu}}\log(m\ell) - K_{\mathcal{T}_{\mu}},\tag{4.21}$$

where the sum is over functions $c_k^{\mathcal{T}_{\mu}}(\ell; n)$ known as cumulants and $K_{\mathcal{T}_{\mu}}$ is a constant that depends on the vacuum expectation value $\langle \mathcal{T}_{\mu} \rangle$. The structure of the right-hand side in the above equation is dictated by the form of correlators of primary fields in CFT. The cumulant expansion of the two-point function of the standard BPTF in the ground state of the free massive boson was performed in [117], and we borrow ideas from that work. The cumulants are multiple integrals of linear combinations of products of form factors. More precisely, we have the following structure

$$c_k^{\mathcal{T}_{\mu}}(\ell;n) = \frac{1}{k!(2\pi)^k} \sum_{j_1,\dots,j_k=1}^n \int_{-\infty}^{\infty} \mathrm{d}\theta_1 \cdots \int_{-\infty}^{\infty} \mathrm{d}\theta_k \ h_k^{\mathcal{T}_{\mu}|j_1\dots,j_k}(\theta_1,\dots,\theta_k,n) e^{-m\ell\sum_{i=1}^k \cosh\theta_i}, \quad (4.22)$$

where the functions $h_k^{\mathcal{O}|j_1...j_k}(\theta_1, \cdots, \theta_k, n)$ are given in terms of the form factors of the field involved, and j_i are the copy numbers. For example:

$$h_{2}^{\mathcal{T}_{\mu}|j_{1}j_{2}}(\theta_{1},\theta_{2},n) = \langle \mathcal{T}_{\mu} \rangle^{-2} \left| F_{2}^{\mathcal{T}_{\mu}|j_{1}j_{2}}(\theta_{1},\theta_{2},n) \right|^{2},$$

$$h_{4}^{\mathcal{T}_{\mu}|j_{1}j_{2}j_{3}j_{4}}(\theta_{1},\theta_{2},\theta_{3},\theta_{4},n) = \langle \mathcal{T}_{\mu} \rangle^{-2} \left| F_{4}^{\mathcal{T}_{\mu}|j_{1}j_{2}j_{3}j_{4}}(\theta_{1},\theta_{2},\theta_{3},\theta_{4},n) \right|^{2}$$

$$-h_{2}^{\mathcal{T}_{\mu}|j_{1}j_{2}}(\theta_{1},\theta_{2},n)h_{2}^{\mathcal{T}_{\mu}|j_{3}j_{4}}(\theta_{3},\theta_{4},n)$$

$$-h_{2}^{\mathcal{T}_{\mu}|j_{1}j_{3}}(\theta_{1},\theta_{3},n)h_{2}^{\mathcal{T}_{\mu}|j_{2}j_{4}}(\theta_{2},\theta_{4},n)$$

$$-h_{2}^{\mathcal{T}_{\mu}|j_{1}j_{4}}(\theta_{1},\theta_{4},n)h_{2}^{\mathcal{T}_{\mu}|j_{2}j_{3}}(\theta_{2},\theta_{3},n), \qquad (4.23)$$

and so on, whereas all odd particle terms are zero. Similar formulae can be written for the cumulants of \mathcal{T}_{σ} where only odd particle cumulants are non-vanishing. A general diagrammatic construction of the functions appearing in a cumulant expansion can be found for instance in [203]. For a generic local field \mathcal{O} , it is standard to require that

$$h_k^{\mathcal{O}|j_1\dots j_k}(\theta_1, \cdots, \theta_k) \sim e^{-\theta_i}, \tag{4.24}$$

as $\theta_i \to \infty$. Given the properties of the form factors presented in the previous section, we see that this asymptotic behaviour is not satisfied for the cumulants of the two-point function of \mathcal{T}_{μ} , or indeed for the cumulants of the two-point function of μ as shown in [62]. In fact, the cumulant expansion is still convergent in both cases, but the leading behaviour for small $m\ell$ is harder to extract than in theories where (4.24) holds.

4.2.2 Two-particle contribution

One of the simplest ways to check the validity of the two-point function expansion consists in recovering the conformal dimension of the field by exact resummation of all terms which are proportional to $\log(m\ell)$ for $m\ell \ll 1$, that is the first term in (4.21). Let us start by considering the simplest contribution to the connected part of the two-point function $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle/\langle \mathcal{T}_{\mu}\rangle^2$, which has already been studied in the literature [107]. The first non-vanishing contribution to the cumulant expansion comes from $h_2^{\mathcal{T}_{\mu}|j_1j_2}(\theta_1, \theta_2, n)$, which is nothing but the normalised squared modulus of the two-particle form factor. Using (4.9), the latter can be rewritten as

$$\sum_{i,j=1}^{n} \left| F_2^{\mathcal{T}_{\mu}|ij}(\theta_1, \theta_2) \right|^2 = n \sum_{j=0}^{n-1} \left| F_2^{\mathcal{T}_{\mu}|11}(\theta_1 + 2\pi i j, \theta_2) \right|^2 = n \langle \mathcal{T}_{\mu} \rangle^2 \sum_{j=0}^{n-1} w((-\theta_{12})^j) w(\theta_{12}^j).$$
(4.25)

Thus we have

$$c_{2}^{\mathcal{T}_{\mu}}(\ell;n) = n \sum_{j=0}^{n-1} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \frac{\mathrm{d}\theta_{1} \mathrm{d}\theta_{2}}{2(2\pi)^{2}} w((-\theta_{12})^{j}) w(\theta_{12}^{j}) e^{-m\ell \cosh \theta_{1} - m\ell \cosh \theta_{2}}$$

$$= \frac{n}{(2\pi)^{2}} \sum_{j=0}^{n-1} \int_{-\infty}^{\infty} \mathrm{d}\theta \, w((-\theta)^{j}) w(\theta^{j}) \, K_{0}(2m\ell \cosh \frac{\theta}{2}), \qquad (4.26)$$

where the second line is obtained by performing a change of variable $\theta = \theta_1 - \theta_2$, $\Theta = \frac{\theta_1 + \theta_2}{2}$ and using the integral representation of the Bessel function [91]

$$K_0(z) = \int_0^\infty dx \, e^{-z \cosh x}.$$
 (4.27)

The sum over the copy index j was computed in [107] via contour integration and it is given by

$$\sum_{j=0}^{n-1} w((-\theta)^j) w(\theta^j) = -i \tanh \frac{\theta}{2} (w(2\theta + i\pi) + w(2\theta - i\pi)) - \frac{1}{n}.$$
(4.28)

This function tends asymptotically to the value $\frac{1}{n}$ for $|\theta| \to \infty$. This means that the usual procedure consisting of expanding the Bessel function for $m\ell \ll 1$ and isolating the $\log(m\ell)$ leading term, thus effectively removing the Bessel function from the integrand in (4.26), now leads to a divergent integral. However, we can rewrite (4.26) as

$$c_{2}^{\mathcal{T}_{\mu}}(\ell;n) = \frac{n}{(2\pi)^{2}} \int_{-\infty}^{\infty} d\theta \left[\sum_{j=0}^{n-1} w((-\theta)^{j}) w(\theta^{j}) - \frac{1}{n} \right] K_{0}(2m\ell\cosh\frac{\theta}{2}) + \frac{1}{(2\pi)^{2}} \int_{-\infty}^{\infty} d\theta K_{0}(2m\ell\cosh\frac{\theta}{2}).$$
(4.29)

In this form, the integral in the first line can be approximated for $m\ell \ll 1$ by expanding the Bessel function, giving a leading contribution which is proportional to $\log(m\ell)$, while the integral in the second line can be computed exactly to

$$\int_{-\infty}^{\infty} \mathrm{d}\theta \, K_0(2m\ell \cosh\frac{\theta}{2}) = 2K_0(m\ell)^2 \stackrel{m\ell \ll 1}{\simeq} -2(\log(m\ell))^2, \tag{4.30}$$

so that, in this case, the leading small $m\ell$ contribution diverges as $(\log(m\ell))^2$. Thus, although the cumulant (4.26) is still well-defined, its leading small $m\ell$ behaviour is now dominated by $(\log(m\ell))^2$ instead of $\log(m\ell)$. This is a consequence of the property (4.24) not holding in this case. Nonetheless, terms of order $(\log(m\ell))^2$ should cancel out when including further contributions in the form factor series as one expects to recover the $1/r^{4\Delta\tau_{\mu}}$ behaviour of the two-point function at short distances. In the next sections we will show that this is indeed the case, providing a way to recover the expected scaling (4.21) from our cumulant expansion.

4.3 Higher particle contributions: closed formulae

Existing studies of the branch-point twist field two-point function for free fermions [63] and bosons [117] have revealed that the form of higher cumulants can be considerably simplified. This is because under sum over particle types and integration over the rapidities, many of the terms in the cumulant either cancel each other out or can be shown to be identical. In fact, it is possible to show that just as for the standard BPTF, and for the same reasons already discussed in [63, 117] the cumulants of the two-point function of \mathcal{T}_{μ} take the generic form

$$c_{2k}^{\mathcal{T}_{\mu}}(\ell;n) = \frac{n}{2k(2\pi)^{2k}} \sum_{j_{1},\dots,j_{2k-1}=0}^{n-1} \left[\prod_{i=1}^{2k} \int_{-\infty}^{+\infty} \mathrm{d}\theta_{i} \, e^{-m\ell\cosh\theta_{i}} \right] \\ \times (-1)^{k} \left(w(\theta_{12}^{-j_{1}}) \prod_{i=1}^{k-1} w(\theta_{2i+1}^{j_{2i}-j_{2i+1}})) \right) \left(w(\theta_{12k}^{j_{2k-1}}) \prod_{i=1}^{k-1} w(\theta_{2i}^{-j_{2i-1}+j_{2i}}) \right).$$
(4.31)

By using the fact that $w(\theta^{-j}) = -w((-\theta)^j)$, we can change the sign of half of the factors in the second line, cancelling out the factor $(-1)^k$, so that the integrand becomes:

$$\sum_{j_1,\dots,j_{2k-1}=0}^{n-1} w((-\theta_{12})_1^j) w(\theta_{1\,2k}^{j_{2k-1}}) \prod_{i=1}^{k-1} w(\theta_{2i+1\,2i+2}^{j_{2i}-j_{2i+1}}) w((-\theta_{2i\,2i+1})^{j_{2i-1}-j_{2i}}).$$
(4.32)

In order to evaluate the integrals (4.31), it is convenient to perform a change of variables whereby we first change the sign of all the rapidities θ_i with *i* even, without any change in the integration measure.

Then, defining $\hat{\theta}_{ij} \equiv \theta_i + \theta_j$ the integrand becomes a function of rapidity sums only:

$$\sum_{j_1,\dots,j_{2k-1}=0}^{n-1} w((-\hat{\theta}_{12})^{j_1}) w(\hat{\theta}_{23}^{j_1-j_2}) w(\hat{\theta}_{34}^{j_2-j_3}) \dots w(\hat{\theta}_{2k-1\,2k}^{j_{2k-2}-j_{2k-1}}) w(\hat{\theta}_{1\,2k}^{j_{2k-1}}).$$
(4.33)

We will refer to this as a fully connected sum, meaning that all terms are cyclically "connected" both at the level of the rapidities and the summation indices.

4.3.1 Recursive formulae

The sum (4.33) can be computed recursively, leading to generalisations of the following result:

$$f_1(x,y;n) := \sum_{j=0}^{n-1} w((-x)^j) w(y^j) = -\frac{i}{2} \frac{\sinh\left(\frac{x+y}{2}\right)}{\cosh\frac{x}{2}\cosh\frac{y}{2}} [w(x+y+i\pi) + w(x+y-i\pi)] - \frac{1}{n}, \quad (4.34)$$

which is presented here for the first time, although the case x = y was obtained in [107] and has already been reported in (4.28). It is also useful to know that

$$\sum_{j=0}^{n-1} w(x^j) = i \tanh \frac{x}{2}.$$
(4.35)

A derivation of formulae (4.34), (4.35) and their generalisations to multiple sums (see below) is presented in Appendix 4.A.

For the branch-point twist field of free fermions and bosons [63, 117] a formula almost identical to (4.34) also holds, albeit without the term $-\frac{1}{n}$. This term in fact makes the generalisation of (4.34) to multiple sums more complex for \mathcal{T}_{μ} than it is for \mathcal{T} . It can nonetheless be done as follows. Let us consider, as an example, the next sum in the series, namely a sum of the form

$$\sum_{j_1,j_2=0}^{n-1} w((-x)^{j_1})w(y^{j_1-j_2})w(z^{j_2}) = \sum_{j=0}^{n-1} f_1(x,y^{-j},n)w(z^j).$$
(4.36)

Repeated use of (4.34) and (4.35) leads to

$$\sum_{j_{1},j_{2}=0}^{n-1} w((-x)^{j_{1}})w(y^{j_{1}-j_{2}})w(z^{j_{2}}) = -\frac{i}{n} \left(\tanh \frac{x}{2} + \tanh \frac{y}{2} + \tanh \frac{z}{2} \right)$$

$$+ \frac{1}{4} \frac{\cosh \left(\frac{x+y+z}{2}\right)}{\cosh \frac{x}{2} \cosh \frac{y}{2} \cosh \frac{z}{2}} \left[2w(x+y+z) + w(x+y+z+2i\pi) + w(x+y+z-2i\pi) \right].$$

$$(4.37)$$

This special case gives a good indication of the kind of structures that emerge. We observe that the contribution in the second line of the above expression has exactly the same structure as found for the BPTF in the free fermion theory [63]. The terms in the first line form a symmetric polynomial on the

variables $\tanh \frac{x}{2}, \tanh \frac{y}{2}, \tanh \frac{z}{2}$. The general structure for higher sums goes as follows. let us define:

$$f_k(x_1, \dots, x_{2k}, n) := \frac{2i(-1)^k \sinh \frac{x}{2}}{\prod_{i=1}^{2k} 2 \cosh \frac{x_i}{2}} \mathcal{F}_k(x; n),$$
(4.38)

$$g_k(x_1, \dots, x_{2k+1}, n) := \frac{2(-1)^{k+1} \cosh \frac{x}{2}}{\prod_{i=1}^{2k+1} 2 \cosh \frac{x_i}{2}} \mathcal{G}_k(x; n),$$
(4.39)

where $x := \sum_{i} x_i$ and

$$\mathcal{F}_k(x;n) := \sum_{j=1}^k \binom{2k-1}{k-j} \left[w(x^{j-\frac{1}{2}}) + w(x^{-j+\frac{1}{2}}) \right], \tag{4.40}$$

$$\mathcal{G}_{k}(x;n) := \binom{2k}{k} w(x) + \sum_{j=1}^{k} \binom{2k}{k-j} \left[w(x^{j}) + w(x^{-j}) \right], \tag{4.41}$$

with

$$\lim_{|x|\to\infty} \mathcal{F}_k(x;n) = \operatorname{sgn}(x)\frac{i}{n}2^{2k-1}, \quad \lim_{|x|\to\infty} \mathcal{G}_k(x;n) = \operatorname{sgn}(x)\frac{i}{n}2^{2k},$$
(4.42)

and

$$\mathcal{F}_{k}(x;1) = 2^{2k-1}i \coth\frac{x}{2}, \quad \mathcal{G}_{k}(x;1) = 2^{2k-1}i \tanh\frac{x}{2}.$$
(4.43)

We can then compute the sum (4.33) to

$$\sum_{j_1,\dots,j_{2k-1}=0}^{n-1} w((-x_1)^j) w(x_2^{j_1-j_2}) \dots w(x_{2k-1}^{j_{2k-2}-j_{2k-1}}) w(x_{2k}^{j_{2k-1}})$$
$$= f_k(x_1,\dots,x_{2k},n) + \frac{(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh \frac{x_1}{2},\dots, \tanh \frac{x_{2k}}{2} \right), \qquad (4.44)$$

whereas a similar sum involving an even number of indices can be evaluated to

$$\sum_{j_1,\dots,j_{2k}=0}^{n-1} w((-x_1)^j) w(x_2^{j_1-j_2}) \dots w(x_{2k}^{j_{2k-1}-j_{2k}}) w(x_{2k+1}^{j_{2k}})$$
$$= g_k(x_1,\dots,x_{2k+1},n) + i \frac{(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j+1}^{(2k+1)} \left(\tanh \frac{x_1}{2},\dots, \tanh \frac{x_{2k+1}}{2} \right).$$
(4.45)

In both formulae, $\sigma_j^{(k)}(a_1, \ldots, a_k)$ is the elementary symmetric polynomial of order j in k variables, defined as

$$\sigma_0^{(k)}(a_1, \dots, a_k) = 1, \quad \text{and} \quad \sigma_j^{(k)}(a_1, \dots, a_k) = \sum_{1 \le i_1 < i_2 < \dots < i_j \le k} a_{i_1} a_{i_2} \cdots a_{i_j}. \quad (4.46)$$

Equations (4.44) and (4.45) can be proven by induction in k, following a procedure already employed for the standard BPTF of the free fermion in [63]. The proofs are presented in Appendix 4.A.
CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

An interesting property of the formula (4.44) and a consistency check of its validity is the fact that the cumulant expansion of $\langle \mu(0)\mu(\ell)\rangle/\langle \mu\rangle^2$ is recovered for n = 1. Indeed, from (4.43) it follows that

$$f_k(x_1, \dots, x_{2k}, 1) = \frac{(-1)^{k+1} \cosh \frac{x}{2}}{\prod_{i=1}^{2k} \cosh \frac{x_i}{2}} = (-1)^{k+1} \sum_{j=0}^k \sigma_{2j}^{(2k)}(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k}}{2}).$$
(4.47)

Then, in the limit $n \to 1$ the only term remaining from the sum (4.44) is the symmetric polynomial $\sigma_{2k}^{(2k)}(\tanh \frac{x_1}{2}, \ldots, \tanh \frac{x_{2k}}{2})$ which is just the product of its arguments. This exactly reproduces the cumulant expansion of $\log \langle \mu(0)\mu(\ell) \rangle$ given in [62], formula (3.12a). Similarly, it can be shown that

$$g_k(x_1,\ldots,x_{2k+1},1) = \frac{i(-1)^{k+1}\sinh\frac{x}{2}}{\prod_{i=1}^{2k+1}\cosh\frac{x_i}{2}} = i(-1)^{k+1}\sum_{j=0}^k \sigma_{2j+1}^{(2k+1)}(\tanh\frac{x_1}{2},\ldots,\tanh\frac{x_{2k+1}}{2}).$$
(4.48)

4.3.2 Main result of this section

Putting together the expression (4.31) with the sum formula (4.44), we have the following exact result for the cumulant expansion of the logarithm of the ground state two-point function of the CTF \mathcal{T}_{μ} in the disordered phase of the Ising model:

$$\log\left(\frac{\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle}{\langle \mathcal{T}_{\mu}\rangle^{2}}\right) = \sum_{k=1}^{\infty} c_{2k}^{\mathcal{T}_{\mu}}(\ell;n)$$

$$= \sum_{k=1}^{\infty} \frac{n}{2k(2\pi)^{2k}} \left[\prod_{i=1}^{2k} \int_{-\infty}^{+\infty} d\theta_{i} e^{-m\ell\cosh\theta_{i}}\right] \left[f_{k}(\hat{\theta}_{12},\ldots,\hat{\theta}_{2k-12k},\hat{\theta}_{12k},n) + \frac{(-1)^{k}}{n} \sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh\frac{\hat{\theta}_{12}}{2},\ldots,\tanh\frac{\hat{\theta}_{2k-12k}}{2},\tanh\frac{\hat{\theta}_{12k}}{2}\right)\right]. \quad (4.49)$$

We now proceed to test the validity of this expression by examining its leading short-distance behaviour beyond the two-particle contribution of Section 4.2.2.

4.4 Conformal dimension from the cumulant expansion

The conformal dimension of the field \mathcal{T}_{μ} was already recovered by Δ -sum rule in [107], but the computation in that case only involved two-particle form factors. Below we provide a more extensive test of all the form factors by obtaining the conformal dimension from the complete cumulant expansion (4.49). Each cumulant is expected to contain a leading contribution which is proportional to $\log m\ell$ whereas other divergent terms should cancel out, so that the overall sum gives (4.21) with dimension given by (4.20).

First, let us return to the fully connected sum (4.33) and change variables once more. We define

$$x_i = \hat{\theta}_{i\,i+1}$$
 for $i = 1, \dots, 2k - 1, \quad x_{2k} = \theta_{2k},$ (4.50)

so that:

$$\theta_i = \sum_{j=i}^{2k} (-1)^{j-i} x_j, \quad \sum_{i=1}^{2k} \theta_i = \sum_{i=1}^k x_{2i-1}, \quad \hat{\theta}_{1\,2k} = \sum_{i=1}^{2k-1} (-1)^{i-1} x_i. \tag{4.51}$$

The Jacobian of the transformation from the θ variables to the x variables is an upper triangular matrix with the diagonal terms being all +1, so the measure acquires no extra factor. By applying this change of variables to (4.33) and expressing the result in the new variables (4.50), we obtain:

$$\sum_{j_1,\dots,j_{2k-1}=0}^{n-1} w((-x_1)^j) w(x_2^{j_1-j_2}) \dots w(x_{2k-1}^{j_{2k-2}-j_{2k-1}}) w\left(\left(\sum_{i=1}^{2k-1} (-1)^{i-1} x_i\right)^{j_{2k-1}}\right)\right)$$
$$= (-1)^k \frac{2i \sinh\left(\sum_{i=1}^k x_{2i-1}\right)}{2\cosh\left(\frac{\sum_{i=1}^{2k-1} (-1)^{i-1} x_i}{2}\right) \prod_{i=1}^{2k-1} 2\cosh\left(\frac{x_i}{2}\right)} \mathcal{F}_k\left(2\sum_{i=1}^k x_{2i-1}; n\right)$$
$$+ \frac{(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh\frac{x_1}{2}, \dots, \tanh\frac{x_{2k-1}}{2}, \tanh\frac{\sum_{i=1}^{2k-1} (-1)^{i-1} x_i}{2}\right), \quad (4.52)$$

with $\mathcal{F}_k(x;n)$ the function defined by (4.40). Furthermore, recalling equation (4.47), it is possible to express the sum over symmetric polynomials in terms of products of hyperbolic functions as

$$\sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k-1}}{2}, \tanh \frac{\sum_{i=1}^{2k-1} (-1)^{i-1} x_i}{2} \right)$$
$$= \frac{\cosh \left(\sum_{i=1}^k x_{2i-1} \right)}{\cosh \left(\frac{\sum_{i=1}^{2k-1} (-1)^{i-1} x_i}{2} \right) \prod_{i=1}^{2k-1} \cosh \left(\frac{x_i}{2} \right)} - \tanh \frac{\sum_{i=1}^{2k-1} (-1)^{i-1} x_i}{2} \prod_{i=1}^{2k-1} \tanh \frac{x_i}{2}. \quad (4.53)$$

This rewriting will prove useful later on.

4.4.1 Exponential factors

Now let us look at the exponential factors in the integrand of (4.31) and see what they look like in terms of the new variables x_i . From the first relation in (4.51) one has:

$$\sum_{j=i}^{2k-1} (-1)^{j-i} x_j = \begin{cases} \theta_i - \theta_{2k} & \text{for } i \text{ even} \\ \theta_i + \theta_{2k} & \text{for } i \text{ odd} \end{cases},$$
(4.54)

so that

$$\sum_{i=1}^{2k} \cosh \theta_i = \cosh \theta_{2k} + \sum_{i=1}^{2k-1} \cosh(\theta_i - \theta_{2k} + \theta_{2k})$$

= $\cosh \theta_{2k} + \cosh \theta_{2k} \left(\sum_{i \text{ even}} \cosh(\theta_i - \theta_{2k}) + \sum_{i \text{ odd}} \cosh(\theta_i + \theta_{2k}) \right)$
+ $\sinh \theta_{2k} \left(\sum_{i \text{ even}} \sinh(\theta_i - \theta_{2k}) - \sum_{i \text{ odd}} \sinh(\theta_i + \theta_{2k}) \right)$
= $\cosh x_{2k} \left[1 + \sum_{i=1}^{2k-1} \cosh\left(\sum_{j=i}^{2k-1} (-1)^{j-i} x_j \right) \right] + \sinh x_{2k} \left[\sum_{i=1}^{2k-1} (-1)^i \sinh\left(\sum_{j=i}^{2k-1} (-1)^{j-i} x_j \right) \right].$ (4.55)

Therefore, since none of the functions in (4.52) depends on x_{2k} the integral on this variable can be carried out by making use of the identity¹

$$\int_{-\infty}^{+\infty} dt \, \exp(-A \cosh t - B \sinh t) = 2K_0 \left(\sqrt{A^2 - B^2}\right), \quad \text{for } A > B, \tag{4.56}$$

giving

$$\int_{-\infty}^{+\infty} \mathrm{d}x_{2k} \, e^{-m\ell \sum_{i=1}^{2k} \cosh \theta_i} = 2K_0(m\ell d_{2k-1}),\tag{4.57}$$

with

$$d_{2k-1}^{2} = \left[1 + \sum_{i=1}^{2k-1} \cosh\left(\sum_{j=i}^{2k-1} (-1)^{j-i} x_{j}\right)\right]^{2} - \left[\sum_{i=1}^{2k-1} (-1)^{i} \sinh\left(\sum_{j=i}^{2k-1} (-1)^{j-i} x_{j}\right)\right]^{2}.$$
 (4.58)

The $m\ell \ll 1$ expansion of the modified Bessel function is:

$$K_0(m\ell d_{2k-1}) = -\log m\ell + \log 2 - \ln d_{2k-1} - \gamma + o(m\ell d_{2k-1}), \tag{4.59}$$

from which the leading short-distance contributions to the cumulant expansion can be obtained. It is worth mentioning that one could also resum contributions proportional to the constant term $\log 2 - \gamma$ in (4.59) and those should contribute to the $K_{\mathcal{T}_{\mu}}$ -term in (4.21), that is to the logarithm of $\langle \mathcal{T}_{\mu} \rangle$. A similar computation was carried out in [117] for $\langle \mathcal{T} \rangle$ in the free boson theory.

4.4.2 Short-distance behaviour of the cumulant expansion

Putting together (4.52), (4.53) and (4.57) in (4.31) we can split the cumulant into three contributions

$$c_{2k}^{\mathcal{T}_{\mu}}(\ell;n) = c_{2k}^{(1)}(\ell;n) + c_{2k}^{(2)}(\ell) + c_{2k}^{\mu}(\ell).$$
(4.60)

¹This identity is obtained by rewriting $A \cosh t + B \sinh t = \sqrt{A^2 - B^2} \cosh \left(t + \ln \sqrt{\frac{A+B}{A-B}} \right)$ and then shifting the integration variable t.

We will define these contributions as follows. First:

$$c_{2k}^{(1)}(\ell;n) = \frac{2(-1)^{k}in}{k(4\pi)^{2k}} \int_{-\infty}^{+\infty} \mathrm{d}x_{1} \cdots \int_{-\infty}^{+\infty} \mathrm{d}x_{2k-1} K_{0}(m\ell d_{2k-1}) \\ \times \frac{\sinh\left(\sum_{i=1}^{k} x_{2i-1}\right)}{\cosh\left(\frac{\sum_{i=1}^{2k-1}(-1)^{i-1}x_{i}}{2}\right) \prod_{i=1}^{2k-1} \cosh\left(\frac{x_{i}}{2}\right)} \hat{\mathcal{F}}_{k}\left(2\sum_{i=1}^{k} x_{2i-1};n\right),$$
(4.61)

with

$$\hat{\mathcal{F}}_k(x;n) := \mathcal{F}_k(x;n) - \operatorname{sgn}(x)\frac{i}{n}2^{2k-1}.$$
(4.62)

This shift is motivated by the asymptotics (4.42) and ensures that the function $\hat{\mathcal{F}}_k(x;n)$ goes to zero for |x| large. This in turn ensures the convergence of the integrals even when the Bessel function is approximated by its leading short-distance contribution $-\log(m\ell)$.

The next contribution is then a combination of the first term in (4.53) and the term introduced by the shift (4.62):

$$c_{2k}^{(2)}(\ell) = \frac{(-1)^{k}}{k(2\pi)^{2k}} \int_{-\infty}^{+\infty} \mathrm{d}x_{1} \cdots \int_{-\infty}^{+\infty} \mathrm{d}x_{2k-1} K_{0}(m\ell d_{2k-1}) \\ \times \left[\frac{\cosh\left(\sum_{i=1}^{k} x_{2i-1}\right) - \sinh\left(\sum_{i=1}^{k} x_{2i-1}\right) \operatorname{sgn}(\sum_{i=1}^{k} x_{2i-1})}{\cosh\left(\frac{\sum_{i=1}^{2k-1} (-1)^{i-1} x_{i}}{2}\right) \prod_{i=1}^{2k-1} \cosh\left(\frac{x_{i}}{2}\right)} \right].$$
(4.63)

Note that this contribution is *n*-independent. Finally, the contribution $c_{2k}^{\mu}(\ell)$ is nothing but the cumulant of the expansion of $\langle \mu(0)\mu(\ell)\rangle/\langle \mu\rangle^2$ resulting from the last term (the product of tanh functions) in (4.53):

$$c_{2k}^{\mu}(\ell) = \frac{(-1)^{k+1}}{k(2\pi)^{2k}} \int_{-\infty}^{+\infty} \mathrm{d}x_1 \cdots \int_{-\infty}^{+\infty} \mathrm{d}x_{2k-1} K_0(m\ell d_{2k-1}) \\ \times \tanh\left(\frac{\sum_{i=1}^{2k-1}(-1)^{i-1}x_i}{2}\right) \prod_{i=1}^{2k-1} \tanh\left(\frac{x_i}{2}\right), \tag{4.64}$$

as it indeed coincides with the cumulant presented in [62] when expressed in terms of the variables (4.50).

4.4.3 Leading contribution to $c_{2k}^{(1)}(\ell;n)$

In order to evaluate the integral (4.61) we can perform yet another change of variables:

$$y = \sum_{i=1}^{k} x_{2i-1} \quad \Rightarrow \quad x_{2k-1} = y - \sum_{i=1}^{k-1} x_{2i-1}, \quad \sum_{i=1}^{2k-1} (-1)^{i-1} x_i = y - \sum_{i=1}^{k-1} x_{2i}, \tag{4.65}$$

CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

so that, at short distances $\sum_k c_{2k}^{(1)}(\ell;n) \simeq -z_n \log(m\ell)$ with

$$z_{n} = \sum_{k=1}^{\infty} \frac{(-1)^{k} 2ni}{k(4\pi)^{2k}} \int_{-\infty}^{+\infty} dy \sinh y \,\hat{\mathcal{F}}_{k}(2y;n) \int_{-\infty}^{+\infty} dx_{1} \cdots \int_{-\infty}^{+\infty} dx_{2k-2} \\ \left[\operatorname{sech}\left(\frac{y - \sum_{i=1}^{k-1} x_{2i-1}}{2}\right) \prod_{i=1}^{k-1} \operatorname{sech}\left(\frac{x_{2i-1}}{2}\right) \right] \left[\operatorname{sech}\left(\frac{y - \sum_{i=1}^{k-1} x_{2i}}{2}\right) \prod_{i=1}^{k-1} \operatorname{sech}\left(\frac{x_{2i}}{2}\right) \right] \\ = \sum_{k=1}^{\infty} \frac{(-1)^{k} 2ni}{k(4\pi)^{2k}} \int_{-\infty}^{+\infty} dy \sinh y \,\hat{\mathcal{F}}_{k}(2y;n) G_{k}^{2}(y),$$
(4.66)

where, exactly as in [63, 117]:

$$G_{k}(y) = \int_{-\infty}^{+\infty} \mathrm{d}x_{1} \cdots \int_{-\infty}^{+\infty} \mathrm{d}x_{k-1} \left[\operatorname{sech}\left(\frac{y - \sum_{i=1}^{k-1} x_{i}}{2}\right) \prod_{i=1}^{k-1} \operatorname{sech}\left(\frac{x_{i}}{2}\right) \right] = \int_{-\infty}^{+\infty} \mathrm{d}a \frac{(2\pi)^{k-1} e^{iay}}{\cosh^{k} \pi a}.$$
(4.67)

The functions $G_k(y)$ can be evaluated explicitly to

$$G_k(y) = \frac{(2\pi)^{k-1}}{(k-1)!} \begin{cases} \frac{y}{\pi \sinh \frac{y}{2}} \prod_{j=1}^{\frac{k}{2}-1} (\frac{y^2}{\pi^2} + (2j)^2) & \text{for k even} \\ \frac{1}{\cosh \frac{y}{2}} \prod_{j=1}^{\frac{k-1}{2}} (\frac{y^2}{\pi^2} + (2j-1)^2) & \text{for k odd} \end{cases}$$
(4.68)

By replacing $\mathcal{F}_k(x;n)$ with $\hat{\mathcal{F}}_k(x;n)$ in (4.61), we have ensured that the integrals (4.66) are convergent since $G_k^2(y) \sinh y$ is asymptotically polynomial in y and $\hat{\mathcal{F}}_k(2y;n)$ is exponentially decaying. They can be evaluated with great precision and fitted to the function

$$z_n = \frac{1}{12} \left(n - \frac{1}{n} \right) + \frac{1}{4n} + z' = 4\Delta_{\mathcal{T}_{\mu}} + z', \qquad (4.69)$$

with z' = -0.217(4). This additional constant should be cancelled by contributions coming from $c_{2k}^{(2)}(\ell) + c_{2k}^{\mu}(\ell)$.

Numerical results for z_n are shown in Fig. 4.1. It is interesting to observe that there is very good agreement with the formula (4.69) for n integer and also for n not integer, greater than 2. However for 1 < n < 2 the numerical data differ from (4.69), suggesting that the analytic continuation of (4.66) to n = 1 from n real greater than 1 is non-trivial. This is in agreement with results found in [107] where the limit $n \to 1$ of the two-particle form factor contribution produced a delta-function term.

4.4.4 Leading contribution to $c^{(2)}_{2k}(\ell) + c^{\mu}_{2k}(\ell)$

We now consider the leading contribution to the second term in the cumulant (4.60). This is independent of n, and employing the same change of variables as above it is easy to write an expression which is given by a convergent integral involving the functions $G_k(y)$. Letting



Figure 4.1 Left: The function z_n evaluated numerically through the sum (4.66) for integer values of n = 1, ..., 10 (red squares) against the formula (4.69) (blue solid line). Right: The same comparison for $n \in [1,3]$ including non integer values. When evaluating the sum (4.66) numerically we truncate at some value of k. This value of k is different for each value of n and is chosen so that the sum is stable up to 5 decimal digits.

 $\sum_k c_{2k}^{(2)}(\ell) \simeq -z'' \log(m\ell)$, we obtain

$$z'' = \sum_{k=1}^{\infty} \frac{2(-1)^k}{k(2\pi)^{2k}} \int_0^{+\infty} \mathrm{d}y \, e^{-y} \, G_k^2(y) = -0.0326(1), \tag{4.70}$$

and we note that

$$z' + z'' = -0.250(0) \simeq -\frac{1}{4}.$$
(4.71)

Remarkably, this value is precisely what we need to recover the correct dimension of the field \mathcal{T}_{μ} . This is because [62]

$$\sum_{k=1}^{\infty} c_{2k}^{\mu}(\ell) \simeq -\frac{1}{4} \log(m\ell), \qquad (4.72)$$

as this is the sum over cumulants corresponding to the two-point function $\langle \mu(0)\mu(\ell)\rangle/\langle \mu\rangle^2$ and μ has dimension $4\Delta_{\mu} = 1/4$. Therefore, the overall leading short-distance behaviour of the $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle/\langle \mathcal{T}_{\mu}\rangle^2$ cumulants correctly predicts the conformal dimension (4.20). This highly non-trivial result provides strong support for the formula (4.49). In addition, the structure of the cumulants means that we can also write

$$\frac{\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle}{\langle \mathcal{T}_{\mu}\rangle^2} = \mathcal{R}(\ell;n)\,\frac{\langle \mu(0)\mu(\ell)\rangle}{\langle \mu\rangle^2},\tag{4.73}$$

where $\mathcal{R}(\ell; n) := \prod_{k=1}^{\infty} e^{c_{2k}^{(1)}(\ell; n)} e^{c_{2k}^{(2)}(\ell)}$ has the property $\mathcal{R}(\ell; 1) = 1$.

Recalling the observation of Section 4.2.2, namely that the cumulant expansion of \mathcal{T}_{μ} posed some convergence issues, we note that those issues did not feature in the computations of this section. This is because by writing the cumulant as we have done, all convergence issues have been "hidden" in the contribution $c_{2k}^{\mu}(\ell)$. Indeed, a naive expansion of the Bessel function in (4.64) leads to a divergent integral. Nonetheless, as shown in [62], the short-distance limit of this quantity can be obtained via a semiclassical approach and it ultimately leads to the expected result (4.72).

4.5 Analytic continuation to $n \in \mathbb{R}^{\geq 1}$

All results obtained so far are valid for $n \in \mathbb{N}$. This is always the case in the replica picture where n represents a replica number. However, the entanglement measures that our two-point function describes are typically defined for generic positive n. Therefore it is useful to write an expression for the correlation function which is valid for $n \in \mathbb{R}^{\geq 1}$. Let us start by studying the analytic continuation of the leading short-distance terms.

4.5.1 Analytic continuation of leading short-distance contributions

The plot in figure 4.1 (right) strongly suggests that our formula needs to be analytically continued in the region 1 < n < 2. A similar problem was addressed in [63, 117], where it was shown that as *n* approaches 1 from n > 1 some of the poles of the cumulants will cross or pinch the real line and provide additional contributions to the cumulant expansion which are non-vanishing for $n \in \mathbb{R}$ and need to be added. The correct analytic continuation is obtained when these contributions are correctly accounted for. The discussion is nearly identical as for the free boson case [117], albeit involving different functions.

As we have seen, only the contribution $c_{2k}^{(1)}(\ell;n)$ to the cumulant is *n*-dependent. Therefore we only need to analytically continue the coefficient of the leading short-distance contribution to this term, that is the quantity z_n defined in (4.66). For non-integer *n* larger than 1, z_n picks up additional contributions which account for the residues of the poles of $\hat{\mathcal{F}}_k(2y;n)$ that cross the real axis as $n \to 1^+$. The sum (4.40) in the function $\hat{\mathcal{F}}_k(2y,n)$ has poles at²

$$2y \pm (2j-1)i\pi = (2mn+1)i\pi$$
 and $2y \pm (2j-1)i\pi = (2mn-1)i\pi$ for $m \in \mathbb{Z}$, (4.74)

for every j = 1, ..., k. These are due to the kinematic poles of the two-particle form factor (4.12) at $i\pi \pm \theta = 2mni\pi$. Solving the above equation for y gives rise to four families of poles

$$y_1 = (mn+1-j)i\pi, \quad y_2 = (mn-j)i\pi, \quad m \in \mathbb{Z},$$
 (4.75)

$$y_3 = (mn - 1 + j)i\pi, \qquad y_4 = (mn + j)i\pi, \qquad m \in \mathbb{Z},$$
 (4.76)

²The twist field approach assumes n integer larger than 1 (since n is a copy number). For that reason it is natural to look for an analytic continuation to n = 1 from n > 1. However, once found, the analytic continuation is unique and thus valid for all n.

with corresponding residues of the function inside the sum (4.66) given by:

$$R_1(k,j,m,n) = \frac{n(-1)^{k+j}}{k(4\pi)^{2k}} \begin{pmatrix} 2k-1\\ k-j \end{pmatrix} \sinh(i\pi mn) G_k^2((nm-j+1)i\pi),$$
(4.77)

$$R_2(k,j,m,n) = -\frac{n(-1)^{k+j}}{k(4\pi)^{2k}} \begin{pmatrix} 2k-1\\ k-j \end{pmatrix} \sinh(i\pi mn)G_k^2((nm-j)i\pi),$$
(4.78)

$$R_{3}(k,j,m,n) = \frac{n(-1)^{k+j}}{k(4\pi)^{2k}} \begin{pmatrix} 2k-1\\ k-j \end{pmatrix} \sinh(i\pi mn) G_{k}^{2}((nm+j-1)i\pi), \qquad (4.79)$$

$$R_4(k,j,m,n) = -\frac{n(-1)^{k+j}}{k(4\pi)^{2k}} \begin{pmatrix} 2k-1\\ k-j \end{pmatrix} \sinh(i\pi mn) G_k^2((nm+j)i\pi).$$
(4.80)

These functions are all zero for n integer but they contribute for non-integer n. Let us now investigate which of these poles cross the real line in the limit $n \to 1^+$.

Since there are many indices involved, let us start by considering just one example: $n = \frac{4}{3}$ and up to k = 2 in the sum (4.66). According to the formula (4.20) $4\Delta_{\mathcal{T}_{\mu}} = 0.236111$ in this case but the numerical evaluation of (4.66), after subtracting the constant z', gives the value 0.243211 which slightly overestimates the result. The disagreement is not simply due to numerical imprecision. The function $\hat{\mathcal{F}}_3(2y, 4/3)$ has poles that cross the integration line as $n \to 4/3$. From (4.76) and the definition (4.40) we see that for k = 1 the sum runs only over the value j = 1. For j = 1 the four families of poles labeled by the integer m are:

$$y_1 = imn\pi, \qquad y_2 = (mn-1)i\pi, \qquad m \in \mathbb{Z}, \tag{4.81}$$

$$y_3 = imn\pi, \quad y_4 = (mn+1)i\pi, \quad m \in \mathbb{Z}.$$
 (4.82)

It is clear that all these poles are always above the real line (for m > 0) or below the real line (for m < 0), that is they never cross the real line, as n approaches $\frac{4}{3}$. Therefore, there is no correction coming from the k = 1 contribution. Let us consider k = 2. Now j = 1, 2. For j = 1 the poles are the same as above and never cross the real line. For j = 2 we have the following four families:

$$y_1 = i(mn-1)\pi, \qquad y_2 = (mn-2)i\pi, \qquad m \in \mathbb{Z}$$
 (4.83)

$$y_3 = i(mn+1)\pi, \quad y_4 = (mn+2)i\pi, \quad m \in \mathbb{Z}.$$
 (4.84)

We have already seen above that the poles y_1 and y_3 never cross the real line, so we can only have some contributions from y_2 and y_4 . For m > 0 and n positive and large both families of poles are above the real line. However, for $n \to \frac{4}{3}$ we see that the pole $(mn - 2)i\pi$ crosses the real line for m = 1. Similarly, for m < 0 and n positive and large all poles are in the lower half plane but the pole $(mn+2)i\pi$ crosses the real line for $n \to \frac{4}{3}$ and m = -1.

In summary, there are two poles for j = 2 located at $\pm \frac{2\pi i}{3}$. The corresponding residue contributions are

$$2\pi i (R_2(2,2,1,4/3) - R_4(2,2,-1,4/3)) = -0.00680653.$$
(4.85)

Therefore, the addition of the residua of these two poles improves the estimate of the conformal dimension from $4\Delta_{\tau_{\mu}} = 0.243211$ to $4\Delta_{\mu} = 0.243211 - 0.00680653 = 0.236404$ which is much closer to the exact value (note that the formula (4.66) gives $-4\Delta_{\tau_{\mu}}$, hence the minus sign of (4.85)). The addition of poles for higher values of k, and hence of j, will bring this value ever closer to formula (4.20) as shown in Fig 4.2. In the general n case, in order to fully identify those poles that will cross



Figure 4.2 The function z_n evaluated numerically through the sum (4.66) for $n \in [1, 3]$ (red squares) against the formula (4.69) (green dashed line) and its analytically continued values (blue triangles) given by (4.87).

the real line we find once more four cases:

$$y_{1}:mn+1-j<0 \quad \Rightarrow \quad 1 \leq m < \frac{j-1}{n},$$

$$y_{2}:mn-j<0 \quad \Rightarrow \quad 1 \leq m < \frac{j}{n},$$

$$y_{3}:mn-1+j<0 \quad \Rightarrow \quad -\frac{j-1}{n} < m \leq -1,$$

$$y_{4}:mn+j<0 \quad \Rightarrow \quad -\frac{j}{n} < m \leq -1,$$

$$(4.86)$$

This gives the analytically continued values \hat{z}_n

$$\hat{z}_{n} = z_{n} + \sum_{k=1}^{\infty} \sum_{j=1}^{k} \sum_{m=1}^{\left[\frac{j-1}{n}\right]-q_{1}} \frac{in(-1)^{k+j+1}}{k(4\pi)^{2k-1}} \begin{pmatrix} 2k-1\\k-j \end{pmatrix} \sinh\left(i\pi nm\right) G_{k}^{2}\left((nm-j+1)i\pi\right) \\ + \sum_{k=1}^{\infty} \sum_{j=1}^{k} \sum_{m=1}^{\left[\frac{j}{n}\right]-q_{2}} \frac{in(-1)^{k+j+1}}{k(4\pi)^{2k-1}} \begin{pmatrix} 2k-1\\k-j \end{pmatrix} \sinh\left(i\pi nm\right) G_{k}^{2}\left((nm-j)i\pi\right), \quad (4.87)$$

where we used the fact that the residues $R_2(k, j, m, n) = -R_4(k, j, m, n)$ and $R_1(k, j, m, n) = -R_3(k, j, m, n)$ (which produces a factor 2) and are multiplied by a factor $2\pi i$ as required by the residue theorem. The shifts q_1, q_2 take the value 1 when $n[\frac{j-1}{n}] = j - 1$ and $n[\frac{j}{n}] = j$, respectively and are zero otherwise (they can be removed by requiring *n* to be non-integer). Here the symbol [.] represents the integer part. Fig. 4.2 shows the same functions as in Fig. 4.1 (right) plus an additional set of values, which are the analytically continued values of z_n (in blue). As we can see these now agree perfectly with the fit (4.69), even for non-integer *n* between 1 and 2.

4.5.2 Analytic continuation of the *n*-derivative

Applications of the correlation function (4.49) in the context of entanglement measures frequently require the computation of its derivative with respect to n followed by the limit $n \to 1$. As discussed in [20, 63] and [107] the derivative with respect to n of the function (4.38) has a discontinuity. More precisely, as n approaches 1 and poles cross the real line, the derivative is not uniformly convergent as a function of θ and this leads to terms involving δ -functions. The simplest examples of this phenomenon are seen for the two-particle contribution to the two-point function of \mathcal{T} [20] and of \mathcal{T}_{μ} [107]. Here we show how this generalises to the whole cumulant sum. Notice that we only need to consider the contribution from the function $c_{2k}^{(1)}(\ell; n)$ in (4.61) since all other terms are independent of n and so the derivative is zero. For this term, we actually only need to consider $\mathcal{F}_k(x; n)$ as the additional term in $\hat{\mathcal{F}}_k(x; n)$ version is also n-independent. Thus, we define

$$s_{2k}^{\mathcal{T}_{\mu}}(\ell) := -\lim_{n \to 1} \frac{d}{dn} c_{2k}^{(1)}(\ell; n)$$

$$= \frac{2(-1)^{k+1}i}{k(4\pi)^{2k}} \int_{-\infty}^{+\infty} dx_1 \cdots \int_{-\infty}^{+\infty} dx_{2k-1} K_0(m\ell d_{2k-1})$$

$$\times \frac{\sinh\left(\sum_{i=1}^k x_{2i-1}\right)}{\cosh\left(\frac{\sum_{i=1}^{2k-1}(-1)^{i-1}x_i}{2}\right) \prod_{i=1}^{2k-1} \cosh\left(\frac{x_i}{2}\right)} \lim_{n \to 1} \frac{d}{dn} \left[n\mathcal{F}_k\left(2\sum_{i=1}^k x_{2i-1}; n\right) \right]. \quad (4.88)$$

CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

One way to treat the derivative is to recall the k = 1 result that was derived in [107], namely

$$\lim_{n \to 1} \frac{d}{dn} n f_1(x, x, n) = -\frac{i}{2} \frac{\sinh x}{\cosh^2 \frac{x}{2}} \lim_{n \to 1} \frac{d}{dn} n [w(2x + i\pi) + w(2x - i\pi)]$$
$$= \frac{x}{\cosh^2 \frac{x}{2} \sinh x} - \frac{\pi^2}{2} \delta(x), \qquad (4.89)$$

that is, there is a finite part and a distribution part that accounts for the behaviour around x = 0. Recall that the function $f_1(x, y, n)$ is defined in (4.34). This extends to higher cumulants in similar ways, so that we can write

$$s_{2k}^{\mathcal{T}_{\mu}}(\ell) = s_{2k}^{\text{fin}}(\ell) + s_{2k}^{\delta}(\ell), \qquad (4.90)$$

where the two contributions represent the finite and δ -function contributions. The finite part can be easily computed by noting that

$$\lim_{n \to 1} \frac{d}{dn} n \sinh x \mathcal{F}(2x; n) = \frac{i2^{2k-1}x}{\sinh x},\tag{4.91}$$

and therefore

$$s_{2k}^{\text{fin}}(\ell) = \frac{(-1)^k}{k(2\pi)^{2k}} \int_{-\infty}^{+\infty} \mathrm{d}x_1 \cdots \int_{-\infty}^{+\infty} \mathrm{d}x_{2k-1} K_0(m\ell d_{2k-1}) \\ \times \frac{\sum_{i=1}^k x_{2i-1}}{\sinh\left(\sum_{i=1}^k x_{2i-1}\right) \cosh\left(\frac{\sum_{i=1}^{2k-1}(-1)^{i-1}x_i}{2}\right) \prod_{i=1}^{2k-1} \cosh\left(\frac{x_i}{2}\right)}.$$
 (4.92)

The δ -function contribution is a generalisation of the k = 1 case seen above and can be obtained by identical arguments as those presented in [63]. In fact, the result is also identical to formula (4.6) in [63], that is,

$$s_{2k}^{\delta}(\ell) = \frac{\pi^{2}(-1)^{k}}{k(4\pi)^{2k}} \int_{-\infty}^{+\infty} dx_{1} \cdots \int_{-\infty}^{+\infty} dx_{2k-1} \,\delta(\sum_{i=1}^{k} x_{2i-1}) \\ \times \left[\begin{pmatrix} 2k-2\\k-1 \end{pmatrix} \frac{2K_{0}(2m\ell d_{2k-1})}{\cosh\left(\frac{\sum_{i=1}^{2k-1}(-1)^{i-1}x_{i}}{2}\right) \prod_{i=1}^{2k-1} \cosh\left(\frac{x_{i}}{2}\right)} \right] \\ -\frac{\pi^{2}(-1)^{k}}{k(4\pi)^{2k}} \int_{-\infty}^{+\infty} dx_{1} \cdots \int_{-\infty}^{+\infty} dx_{2k} \,\delta(\sum_{i=1}^{k} x_{2i-1}) \\ \times \sum_{j=1}^{k} \sum_{m=1}^{j-1} \sum_{q=\pm} \left[\begin{pmatrix} 2k-1\\k-j \end{pmatrix} (-1)^{j} \frac{\prod_{i=1}^{2k} e^{-rm \cosh\left(\sum_{j=1}^{2k}(-1)^{j-i}x_{i}+i\pi q\frac{j-m}{2k}\right)}}{\cosh\left(\frac{\sum_{i=1}^{2k-1}(-1)^{i-1}x_{i}}{2}\right) \prod_{i=1}^{2k-1} \cosh\left(\frac{x_{i}}{2}\right)} \right]. \quad (4.93)$$

4.6 Concluding remarks

In this paper we studied the normalised two-point function $\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle/\langle \mathcal{T}_{\mu}\rangle^2$ of the composite twist field \mathcal{T}_{μ} and its conjugate. The motivation to study this object comes from recent investigations of symmetry-resolved entanglement entropy in QFTs possessing an internal \mathbb{Z}_2 symmetry [19, 94, 107]. More fundamentally, our work contributes to developing the understanding of correlation functions in the replica Ising field theory, a theory that contains a large number of symmetry fields or twist fields which are not present in the standard, non-replicated model.

We employed traditional IQFT techniques, namely the the form factor bootstrap program adapted to composite twist fields [107], to expand the logarithm of the correlation function into a series of cumulants. The main result of this chapter consists in deriving closed-form expressions for these cumulants which result from a number of multiple sum formulae involving the two-particle form factors of the field \mathcal{T}_{μ} , presented in Appendix 4.A.

Employing the cumulant expansion we found the following structure

$$\frac{\langle \mathcal{T}_{\mu}(0)\tilde{\mathcal{T}}_{\mu}(\ell)\rangle}{\langle \mu(0)\mu(\ell)\rangle}\frac{\langle \mu\rangle^2}{\langle \mathcal{T}_{\mu}\rangle^2} = \mathcal{R}(\ell;n) \quad \text{with} \quad \mathcal{R}(\ell;1) = 1,$$
(4.94)

where μ is the disorder field of the Ising field theory, and we provided an explicit expression for $\mathcal{R}(\ell; n)$. By exact resummation of leading contributions to the cumulant expansion, we showed that at short distances this two-point function scales as a power law in r with exponent consistent with the CFT dimension (4.20). Furthermore, we provided the analytic continuation of our formulae to real replica number, generalising results found in [107] and [20]. As a byproduct of our investigation, we have also showed how the form factors of the composite field \mathcal{T}_{σ} can be obtained from those of \mathcal{T}_{μ} via clustering in momentum space, in much the same way as the form factors of the fields σ and μ are related.

As mentioned above, our result has applications in the context of the symmetry-resolved entanglement entropy and directly leads to a more complete formula for the latter in the Ising model. We further expect the results of this investigation to apply with some modifications to other composite fields, at least for free theories, for instance those associated with U(1) symmetry in doubled free models which were studied in [105, 123, 124].

4.A Summation formulae

In this appendix we derive the identities presented in Section 4.3.1. Let us first prove equations (4.35) and (4.34), which are both obtained via contour integration. To show (4.35), consider the

CHAPTER 4. TWO-POINT FUNCTION OF COMPOSITE TWIST FIELDS IN THE DISORDERED PHASE OF THE ISING FIELD THEORY

integral in the complex plane:

$$\frac{1}{2\pi i} \oint_{\mathcal{C}} \mathrm{d}z \,\pi \cot(\pi z) w(x^z),\tag{4.95}$$

where C is the rectangular contour with vertices $-\epsilon + iL$, $-\epsilon - iL$, $n - \epsilon - iL$, $n - \epsilon + iL$. The vertical contributions cancel off because the integrand is invariant under the shift $z \to z + n$. The same holds for the horizontal contributions in the large L limit, as

$$\lim_{L \to \infty} \cot \pi (t \pm iL) = \mp i, \quad \lim_{L \to \infty} w \left(x^{t \pm iL} \right) = \mp \frac{i}{n}, \quad t \in \mathbb{R},$$
(4.96)

and therefore the sum of the residues must vanish. Within the integration contour, the function $\pi \cot(\pi z)$ has simple poles at $z = 0, 1, \ldots, n-1$ with unite residue. The kinematic poles of $w(x^z)$ are at $z = \frac{1}{2} - \frac{x}{2\pi i}, z = n - \frac{1}{2} - \frac{x}{2\pi i}$, with residue $\frac{1}{2\pi}$. At both these points, $\cot(\pi z) = -i \tanh \frac{x}{2}$. Putting all the pieces together, one has therefore:

$$0 = \sum_{j=0}^{n-1} w(x^j) - i \tanh \frac{x}{2}.$$
(4.97)

Using the very same strategy, one can prove (4.34). The integral to evaluate is now

$$\frac{1}{2\pi i} \oint_{\mathcal{C}} \mathrm{d}z \,\pi \cot(\pi z) w((-x)^z) w(y^z),\tag{4.98}$$

along the same contour C as before. In this case, however, the horizontal contributions do not cancel off, as

$$\lim_{L \to \infty} w((-x)^{t \pm iL}))w(y^{t \pm iL}) = -\frac{1}{n^2}, \quad t \in \mathbb{R},$$
(4.99)

and thus the integral evaluates to $-\frac{1}{n}$ in the large *L* limit. Summing over the residues of the poles of $\pi \cot(\pi z)$ gives the left-hand side of (4.34), while the kinematic poles are now at $z = \frac{1}{2} + \frac{x}{2\pi i}, -\frac{1}{2} + n + \frac{x}{2\pi i}$ and $z = \frac{1}{2} - \frac{y}{2\pi i}, -\frac{1}{2} + n - \frac{y}{2\pi i}$, with residues:

$$\begin{split} & \underset{z=\frac{1}{2}+\frac{x}{2\pi i}}{\operatorname{Res}} w((-x)^z)w(y^z) = \frac{1}{2\pi}w(x+y+i\pi), & \underset{z=n-\frac{1}{2}+\frac{x}{2\pi i}}{\operatorname{Res}} w((-x)^z)w(y^z) = \frac{1}{2\pi}w(x+y-i\pi), \\ & \underset{z=\frac{1}{2}-\frac{y}{2\pi i}}{\operatorname{Res}} w((-x)^z)w(y^z) = -\frac{1}{2\pi}w(x+y-i\pi), & \underset{z=n-\frac{1}{2}-\frac{y}{2\pi i}}{\operatorname{Res}} w((-x)^z)w(y^z) = -\frac{1}{2\pi}w(x+y+i\pi). \end{split}$$

By evaluating the cotangent at the kinematic poles and putting all the pieces together, we obtain

$$\begin{aligned} -\frac{1}{n} &= \sum_{j=0}^{n-1} \operatorname{Res}[\pi \cot(\pi z)w((-x)^z)w(y^z)] \\ &= \sum_{j=0}^{n-1} w((-x)^j)w(y^j) + \frac{i}{2} \frac{\sinh\left(\frac{x+y}{2}\right)}{\cosh\left(\frac{x}{2}\right)\cosh\left(\frac{y}{2}\right)} [w(x+y+i\pi) + w(x+y-i\pi)], \end{aligned}$$

which is indeed (4.34).

Using the above results, it is possible to prove (4.44) and (4.45) by induction. It is useful to observe beforehand that the following expansions hold:

$$\frac{\sinh\left(\sum_{i=1}^{k} x_i\right)}{\prod_{i=1}^{k} \cosh x_i} = \sum_{j=0}^{\left\lfloor\frac{k-1}{2}\right\rfloor} \sigma_{2j+1}^{(k)} (\tanh x_1, \dots, \tanh x_k),$$
(4.100)

$$\frac{\cosh\left(\sum_{i=1}^{k} x_i\right)}{\prod_{i=1}^{k} \cosh x_i} = \sum_{j=0}^{\left[\frac{k}{2}\right]} \sigma_{2j}^{(k)}(\tanh x_1, \dots, \tanh x_k).$$
(4.101)

Following the procedure employed in [63, 117], we will prove that (4.44) implies (4.45). If (4.44) holds, than we can shift x_{2k} by $-2i\pi p$, multiply the left-hand side by a factor $w(x_{2k+1}^p)$ and sum over p to obtain:

$$\sum_{j_1,\dots,j_{2k-1},p=0}^{n-1} w((-x_1)^j) w(x_2^{j_1-j_2}) \dots w(x_{2k-1}^{j_{2k-2}-j_{2k-1}}) w(x_{2k}^{j_{2k-1}-p}) w(x_{2k+1}^p)$$
$$= \sum_{p=0}^{n-1} f_k(x_1,\dots,x_{2k}^{-p},n) w(x_{2k+1}^p) + \sum_{p=0}^{n-1} \frac{(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh\frac{x_1}{2},\dots,\tanh\frac{x_{2k}^{-p}}{2} \right) w(x_{2k+1}^p), \quad (4.102)$$

Let us focus on the first term in the second line, which yields two contributions due to the presence of a constant term in the right-hand side of (4.34). Indeed, defining $x = \sum_{i=1}^{2k} x_i$, we obtain:

$$\sum_{p=0}^{n-1} f_k(x_1, \dots, x_{2k}^{-p}, n) w(x_{2k+1}^p)$$

$$= \frac{2i(-1)^k \sinh \frac{x}{2}}{\prod_{i=1}^{2k} 2 \cosh \frac{x_i}{2}} \sum_{p=0}^{n-1} \sum_{j=1}^k {\binom{2k-1}{k-j}} \left[w \left(x^{j-p} - i\pi \right) + w \left(x^{-j-p} + i\pi \right) \right] w(x_{2k+1}^p)$$

$$= g_k(x_1, \dots, x_{2k+1}, n) + \frac{4i(-1)^k \sinh \frac{x}{2}}{n \prod_{i=1}^{2k} 2 \cosh \frac{x_i}{2}} \sum_{j=1}^k {\binom{2k-1}{k-j}}$$

$$= g_k(x_1, \dots, x_{2k+1}, n) + \frac{i(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j+1}^{(2k)} (\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k}}{2}).$$
(4.103)

The emergence of the function g_k in going from the second to the third line was already proved in Appendix A of [63]. In going from the third to the fourth line we used the identity (4.100) and $\sum_{j=1}^{k} \binom{2k-1}{k-j} = 2^{2k-2}$. We now consider the second term in the second line of (4.102): using the fact that $\tanh \frac{x^{\pm p}}{2} = \tanh \frac{x}{2}$ and the sum (4.35), we have:

$$\sum_{p=0}^{n-1} \frac{(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k}^{-p}}{2} \right) w(x_{2k+1}^p)$$
$$= \frac{i(-1)^k}{n} \sum_{j=0}^{k-1} \sigma_{2j}^{(2k)} \left(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k}}{2} \right) \tanh \frac{x_{2k+1}}{2}.$$
(4.104)

Now we observe that the elementary symmetric polynomial of degree j in k variables can be

decomposed as $\sigma_j^{(k)}(a_1, \dots, a_k) = \sigma_j^{(k-1)}(a_1, \dots, a_{k-1}) + \sigma_{j-1}^{(k-1)}(a_1, \dots, a_{k-1}) a_k$, hence: $\sum_{j=0}^{k-1} \left[\sigma_{2j+1}^{(2k)}(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k}}{2}) + \sigma_{2j}^{(2k)}\left(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k}}{2}\right) \tanh \frac{x_{2k+1}}{2} \right]$ $= \sum_{i=0}^{k-1} \sigma_{2j+1}^{(2k+1)}(\tanh \frac{x_1}{2}, \dots, \tanh \frac{x_{2k+1}}{2}).$ (4.105)

Thus the sum of (4.103) and (4.104) yields (4.45). In an analogous way it is possible to prove (4.44) starting from (4.45).

CHAPTER

FIVE

ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

Global symmetries of quantum many-body systems can be spontaneously broken. Whenever this mechanism happens, the ground state is degenerate and one encounters an ordered phase. In this chapter, based on [5], we investigate this phenomenon by examining the entanglement asymmetry of a region in a one-dimensional many-body system in its ordered phase. This quantity has been recently introduced in the context of U(1) symmetry breaking, and we extend its definition to encompass arbitrary finite groups G. We also establish a field-theoretic framework in the replica theory using twist operators. We explicitly demonstrate our construction in the ordered phase of the Ising field theory in 1+1 dimensions, where a \mathbb{Z}_2 symmetry is spontaneously broken, and we employ a form factor bootstrap approach to characterise a family of composite twist fields. Analytical predictions are provided for the entanglement asymmetry of an interval in the Ising model as the length of the interval becomes large. We also propose a general conjecture relating the entanglement asymmetry and the number of degenerate vacua, which we expect to be valid for a large class of states, and we prove it explicitly in a simple case.

5.1 Introduction

Symmetry is nowadays considered a cornerstone of modern Physics. Its breaking is responsible for a plethora of interesting phenomena, such as ferromagnetism [204], superconduction, superfluidity [205]. Spontaneous symmetry breaking is the phenomenon by which a symmetry possessed by a quantum system, described in terms of its Hamiltonian/Lagrangian or its equations of motion, is not mirrored by the ground state of the system. For instance, it is very well known that at low-enough temperatures

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

some ferromagnetic materials magnetise spontaneously in a given direction, which depends solely on the way those materials are cooled down. A similar mechanism has been observed at zero temperature in frustrated quantum systems, e.g. quantum spin chains, where spontaneous symmetry breaking arises if an external parameter, such as a magnetic field or a chemical potential, is varied [206].

Zero-temperature phases with different symmetries can be separated by a quantum phase transition [207]. Close to the transition, quantum correlations are dominant, giving rise to a large amount of entanglement among the regions of the systems. In the last decade, much attention has been devoted to the relation between symmetries and entanglement [19, 92, 94], especially in the context of zero-temperature states close to phase transitions, which can be investigated quantitatively via QFT [104, 124–126]. As a result of this interest, a notion of symmetry-resolved entanglement for one-dimensional quantum many-body systems was introduced in [19, 92, 94] and further developed in [96, 97, 121, 185, 194, 208].

In contrast, until recently, little research focused on exploring the relationship between symmetry breaking and entanglement. In [29] a new measure of entanglement, dubbed *entanglement* asymmetry, was introduced to probe symmetry breaking in many-body systems. Originally, this quantity was used in the context of quench dynamics to analyse the restoration of a U(1) symmetry in a symmetry-breaking state of a quantum spin chain, evolved using a symmetric Hamiltonian [29–31]. Following these initial works, entanglement asymmetry was investigated in several other models, as we reviewed in Section 1.3. However, a general framework for the study of entanglement asymmetry in quantum field theories is still missing. Therefore, the purpose of this work is to apply the proposed approach of [29] to characterise the spontaneous symmetry-breaking pattern in equilibrium quantum many-body systems that can be described by 1+1D QFT.

We initiate this program by giving a definition of entanglement asymmetry that can be applied to any finite or compact Lie group, extending the construction already provided for U(1) in [29] and for \mathbb{Z}_N in [112]. Moreover, we provide a field-theoretic treatment of the one-dimensional quantum Ising model via form factor bootstrap. Our approach combines the expression of the Rényi entropies in terms of twist fields via the replica trick [18, 20] and its extension in the presence of additional Aharonov–Bohm fluxes [19], which stems from the action of the group and gives rise to composite (charged) twist fields [157]. A vast literature regarding integrable field theories where similar fields were considered is present, and we refer the reader to [1–4, 105–108] for further details. However, most of these works refer to paramagnetic phases of field theories, where a single symmetric vacuum is present: there, different ways of inserting the same total Aharonov–Bohm flux among the replicas give rise to the same result (see e.g [105, 107, 108]). An example of this is provided by complex free theories with U(1) symmetry, in which the flux can be equivalently inserted between the *n*th and the first replica or fractionalised and uniformly distributed among all replicas. These distinct choices correspond to the insertion of distinct operators, and symmetry-broken states explicitly spot these differences. This mechanism, which is new to our knowledge, lies at the core of entanglement asymmetry, as we will show.

Before delving into the main content of this work, we first provide a definition of Rényi entanglement asymmetry, inspired by [29], which applies to any finite group G. Let us consider a (possibly mixed) state ρ of a bipartite system $A \cup \overline{A}$, described by the Hilbert space

$$\mathcal{H} = \mathcal{H}_A \otimes \mathcal{H}_{\bar{A}}.\tag{5.1}$$

We assume that a finite group G acts unitarily on \mathcal{H} via a linear map $G \ni g \mapsto \hat{g} \in \text{End}(\mathcal{H})$ which satisfies

$$\hat{g} = \hat{g}_A \otimes \hat{g}_{\bar{A}} \in \operatorname{End}\left(\mathcal{H}_A\right) \otimes \operatorname{End}\left(\mathcal{H}_{\bar{A}}\right).$$
(5.2)

That is, A and \overline{A} are not mixed by G, which plays the role of a global symmetry for the system (see also [95]). Given ρ , we construct the reduced density matrix over A in the usual way as

$$\rho_A := \operatorname{Tr}_{\bar{A}}\left(\rho\right),\tag{5.3}$$

and we aim to understand whether ρ_A is symmetric under the group G. This is equivalent to asking whether the equality

$$\rho_A = \hat{g}_A \rho_A \hat{g}_A^{-1}, \tag{5.4}$$

always holds or it is violated for some $g \in G$, thus signaling a breaking of the symmetry. To do so, we introduce a fictitious density matrix $\tilde{\rho}_A$ defined as

$$\tilde{\rho}_A := \frac{1}{|G|} \sum_{g \in G} \hat{g}_A \rho_A \hat{g}_A^{-1}, \tag{5.5}$$

where |G| denotes the order of the group G. The quantity $\tilde{\rho}_A$ can be regarded as the symmetrisation of ρ_A under the adjoint action of the group (see e.g. a standard textbook in linear representations of finite groups [209]), as it is easy to show that $\tilde{\rho}_A$ is symmetric under G:

$$\hat{g}_A \tilde{\rho}_A \hat{g}_A^{-1} = \tilde{\rho}_A, \quad \forall g \in G.$$
(5.6)

Moreover, it is not difficult to check that

$$\tilde{\rho}_A = \rho_A \quad \text{iff} \quad \rho_A = \hat{g}_A \rho_A \hat{g}_A^{-1}, \quad \forall g \in G.$$
 (5.7)

Therefore, following the logic of [29], it is rather natural to compare the two states ρ_A and $\tilde{\rho}_A$ in order

to probe (spontaneous or explicit) symmetry breaking at the level of the subsystem A. We do so via the introduction of the *Rényi entanglement asymmetry*, defined as

$$\Delta S_n := \frac{1}{1-n} \log \operatorname{Tr}\left(\tilde{\rho}_A^n\right) - \frac{1}{1-n} \log \operatorname{Tr}\left(\rho_A^n\right),$$
(5.8)

that is the difference of Rényi entropies of the two states. Similarly, in the limit $n \to 1$ we get the difference of von Neumann entropies

$$\Delta S_1 := -\mathrm{Tr}\left(\tilde{\rho}_A \log \tilde{\rho}_A\right) + \mathrm{Tr}\left(\rho_A \log \rho_A\right),\tag{5.9}$$

and we refer to ΔS_1 as the *entanglement asymmetry*. We mention that the definitions above can be generalised to compact Lie groups in a straightforward way [210]. For instance, given the (normalised) Haar measure $\int_G dg$ of a compact Lie Group G [209], it is sufficient to replace (5.5) with

$$\tilde{\rho}_A := \int_G \mathrm{d}g \, \hat{g}_A \rho_A \hat{g}_A^{-1},\tag{5.10}$$

that is compatible with the original formulation of asymmetry valid for U(1) [29]. While most of the theory we discuss in this work is unchanged for continuous groups, finite groups are the only relevant ones in the context of spontaneous symmetry breaking in zero-temperature one-dimensional systems, due to the Mermin-Wagner theorem[211].

In this work we analyse in detail the Ising field theory [212, 213] in its ferromagnetic phase, probing the symmetry-breaking pattern

$$\mathbb{Z}_2 \to \{1\},\tag{5.11}$$

via the (Rényi) entanglement asymmetry associated with the group $G = \mathbb{Z}_2$. In particular, we consider one (of the two) spontaneously broken ground states, and we compute the asymmetry of an interval of size ℓ which is large compared to the correlation length $\sim m^{-1}$ of the model. We do so with a replica trick, relating the Rényi entanglement asymmetry for $n \geq 2$ integer to the expectation values of some \mathbb{Z}_2 composite twist fields. We describe systematically the form factors of these fields, and we obtain analytical results for their correlation functions in the two-particle approximation. The main result for the Ising QFT can be summarised by the following expression:

$$\Delta S_n \simeq \log 2, \quad m\ell \gg 1, \tag{5.12}$$

which is valid for any integer $n \ge 2$ up to exponentially small corrections which we compute at first order.

We structure this chapter as follows. In Section 5.2 we provide an explicit construction of the

composite twist operators, valid for finite-dimensional Hilbert spaces, and we relate their expectation values to the asymmetry of a subsystem. In Section 5.3, we review the scattering properties of the Ising field theory, and we characterise the form factors of the standard twist field in the ferromagnetic phase. These form factors are eventually employed in the computation of the Rényi entropies in this phase. Then, in Section 5.4, the core of our work, we extend our analysis to a family of composite \mathbb{Z}_2 twist fields. In particular, we focus on those fields with vanishing net \mathbb{Z}_2 flux across the replicas, and we establish a connection between their form factors and the ones of the standard twist fields. This analysis allows us to compute the entanglement asymmetry of a large interval. Finally, in Section 5.5 we identify a fundamental mechanism behind the large-volume behavior of the entanglement asymmetry of arbitrary (clustering) states, and we provide a general conjecture valid for any finite group G. We leave conclusions and outlook to Section 5.6. Appendix 5.A contains the details of some two-particle form factor calculations, while in Appendix 5.B we provide some useful bounds on the Rényi asymmetries for finite groups. Finally, in Appendix 5.C we discuss some results on the entanglement entropy in the ordered and disordered phase of the transverse-field Ising chain.

5.2 Twist operators and entanglement asymmetry

In this section, following the ideas of [214, 215], we introduce a family of operators in the replica theory, the twist operators, which allow us to express the entanglement measures of interest as expectation values. The branch-point twist fields in 1+1D replica theories, together with their composite versions and their generalisations to higher-dimensional theories, have been widely used throughout this thesis. However, the point of view we adopt here is slightly different. While in the previous chapters we defined the twist operators via their equal-time commutation relations with local fields of the theory, in this chapter we give a rigorous characterisation of these operators for finite-dimensional Hilbert spaces, extending the analysis of [214, 215] with the additional introduction of Aharonov-Bohm fluxes arising in the presence of the action of a group G. While the technical details of our construction do not immediately apply to infinite chains or quantum field theories, the main important properties are expected to remain valid in the continuum limit, as explained in [214].

5.2.1 Characterisation of the twist operators

Let us consider a finite-dimensional Hilbert space \mathcal{H} describing a bipartition $A \cup \overline{A}$ as in (5.1). We take *n* copies of the above, so that the total Hilbert space of the replica model is $\mathcal{H}^{\otimes n}$. We aim to define a twist operator \mathcal{T}_A , associated with a cyclic permutation among the replicas restricted to the

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

subsystem A. We do so by requiring that on any factorised state of the replica model

$$|v_1, \dots, v_n\rangle \otimes |\bar{v}_1, \dots, \bar{v}_n\rangle := \left(\otimes_{j=1}^n |v_j\rangle\right) \otimes \left(\otimes_{j=1}^n |\bar{v}_j\rangle\right), \quad |v_j\rangle \in \mathcal{H}_A, |\bar{v}_j\rangle \in \mathcal{H}_{\bar{A}}, \tag{5.13}$$

the action of \mathcal{T}_A is¹

$$\mathcal{T}_A(|v_1, v_2, \dots, v_n\rangle \otimes |\bar{v}_1, \dots, \bar{v}_n\rangle) := |v_n, v_1, \dots, v_{n-1}\rangle \otimes |\bar{v}_1, \dots, \bar{v}_n\rangle.$$
(5.14)

Physically, \mathcal{T}_A implements the permutation $j \to j + 1$ on A, where j is a replica index.

In the presence of a global symmetry associated with a group G, it is possible to "charge" the twist operator defined above via the action of G. Following the terminology of [19], this corresponds to the additional insertion of Aharonov–Bohm fluxes between the replicas. The action of G on \mathcal{H} , defined in (5.2), is naturally extended to the replica model $\mathcal{H}^{\otimes n}$. Thus, we can construct a composite twist operator $\mathcal{T}_{A}^{\{g_1,\ldots,g_n\}}$ obtained as the combination of the replica shift $j \to j + 1$ and the insertion of a flux g_j between the *j*th and the (j + 1)th replicas. We do that by defining

$$\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}} := \mathcal{T}_{A} \circ (\hat{g}_{1,A} \otimes \cdots \otimes \hat{g}_{n,A} \otimes \mathbb{1}_{\bar{A}}^{\otimes n}), \tag{5.15}$$

from which it follows that

$$\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}}\left(|v_{1},\ldots,v_{n}\rangle\otimes|\bar{v}_{1},\ldots,\bar{v}_{n}\rangle\right) = \hat{g}_{n,A}|v_{n}\rangle\otimes\hat{g}_{1,A}|v_{1}\rangle\otimes\cdots\otimes|\bar{v}_{1},\ldots,\bar{v}_{n}\rangle,\tag{5.16}$$

where $\hat{g}_{j,A}$ refers to the action of g_j restricted to A and $\mathbb{1}$ is the action of the identity element $1 \in G$. An operator is therefore associated to any *n*-tuple $\{g_1, \ldots, g_n\}$ and, in particular, \mathcal{T}_A is recovered for $\{g_1, \ldots, g_n\} = \{1, \ldots, 1\}.$

In the remaining part of this section, we investigate some useful properties of the composite twist operators. Namely, we relate them to specific traces (charged moments) that appear in the computation of the Rényi entanglement asymmetry. Then, we show that distinct twist operators can be related to each other via global unitary transformations induced by the group elements. Finally, we present the mutual locality relations of twist operators with local observables.

1. Computation of traces. Let $\rho \in \text{End}(\mathcal{H})$ be the density matrix of a (possibly mixed) state and $\rho_A = \text{Tr}_{\mathcal{H}_{\bar{A}}}\rho$. We show that the following identities hold:

$$\operatorname{Tr}_{\mathcal{H}^{\otimes n}}\left(\rho^{\otimes n}\mathcal{T}_{A}\right) = \operatorname{Tr}_{\mathcal{H}_{A}}\rho_{A}^{n},\tag{5.17}$$

¹By linearity this definition allows us to express the action of \mathcal{T}_A on any vector in $\mathcal{H}^{\otimes n}$.

$$\operatorname{Tr}_{\mathcal{H}^{\otimes n}}\left(\rho^{\otimes n}\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}}\right) = \operatorname{Tr}_{\mathcal{H}_{A}}\left(\rho_{A}\hat{g}_{n,A}\rho_{A}\hat{g}_{n-1,A}\ldots\rho_{A}\hat{g}_{1,A}\right).$$
(5.18)

These are relations between the (charged) moments of ρ_A and the expectation values of the (composite) twist operators in the replica model. Let us stress that in the left-hand side the traces are taken over $\mathcal{H}^{\otimes n}$, while in the right-hand side the traces are only over one copy of the subsystem A. Since the first relation is a special case of the second one, obtained when all the $g_j = 1$, we focus on (5.18).

A straightforward proof can be provided if one picks two orthonormal bases for $\mathcal{H}_A, \mathcal{H}_{\bar{A}}$ and compute the trace in those bases. We denote by $|e_j\rangle, |\bar{e}_j\rangle$ the generic basis elements of $\mathcal{H}_A, \bar{\mathcal{H}}_A$ in the *j*th replica respectively, and we simply expand

$$\operatorname{Tr}_{\mathcal{H}^{\otimes n}}\left(\rho^{\otimes n}\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}}\right)$$

$$=\sum_{\substack{e_{1},\ldots,e_{n}\\\bar{e}_{1},\ldots,\bar{e}_{n}}}\langle e_{1},\ldots,e_{n}|\otimes\langle\bar{e}_{1},\ldots,\bar{e}_{n}|\rho^{\otimes n}\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}}|e_{1},\ldots,e_{n}\rangle\otimes|\bar{e}_{1},\ldots,\bar{e}_{n}\rangle$$

$$=\sum_{e_{1},\ldots,e_{n}}\langle e_{1}|\rho_{A}\hat{g}_{n,A}|e_{n}\rangle\langle e_{2}|\rho_{A}\hat{g}_{1,A}|e_{1}\rangle\ldots\langle e_{n}|\rho_{A}\hat{g}_{n-1,A}|e_{n-1}\rangle$$

$$=\operatorname{Tr}_{\mathcal{H}_{A}}\left(\rho_{A}\hat{g}_{n,A}\rho_{A}\hat{g}_{n-1,A}\ldots\hat{g}_{2,A}\rho_{A}\hat{g}_{1,A}\right),$$
(5.19)

which proves equation (5.18).

2. Unitary transformations

An important observation is that different composite twist operators $\mathcal{T}_A^{\{g_1,\ldots,g_n\}}$, corresponding to different choices of $\{g_1,\ldots,g_n\}$, can be related to each other via global unitary transformations. Specifically, given $h_j \in G$, $j = 1, \ldots, n$, we show that

$$\left(\hat{h}_1 \otimes \cdots \otimes \hat{h}_n\right) \mathcal{T}_A^{\{g_1,\dots,g_n\}} \left(\hat{h}_1 \otimes \cdots \otimes \hat{h}_n\right)^{-1} = \mathcal{T}_A^{\{g'_1,\dots,g'_n\}}, \quad g'_j := h_{j+1}g_j h_j^{-1}.$$
(5.20)

In the relation above, the action of h_j is not restricted to the subsystem A. Indeed, the region A is not affected by the twist operator appearing in Eq. (5.20), and the combined action of h_j, h_j^{-1} gives the identity on that subsystem.

To prove (5.20) it is sufficient to show that the left- and right-hand side act in the same way on

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

factorised vectors of $\mathcal{H}^{\otimes n}$. Therefore, for a given state (5.13) we just compute

$$\begin{pmatrix} \hat{h}_{1} \otimes \cdots \otimes \hat{h}_{n} \end{pmatrix} \mathcal{T}_{A}^{\{g_{1},\dots,g_{n}\}} \begin{pmatrix} \hat{h}_{1} \otimes \cdots \otimes \hat{h}_{n} \end{pmatrix}^{-1} |v_{1},\dots,v_{n}\rangle \otimes |\bar{v}_{1},\dots,\bar{v}_{n}\rangle$$

$$= \begin{pmatrix} \hat{h}_{1} \otimes \cdots \otimes \hat{h}_{n} \end{pmatrix} \begin{pmatrix} \hat{g}_{n,A} \hat{h}_{n,A}^{-1} |v_{n}\rangle \otimes \cdots \otimes \hat{g}_{n-1,A} \hat{h}_{n-1,A}^{-1} |v_{n-1}\rangle \otimes \hat{h}_{1,\bar{A}}^{-1} |\bar{v}_{1}\rangle \otimes \cdots \otimes \hat{h}_{n,\bar{A}}^{-1} |\bar{v}_{n}\rangle$$

$$= \begin{pmatrix} \hat{h}_{1,A} \hat{g}_{n,A} \hat{h}_{n,A}^{-1} |v_{n}\rangle \otimes \cdots \otimes \hat{h}_{n,A} \hat{g}_{n-1,A} \hat{h}_{n-1,A}^{-1} |v_{n-1}\rangle \end{pmatrix} \otimes |\bar{v}_{1},\dots,\bar{v}_{n}\rangle$$

$$= \mathcal{T}_{A}^{\{h_{2}g_{1}h_{1}^{-1},\dots,h_{1}g_{n}h_{n}^{-1}\}} |v_{1},\dots,v_{n}\rangle \otimes |\bar{v}_{1},\dots,\bar{v}_{n}\rangle,$$

$$(5.21)$$

that is an elementary proof of (5.20).

We now discuss a number of remarkable consequences of the above equation. First, the relation $g'_j = h_{j+1}g_jh_j^{-1}$ implies that

$$g'_n g'_{n-1} \dots g'_1 = 1$$
 iff $g_n g_{n-1} \dots g_1 = 1$, (5.22)

regardless of the choice of $\{h_j\}$. Physically, this means that a twist operator $\mathcal{T}_A^{\{g_1,\ldots,g_n\}}$ with a vanishing net Aharonov-Bohm flux accumulated across the replicas is unitarily equivalent via (5.20) to all and only the other operators $\mathcal{T}_A^{\{g'_1,\ldots,g'_n\}}$ with the same property. To characterise this unitary transformation it is sufficient to show that for any $\{g'_i\}$ satisfying $g'_n \ldots g'_1 = 1$ there is a *n*-tuple $\{g_j\}$ such that

$$g'_j = g_{j+1}g_j^{-1}. (5.23)$$

A solution to (5.23) is indeed

$$g_1 = 1, \quad g_2 = g'_1, \quad g_3 = g'_2 g'_1, \quad \dots \quad g_n = g'_{n-1} \dots g'_1.$$
 (5.24)

Therefore, with the previous choices of $\{g_j\}, \{g'_j\}$, it holds

$$(\hat{g}_1 \otimes \cdots \otimes \hat{g}_n) \mathcal{T}_A (\hat{g}_1 \otimes \cdots \otimes \hat{g}_n)^{-1} = \mathcal{T}_A^{\{g'_1, \dots, g'_n\}},$$
(5.25)

which means that \mathcal{T}_A is always unitarily equivalent to a given composite twist operator with a vanishing net flux.

Finally, we notice that distinct unitary transformations can give rise to the same twist operator, since equation (5.23) has multiple solutions. For example, it is easy to show that

$$\left(\hat{g}^{\otimes n}\right)\mathcal{T}_{A}\left(\hat{g}^{\otimes n}\right)^{-1}=\mathcal{T}_{A},$$
(5.26)

for any choice of $g \in G$, which means that the standard twist operator \mathcal{T}_A is neutral under a global unitary transformation. More generally, equation (5.23) has precisely |G| solutions: indeed, if $\{g_j\}$ is a solution of (5.23) for a given *n*-tuple $\{g'_j\}$, then also $\{g_jg\}$ satisfies the same relation for any $g \in G$.

3. Action on local observables

The last properties we show are the commutation relations between the twist operators and the local observables of the system. This is particularly useful to make an explicit connection between the construction proposed above and the defining properties of twist operators in QFT based on the algebra of local observables. Let $\mathcal{O} \in \text{End}(\mathcal{H}_A)$ be an observable of the subsystem A. \mathcal{O} can be naturally embedded in the space of observables of the whole system $A \cup \overline{A}$ by mapping it to $\mathcal{O} \otimes \mathbb{1}_{\overline{A}} \in \text{End}(\mathcal{H})$. Similarly, we can embed \mathcal{O} in the *j*th replica of the replica model by defining

$$\mathcal{O}^{j} := \underbrace{\mathbb{1}_{A} \otimes \dots \mathcal{O}}_{j} \otimes \dots \mathbb{1}_{A} \otimes \mathbb{1}_{\overline{A}}^{\otimes n} \in \operatorname{End}(\mathcal{H}^{\otimes n}).$$
(5.27)

As a consequence of the definitions, it is not difficult to show that

$$\mathcal{T}_A \mathcal{O}^j = \mathcal{O}^{j+1} \mathcal{T}_A, \tag{5.28}$$

and we say that the twist operator acts as a replica shift $j \to j + 1$ on the observable \mathcal{O} of A. Similarly, one has

$$\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}}\mathcal{O}^{j} = \left(\hat{g}_{j}^{\otimes n}\right)\mathcal{O}^{j+1}\left(\hat{g}_{j}^{\otimes n}\right)^{-1}\mathcal{T}_{A}^{\{g_{1},\ldots,g_{n}\}}.$$
(5.29)

Physically, this means that the action of the composite twist operators on a local observable of A amounts to the usual replica shift followed by the action of the group element g_j on \mathcal{O} . Vice versa, local observables of \overline{A} are not affected by the presence of the twist operators. For instance, given $\mathcal{O} \in \text{End}(\mathcal{H}_{\overline{A}})$, and \mathcal{O}^j its embedding in the replica model, one easily shows

$$\mathcal{T}_A^{\{g_1,\dots,g_n\}}\mathcal{O}^j = \mathcal{O}^j \mathcal{T}_A^{\{g_1,\dots,g_n\}}.$$
(5.30)

Therefore, as expected, the action of the twist operator on A commutes with the observables of the complement \bar{A} .

5.2.2 Entanglement asymmetry via twist operators

In this section, we make a connection between the twist operators introduced previously and the definition of the Rényi entanglement asymmetry in Eq. (5.8). To do so, we characterise the Rényi asymmetry of a state $\rho \in \text{End}(\mathcal{H}_A \otimes \mathcal{H}_{\bar{A}})$ for an integer $n \geq 2$. First, we recall that (5.17) allows us to express the Rényi entropy of ρ_A via the twist operator. Then, by making use of the definition of

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

 $\tilde{\rho}_A$ in Eq. (5.5) and after a change of variable, we obtain

$$\operatorname{Tr}\left(\tilde{\rho}_{A}^{n}\right) = \frac{1}{|G|^{n}} \sum_{g_{1},\dots,g_{n}\in G} \operatorname{Tr}\left(\hat{g}_{1,A}\rho_{A}\hat{g}_{1,A}^{-1}\hat{g}_{2,A}\rho_{A}\hat{g}_{2,A}^{-1}\dots\hat{g}_{n,A}\rho_{A}\hat{g}_{n,A}^{-1}\right)$$

$$= \frac{1}{|G|^{n-1}} \sum_{g_{1}',\dots,g_{n-1}'\in G} \operatorname{Tr}\left(\rho_{A}\hat{g}_{1,A}'\dots\rho_{A}\hat{g}_{n-1,A}'\rho_{A}(\hat{g}_{1,A}'\dots\hat{g}_{n-1,A}')^{-1}\right).$$
(5.31)

Here, the second line follows from the change of variable $g'_j = g_j^{-1}g_{j+1}$. Notably, the g'_j are not independent since they satisfy the constraint $\prod_j g'_j = 1$. This allows us to express $\text{Tr}(\tilde{\rho}^n_A)$ as a sum of $|G|^{n-1}$ terms rather than $|G|^n$ terms in the second line. Finally, we employ the property (5.18) and we express the moments of $\tilde{\rho}_A$ using the composite twist operators as

$$\operatorname{Tr}\left(\tilde{\rho}_{A}^{n}\right) = \frac{1}{|G|^{n-1}} \sum_{g_{1},\dots,g_{n-1}\in G} \operatorname{Tr}\left(\rho^{\otimes n} \mathcal{T}_{A}^{\{g_{1},\dots,g_{n-1},(g_{1}\dots,g_{n-1})^{-1}\}}\right),\tag{5.32}$$

and, eventually, from (5.8) we get

$$\Delta S_n = \log|G| + \frac{1}{1-n} \log\left(\sum_{g_1,\dots,g_{n-1}\in G} \frac{\operatorname{Tr}\left(\rho^{\otimes n} \mathcal{T}_A^{\{g_1,\dots,g_{n-1},(g_1\dots,g_{n-1})^{-1}\}}\right)}{\operatorname{Tr}\left(\rho^{\otimes n} \mathcal{T}_A\right)}\right).$$
 (5.33)

We point out that all the possible $|G|^{n-1}$ composite twist operators with a vanishing net Aharonov–Bohm flux appear explicitly² in the sum in (5.33).

5.3 Form factor program for the twist fields in the Ising ferromagnetic phase

In this section, our focus is on the integrable QFT of the Ising model in 1+1 dimensions in its ferromagnetic phase. We aim to provide an explicit characterisation of the twist operators introduced in Section 5.2 by means of form factor bootstrap. In particular, we consider n replicas of the Ising model, and we investigate a family of branch-point twist fields (BPTF). This allows us to obtain the Rényi entropy of the symmetry-broken phase, which is different from the one of the disordered phase investigated in [20].

Before delving into the details of the Ising model, it is worth making a connection between the twist operators of Section 5.2 and the BPTF widely used in 1+1D quantum field theories (see e.g. [18] for CFTs, and [20] for massive integrable QFTs). For any half-line $A = (x, \infty)$, we regard the

²Namely, the operators appearing in the sum are all and only the ones with zero net flux across all replicas. This condition is required for the consistency of our construction. Indeed, an operator with a net flux different from zero would signal that the state has a definite charge value under the symmetry G, but this would in turn imply vanishing entanglement asymmetry.

corresponding twist operator as a (semi-local) field inserted at x and we denote it by $\mathcal{T}(x)$, namely

$$\mathcal{T}_A \sim \mathcal{T}(x), \quad A = (x, \infty),$$
(5.34)

up to a non-universal proportionality constant that has been neglected. These fields are the building blocks needed to reconstruct the entanglement properties of any region, and through their insertion at different points, one can express the twist operator of any union of disjoint intervals [18]. A paradigmatic example is the case of a single finite interval, where it is possible to express

$$\mathcal{T}_A \sim \mathcal{T}(0)\mathcal{T}^{\dagger}(\ell), \quad A = (0,\ell),$$
(5.35)

with $\mathcal{T}, \mathcal{T}^{\dagger}$ a pair of Hermitian conjugated twist fields. This correspondence was generalised to U(1) composite twist fields in [2, 3].

Having made the identification (5.34) or (5.35), it is worth noting that the expression for ΔS_n in (5.33) is invariant after rescaling of the space coordinate. Indeed, under a transformation $x \to \lambda x$ the twist operators transform with the same scaling dimension d:

$$\mathcal{T}_{A} \to \lambda^{-d} \mathcal{T}_{A}, \quad \mathcal{T}_{A}^{\{g_{1},\dots,g_{n-1},(g_{1},\dots,g_{n-1})^{-1}\}} \to \lambda^{-d} \mathcal{T}_{A}^{\{g_{1},\dots,g_{n-1},(g_{1},\dots,g_{n-1})^{-1}\}}.$$
(5.36)

This observation is fundamental for quantum field theories in which spontaneous symmetry breaking occurs, and it can be argued from the fact that the two fields are related by a global unitary transformation. That is, equation (5.25) implies that the charged and uncharged twist operators must have the same (UV) scaling dimension. Indeed, in the limit of a small region their correlation functions are expected to be the same as those computed at the UV critical point. But because the critical point is invariant under the symmetry group G, these expectation values are equal, and so must be the scaling dimensions of the two fields. Furthermore, because of equation (5.25), the normalisation of \mathcal{T}_A unambiguously determines the normalisation of $\mathcal{T}_A^{g_1,\ldots,g_{n-1},(g_1,\ldots,g_{n-1})^{-1}}$. Consequently, the ratio appearing in (5.33), and therefore the entanglement asymmetry, is universal. It is important to emphasise that this is not the case for entanglement entropy (see for instance [10, 216]) due to the presence of UV divergences.

Despite the extensive literature existing on twist fields in 1+1D integrable QFT, the vast majority of those works [18, 21, 200, 217, 218] focus on the entanglement properties in the the paramagnetic (disordered) phase only, where a single vacuum is present. However, in the presence of spontaneous symmetry breaking, multiple vacua are present, and one must be cautious when specifying them. Moreover, as there might be elementary excitations (kinks) interpolating between different vacua, analytic continuations in rapidity space may result in a mixing of the corresponding form factors, a

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

mechanism that is well understood for the q-state Potts field theory [219], which describes the Ising model for q = 2. To the best of our knowledge, the consequence of vacuum degeneracy in the context of entanglement for field theories has not been explored yet. Our objective is to address this gap by examining the simplest, yet non-trivial case: the Ising model. In particular, in this section we aim to formulate and solve the form factor bootstrap equations for the standard twist fields in the ferromagnetic case. These results will be eventually employed to compute the Rényi entropy of a symmetry-broken ground state.

We first review some basic properties of the massive Ising field theory [60–62, 81, 212, 213], and then we discuss its *n*-replica model, where the twist fields arise. The theory can be regarded as a relevant perturbation of the Ising CFT [166, 198] with Euclidean action³

$$S = S_{\rm CFT} + \lambda \int dx d\tau \ \varepsilon(x, \tau).$$
(5.37)

Here S_{CFT} is the action at criticality, associated to a CFT with central charge c = 1/2, λ is a parameter with the dimension of a mass and $\varepsilon(x,\tau)$ is a scalar field of dimension 1 representing an energy density. Depending on the sign of λ , the theory is in its paramagnetic (disordered, $\lambda < 0$) or ferromagnetic (ordered, $\lambda > 0$) phase. We focus on the ferromagnetic phase where spontaneous symmetry breaking arises, and two degenerate vacua $|+\rangle$, $|-\rangle$ related by \mathbb{Z}_2 symmetry are present. The elementary excitations above the ground state(s) are multi-kink configurations interpolating between the two vacua [219]. In particular, a single-particle (kink) state with rapidity θ , denoted by

$$|K_{+-}(\theta)\rangle,\tag{5.38}$$

interpolates between $|+\rangle$ at $-\infty$ and $|-\rangle$ at ∞ . To make contact with the quantum Ising chain, one should regard a one-kink state with definite momentum as the $L \to \infty$ limit of the plane-wave superposition:

$$|K_{+-}(p)\rangle = \mathcal{N} \sum_{j=-L+1}^{L} e^{ipj} |\tilde{K}_{+-}(j)\rangle,$$
 (5.39)

where p is the quantised momentum, \mathcal{N} is a normalisation factor and $|\tilde{K}_{+-}(j)\rangle$ is the domain wall created by the lattice disorder operator $\mu_j = \prod_{j \leq i} \sigma_i^z$ on the state $\otimes_{i=-L}^L |+\rangle_i$:

$$|\tilde{K}_{+-}(j)\rangle = \mu_j \otimes_{i=-L}^{L} |+\rangle_i = |+\rangle_{-L} \cdots \otimes |+\rangle_{j-1} \otimes |-\rangle_j \cdots \otimes |-\rangle_L.$$
(5.40)

In the continuum limit, the momentum p is parametrised by the rapidity as $p = m \sinh \theta$. In general,

³After changing coordinates from Euclidean to holomorphic and expressing the energy density in terms of the fermion field, equation (5.37) reduces to the Majorana action (4.1).

a configuration with N kinks has the form

$$|K_{\alpha_1\alpha_2}(\theta_1)K_{\alpha_2\alpha_3}(\theta_2)\dots K_{\alpha_N\alpha_{N+1}}(\theta_N)\rangle, \tag{5.41}$$

where the indices $\alpha_j \in \{+, -\}$ label the vacua and $\theta_j \in \mathbb{R}$, j = 1, ..., N are the rapidities of the excitations. We say that the configuration is neutral, or topologically trivial, if $\alpha_1 = \alpha_{N+1}$. It is convenient to identify four topological sectors, labelled by the values of (α_1, α_{N+1}) : physically, these sectors correspond to different choices of the boundary conditions (i.e. spin up or spin down) at $x = -\infty$ and $x = +\infty$, and they can be regarded as constraints for the states interpolating between them.

The multi-kink configurations are the (asymptotic) excited states of the theory, and their energy-momentum is

$$(E,P) = \left(m\sum_{j=1}^{N}\cosh\theta_j, m\sum_{j=1}^{N}\sinh\theta_j\right).$$
(5.42)

Here $m = |\lambda|$ is the mass of each kink, and it plays the role of inverse correlation length away from criticality. When two kinks scatter, they acquire a scattering phase $e^{i\delta} = -1$ which does not depend on the rapidity difference, a property ultimately related to the equivalence (see for instance [198]) between the Ising model and free fermionic Majorana field theory⁴.

For any field $\mathcal{O}(x)$ one can define the form factors, i.e. the matrix elements of \mathcal{O} between the vacua and the multi-kink states, as

$$\langle \pm |\mathcal{O}(0)| K_{\alpha_1 \alpha_2}(\theta_1) K_{\alpha_2 \alpha_3}(\theta_2) \dots K_{\alpha_N \alpha_{N+1}}(\theta_N) \rangle.$$
(5.43)

In principle, depending on the field \mathcal{O} , there may be several non-vanishing form factors; however, their number is drastically reduced by topological constraints, as discussed in [219] for the *q*-state Potts model. In the Ising field theory, for the order field σ only the form factors with neutral states may be non-vanishing, and they are

$$\langle \pm | \sigma(0) | K_{\pm \alpha_2}(\theta_1) K_{\alpha_2 \alpha_3}(\theta_2), \dots K_{\alpha_N \pm}(\theta_N) \rangle, \qquad (5.44)$$

together with the vacuum expectation values (VEVs) $\langle \pm |\sigma(0)|\pm \rangle$. In contrast, the disorder field $\mu(0)$ is topologically non-trivial, as it introduces a spin-flip along the half-line $[0, +\infty)$ that mixes the two

⁴There are some technical differences between the Ising model and the field theory of free Majorana fermions, which can be traced back to the presence of distinct spin sectors. However, since this feature does not play any role in this work, we do not discuss it further.

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY



Figure 5.1 (Left) Order field $\sigma(x)$: the two-particle form factor $\langle +|\sigma(0)|K_{+-}(\theta_1)K_{-+}(\theta_2)\rangle$ is shown. (Right) Disorder field $\mu(x)$: the three-particle form factor $\langle +|\mu(0)|K_{+-}(\theta_1)K_{-+}(\theta_2)K_{+-}(\theta_3)\rangle$ is represented. The red line corresponds to a spin-flip exchanging the two vacua. Space is in the horizontal direction and time in the vertical direction.

vacua, and the form factors which are not ruled out by topological constraints are

$$\langle \pm | \mu(0) | K_{\pm \alpha_2}(\theta_1) K_{\alpha_2 \alpha_3}(\theta_2) \dots K_{\alpha_N \mp}(\theta_N) \rangle.$$
(5.45)

In Figure 5.1 we represent pictorially the form factors of the order and disorder field. While $\sigma(x)$ is a local field, $\mu(x)$ is semi-local, and we depict this semi-locality with a red line in Figure 5.1. The locality properties of the two fields fix the boundary conditions at infinity for incoming states interpolating with the outgoing vacuum. In particular, only an even/odd number of kinks gives rise to non-vanishing form factors for $\sigma(x), \mu(x)$ respectively.

In the next sections, we first extend the analysis above to the twist fields of the replica model, identifying their topological constraints. Then, via the bootstrap program, we characterise analytically their non-vanishing form factors, and we give an analytical expression for the two-particle ones.

5.3.1 Standard twist fields

The replica theory of the Ising model in the ordered phase is obtained by taking n decoupled copies of the model and considering the resulting theory itself as a QFT. The (asymptotic) spectrum of the replica QFT is expressed in terms of single-replica vacua and kink states, as no correlations are present between distinct copies. In particular, in the ordered phase, there are 2^n degenerate vacua which are obtained as tensor products of single-replica vacua:

$$|+\rangle^{\otimes n}, |+\rangle^{\otimes n-1} \otimes |-\rangle, \dots, |-\rangle^{\otimes n}.$$
(5.46)

The excitations can be described via the insertion of kinks in each replica, similarly to what happens with quasi-particle excitations in the paramagnetic phase (see [20]). Moreover, the scattering matrix

between two kinks at replicas i and j with rapidities θ_i , θ_j respectively is

$$S_{ij}(\theta_i - \theta_j) = \begin{cases} -1 & i = j, \\ 1 & i \neq j, \end{cases}$$
(5.47)

since no interactions between distinct replicas are present. This further implies that two kinks on the same replica satisfy the Faddeev-Zamolodchikov algebra:

$$K_{\pm\mp}(\theta_1)K_{\mp\pm}(\theta_2) = -K_{\pm\mp}(\theta_2)K_{\mp\pm}(\theta_1),$$
(5.48)

which is consistent with the algebra of the scattering kinks in the q-state Potts model [219, 220] for q = 2.

The standard BPTF is as usual introduced as a semi-local field of the replica model implementing the replica cyclic permutation $j \rightarrow j + 1$. In particular, from the commutation relations (5.28), (5.30) and the identification (5.35) we obtain

$$\mathcal{T}(x)\mathcal{O}_{j}(y) = \begin{cases} \mathcal{O}_{j+1}(y)\mathcal{T}(x) & x < y, \\ \mathcal{O}_{j}(y)\mathcal{T}(x) & \text{otherwise,} \end{cases}$$
(5.49)

for any local field⁵ $\mathcal{O}_j(x)$ inserted in the *j*th replica. In particular, this reproduces the commutation relation (4.3) between the BPTF and the fermion field in the paramagnetic phase. According to this commutation relation (see Refs [18, 20]), we say that $\mathcal{T}(x)$ inserts a branch cut on the semi-infinite line (x, ∞) , which connects the *j*th replica with the (j + 1)th. It is important to stress that Eq. (5.49), rather than unambiguously identifying a single field $\mathcal{T}(x)$, selects a space of fields with given monodromy properties (details can be found in [64]). However, we are interested in the lightest fields satisfying (5.49), which correspond to a deformation of the primary twist fields of CFTs [18]. As seen in the previous chapters, these are scalar fields with conformal dimension $\Delta_{\mathcal{T}}$ given by (1.27).

In the remaining part of the section, we study the form factors of branch-point twist fields in the ferromagnetic phase of the Ising model. These are matrix elements of \mathcal{T} , \mathcal{T}^{\dagger} between the multi-kink state and one of the 2^n vacua. For our purposes, the most interesting form factors are the ones with the vacuum $\langle +|^{\otimes n}$. However, as we will see, other form factors are generated from the latter using the bootstrap equations.

⁵One should be careful about the definition of local fields, and more in general local observables. Here, we just mean that \mathcal{O} belongs to the algebra generated by the fields ε (the energy density) and the order operator σ . This requirement is very natural in the lattice counterpart, as it corresponds to the usual notion of locality in the computational basis of the quantum spin chain.

5.3.1.1 Zero-particle form factors

The simplest form factor that is expected to be non-vanishing is the VEV

$$^{\otimes n}\langle +|\mathcal{T}(0)|+\rangle^{\otimes n}.$$
(5.50)

We represent it pictorially in Figure 5.2: the *n* replicas are connected together along the branch cuts inserted by the twist fields \mathcal{T} at the origin of space-time, the state $|+\rangle^{\otimes n}$ ($^{\otimes n}\langle+|$) is the incoming (outgoing) state represented below (above) the origin in each replica. In particular, since the standard twist fields exchange replicas but do not act as spin-flips, if \pm is present in the *j*th replica above the branch cut, then \pm has to be present also in the (j + 1)th replica just below the branch cut. In other words, fixing the boundary condition of the outgoing state at spacial infinity unambiguously fixes the boundary conditions of the incoming state. Specifically, the boundary conditions at $x = -\infty$ remain the same for both outgoing and incoming states, while the boundary conditions at $x = +\infty$ are connected through a replica shift $j \to j+1$. For a similar discussion concerning the (single-replica) q-state Potts model, we refer the reader to [219].

According to these considerations, it is not difficult to show that the only other non-vanishing zero-particle form factor is

$$^{\otimes n}\langle -|\mathcal{T}(0)|-\rangle^{\otimes n}.$$
(5.51)

Its value can be related to the one of $\otimes^n \langle + | \mathcal{T}(0) | + \rangle^{\otimes n}$ via a global \mathbb{Z}_2 symmetry. Indeed, let us consider the global unitary transformation μ^6 which generates the \mathbb{Z}_2 symmetry exchanging the two vacua. In the replica theory, it holds

$$\mu^{\otimes n} |\pm\rangle^{\otimes n} = |\mp\rangle^{\otimes n}. \tag{5.52}$$

Since the twist field is neutral with respect to the action of the global spin-flip $\mu^{\otimes n}$ (see Eq. (5.26)), this immediately implies that the two zero-particle form factors above are equal: this value is referred to as

$$\tau := {}^{\otimes n} \langle \pm | \mathcal{T}(0) | \pm \rangle^{\otimes n}.$$
(5.53)

5.3.1.2 Vanishing one-particle form factors and further constraints

Here, we show that the one-particle form factors are all vanishing. To do that, it is sufficient to observe that these form factors are not compatible with the topological constraints arising from the boundary conditions at spacial infinity.

⁶Notice that the operator μ acts as a spin-flip everywhere while the field $\mu(x)$ acts on the half-line $(x, +\infty)$. Roughly, the relation between these operators is (up to a proportionality constant) $\mu \propto \mu(-\infty)\mu^{\dagger}(+\infty)$.



Figure 5.2 The form factors of the twist field $\mathcal{T}(0)$. Consecutive replicas are connected to each other via the branch cut, represented by a dashed line starting at x = 0. Top: vacuum expectation value ${}^{\otimes n}\langle +|\mathcal{T}(0)|+\rangle^{\otimes n}$. Bottom: Two-particle form factors ${}^{\otimes n}\langle +|\mathcal{T}(0)|K_{+-}(\theta_1)K_{-+}(\theta_2),+,\ldots,+\rangle$ with two kinks in the same replica.

Let us consider a generic outgoing vacuum state $\langle \alpha_1, \ldots, \alpha_n |$, with $\alpha_j = \pm$. If $\mathcal{T}(0)$ interpolates between a multi-particle incoming state and the vacuum above, we can immediately obtain the boundary conditions for the incoming state. Indeed, it is easy to show that the boundary conditions for the latter in the *j*th replica at $x = -\infty$, $x = \infty$ have to be α_j , α_{j-1} respectively. This property rules out the presence of matrix elements with an odd number of kinks having $\langle \alpha_1, \ldots, \alpha_n |$ as outgoing state, and, in particular, the one-particle form factor.

Further constraints are also implied, and, for the sake of concreteness, we only discuss them for the outgoing vacuum $\langle +, \ldots, + |$. Since + should be present at $x = \pm \infty$ of the incoming state in the *j*th replica, only an even number of kinks has to be present for any replica. For example, two-particle states with two kinks inserted at different replicas have vanishing form factors. It is worth noting that this property is specific to the ferromagnetic phase of the Ising model, while the same mechanism does not arise in the paramagnetic one.

5.3.2 Two-particle form factors

The two-particle (or two-kink) form factors of $\mathcal{T}(x)$ are responsible for the leading corrections to the saturation value of the Rényi entropy. They are generally non-vanishing, as they are not ruled out by symmetry or topological constraints, and can be exactly determined via the bootstrap equations. We perform an analysis similar to the one in [20], where the paramagnetic phase was considered: many analogies arise due to the same scattering properties in the two phases, but there are also important conceptual differences stemming from the vacuum degeneracy. The two-kink form factors can be determined by their monodromy conditions and the residues at their kinematic poles. We shall first derive the monodromy equations, and then move to the equations for the kinematic poles.

Let us focus on the following two-kink form factor

$$F_{11}^{(n)}(\theta_1 - \theta_2) := {}^{\otimes n} \langle + |\mathcal{T}(0)| K_{+-}(\theta_1) K_{-+}(\theta_2), +, \dots, + \rangle,$$
(5.54)

which depends on the difference $\theta_1 - \theta_2$ only, due to relativistic invariance. In general, this is expected to be non-vanishing, since, as argued in Section 5.3.1.2, two kinks are presented in the same replica (the first). We give a pictorial representation of $F_{11}^{(n)}(\theta_1 - \theta_2)$ in Fig. 5.2. As a consequence of replica symmetry there is no explicit dependence on the replica j where the pair of kinks is inserted, and we have

$$F_{jj}^{(n)}(\theta_1 - \theta_2) := {}^{\otimes n} \langle + |\mathcal{T}(0)| \underbrace{+, \dots, +, K_{+-}(\theta_1)K_{-+}(\theta_2)}_{j}, +, \dots, + \rangle = F_{11}^{(n)}(\theta_1 - \theta_2), \tag{5.55}$$

with $j = 1, \ldots, n$.

Let us discuss the monodromy equations of $F_{11}^{(n)}(\theta)$. The rapidity shift $\theta_1 \to \theta_1 + 2\pi i$ in Eq. (5.54) has the effect of moving the corresponding kink from the first to the second replica. In addition, the topological sectors are mixed with each other, and we get

$$^{\otimes n} \langle + |\mathcal{T}(0)| K_{+-}(\theta_1 + 2\pi i) K_{-+}(\theta_2), +, \dots, + \rangle$$

= $\langle -, +, \dots, + |\mathcal{T}(0)| K_{-+}(\theta_2), K_{+-}(\theta_1), +, \dots, + \rangle.$ (5.56)

This means that the outgoing vacuum is modified under the above rapidity shift. This is not particularly surprising, since we already argued in Section 5.3.1.2 that the two-particle form factors of $\langle +, \ldots, + |$ with kinks in distinct replicas vanish. Similarly, for $j = 1, \ldots, n-1$, we have

$$^{\otimes n} \langle + | \mathcal{T}(0) | K_{+-}(\theta_1 + 2\pi i j) K_{-+}(\theta_2), +, \dots, + \rangle$$

= $\langle \underbrace{-, \dots, -}_{j}, +, \dots, + | \mathcal{T}(0) | \underbrace{K_{-+}(\theta_2), -, \dots, -, K_{+-}(\theta_1)}_{j+1}, +, \dots, + \rangle,$ (5.57)

and the kink moves from the first to the (j + 1)th replica under $\theta_1 \rightarrow \theta_1 + 2\pi i j$. Under the shift $\theta_1 \rightarrow \theta_1 + 2\pi i n$, the kink goes back to the first replica, but all the signs are exchanged, namely

$$^{\otimes n} \langle + | \mathcal{T}(0) | K_{+-}(\theta_1 + 2\pi i n) K_{-+}(\theta_2), +, \dots, + \rangle$$

= $^{\otimes n} \langle - | \mathcal{T}(0) | K_{-+}(\theta_2) K_{+-}(\theta_1), -, \dots, - \rangle.$ (5.58)

This expression can be further manipulated by using the \mathbb{Z}_2 neutrality of the twist fields together with

the (fermionic) scattering properties (5.48), yielding:

$$^{\otimes n} \langle + | \mathcal{T}(0) | K_{+-}(\theta_1 + 2\pi i n) K_{-+}(\theta_2), +, \dots, + \rangle$$

= $-^{\otimes n} \langle + | \mathcal{T}(0) | K_{+-}(\theta_1) K_{-+}(\theta_2), +, \dots, + \rangle,$ (5.59)

that is $F_{11}^{(n)}(\theta+2in\pi) = -F_{11}^{(n)}(\theta)$. The monodromy equations thus lead to an anti-periodicity condition on the fundamental two-kink form factor.

Following closely the arguments in [21], we argue that the form factor $F_{11}^{(n)}(\theta)$ has two kinematic poles at $\theta = i\pi$ and $\theta = 2\pi i n - i\pi$. Their residues are

$$\operatorname{Res}_{\theta=i\pi} F_{11}^{(n)}(\theta) = i\tau, \quad \operatorname{Res}_{\theta=(2n-1)i\pi} F_{11}^{(n)}(\theta) = -i\tau, \tag{5.60}$$

with τ the vacuum expectation value (5.53). Combining all these results, we conclude that the monodromy and the bootstrap equations of $F_{11}^{(n)}(\theta)$ are the same encountered in the paramagnetic phase [20]. Therefore, a solution⁷ to these equations can be easily provided:

$$F_{11}^{(n)}(\theta) = \frac{i\tau \cos\left(\frac{\pi}{2n}\right)\sinh\left(\frac{\theta}{2n}\right)}{n \sinh\left(\frac{\theta-i\pi}{2n}\right)\sinh\left(\frac{\theta+i\pi}{2n}\right)}.$$
(5.61)

5.3.3 Rényi entropy and its analytic continuation

With the previous results, we can finally present a form-factor expansion for the two-point function of the twist fields, valid in the limit $m\ell \gg 1$. This expansion is then used to obtain the Rényi entropies. Eventually, we analytically continue the replica index $n \to 1$ to get the entanglement entropy.

Starting from the replica state $|+\rangle^{\otimes n}$, the quantity we need to compute is

$$\operatorname{Tr}\left(\rho^{\otimes n}\mathcal{T}_{A}\right) = {}^{\otimes n}\langle +|\mathcal{T}(0)\mathcal{T}^{\dagger}(\ell)|+\rangle^{\otimes n}.$$
(5.62)

We insert a resolution of the identity between the branch-point twist fields and, at the two-particle approximation, we get

$$^{\otimes n}\langle +|\mathcal{T}(0)\mathcal{T}^{\dagger}(\ell)|+\rangle^{\otimes n} \simeq |\tau|^{2} + \sum_{j=1}^{n} \int_{\theta_{1}>\theta_{2}} \frac{\mathrm{d}\theta_{1}\,\mathrm{d}\theta_{2}}{(2\pi)^{2}} |F_{jj}^{(n)}(\theta_{1}-\theta_{2})|^{2} e^{-m\ell(\cosh\theta_{1}+\cosh\theta_{2})}.$$
(5.63)

In deriving the expression above, we used the fact that the only vacuum that can be connected to $\otimes^n \langle + |$ by $\mathcal{T}(0)$ is $|+\rangle^{\otimes n}$. Moreover, only pairs of kinks inserted in the same replica j give a non-vanishing

⁷It is well known that the bootstrap equations have multiple solutions, but the Δ -theorem [202] provides a severe additional constraint. Here, as in [20], the solution with the mildest growth for $|\theta| \rightarrow \infty$ is given. This one is expected to describe the twist field with the lowest scaling dimension, corresponding to a primary field of the associated CFT.

CHAPTER 5. ENTANGLEMENT ASYMMETRY IN THE ORDERED PHASE OF THE ISING FIELD THEORY

contribution in the expansion above.

By performing the usual change of variable $\theta = \theta_1 - \theta_2$, $\Theta = \frac{\theta_1 + \theta_2}{2}$ and after an integration over Θ and a sum over the copy index j, which trivially follows from $|F_{jj}^{(n)}(\theta)|^2 = |F_{11}^{(n)}(\theta)|^2$, the above equation reduces to

$$^{\otimes n}\langle +|\mathcal{T}(0)\mathcal{T}^{\dagger}(\ell)|+\rangle^{\otimes n} \simeq |\tau|^{2} + \frac{n}{4\pi^{2}} \int_{-\infty}^{+\infty} \mathrm{d}\theta \, K_{0}\left(2m\ell\cosh\left(\frac{\theta}{2}\right)\right) |F_{11}^{(n)}(\theta)|^{2}.$$
(5.64)

In the large volume limit $m\ell \to \infty$ the correlation function factorises and the square of the vacuum expectation value $|\tau|^2$ is recovered: the first non-trivial correction, which decays exponentially to zero as $\sim e^{-2m\ell}$, comes from the second term of equation (5.64) and is due to the pairs of kinks. This result should be compared to the (two-particle approximation of) the two-point function obtained in the paramagnetic phase [20]. The crucial difference is that in the paramagnetic phase form factors with single-particle excitations inserted in distinct replicas are allowed, which amounts to replacing

$$|F_{11}^{(n)}(\theta)|^2 \to \sum_{j=0}^{n-1} |F_{11}^{(n)}(-\theta + 2\pi i j)|^2$$
(5.65)

in equation (5.64). As we show below, this difference has important consequences in the replica limit $n \rightarrow 1$.

Now, we can finally express the Rényi entropy of an interval in the state $|+\rangle$ as

$$S_n = \frac{1}{1-n} \log^{\otimes n} \langle + |\mathcal{T}(0)\mathcal{T}^{\dagger}(\ell)| + \rangle^{\otimes n} + \mathcal{C}_n$$

$$\simeq \frac{1}{1-n} \log \left(1 + \frac{n}{4\pi^2} \int_{-\infty}^{+\infty} \mathrm{d}\theta \, K_0 \left(2m\ell \cosh\left(\frac{\theta}{2}\right) \right) \frac{|F_{11}^{(n)}(\theta)|^2}{|\tau|^2} \right) + \mathcal{C}_n, \tag{5.66}$$

where the constant C_n , which contains the non-universal twist field normalisation and the short-distance cut-off, will not play any role in the following. In the large $m\ell$ limit, the support of the integral in equation (5.66) is localised near $\theta \simeq 0$, and a saddle-point analysis reveals that the two-particle contribution decreases exponentially, scaling as $\sim e^{-2m\ell}$. While these considerations hold for integer values of $n \geq 2$, the analytical continuation $n \to 1$ is more subtle, as we explain below.

Indeed, we observe that the ratio $\frac{|F_{11}^{(n)}(\theta)|^2}{|\tau|^2}$ does not explicitly depend on the field normalisation, and it is fixed by bootstrap. Moreover, since for n = 1 the twist field becomes the identity operator [20], its two-particle form factors vanish. This suggests that under analytic continuation over n, $F_{11}^{(n)}(\theta)$ converges to zero uniformly in the distributional sense (for θ real) as $n \to 1$ (see [20]), a property that can be explicitly checked from the expression (5.61). Therefore, $\frac{|F_{11}^{(n)}(\theta)|^2}{|\tau|^2}$ behaves as $\mathcal{O}((n-1)^2)$ for (n-1) small and, accordingly, the two-particle contribution to the von Neumann entropy, obtained as the limit $n \to 1$ in equation (5.66), vanishes identically. This circumstance is quite unexpected and, to the best of our knowledge, it has not been previously observed. Indeed, a two-particle contribution is always present in the paramagnetic phase of a large class of theories with a single vacuum, as shown in [21].

Finally, we stress that our analysis does not imply that the entanglement entropy is exactly independent of ℓ . In fact, it is reasonable to expect a non-vanishing contribution from the four-kink form factors, thus yielding

$$S_1 \simeq \text{const.} + O(e^{-4m\ell}). \tag{5.67}$$

However, the investigation of higher-kink form factors is beyond the purpose of this work.

5.4 Form factors for the composite twist fields and entanglement asymmetry in the ferromagnetic phase of Ising QFT

In this section, we analyse the \mathbb{Z}_2 composite twist fields that arise in the computation of the entanglement asymmetry. These are the building blocks used to reconstruct the composite twist operators introduced in Section 5.2, and, as for the standard twist fields, the correspondence goes as

$$\mathcal{T}_{A}^{\{g_1,\dots,g_n\}} \sim \mathcal{T}^{\{g_1,\dots,g_n\}}(x), \quad A = (x,\,\infty),$$
(5.68)

$$\mathcal{T}_{A}^{\{g_{1},\dots,g_{n}\}} \sim \mathcal{T}^{\{g_{1},\dots,g_{n}\}}(0) \left(\mathcal{T}^{\{g_{1},\dots,g_{n}\}}\right)^{\dagger}(\ell), \quad A = (0,\ell),$$
(5.69)

and similarly for any union of disjoint intervals. Physically, $\mathcal{T}^{\{g_1,\ldots,g_n\}}(x)$ introduces a branch cut along (x,∞) which connects the *j*th and (j+1)th replicas via the additional insertion of an Aharonov–Bohm flux g_j . In particular, to compute the entanglement asymmetry, we only need those composite twist fields with vanishing net flux (satisfying $g_n \cdots g_1 = 1$). A fundamental observation pointed out in Section 5.2 is that such composite twist fields can be related to the standard one via global unitary transformations induced by the symmetry. In particular, given the form factors of $\mathcal{T}(x)$, the ones of $\mathcal{T}^{\{g_1,\ldots,g_n\}}(x)$ can be easily reconstructed once the action of the symmetry on the multi-particle states is known. Before entering the core of this section, we recall that \mathbb{Z}_2 composite twist fields with non-vanishing net flux have been characterised in the paramagnetic phase of the Ising model [4, 107] and used to compute the symmetry-resolved entanglement entropy: those fields are not related by global unitary transformations to $\mathcal{T}(x)$ and the discussion of this section does not apply to them.

Our analysis refers to the ferromagnetic phase of the Ising field theory, where the group is $G = \mathbb{Z}_2$ which corresponds to a spin-flip exchanging the two vacua. We parametrise the elements of \mathbb{Z}_2 as

$$\mathbb{Z}_2 = \{1, \mu\},\tag{5.70}$$
with $\mu^2 = 1$, and in the forthcoming discussion, with a slight abuse of notation, we do not distinguish the elements of \mathbb{Z}_2 from the corresponding unitary operators (i.e. we drop the "hat" notation that we adopted in Section 5.2). It is easy to show that the *n*-tuples $\{g_j\}$ with vanishing net flux are the ones for which an even number of elements μ is present. Since |G| = 2, there are 2^{n-1} of those *n*-tuples (see Eq. (5.32)), and, for the sake of completeness, we list them explicitly in a few cases

- n = 2: $\{g_1, g_2\}$ can take the values $\{1, 1\}$ and $\{\mu, \mu\}$.
- n = 3: $\{g_1, g_2, g_3\}$ can take the values $\{1, 1, 1\}, \{1, \mu, \mu\}$ and its two cyclic permutations.
- n = 4: $\{g_1, g_2, g_3, g_4\}$ can take the values $\{1, 1, 1, 1\}$, $\{\mu, \mu, \mu, \mu\}$, $\{1, \mu, 1, \mu\}$ and its cyclic permutation, $\{1, 1, \mu, \mu\}$ and its three cyclic permutations.

We first discuss the zero-particle form factors of the composite twist fields. We anticipate that $\mathcal{T}^{\{g_1,\ldots,g_n\}}(0)$ does not interpolate between $\otimes^n \langle + |$ and any other vacuum (except in the trivial case $\{g_j = 1\}$), a property which is crucial for the large-distance behavior of the entanglement asymmetry. Then, we move to the analysis of the two-particle form factors, providing exact analytical results. Finally, we collect these results together and give an expression for the Rényi entanglement asymmetry of a large interval.

5.4.1 Zero-particle form factors

We first recall the relation (5.25) and apply it to the composite twist fields, obtaining

$$(g'_1 \otimes \dots \otimes g'_n) \mathcal{T}(0) (g'_1 \otimes \dots \otimes g'_n)^{-1} = \mathcal{T}^{\{g_1, \dots, g_n\}}(0), \quad g_j = g'_{j+1}(g'_j)^{-1},$$
 (5.71)

which is valid for $g_n \cdots g_1 = 1$. As previously established, the only non-vanishing vacuum expectation values of $\mathcal{T}(0)$ (see Eq. (5.53)) are $^{\otimes n}\langle \pm | \mathcal{T}(0) | \pm \rangle^{\otimes n}$. Similarly, thanks to (5.71), the composite twist fields have only two non-vanishing zero-particle form factors: these are overlaps between the same vacuum, which however is in general different from $|\pm\rangle^{\otimes n}$, as we show below.

If $g'_j = 1$ or $g'_j = \mu$ for every j = 1, ..., n in Eq. (5.71), we have $\mathcal{T}^{\{g_1,...,g_n\}}(0) = \mathcal{T}(0)$. In all the other cases, namely if at least two elements of the *n*-tuple $\{g'_j\}$ are different, we get a non-trivial composite twist field $\mathcal{T}^{\{g_1,...,g_n\}}(x)$, i.e. at least one of the $g_j = \mu$. For these non-trivial cases, it holds

$$^{\otimes}\langle \pm |\mathcal{T}^{\{g_1,\dots,g_n\}}(0)|\pm\rangle^{\otimes n} = ^{\otimes n}\langle \pm | \left(g'_1\otimes\dots\otimes g'_n\right)\mathcal{T}(0) \left(g'_1\otimes\dots\otimes g'_n\right)^{-1} |\pm\rangle^{\otimes n} = 0, \tag{5.72}$$

since $(g'_1 \otimes \cdots \otimes g'_n)^{-1} |\pm\rangle^{\otimes n}$ is always different from $|\pm\rangle^{\otimes n}$. This is one of the main results we will need later.

Other vacua have nevertheless non-vanishing amplitudes. Even if the latter do not appear in the



Figure 5.3 Pictorial representation of the form factor $\langle +, -|\mathcal{T}^{\{\mu,\mu\}}(0)|+, -\rangle$. The blue dotted arrow in the top left figure shows that the second replica is reached from the first one when the branch cut, depicted by a dashed black line, is crossed clockwise. The red line represents the spin-flip, which exchanges + and -. $\mathcal{T}^{\{\mu,\mu\}}(0)$ can be equivalently obtained via global spin-flips acting on $\mathcal{T}(0)$, as shown in the top right figure. Since the global spin-flip lines commute with the Hamiltonian, they can be translated and positioned at infinity, so that they act on the ingoing/outgoing states. This equivalence allows us to establish the relation $\langle +, -|\mathcal{T}^{\{\mu,\mu\}}(0)|+, -\rangle = \langle +, +|\mathcal{T}(0)|+, +\rangle$.

computation of the entanglement asymmetry (because in equation (5.33) we take $\rho = |+\rangle\langle+|$), it is useful to analyse a simple case explicitly. We consider for n = 2 the twist field $\mathcal{T}^{\{\mu,\mu\}}(0)$. This is related to the standard twist field via the transformation (5.71), with $\{g'_1 = 1, g'_2 = \mu\}$. Therefore, from (5.71) we get

$$\langle \pm, \mp | \mathcal{T}^{\{\mu,\mu\}}(0) | \pm, \mp \rangle = \langle \pm, \pm | \mathcal{T}(0) | \pm, \pm \rangle, \tag{5.73}$$

and these are the only non-vanishing vacuum expectation values of $\mathcal{T}^{\{\mu,\mu\}}(x)$.

We can interpret this result as follows. If the boundary condition + is chosen in the first replica for both the outgoing and the incoming states then, due to the spin-flip induced by the field, the boundary condition – has to be present in the second replica. We represent this mechanism in Figure 5.3, which shows pictorially the relation between $\mathcal{T}(x)$ and $\mathcal{T}^{\{\mu,\mu\}}(x)$.

5.4.2 Two-particle form factors

The analysis for the two-particle form factors is slightly more involved. In principle, one can just use the form factors of Section 5.3.2, with vacua as outgoing states and pairs of kinks as incoming ones, use the relation (5.71) and systematically reconstruct every non-vanishing two-particle form factor. In practice, this procedure becomes rather cumbersome as the number of composite twist operators grows exponentially with n. For our purpose, it is more convenient to focus on the outgoing state



Figure 5.4 Two-particle form factor $^{\otimes 4}\langle +|\mathcal{T}^{\{\mu,1,\mu,1\}}(0)|+, K_{+-}(\theta_1), +, K_{+-}(\theta_2)\rangle$. Since a single kink K_{+-} is present in the second replica, a spin-flip, depicted as a red line, has to be inserted in the first replica to get a non-vanishing form factor. Similarly, the presence of a kink in the fourth replica implies the spin-flip in the third replica.

 $^{\otimes n}\langle +|$ and the incoming states

$$|+, \dots, K_{+-}(\theta_1), \dots, K_{+-}(\theta_2), \dots, +\rangle,$$
 (5.74)

with the kinks $K_{+-}(\theta_1)$, $K_{+-}(\theta_2)$ inserted in the replicas j and j' respectively and + in every other replica. We do so as these are the only contributions entering our computation of the entanglement asymmetry. Then, we investigate which *n*-tuples $\{g_j\}$ give rise to a non-vanishing form factor of $\mathcal{T}^{\{g_1,\ldots,g_n\}}(x)$ for the states above. We also explicitly assume $j \neq j'$ (say j < j'), since, for j = j', that is when the two kinks are in the same replica, only the trivial twist field $\mathcal{T}(0)$ can interpolate between this state and $\otimes^n \langle +|$, a case already discussed in Section 5.3.2. Let us now discuss the consequences of the presence of a single kink K_{+-} in the *j*th replica. Since the boundary conditions are \pm for $x = \mp \infty$, a spin-flip has to connect the (j - 1)th and the *j*th replica. This is the only way to match the + above the branch cut in the (j - 1)th replica and the – below the branch cut in the *j*th one. Vice versa, if no kinks are present in the *j*th replica, no spin-flips between the (j - 1)th and the *j*th replicas can be present. An example of the mechanism above is shown in Fig. 5.4. As a result, the only composite twist fields having non-vanishing form factors with two kinks in different replicas are precisely those with two spin-flips. Namely, the matrix element

$$^{\otimes n} \langle + | \mathcal{T}(0)^{\{1, \dots, \mu, \dots, \mu, \dots, 1\}} | +, \dots, K_{+-}(\theta_1), \dots, K_{+-}(\theta_2), \dots, + \rangle,$$
(5.75)

with μ inserted in the (j-1)th, (j'-1)th replicas and $K_{+-}(\theta_1)$, $K_{+-}(\theta_2)$ in the *j*th and *j*'th replica respectively, is non-vanishing. Furthermore, it is not difficult to show that the value of (5.75) is precisely $F_{11}^{(n)}(2\pi i(j'-j)-\theta_1+\theta_2)$, and we leave some details to Appendix 5.A.

5.4.3 Entanglement asymmetry

In this section we put together all the results obtained so far and we compute the entanglement asymmetry of an interval of length ℓ for the state $|+\rangle$. We focus on the limit $m\ell \gg 1$ keeping only the two-particle contribution. In the two-particle approximation, the only twist fields appearing in the computation are the standard twist field $\mathcal{T}(x)$ and the composite fields with two spin-flips $\mathcal{T}^{\{1,\dots,\mu,\dots,\mu,\dots,1\}}(x)$. The reason is that the other fields do not have non-vanishing two-particle form factors with $\otimes^n \langle +|$, and their contribution is therefore subleading.

For $A = (0, \ell)$ and $\rho = |+\rangle \langle +|$ we can write

$$\sum_{g_{1},\dots,g_{n-1}\in\{1,\mu\}} \operatorname{Tr}\left(\rho^{\otimes n}\mathcal{T}_{A}^{\{g_{1},\dots,g_{n-1},(g_{1},\dots,g_{n-1})^{-1}\}}\right)$$

$$=\sum_{g_{1},\dots,g_{n-1}\in\{1,\mu\}} {}^{\otimes n}\langle +|\mathcal{T}_{A}^{\{g_{1},\dots,g_{n-1},(g_{1},\dots,g_{n-1})^{-1}\}}|+\rangle^{\otimes n}$$

$$\simeq^{\otimes n}\langle +|\mathcal{T}(0)\mathcal{T}^{\dagger}(\ell)|+\rangle^{\otimes n} +\sum_{1\leq j< j'\leq n} {}^{\otimes n}\langle +|\mathcal{T}^{\{1,\dots,\mu,\dots,\mu,\dots,1\}}(0)\left(\mathcal{T}^{\{1,\dots,\mu,\dots,\mu,\dots,1\}}\right)^{\dagger}(\ell)|+\rangle^{\otimes n},$$
(5.76)

where the spin-flips μ are inserted at positions j and j'. In the large-volume limit $m\ell \to \infty$ the two-point function of the standard twist field converges to $|\tau|^2$ (the square modulus of the vacuum expectation value). In contrast, the one of any composite twist field goes to zero exponentially fast as $\sim e^{-2m\ell}$, because these fields have vanishing vacuum expectation value and the first non-trivial contribution comes from the two-particle form factors. Expanding in the basis of the multi-kink states, we arrive at

$$\sum_{g_1,\dots,g_{n-1}\in\{1,\mu\}} \frac{\otimes n \langle +|\mathcal{T}_A^{\{g_1,\dots,g_{n-1},(g_1,\dots,g_{n-1})^{-1}\}}|+\rangle^{\otimes n}}{\otimes n \langle +|\mathcal{T}_A|+\rangle^{\otimes n}}$$
(5.77)

$$\simeq 1 + \frac{n}{4\pi^2} \sum_{j=1}^{n-1} \int_{\mathbb{R}} \mathrm{d}\theta \, K_0\left(2m\ell \cosh\left(\frac{\theta}{2}\right)\right) \frac{|F_{11}^{(n)}(2\pi i j - \theta)|^2}{|\tau|^2}.$$
(5.78)

Finally, we employ the definition of Rényi asymmetry entanglement, and from (5.33) we get

$$\Delta S_n \simeq \log 2 - \frac{1}{n-1} \log \left(1 + \frac{n}{4\pi^2} \sum_{j=1}^{n-1} \int_{\mathbb{R}} \mathrm{d}\theta \, K_0 \left(2m\ell \cosh\left(\frac{\theta}{2}\right) \right) \frac{|F_{11}^{(n)}(2\pi i j - \theta)|^2}{|\tau|^2} \right), \tag{5.79}$$

an expression valid for any integer $n \geq 2$. Before discussing the analytic continuation over n, some comments are in order. First, the Rényi entanglement asymmetry is a universal quantity as it does not depend on the normalisation of the twist fields: this is manifest in the appearance of the two-particle form factor $F_{11}^{(n)}(\theta)$ through its ratio with the VEV only. Moreover, for large $m\ell$, the quantity ΔS_n approaches its asymptotic value log 2 from below, and this limit is compatible with the inequality

$$0 \le \Delta S_n \le \log |G|,\tag{5.80}$$

which we expect to have general validity, and we prove in some simple cases in Appendix 5.B. These conclusions appear to be quite general, as they do not depend on the explicit expression of the

two-particle form factors in Eq. (5.61): instead, they strongly rely on the fact that the VEV of composite twist fields vanish, a property which results from topological constraints only.

In order to obtain the entanglement asymmetry we need to take the limit $n \to 1^+$ of ΔS_n , which requires to analytically continue the sum

$$\sum_{j=1}^{n-1} |F_{11}^{(n)}(2i\pi j - \theta)|^2,$$
(5.81)

a task that has been accomplished in [20]. Following the same steps and employing the notation thereof, we define the function

$$f(\theta, n) = \frac{1}{|\tau|^2} \sum_{j=0}^{n-1} |F_{11}^{(n)}(2\pi i j - \theta)|^2,$$
(5.82)

and we denote with $\tilde{f}(\theta, n)$ its analytic continuation from $n \in \mathbb{N} \setminus \{1\}$ to $n \in [1, \infty)$. In [20] it was shown that

$$\lim_{n \to 1^+} \tilde{f}(\theta, n) = 0, \quad \lim_{n \to 1^+} \frac{\partial f(\theta, n)}{\partial n} = \frac{\pi^2}{2} \delta(\theta), \tag{5.83}$$

a relation valid in the distributional sense over real values of θ . Here, we are interested in the sum

$$g(\theta, n) = \frac{1}{|\tau|^2} \sum_{j=1}^{n-1} |F_{11}^{(n)}(2\pi i j - \theta)|^2 = f(\theta, n) - \frac{|F_{11}^{(n)}(\theta)|^2}{|\tau|^2},$$
(5.84)

namely, the term with j = 0 in Eq. (5.82) is not present in our calculation. Since, as we showed in Section 5.3.3, $|F_{11}^{(n)}(\theta)| \sim O((n-1)^2)$, the same properties established above for $\tilde{f}(\theta, n)$ are valid for $\tilde{g}(\theta, n)$, including its analytic continuation. In particular, we have

$$\lim_{n \to 1^+} \tilde{g}(\theta, n) = 0, \quad \lim_{n \to 1^+} \frac{\partial \tilde{g}(\theta, n)}{\partial n} = \frac{\pi^2}{2} \delta(\theta).$$
(5.85)

Inserting this result in (5.79) we finally get

$$\Delta S_1 = \log 2 - \frac{1}{4\pi^2} \lim_{n \to 1^+} \int_{\mathbb{R}} d\theta \, K_0 \left(2m\ell \cosh\left(\frac{\theta}{2}\right) \right) \frac{\partial}{\partial n} (n\tilde{g}(\theta, n)) + O(e^{-4m\ell})$$
$$= \log 2 - \frac{K_0(2m\ell)}{8} + \mathcal{O}(e^{-4m\ell}), \tag{5.86}$$

that is the main result of this section. As a concluding remark, we observe that from the large-z asymptotics of the modified Bessel function [91], that is $K_0(z) \sim \sqrt{\frac{\pi}{2z}}e^{-z}$, one immediately sees that the two-particle contribution to the entanglement asymmetry is exponentially suppressed in the size of the region.

5.5 Further generalisations

In this section, we propose a general conjecture regarding the entanglement asymmetry for any finite group G. We believe that our finding $\Delta S_n \simeq \log 2$ is not specific to the integrability features of the Ising field theory, and it mostly relies on the symmetry-breaking pattern $\mathbb{Z}_2 \to \{1\}$ of this model. We also provide a paradigmatic case in which our conjecture can be proven with elementary techniques, that is, the case of zero-entanglement states.

Let us first recall that a state ρ is symmetric under $g \in G$ if the following equality holds

$$\rho = g\rho g^{-1}.\tag{5.87}$$

Clearly, the set of $g \in G$ leaving ρ invariant is a group itself. Therefore, we define

$$H := \{ h \in G | \rho = h\rho h^{-1} \} \subset G, \tag{5.88}$$

and we say that the symmetry-breaking pattern

$$G \to H,$$
 (5.89)

arises for the state ρ . We conjecture that the Rényi asymmetry of a large subsystem A in the state ρ is

$$\Delta S_n \simeq \log \frac{|G|}{|H|}.\tag{5.90}$$

In particular, for the Ising model $G = \mathbb{Z}_2$, $H = \{1\}$ (the trivial group), and our conjecture is compatible with the main result (5.12) of this paper. In general, |G|/|H| is precisely the number of ground states (vacua) in the ordered phase of a theory. For example, for the *q*-state Potts model [219], *q* distinct vacua are present and our conjecture (5.90) gives $\Delta S_n \simeq \log q$. We remark that equation (5.90) refers to finite groups only, and it does not apply to the case of U(1) that has been previously studied in Ref. [29] (see also Appendix 5.B and [210]).

While we do not have a rigorous general proof of (5.90), we think that a few physical hypotheses should be sufficient to prove it:

- Homogeneity: a finite region A has to encode the global symmetries of the total system, that is the case for homogeneous systems.
- Short-range correlations: we require that correlations among points of the region A decay fast enough. This is the case for ground states, and equilibrium states of short-range Hamiltonians. We believe that this hypothesis is sufficient to prove the existence of a large-volume limit for the

entanglement asymmetry.

Before discussing an explicit case where the conjecture (5.90) can be proven, we would like to point out the intuition behind it. Let us first express, for $n \ge 2$

$$\frac{\text{Tr}(\tilde{\rho}_{A}^{n})}{\text{Tr}(\rho_{A}^{n})} = \frac{1}{|G|^{n-1}} \sum_{g_{1},\dots,g_{n-1}\in G} \frac{\text{Tr}(\rho_{A}g_{1}\rho_{A}\dots g_{n-1}\rho_{A}(g_{1}\dots g_{n-1})^{-1})}{\text{Tr}(\rho_{A}^{n})},$$
(5.91)

where both the index A and the hat have been dropped from the operators for notational convenience. Whenever $g_j \in H$ in the sum above, that is

$$\rho g_j = g_j \rho, \quad j = 1, \dots, n-1,$$
(5.92)

one can commute the elements inside the trace in the numerator, obtaining

$$\frac{\text{Tr}(\rho_A g_1 \rho_A \dots g_{n-1} \rho_A (g_1 \dots g_{n-1})^{-1})}{\text{Tr}(\rho_A^n)} = 1, \quad g_j \in H.$$
(5.93)

There are precisely $|H|^{n-1}$ ways one can choose such (n-1)-tuples $\{g_1, \ldots, g_{n-1}\}$ such that every $g_j \in H$. The other terms in the sum (5.91) are expected to vanish in the large volume limit. Indeed, if $[\rho_A, g_j] \neq 0$ the eigenspaces of g_j and ρ_A are distinct: in the large volume limit, we expect that the eigenspaces above are at "generic positions" with respect to each other at large size, leading to a fast decay of the traces in which ρ_A and g_j are inserted consecutively: a related mechanism is known in the context of *free independence* between random matrices, and we refer the interested reader to [221] for a review. The considerations above lead to (5.90) after simple manipulations.

5.5.1 Zero-entanglement states

For zero-entanglement states, the conjecture (5.90) can be proven for any finite group G, as we show below. While this example may not be a realistic description of physical states with short-range correlations, it correctly captures the important features we mentioned above.

Let us consider the pure state

$$\rho = |0\dots0\rangle_A \otimes |0\dots0\rangle_{\bar{A}},\tag{5.94}$$

in which $|0\rangle$ is an on-site configuration belonging to a finite-dimensional Hilbert space. The reduced density matrix of the state above is

$$\rho_A = |0\dots0\rangle_A {}_A \langle 0\dots0|, \qquad (5.95)$$

and, since ρ_A is pure, no entanglement between A and \overline{A} is present. In particular, this means $\text{Tr}(\rho_A^n) =$ 1. Let us then assume that a finite group G acts unitarily as a global symmetry of the system. It is not difficult to show that the subgroup H that leaves the state invariant is:

$$H = \{ h \in G \text{ s.t } |\langle 0|h|0\rangle| = 1 \},$$
(5.96)

namely $|0\rangle$ and $h|0\rangle$ are proportional through a phase. In contrast, whenever $g \notin H$, one has

$$|\langle 0|g|0\rangle| < 1. \tag{5.97}$$

By considering a generic *n*-tuple $\{g_j \in G\}_{j=1,\dots,n}$, we can compute

$$\operatorname{Tr}\left(\rho_{A}g_{1}\dots\rho_{A}g_{n}\right)=\prod_{j=1}^{n}{}_{A}\langle 0\dots0|g_{j}|0\dots0\rangle_{A}=\prod_{j=1}^{n}\langle 0|g_{j}|0\rangle^{|A|},$$
(5.98)

with |A| the number of sites of A. As long as at least one element satisfies $g_j \notin H$, one has

$$\lim_{|A| \to \infty} |\langle 0|g_j|0\rangle|^{|A|} = 0, \tag{5.99}$$

which trivially implies

$$\lim_{|A|\to\infty} \operatorname{Tr}\left(\rho_A g_1 \dots \rho_A g_n\right) = 0.$$
(5.100)

On the other hand, for *n*-tuples of elements of H, equation (5.93) holds. Putting together these results with (5.91), one finally gets (5.90) for the case of a large subsystem A.

5.6 Concluding remarks

In this paper we generalised the notion of Rényi entanglement asymmetry, first proposed in [29], to include the action of an arbitrary finite group G, and we characterised its value in the symmetry-broken ground state of the Ising field theory. In particular, we employed the replica trick to describe the quantities of interest as expectation values of composite twist operators, and we provided analytical expressions of their form factors using integrable bootstrap.

In addition, we proposed a general conjecture (5.90), which we expect to hold for a large class of quantum states and for any finite group G. Remarkably, if our conjecture is correct, the entanglement asymmetry of a large but finite region becomes a useful tool to "count" the number of ground states in a spontaneously symmetry-broken phase via a local probe.

Many interesting directions are left to explore. It would be interesting to investigate other integrable QFTs using the approach we developed in this paper. For instance, the q-state Potts model in its ferromagnetic phase [219] stands out as the simplest generalisation of the Ising model. Moreover, one could analyse massless flows [222] in which the symmetry is broken explicitly along

the renormalisation group flow, and distinct symmetries are present in the IR and UV fixed points. It may also be worth considering field theories where the symmetry breaking arises due to the presence of boundary conditions or impurities (see e.g. [223, 224]) and translational invariance is absent.

Furthermore, the relation between the paramagnetic and the ferromagnetic phase of the Ising model is, from the point of view of entanglement content, quite puzzling. As we have shown, there are formal analogies in the form factor bootstrap of the twist fields, but the relation is subtle and, in general, the entropies are different in the two phases (see Appendix 5.C). It is well-known that the Kramers-Wannier duality [225], which is nowadays understood in the language of non-invertible defects [226], relates the ordered and the disordered phase non-locally: this duality might be a good candidate to explain the relationship between the twist fields in the two phases. In particular, it would be interesting to understand what is the "dual" of the entanglement of a region under the Kramers-Wannier duality.

We also consider the study of low-energy states in the ferromagnetic phase of one-dimensional quantum systems (i.e. domain walls interpolating between distinct vacua) to be promising. Indeed, it is not obvious a priori whether the results found in [114–116, 144] for quasi-particles should also hold for these topological excitations typical of the ordered phase.

Finally, a general, comprehensive proof of conjecture (5.90) is still missing. Recently, its validity has been rigorously shown for matrix product states (see e.g. [227]) in [210], where the authors also proposed a generalisation of equation (5.90) for compact Lie groups. However, many physically relevant states, such as (generalised) Gibbs ensembles or regularised boundary states of CFT [228] are yet to be studied. Moreover, it would be interesting to understand whether Eq. (5.90) is also valid for the steady states arising in the long-time dynamics, which is the main motivation behind the original formulation of the entanglement asymmetry [29].

5.A Two-particle form factors of composite twist fields

In this appendix we show in some simple cases the validity of the relation

$$^{\otimes n}\langle +|\mathcal{T}(0)^{\{1,\dots,\mu,\dots,\mu,\dots,1\}}|+,\dots,K_{+-}(\theta_1),\dots,K_{+-}(\theta_2),\dots,+\rangle = F_{11}^{(n)}(2\pi i(j'-j)-\theta_1+\theta_2), \quad (5.101)$$

which was presented at the end of Section 5.4.2. A proof of the equation above for generic values of n, j, j' is easy to obtain by making use of the monodromy equation, global spin-flip invariance and the algebra (5.48). However, the general mechanism is better highlighted by looking at some explicit examples.

Let us consider n = 2, 3, 4 and the following fields (see Section 5.2)

$$\mathcal{T}^{\{\mu,\mu\}}(x) = (1 \otimes \mu)\mathcal{T}(x)(1 \otimes \mu) \quad \text{for } n = 2,$$
(5.102)

$$\mathcal{T}^{\{1,\mu,\mu\}}(x) = (1 \otimes 1 \otimes \mu)\mathcal{T}(x)(1 \otimes 1 \otimes \mu) \quad \text{for } n = 3,$$
(5.103)

$$\mathcal{T}^{\{\mu,1,\mu,1\}}(x) = (1 \otimes \mu \otimes \mu \otimes 1)\mathcal{T}(x)(1 \otimes \mu \otimes \mu \otimes 1) \quad \text{for } n = 4.$$
(5.104)

For n = 2 we have

$$\langle +, + | \mathcal{T}^{\{\mu,\mu\}}(0) | K_{+-}(\theta_1), K_{+-}(\theta_2) \rangle$$

$$= \langle +, + | (1 \otimes \mu) \mathcal{T}(0)(1 \otimes \mu) | K_{+-}(\theta_1), K_{+-}(\theta_2) \rangle$$

$$= \langle +, - | \mathcal{T}(0) | K_{+-}(\theta_1), K_{-+}(\theta_2) \rangle$$

$$= \langle -, - | \mathcal{T}(0) | K_{-+}(\theta_2 + 2\pi i) K_{+-}(\theta_1), - \rangle$$

$$= \langle +, + | \mathcal{T}(0) | K_{+-}(\theta_2 + 2\pi i) K_{-+}(\theta_1), + \rangle = F_{11}^{(2)} (2\pi i + \theta_2 - \theta_1),$$
(5.105)

where we used the monodromy equation for $\mathcal{T}(x)$ (discussed in Section 5.3) in going from the third to the fourth line and applied an overall spin-flip transformation to each replica to obtain the last line. Similar calculations are shown below for n = 3:

$$\langle +, +, + | \mathcal{T}^{\{1,\mu,\mu\}}(0) | K_{+-}(\theta_2), +, K_{+-}(\theta_1) \rangle$$

$$= \langle +, +, + | (1 \otimes 1 \otimes \mu) \mathcal{T}(0) (1 \otimes 1 \otimes \mu) | K_{+-}(\theta_2), +, K_{+-}(\theta_1) \rangle$$

$$= \langle +, +, - | \mathcal{T}(0) | K_{+-}(\theta_2), +, K_{-+}(\theta_1) \rangle$$

$$= \langle -, -, - | \mathcal{T}(0) | K_{-+}(\theta_1 + 4\pi i) K_{+-}(\theta_1), -, - \rangle$$

$$= \langle +, +, + | \mathcal{T}(0) | K_{+-}(\theta_1 + 4\pi i) K_{-+}(\theta_2), +, + \rangle$$

$$= F_{11}^{(3)} (4\pi i + \theta_2 - \theta) = F_{11}^{(3)} (2\pi i - \theta_1 + \theta_2).$$
(5.106)

Finally, for n = 4 one has

$$\langle +, +, +, + | \mathcal{T}^{\{\mu, 1, \mu, 1\}}(0) | +, K_{+-}(\theta_1), +, K_{+-}(\theta_2) \rangle$$

$$= \langle +, +, +, + | (1 \otimes \mu \otimes \mu \otimes 1) \mathcal{T}(0) (1 \otimes \mu \otimes \mu \otimes 1) | +, K_{+-}(\theta_1), +, K_{+-}(\theta_2) \rangle$$

$$= \langle +, -, -, + | \mathcal{T}(0) | +, K_{-+}(\theta_1), -, K_{+-}(\theta_2) \rangle$$

$$= \langle +, +, +, + | \mathcal{T}(0) | +, K_{+-}(\theta_2 + 4\pi i) K_{-+}(\theta_1), +, + \rangle$$

$$= F_{11}^{(4)} (4\pi i - \theta_1 + \theta_2).$$
(5.107)

5.B Useful inequalities

In this appendix, we provide some upper and lower bounds for the Rényi entanglement asymmetry of finite groups. In particular, we focus on n = 2, leaving some considerations for the other values of n at the end of this section. The main result, proved below, is

$$0 \le \Delta S_2 \le \log |G|,\tag{5.108}$$

with $\Delta S_2 = 0$ iff $\rho_A = \tilde{\rho}_A$.

We first express ΔS_2 , defined in (5.8), as (see also equation (5.33))

$$\Delta S_2 = \log|G| - \log\left(\sum_{g \in G} \frac{\operatorname{Tr}\left(\rho_A g \rho_A g^{-1}\right)}{\operatorname{Tr}\left(\rho_A^2\right)}\right).$$
(5.109)

Therefore, to bound ΔS_2 it is sufficient to analyse the possible values taken by

$$\frac{\operatorname{Tr}\left(\rho_A g \rho_A g^{-1}\right)}{\operatorname{Tr}\left(\rho_A^2\right)},\tag{5.110}$$

as a function of $g \in G$. We observe that, since ρ_A and $g\rho_A g^{-1}$ are both positive semi-definite, one can easily show that⁸

$$\frac{\operatorname{Tr}\left(\rho_A g \rho_A g^{-1}\right)}{\operatorname{Tr}\left(\rho_A^2\right)} \ge 0, \quad \forall g \in G.$$
(5.111)

Furthermore, because the ratio takes the value 1 for g = 1, we can write

$$\log\left(\sum_{g\in G} \frac{\operatorname{Tr}\left(\rho_A g \rho_A g^{-1}\right)}{\operatorname{Tr}\left(\rho_A^2\right)}\right) = \log\left(1 + \sum_{g\neq 1} \frac{\operatorname{Tr}\left(\rho_A g \rho_A g^{-1}\right)}{\operatorname{Tr}\left(\rho_A^2\right)}\right) \ge 0,\tag{5.112}$$

which, from equation (5.109), implies $\Delta S_2 \leq \log |G|$.

To prove $\Delta S_2 \ge 0$, we apply the von Neumann's trace inequality [229] to ρ_A and $g\rho_A g^{-1}$, obtaining

$$\frac{\operatorname{Tr}\left(\rho_A g \rho_A g^{-1}\right)}{\operatorname{Tr}\left(\rho_A^2\right)} \le 1,\tag{5.113}$$

and the bound is saturated iff ρ_A and $g\rho_A g^{-1}$ share the same eigenvectors (which means they have to be equal, as their eigenvalues are always the same). Inserting the result above in equation (5.109), we obtain

$$\Delta S_2 \ge 0, \tag{5.114}$$

where the equality holds iff $\rho_A = g\rho_A g^{-1}$ for any $g \in G$. Therefore, from the definition (5.5), we have

⁸Given A, B positive semi-definite matrices, one has $\operatorname{Tr}(AB) = \operatorname{Tr}\left((\sqrt{A}\sqrt{B})(\sqrt{A}\sqrt{B})^{\dagger}\right) \geq 0.$

 $\Delta S_2 = 0$ iff $\rho_A = \tilde{\rho}_A$, which proves the main result (5.108) of this appendix.

It is worth considering whether comparable strategies can be used to establish the inequality $0 \le \Delta S_n \le \log |G|$ for other values of n. The first issue we encounter is that

$$\frac{\operatorname{Tr}\left(\rho_A \cdot g_1 \rho_A g_1^{-1} \cdot g_2 \rho_A g_2^{-1} \dots g_{n-1} \rho_A g_{n-1}^{-1}\right)}{\operatorname{Tr}\left(\rho_A^n\right)},\tag{5.115}$$

is not necessarily real for $n \geq 3$. For example, the matrices

$$M_1 = \begin{pmatrix} 1 & 0 \\ 0 & 0 \end{pmatrix}, \quad M_2 = \frac{1}{2} \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix}, \quad M_3 = \frac{1}{2} \begin{pmatrix} 1 & i \\ -i & 1 \end{pmatrix}, \quad (5.116)$$

are positive semi-definite, having 0 and 1 as eigenvalues, but $\operatorname{Tr}(M_1M_2M_3) = \frac{1+i}{4} \notin \mathbb{R}$. This implies that the previous technique, employed for n = 2 to show $\Delta S_2 \leq \log |G|$, does not apply directly to $n \geq 3$.

On the other hand, we can provide a simple proof of $\Delta S_n \geq 0$ valid for any, possibly non-integer, $n \geq 1$. The key idea is that the symmetrised state $\tilde{\rho}_A$ is generically more mixed (less pure) than ρ_A , and, therefore, it has more entropy. More precisely, one can show from (5.5) that the superoperator Φ , defined as

$$\Phi(\rho_A) := \tilde{\rho}_A,\tag{5.117}$$

is completely positive and trace-preserving (CPT). Then, we use the monotonicity of the sandwiched Rényi divergence [230], valid for CPT maps, and we apply it directly to ρ_A and the infinite-temperature state $\frac{\mathbb{1}_A}{\operatorname{Tr}(\mathbb{1}_A)}$. As an immediate consequence, we get

$$\Delta S_n \ge 0, \quad n \ge 1. \tag{5.118}$$

However, with this approach it is not possible to claim that, for a generic value of n, $\Delta S_n = 0$ implies $\rho_A = \tilde{\rho}_A$.

As a last remark, we point out that for a compact Lie group the number of elements is infinite and the Rényi entanglement asymmetry does not have an upper bound. For instance, in a simple example analysed in [29], the U(1) asymmetry grows logarithmically in the subsystem size. A detailed exploration of the entanglement asymmetry for compact Lie groups can be found in [210].

5.C Exact results for the Rényi entropies via corner transfer matrix and Kramers-Wannier duality

In this appendix, we review some exact results available for the entanglement of the transverse-field Ising chain, following closely [231]. Let us consider the quantum one-dimensional Ising model, described by the Hamiltonian

$$H = -\sum_{j} \left(\sigma_j^x \sigma_{j+1}^x + h \sigma_j^z \right), \qquad (5.119)$$

with σ_j^x, σ_j^z the Pauli matrices at position j. This model displays a quantum critical point at h = 1 separating a ferromagnetic phase (h < 1) from a paramagnetic one (h > 1). It is well known that Kramers-Wannier duality [225] relates the spectra of the Hamiltonian at h and h^{-1} , and, more in general, local and semi-local observables are mapped into each other by this duality. The same mechanism is observed in the underlying Ising field theory, where one can relate the paramagnetic and the ferromagnetic phase at the same value of the mass m. However, as shown in [231], the entropies at dual points differ explicitly. In particular, the entanglement of the half-chain has been characterised analytically in the infinite-volume limit via the Corner Transfer Matrix (CTM) method [232]. Here, we only report and discuss the final result found in [231].

Let us define the parameters

$$k := \min(h, h^{-1}), \quad \epsilon = \pi \frac{K(\sqrt{1-k^2})}{K(k)},$$
(5.120)

with K(z) the complete elliptic integral of the first kind [91]. Then, given

$$\epsilon_j := \begin{cases} (2j+1)\epsilon & h > 1, \\ \\ 2j\epsilon & h < 1, \end{cases}$$
(5.121)

for $j = 0, 1, \ldots$ one can express the *n*th Rényi entropy of the half chain at field *h* as

$$S_n(h) = \frac{1}{1-n} \sum_{j=0}^{\infty} \log \frac{1+e^{-\epsilon_j n}}{(1+e^{-\epsilon_j})^n}.$$
(5.122)

Equation (5.120) clearly shows that the value of the parameter ϵ is the same at the dual points h, h^{-1} , even though the "single-particle eigenvalues" ϵ_j are different.

We emphasise that for h < 1, corresponding to the ferromagnetic phase, the ground state is degenerate and one should be careful. For instance, the result (5.122) refers to the symmetric ground state (often dubbed GHZ state), which in our notation corresponds to

$$|\text{GHZ}\rangle = \frac{1}{\sqrt{2}} \left(|+\rangle + |-\rangle\right). \tag{5.123}$$

To obtain the entropies of the symmetry-broken ground state $|\pm\rangle$ one has to explicitly remove the zero-mode j = 0 from the sum in (5.122), as explained in [231], thus getting

$$S_n(h<1) = \frac{1}{1-n} \sum_{j=1}^{\infty} \log \frac{1+e^{-\epsilon_j n}}{(1+e^{-\epsilon_j})^n} = \frac{1}{1-n} \sum_{j=0}^{\infty} \log \frac{1+e^{-\epsilon_j n}}{(1+e^{-\epsilon_j})^n} - \log 2,$$
(5.124)

which means that the Rényi entropy of the doubly degenerate state $|\pm\rangle$ is smaller than that of the GHZ state (5.122), as one would intuitively expect. Remarkably, in the limit $h \to 1$ one has $\epsilon \to 0$ and, accordingly, the Rényi entropy diverges. The origin of this divergence is universal, and it is traced back to the properties of the underlying CFT (see [231] for further details): in particular, the divergence is the same if the critical magnetic field h = 1 is approached from above or below.

Discrepancies between dual points can nevertheless be spotted if one takes into account also the finite terms arising in (5.122) when taking the limit $h \rightarrow 1$. In particular, using the results of [231] one can show that the entropy difference between the paramagnetic phase and the ferromagnetic phase (in the GHZ state) satisfies

$$\lim_{h \to 1^+} \left[S_n(h) - S_n(h^{-1}) \right] = -\frac{\log(2)}{2}.$$
(5.125)

The origin of the constant $-\frac{\log(2)}{2}$ is much more subtle than the $\log(2)$ term in (5.124): indeed, the former appears only close to the critical point, and a full explanation based on the formal analogy between (5.122) and the thermal entropy of a CFT has been provided in [231] (see also [63] for a related discussion).

Below, we interpret these results in the language of the Ising field theory. Let us focus on a given value of the mass m and let $|0\rangle$ be the paramagnetic ground state, $|\pm\rangle$ the symmetry-broken ferromagnetic ground state, and $|\text{GHZ}\rangle$ the state defined by (5.123). To compute the Rényi entropy of a given state in the half-infinite chain, say $A = (0, \infty)$, we have to evaluate the twist field $\mathcal{T}(0)$ over the corresponding replica state. In particular, the difference of Rényi entropy between $|\text{GHZ}\rangle$ and $|+\rangle$ would be just

$$\frac{1}{1-n}\log\frac{\otimes n\langle \mathrm{GHZ}|\mathcal{T}(0)|\mathrm{GHZ}\rangle^{\otimes n}}{\otimes n\langle +|\mathcal{T}(0)|+\rangle^{\otimes n}} = \log 2.$$
(5.126)

Here, to obtain (5.126) we only used the definition (5.123) and the fact that the only non-vanishing VEVs of $\mathcal{T}(0)$ over replica vacua are $\otimes^n \langle + |\mathcal{T}(0)| + \rangle^{\otimes n} = \otimes^n \langle - |\mathcal{T}(0)| - \rangle^{\otimes n}$ (see Section 5.3). This result is compatible with (5.124), and reasonably a similar argument can be employed even beyond the field theoretic regime, when |h - 1| is finite.

Instead, a different mechanism arises if we try to compare $|+\rangle$ and $|0\rangle$. We first notice that the ratio

$$\frac{{}^{\otimes n}\langle +|\mathcal{T}(0)|+\rangle^{\otimes n}}{{}^{\otimes n}\langle 0|\mathcal{T}(0)|0\rangle^{\otimes n}},\tag{5.127}$$

is dimensionless, i.e. it is a universal number which does not depend on m (the only mass scale). Moreover, according to the lattice result (5.125) valid near the critical regime where the field theory is predictive, we expect

$$\frac{{}^{\otimes n}\langle +|\mathcal{T}(0)|+\rangle^{\otimes n}}{{}^{\otimes n}\langle 0|\mathcal{T}(0)|0\rangle^{\otimes n}} = g^{n-1}, \quad g = \sqrt{2}, \tag{5.128}$$

so that the difference of entropy between $|+\rangle$ and $|0\rangle$ would be just⁹

$$\frac{1}{1-n}\log\frac{\otimes n\langle +|\mathcal{T}(0)|+\rangle \otimes n}{\otimes n\langle 0|\mathcal{T}(0)|0\rangle \otimes n} = -\frac{\log 2}{2}.$$
(5.129)

We believe that (5.128) can be proven, once the normalisation of the twist field is fixed, via a form factor approach similar to that of [63]. We also conjecture that the value of the constant g in Eq. (5.128)comes precisely from Affleck-Ludwig boundary entropy [233]: however, it is not clear to us whether an explicit relation between the off-critical paramagnetic/ferromagnetic phases and a boundary CFT can be provided. We only point out that if (5.128) is correct, then it cannot be a trivial consequence of the Kramers-Wannier duality, which holds exactly on the lattice, as the relation (5.125) is only valid close to the critical point.

⁹Equation (5.125) compares the GHZ state and the paramagnetic ground state. If we take $|+\rangle$ instead of the GHZ state, we get $\lim_{h\to 1^+} S_n(h) - S_n(h^{-1}) = +\frac{\log(2)}{2}$.

CHAPTER

SIX

OUTLOOK

The results we presented in this thesis contribute to expand the knowledge in the field of symmetry-resolved entanglement measures in quantum many-body systems, an area which has been the object of intense activity in the past five years. One of the main theoretical tools we employed throughout this work is that of composite twist operators: these are operators that are defined in a very simple way by means of their algebraic relations with the local observables of a theory in a generic *D*-dimensional QFT. In the context of charged zero-density excited states of a QFT, by making some simple approximations, composite twist operators allow us to reproduce, at leading order in the system size, the results which can be obtained by form factor calculations and generalise those to higher-dimensional theories. Furthermore, crucially, we used twist operators to derive a new, general way of computing entanglement asymmetry in a symmetry-breaking ground state of a QFT, paving the way for a deeper investigation of entanglement properties in the ordered phase of a field theory.

Since the notion of symmetry resolution of entanglement was introduced, different techniques have been employed to establish exact results as well as asymptotic behaviours of the symmetry-resolved bipartite and multipartite entanglement measures in one-dimensional systems. These include, most notably, spin chains, conformal field theories, integrable quantum field theories, and quantum cellular automata. The most prominent result that has emerged from these extensive investigations is the equipartition of symmetry-resolved entropy at leading order in the system size, that is, the property that different charge sectors equally contribute to the total entanglement. Our main contribution to the field of symmetry-resolved entanglement consists in obtaining exact and universal leading-order results for the U(1) symmetry-resolved entanglement of zero-density excited states in massive QFTs. Our results exclusively rely on the locality of the excitations, and bridge a significant gap in the existing literature. Next-to-leading-order contributions to the entanglement will reveal how the latter depends on the microscopic details of the theory; however, a perturbative study of the large-volume limit of entanglement measures generally requires the knowledge of finite-volume form factors, and the development of a form factor program for non-local fields at finite volume is a challenging and -so far- virtually untackled task.

Employing a notion of entanglement asymmetry to investigate the properties of symmetry-broken phases in one-dimensional systems is an idea that made its appearence in the community of quantum many-body physics very recently. Prior to our work, only few papers had appeared, primarily focussing on the evolution of U(1) entanglement asymmetry after a quantum quench. In our work, we looked at the ferromagnetic phase of the Ising field theory to propose an algebraic approach for the computation of entanglement asymmetry in field theories, based on composite twist operators. Furthermore, we conjectured a formula for the entanglement asymmetry of a state displaying partial symmetry breaking which, to the best of our knowledge, was verified in any system studied to date.

We now highlight some research directions that emerge from our work. A first topic that deserves further investigation is that of entanglement properties of localised quasi-particle excitations. Indeed, the universality of entanglement entropy (total and symmetry-resolved) of excited states of gapped theories poses a question: how do we distinguish between states at different energy, from the point of view of the entanglement content? A possible way to tackle the problem is by looking at relative entropy as a measure of distinguishability among excited states with the same energy and same entanglement entropy. Moreover, form factor expansions allows us to obtain finite-volume corrections to the correlation functions, and beyond leading order the energy of the excitations (as well as the integrability features of the theory) becomes relevant.

On the other hand, the entanglement entropy of descendant states of a CFT has been well studied in the past decade. It is known that the low-lying and the high-lying states in the spectrum have entropies that depend on the details of the system in functionally different ways. The high-energy states reproduce, in certain limits, the universal result found in gapped theories. How does this crossover between the CFT and massive QFT results happens physically? Specific limits can be taken but a general explanation is missing. Holography may be a key, as high-energy CFT states may be obtained by looking at (2+1)d in a small coupling regime.

Two physical questions of the utmost relevance can be approached using integrability techniques, namely: the interplay between symmetries and boundary effects, and the dynamics of entanglement in QFT. Indeed, on the one hand the twist field and twist operator techniques we developed in [1, 2, 5] pave the way for a study of correlation functions and symmetry-resolved entanglement measures in theories with boundaries, thus addressing the question of whether -or to which extent- entropy equipartition is spoiled by boundary effects. On the other hand, the very same techniques allow us to investigate what happens to entanglement-related physical observable of a QFT after a quench in the mass or in other parameters of the theory.

Finally, the entanglement properties specific to the ordered phase of a many-body system are particularly interesting. In the recent work [8], we observed that the entanglement content of kink-like excitations in the ferromagnetic phase of the Ising model is not captured by the universal results for the entanglement of quasi-particle excitations. This is due to the different localisation properties of domain walls, which induce topological effects. The algebraic framework we developed in [5] and [8] can be employed for the investigation of entanglement properties in the ordered phase of a QFT, where excitations can be described by kinks interpolating between vacua.

REFERENCES

- L. Capizzi, O. A. Castro-Alvaredo, C. De Fazio, M. Mazzoni, and L. Santamaria-Sanz. "Symmetry resolved entanglement of excited states in quantum field theory. Part I. Free theories, twist fields and qubits". In: *JHEP* 127 (2022). DOI: 10.1007/JHEP12(2022)127. arXiv: 2203.12556 [hep-th].
- [2] L. Capizzi, C. De Fazio, M. Mazzoni, L. Santamaria-Sanz, and O. A. Castro-Alvaredo. "Symmetry resolved entanglement of excited states in quantum field theory. Part II. Numerics, interacting theories and higher dimensions". In: *JHEP* 128 (2022). DOI: 10.1007/JHEP12(2022)128. arXiv: 2206.12223 [hep-th].
- [3] L. Capizzi, M. Mazzoni, and O. A. Castro-Alvaredo. "Symmetry resolved entanglement of excited states in quantum field theory. Part III. Bosonic and fermionic negativity". In: JHEP 74 (2023). DOI: 10.1007/JHEP06(2023)074.
- [4] O. A. Castro-Alvaredo and M. Mazzoni. "Two-point functions of composite twist fields in the Ising field theory". In: J. Phys. A: Math. Theor. 56.12 (2023), p. 124001. DOI: 10.1088/1751-8121/acbe82. arXiv: 2301.01745 [hep-th].
- [5] L. Capizzi and M. Mazzoni. "Entanglement asymmetry in the ordered phase of many-body systems: the Ising field theory". In: *JHEP* 2023.12 (2023), pp. 1–36. DOI: 10.1007/JHEP12(2023)144.
- [6] M. Mazzoni, R. Travaglino, and O. A. Castro-Alvaredo. "Expectation Values of Conserved Charges in Integrable Quantum Field Theories out of Thermal Equilibrium". In: arXiv preprint arXiv:2402.14788 (2024). URL: https://doi.org/10.48550/arXiv.2402.14788.
- [7] R. Travaglino, M. Mazzoni, and O. A. Castro-Alvaredo. "Generalised Hydrodynamics of *TT*-Deformed Integrable Quantum Field Theories". In: *JHEP* 90 (2024). DOI: 10.1007/JHEP08(2024)090.
- [8] L. Capizzi and M. Mazzoni. "Entanglement content of kink excitations". In: arXiv preprint arXiv:2409.03048 (2024). URL: https://doi.org/10.48550/arXiv.2409.03048.

- L. Amico, R. Fazio, A. Osterloh, and V. Vedral. "Entanglement in many-body systems". In: *Rev. Mod. Phys.* 80 (2008), pp. 517–576. DOI: 10.1103/RevModPhys.80.517. arXiv: quant-ph/0703044.
- C. G. Callan Jr. and F. Wilczek. "On geometric entropy". In: *Phys. Lett. B* 333 (1994), pp. 55–61. DOI: 10.1016/0370-2693(94)91007-3. arXiv: hep-th/9401072.
- C. Holzhey, F. Larsen, and F. Wilczek. "Geometric and renormalized entropy in conformal field theory". In: Nucl. Phys. B 424 (1994), pp. 443–467. DOI: 10.1016/0550-3213(94)90402-2.
- P. Calabrese and J. L. Cardy. "Entanglement entropy and quantum field theory". In: J. Stat. Mech. 0406 (2004), P002. DOI: 10.1088/1742-5468/2004/06/P06002.
- G. Vidal, J. I. Latorre, E. Rico, and A. Kitaev. "Entanglement in quantum critical phenomena". In: *Phys. Rev. Lett.* 90 (2003), p. 227902. DOI: 10.1103/PhysRevLett.90.227902.
- [14] J. I. Latorre, E. Rico, and G. Vidal. "Ground state entanglement in quantum spin chains". In: *Quant. Inf. Comput.* 4 (2004), pp. 48–92. DOI: 10.5555/2011572.2011576.
- B.-Q. Jin and V. Korepin. "Quantum spin chain, Toeplitz determinants and Fisher-Hartwig conjecture". In: J. Stat. Phys. 116 (2004), pp. 79–95. DOI: 10.1023/B: JOSS.0000037230.
 37166.42.
- [16] J. I. Latorre, C. A. Lutken, E. Rico, and G. Vidal. "Fine-grained entanglement loss along renormalization group flows". In: *Phys. Rev. A* 71 (2005), p. 034301. DOI: 10.1103/PhysRevA. 71.034301.
- [17] C. H. Bennett, H. J. Bernstein, S. Popescu, and B. Schumacher. "Concentrating partial entanglement by local operations". In: *Phys. Rev.* A53 (1996), pp. 2046–2052. DOI: 10.1103/PhysRevA.53.2046.
- [18] P. Calabrese and J. Cardy. "Entanglement entropy and conformal field theory". In: J. Phys. A: Math. Theor. 42 (2009), p. 504005. DOI: 10.1088/1751-8113/42/50/504005. arXiv: 0905.4013 [cond-mat.stat-mech].
- M. Goldstein and E. Sela. "Symmetry-Resolved Entanglement in Many-Body Systems". In: Phys. Rev. Lett. 120.20 (2018), p. 200602. DOI: 10.1103/PhysRevLett.120.200602. URL: https://link.aps.org/doi/10.1103/PhysRevLett.120.200602.
- [20] J. L. Cardy, O. A. Castro-Alvaredo, and B. Doyon. "Form factors of branch-point twist fields in quantum integrable models and entanglement entropy". In: J. Stat. Phys. 130 (2008), pp. 129–168. DOI: 10.1007/s10955-007-9422-x.
- B. Doyon. "Bipartite entanglement entropy in massive two-dimensional quantum field theory".
 In: *PRL* 102.3 (2009), p. 031602. DOI: 10.1103/PhysRevLett.102.031602.

- [22] R. Islam, R. Ma, P. M. Preiss, M. E. Tai, A. Lukin, M. Rispoli, and M. Greiner. "Measuring entanglement entropy in a quantum many-body system". In: *Nature* 528 (2015), pp. 77–83. DOI: 10.1038/nature15750.
- [23] A. Neven, J. Carrasco, V. Vitale, C. Kokail, A. Elben, M. Dalmonte, P. Calabrese, P. Zoller, B. Vermersch, R. Kueng, et al. "Symmetry-resolved entanglement detection using partial transpose moments". In: npj Quantum Information 7.1 (2021), p. 152. DOI: 10.1038/s41534-021-00487-y.
- [24] V. Vitale, A. Elben, R. Kueng, A. Neven, J. Carrasco, B. Kraus, P. Zoller, P. Calabrese, B. Vermersch, and M. Dalmonte. "Symmetry-resolved dynamical purification in synthetic quantum matter". In: SciPost Phys. 12 (2022), p. 106. DOI: 10.21468/SciPostPhys.12.3.106.
- [25] H. M. Wiseman and J. A. Vaccaro. "Entanglement of indistinguishable particles shared between two parties". In: *PRL* 91.9 (2003), p. 097902. DOI: 10.1103/PhysRevLett.91.097902.
- [26] H. Barghathi, C. Herdman, and A. Del Maestro. "Rényi generalization of the accessible entanglement entropy". In: *PRL* 121.15 (2018), p. 150501. DOI: 10.1103/PhysRevLett.121.150501.
- [27] H. Barghathi, E. Casiano-Diaz, and A. D. Maestro. "Operationally accessible entanglement of one-dimensional spinless fermions". In: *Phys. Rev. A* 100 (2019), p. 022324. DOI: 10.1103/ PhysRevA.100.022324.
- [28] O. A. Castro-Alvaredo and L. Santamaría-Sanz. "Symmetry-resolved measures in quantum field theory: A short review". In: *Modern Physics Letters B* (2024), p. 2430002. DOI: 10.1142/ S0217984924300023.
- [29] F. Ares, S. Murciano, and P. Calabrese. "Entanglement asymmetry as a probe of symmetry breaking". In: *Nature Commun.* 14.1 (2023). DOI: 10.1038/s41467-023-37747-8. arXiv: 2207.14693 [cond-mat.stat-mech].
- [30] F. Ares, S. Murciano, E. Vernier, and P. Calabrese. "Lack of symmetry restoration after a quantum quench: an entanglement asymmetry study". In: (2023). arXiv: 2302.03330 [cond-mat.stat-mech].
- [31] B. Bertini, K. Klobas, M. Collura, P. Calabrese, and C. Rylands. "Dynamics of charge fluctuations from asymmetric initial states". In: *Phys. Rev. B* 109.18 (2024), p. 184312. DOI: 10.1103/PhysRevB.109.184312.
- [32] M. Fossati, F. Ares, J. Dubail, and P. Calabrese. "Entanglement asymmetry in CFT and its relation to non-topological defects". In: JHEP 2024.5 (2024), pp. 1–38. DOI: 10.1007/ JHEP05(2024)059.

- [33] M. Lastres, S. Murciano, F. Ares, and P. Calabrese. "Entanglement asymmetry in the critical XXZ spin chain". In: arXiv preprint arXiv:2407.06427 (2024). URL: https://doi.org/10.48550/arXiv.2407.06427.
- [34] F. Ares, S. Murciano, L. Piroli, and P. Calabrese. "Entanglement asymmetry study of black hole radiation". In: *Phys. Rev. D* 110.6 (2024), p. L061901. DOI: 10.1103/PhysRevD.110.L061901.
- [35] E. Schrodinger. "The Present Situation in Quantum Mechanics: A Translation of Schrodinger's' Cat Paradox'Paper". In: Proceedings of the American Philosophical Society 124.5 (1980), pp. 323–38.
- [36] J. Von Neumann and R. T. Beyer. Mathematical foundations of quantum mechanics. Princeton University Press Princeton, 1955.
- [37] A. Einstein, B. Podolsky, and N. Rosen. "Can quantum-mechanical description of physical reality be considered complete?" In: *Phys. Rev.* 47.10 (1935), pp. 777–780. DOI: 10.1103/PhysRev.47.777.
- [38] J. S. Bell. "On the Einstein Podolsky Rosen paradox". In: *Physics Physique Fizika* 1.3 (1964), pp. 195–200. DOI: 10.1103/PhysicsPhysiqueFizika.1.195.
- [39] A. Aspect, P. Grangier, and G. Roger. "Experimental tests of realistic local theories via Bell's theorem". In: *PRL* 47.7 (1981), p. 460. DOI: 10.1103/PhysRevLett.47.460.
- [40] A. Aspect, P. Grangier, and G. Roger. "Experimental realization of Einstein-Podolsky-Rosen-Bohm Gedankenexperiment: a new violation of Bell's inequalities". In: *PRL* 49.2 (1982), p. 91. DOI: 10.1103/PhysRevLett.49.91.
- [41] A. Aspect, J. Dalibard, and G. Roger. "Experimental test of Bell's inequalities using time-varying analyzers". In: *PRL* 49.25 (1982), pp. 1804–1807. DOI: 10.1103/PhysRevLett.49.1804.
- [42] K. Mattle, H. Weinfurter, P. G. Kwiat, and A. Zeilinger. "Dense coding in experimental quantum communication". In: *PRL* 76.25 (1996), p. 4656. DOI: 10.1103/PhysRevLett.76.4656.
- [43] D. Bouwmeester, J.-W. Pan, K. Mattle, M. Eibl, H. Weinfurter, and A. Zeilinger. "Experimental quantum teleportation". In: *Nature* 390.6660 (1997), pp. 575–579. DOI: 10.1038/37539.
- [44] M. B. Plenio and S. Virmani. "An introduction to entanglement measures". In: Quantum Information & Computation 7.1 (2007), pp. 1–51. DOI: 10.5555/2011706.2011707.
- [45] K. Kraus, A. Böhm, J. D. Dollard, and W. Wootters. States, Effects, and Operations Fundamental Notions of Quantum Theory: Lectures in Mathematical Physics at the University of Texas at Austin. Springer, 1983. URL: https://doi.org/10.1007/3-540-12732-1.

- [46] C. E. Shannon. "A mathematical theory of communication". In: The Bell system technical journal 27.3 (1948), pp. 379–423. DOI: 10.1002/j.1538-7305.1948.tb01338.x.
- [47] M. A. Nielsen and I. L. Chuang. Quantum computation and quantum information. Cambridge university press, 2010.
- [48] P. Calabrese and J. L. Cardy. "Evolution of entanglement entropy in one-dimensional Systems". In: J. Stat. Mech. (2005). cond-mat/0503393, P010. DOI: 10.1088/1742-5468/2005/04/ P04010.
- [49] T. Nishioka, S. Ryu, and T. Takayanagi. "Holographic entanglement entropy: an overview". In: J. Phys. A: Math. Theor. 42.50 (2009), p. 504008. DOI: 10.1088/1751-8113/42/50/504008.
- [50] A. Rényi. "On measures of entropy and information". In: Proceedings of the fourth Berkeley symposium on mathematical statistics and probability, volume 1: contributions to the theory of statistics. Vol. 4. University of California Press. 1961, pp. 547–562.
- [51] M. Srednicki. "Entropy and area". In: *PRL* 71.5 (1993), p. 666. DOI: 10.1103/PhysRevLett.
 71.666.
- [52] M. B. Hastings. "An area law for one-dimensional quantum systems". In: J. Stat. Mech 2007.08 (2007), P08024. DOI: 10.1088/1742-5468/2007/08/P08024.
- [53] J. Eisert, M. Cramer, and M. B. Plenio. "Colloquium: Area laws for the entanglement entropy". In: Reviews of modern physics 82.1 (2010), pp. 277–306. DOI: 10.1103/RevModPhys.82.277.
- [54] A. Peres. "Separability criterion for density matrices". In: *PRL* 77.8 (1996), p. 1413. DOI: 10.1103/PhysRevLett.77.1413.
- [55] P. Horodecki. "Separability criterion and inseparable mixed states with positive partial transposition". In: *Physics Letters A* 232.5 (1997), pp. 333–339. DOI: 10.1016/S0375-9601(97)00416-7.
- [56] M. B. Plenio. "Logarithmic negativity: a full entanglement monotone that is not convex". In: *Phys. Rev. Lett.* 95.9 (2005), p. 090503. DOI: 10.1103/PhysRevLett.95.090503.
- [57] M. B. Plenio. "Erratum: Logarithmic Negativity: A Full Entanglement Monotone That Is not Convex". In: *Phys. Rev. Lett.* 95.9 (2005), p. 119902. DOI: 10.1103/PhysRevLett.95.119902.
- [58] L. Dixon, D. Friedan, E. Martinec, and S. Shenker. "The conformal field theory of orbifolds".
 In: Nucl. Phys. B 282 (1987), pp. 13–73. DOI: 10.1016/0550-3213(87)90676-6.
- [59] V. Knizhnik. "Analytic fields on Riemann surfaces. II". In: Comm. Math. Phys. 112.4 (1987), pp. 567–590. DOI: 10.1007/BF01225373.

- [60] B. Schroer and T. T. Truong. "The order/disorder quantum field operators associated with the two-dimensional Ising model in the continuum limit". In: Nucl. Phys. B 144.1 (1978), pp. 80–122. DOI: 0550-3213(78)90499-6.
- [61] B. Zuber and C. Itzykson. "Quantum field theory and the two-dimensional Ising model". In: *Phys. Rev. D* 15 (1977), pp. 2875–2884. DOI: 10.1103/PhysRevD.15.2875.
- V. P. Yurov and A. B. Zamolodchikov. "Correlation functions of integrable 2-D models of relativistic field theory. Ising model". In: Int. J. Mod. Phys. A 6 (1991), pp. 3419–3440. DOI: 0.1142/S0217751X91001660.
- [63] O. A. Castro-Alvaredo and B. Doyon. "Bi-partite entanglement entropy in massive QFT with a boundary: The Ising model". In: J. Statist. Phys. 134 (2009), pp. 105–145. DOI: 10.1007/ s10955-008-9664-2. arXiv: 0810.0219 [hep-th].
- [64] O. Blondeau-Fournier and B. Doyon. "Expectation values of twist fields and universal entanglement saturation of the free massive boson". In: J. Phys. A: Math. Theor. 50.27 (2017), p. 274001. DOI: 10.1088/1751-8121/aa7492.
- [65] P. Calabrese, J. Cardy, and E. Tonni. "Entanglement negativity in quantum field theory". In: *Phys. Rev. Lett.* 109.13 (2012), p. 130502. DOI: 10.1103/PhysRevLett.109.130502.
- [66] P. Calabrese, J. Cardy, and E. Tonni. "Entanglement negativity in extended systems: a field theoretical approach". In: J. Stat. Mech. 2013.02 (2013), P02008. DOI: 10.1088/1742-5468/ 2013/02/P02008.
- [67] F. Verstraete, M. M. Wolf, D. Perez-Garcia, and J. I. Cirac. "Criticality, the area law, and the computational power of projected entangled pair states". In: *PRL* 96.22 (2006), p. 220601. DOI: 10.1103/PhysRevLett.96.220601.
- [68] D. Gioev and I. Klich. "Entanglement entropy of fermions in any dimension and the Widom conjecture". In: PRL 96.10 (2006), p. 100503. DOI: 10.1103/PhysRevLett.96.100503.
- [69] A. R. Its, B.-Q. Jin, and V. E. Korepin. "Entanglement in the XY spin chain". In: J. Phys. A: Math. Gen. 38.13 (2005), p. 2975. DOI: 10.1088/0305-4470/38/13/011.
- [70] A. Its, F. Mezzadri, and M. Mo. "Entanglement entropy in quantum spin chains with finite range interaction". In: Comm. Math. Phys. 284.1 (2008), pp. 117–185. DOI: 10.1007/s00220-008-0566-6.
- [71] E. Ercolessi, S. Evangelisti, and F. Ravanini. "Exact entanglement entropy of the XYZ model and its sine-Gordon limit". In: *Physics Letters A* 374.21 (2010), pp. 2101–2105. DOI: 10.1016/j.physleta.2010.03.014.
- [72] J. Latorre and A. Riera. "A short review on entanglement in quantum spin systems". In: J. Phys. A: Math. Theor. 42.50 (2009), p. 504002. DOI: 10.1088/1751-8113/42/50/504002.

- [73] M. B. Hastings. "Entropy and entanglement in quantum ground states". In: *Phys. Rev. B* 76.3 (2007), p. 035114. DOI: 10.1103/PhysRevB.76.035114.
- [74] G. Mussardo. Statistical field theory: an introduction to exactly solved models in statistical physics. Oxford University Press, USA, 2010. DOI: 10.1093/oso/9780198788102.001.0001.
- [75] B. Doyon. Integrability. https://benjamindoyon.weebly.com/uploads/3/8/5/8/38586121/
 notes-integrability.pdf. Lecture notes for the London Taught Course Centre. 2012.
- [76] D. Bombardelli. "S-matrices and integrability". In: J. Phys. A: Math. Theor. 49.32 (2016),
 p. 323003. DOI: 10.1088/1751-8113/49/32/323003.
- [77] M. Karowski and P. Weisz. "Exact S matrices and form-factors in (1+1)-dimensional field theoretic models with soliton behavior". In: Nucl. Phys. B 139 (1978), pp. 455–476. DOI: 10. 1016/0370-1573(79)90113-3.
- [78] F. Smirnov. Form factors in completely integrable models of quantum field theory. Vol. 14. Adv. Series in Math. Phys. Singapore: World Scientific, 1992.
- [79] G. Mussardo. "Off-critical statistical models: factorized scattering theories and bootstrap program". In: *Physics Reports* 218.5-6 (1992), pp. 215–379. DOI: 10.1016/0370-1573(92)90047-4.
- [80] J. L. Cardy and G. Mussardo. "Form-factors of descendent operators in perturbed conformal field theories". In: Nucl. Phys. B 340 (1990), pp. 387–402. DOI: 10.1016/0550-3213(90)90452–J.
- [81] O. Babelon and D. Bernard. "From Form Factors to Correlation Functions: The Ising Model". In: Phys. Lett. B288 (1992), pp. 113–120. DOI: 10.1016/0370-2693(92)91964-B.
- [82] O. A. Castro-Alvaredo and A. Fring. "Form factors from free fermionic Fock fields, the Federbush model". In: Nuclear Physics B 618.3 (2001), pp. 437–464. URL: https://doi.org/10.1016/S0550-3213(01)00462-X.
- [83] A. Fring, G. Mussardo, and P. Simonetti. "Form factors for integrable lagrangian field theories, the sinh-Gordon model". In: Nucl. Phys. B 393.1-2 (1993), pp. 413–441. DOI: 10.1016/0550– 3213(93)90252-K.
- [84] A. B. Zamolodchikov. "Two-point correlation function in scaling Lee-Yang model". In: Nucl. Phys. B 348.3 (1991), pp. 619–641. DOI: 10.1016/0550-3213(91)90207-E.
- [85] H. Babujian, A. Fring, M. Karowski, and A. Zapletal. "Exact form factors in integrable quantum field theories: The sine-Gordon model". In: Nucl. Phys. B 538.3 (1999), pp. 535–586. DOI: 10.1016/S0550-3213(98)00737-8.

- [86] O. Castro-Alvaredo, A. Fring, and C. Korff. "Form factors of the homogeneous sine-Gordon models". In: *Physics Letters B* 484.1-2 (2000), pp. 167–176. URL: https://doi.org/10.1016/ S0370-2693(00)00565-7.
- [87] O. Castro-Alvaredo and A. Fring. "Identifying the operator content, the homogeneous sine-Gordon models". In: Nuclear Physics B 604.1-2 (2001), pp. 367–390. URL: https://doi.org/10.1016/S0550-3213(01)00055-4.
- [88] A. Fring, G. Mussardo, and P. Simonetti. "Form factors of the elementary field in the Bullough-Dodd model". In: *Physics Letters B* 307.1-2 (1993), pp. 83–90. DOI: 10.1016/0370-2693(93)90196-0.
- [89] D. Bernard and A. Leclair. "Differential Equations for Sine-Gordon Correlation Functions at the Free Fermion Point". In: Nucl. Phys. B 426 (1994). Erratum-ibid.B498, 619–62 (1997), pp. 534–558. DOI: 10.1016/0550-3213(94)90020-5.
- C. Efthimiou and A. LeClair. "Particle-field duality and form factors from vertex operators".
 In: Comm. Math. Phys. 171 (1995), pp. 531–546. DOI: 10.1007/BF02104677.
- [91] I. S. Gradshteyn and I. M. Ryzhik. Table of integrals, series, and products. Academic press, 2014.
- [92] N. Laflorencie and S. Rachel. "Spin-resolved entanglement spectroscopy of critical spin chains and Luttinger liquids". In: J. Stat. Mech. 2014 (2014). DOI: 10.1088/1742-5468/2014/11/ P11013.
- [93] A. Lukin et al. "Probing entanglement in a many-body-localized system". In: Science 364.6437 (2019), pp. 256–260. DOI: 10.1126/science.aau0818.
- [94] J. C. Xavier, F. C. Alcaraz, and G. Sierra. "Equipartition of the entanglement entropy". In: Phys. Rev. B 98.4 (2018), p. 041106. DOI: 10.1103/PhysRevB.98.041106. arXiv: 1804.06357 [cond-mat.stat-mech].
- [95] P. Calabrese, J. Dubail, and S. Murciano. "Symmetry-resolved entanglement entropy in Wess-Zumino-Witten models". In: JHEP 10 (2021), p. 067. DOI: 10.1007/JHEP10(2021)067. arXiv: 2106.15946 [hep-th].
- [96] E. Cornfeld, M. Goldstein, and E. Sela. "Imbalance entanglement: Symmetry decomposition of negativity". In: *Phys. Rev. A* 98.3 (2018), p. 032302. DOI: 10.1103/PhysRevA.98.032302.
 URL: https://link.aps.org/doi/10.1103/PhysRevA.98.032302.
- [97] L. Capizzi and P. Calabrese. "Symmetry resolved relative entropies and distances in conformal field theory". In: JHEP 10 (2021), p. 195. DOI: 10.1007/JHEP10(2021)195. arXiv: 2105.08596
 [hep-th].

- [98] A. Belin, L.-Y. Hung, A. Maloney, S. Matsuura, R. C. Myers, and T. Sierens. "Holographic Charged Rényi Entropies". In: JHEP 2013 (2013), p. 59. DOI: 10.1007/JHEP12(2013)059.
- [99] A. Belin, L.-Y. Hung, A. Maloney, and S. Matsuura. "Charged Rényi entropies and holographic superconductors". In: JHEP 2015.1 (2015), pp. 1–20. DOI: 10.1007/JHEP01(2015)059.
- [100] G. Pastras and D. Manolopoulos. "Charged Rényi entropies in CFTs with Einstein-Gauss-Bonnet holographic duals". In: JHEP 2014.11 (2014), pp. 1–42. DOI: 10.1007/JHEP11(2014)007.
- S. Matsuura, X. Wen, L.-Y. Hung, and S. Ryu. "Charged topological entanglement entropy". In: *Phys. Rev. B* 93.19 (2016), p. 195113. DOI: 10.1103/PhysRevB.93.195113.
- [102] P. Caputa, M. Nozaki, and T. Numasawa. "Charged Entanglement Entropy of Local Operators". In: Phys. Rev. D93 (2016), p. 105032. DOI: 10.1103/PhysRevD.93.105032.
- Y. Aharonov and D. Bohm. "Significance of electromagnetic potentials in the quantum theory".
 In: *Phys. Rev.* 115.3 (1959), p. 485. DOI: https://doi.org/10.1103/PhysRev.115.485.
- [104] L. Capizzi, P. Ruggiero, and P. Calabrese. "Symmetry resolved entanglement entropy of excited states in a CFT". In: J. Stat. Mech. 2007 (2020), p. 073101. DOI: 10.1088/1742-5468/ab96b6. arXiv: 2003.04670 [cond-mat.stat-mech].
- [105] D. X. Horvath, L. Capizzi, and P. Calabrese. "U(1) symmetry resolved entanglement in free 1+1 dimensional field theories via form factor bootstrap". In: *JHEP* 05 (2021), p. 197. DOI: 10.1007/JHEP05(2021)197. arXiv: 2103.03197 [hep-th].
- [106] D. X. Horvath, P. Calabrese, and O. A. Castro-Alvaredo. "Branch Point Twist Field Form Factors in the sine-Gordon Model II: Composite Twist Fields and Symmetry Resolved Entanglement". In: SciPost Phys. 12.3 (2022), p. 088. DOI: 10.21468/SciPostPhys.12.3.088. arXiv: 2105.13982 [hep-th].
- [107] D. X. Horváth and P. Calabrese. "Symmetry resolved entanglement in integrable field theories via form factor bootstrap". In: JHEP 11 (2020), p. 131. DOI: 10.1007/JHEP11(2020)131. arXiv: 2008.08553 [hep-th].
- [108] L. Capizzi, D. X. Horváth, P. Calabrese, and O. A. Castro-Alvaredo. "Entanglement of the 3-state Potts model via form factor bootstrap: total and symmetry resolved entropies". In: JHEP 05 (2022), p. 113. DOI: 10.1007/JHEP05(2022)113. arXiv: 2108.10935 [hep-th].
- [109] Z. Ma, C. Han, Y. Meir, and E. Sela. "Symmetric inseparability and number entanglement in charge-conserving mixed states". In: *Phys. Rev. A* 105.4 (2022), p. 042416. DOI: 10.1103/ PhysRevA.105.042416.
- [110] E. B. Mpemba and D. G. Osborne. "Cool?" In: *Physics Education* 4.3 (1969), pp. 172–175.
 DOI: https://doi.org/10.1088/0031-9120/4/3/312.

- S. Murciano, F. Ares, I. Klich, and P. Calabrese. "Entanglement asymmetry and quantum Mpemba effect in the XY spin chain". In: J. Stat. Mech 2024.1 (2024), p. 013103. DOI: 10. 1088/1742-5468/ad17b4.
- [112] F. Ferro, F. Ares, and P. Calabrese. "Non-equilibrium entanglement asymmetry for discrete groups: the example of the XY spin chain". In: J. Stat. Mech. 2024.2 (2024), p. 023101. DOI: 10.1088/1742-5468/ad138f.
- [113] H. Araki. "Relative entropy of states of von Neumann algebras". In: *Publications of the Research Institute for Mathematical Sciences* 11.3 (1975), pp. 809–833. DOI: DOI10.2977/PRIMS/1195191148.
- [114] O. A. Castro-Alvaredo, C. De Fazio, B. Doyon, and I. M. Szécsényi. "Entanglement Content of Quasiparticle Excitations". In: *Phys. Rev. Lett.* 121.17 (2018), p. 170602. DOI: 10.1103/ PhysRevLett.121.170602.
- [115] O. A. Castro-Alvaredo, C. De Fazio, B. Doyon, and I. M. Szécsényi. "Entanglement content of quantum particle excitations. Part I. Free field theory". In: JHEP 10 (2018), p. 039. DOI: 10.1007/JHEP10(2018)039.
- [116] O. A. Castro-Alvaredo, C. D. Fazio, B. Doyon, and I. M. Szécsényi. "Entanglement content of quantum particle excitations. Part II. Disconnected regions and logarithmic negativity". In: *JHEP* 2019.11 (2019), pp. 1–47. DOI: 10.1007/JHEP11(2019)058.
- [117] D. Bianchini and O. A. Castro-Alvaredo. "Branch Point Twist Field Correlators in the Massive Free Boson Theory". In: Nucl. Phys. B 913 (2016), pp. 879–911. DOI: 10.1016/j.nuclphysb. 2016.10.016.
- [118] N. Feldman and M. Goldstein. "Dynamics of charge-resolved entanglement after a local quench". In: *Phys. Rev.* B100 (2019), p. 235146. DOI: 10.1103/PhysRevB.100.235146.
- [119] R. Bonsignori and P. Calabrese. "Boundary effects on symmetry resolved entanglement". In: J. Phys. A: Math. Theor. A54.1 (2020), p. 015005. DOI: 10.1088/1751-8121/abcc3a.
- S. Murciano, R. Bonsignori, and P. Calabrese. "Symmetry decomposition of negativity of massless free fermions". In: SciPost Physics 10.5 (2021), p. 111. DOI: 10.21468/SciPostPhys.10.5.111.
- H.-H. Chen. "Symmetry decomposition of relative entropies in conformal field theory". In: JHEP 07 (2021), p. 084. DOI: 10.1007/JHEP07(2021)084. arXiv: 2104.03102 [hep-th].
- [122] B. Estienne, Y. Ikhlef, and A. Morin-Duchesne. "Finite-size corrections in critical symmetry-resolved entanglement". In: *SciPost Phys.* 10 (2021), p. 54. DOI: 10.21468/SciPostPhys.10.3.054.

- [123] R. Bonsignori, P. Ruggiero, and P. Calabrese. "Symmetry resolved entanglement in free fermionic systems". In: J. Phys. A: Math. Theor. A52.47 (2019), p. 475302. DOI: 10.1088/1751-8121/ab4b77.
- S. Murciano, G. Di Giulio, and P. Calabrese. "Entanglement and symmetry resolution in two dimensional free quantum field theories". In: JHEP 08 (2020), p. 073. DOI: 10.1007/JHEP08(2020)073. arXiv: 2006.09069 [hep-th].
- S. Zhao, C. Northe, and R. Meyer. "Symmetry-resolved entanglement in AdS3/CFT2 coupled to U (1) Chern-Simons theory". In: *JHEP* 2021.7 (2021), pp. 1–38. DOI: 10.1007/JHEP07(2021) 030.
- K. Weisenberger, S. Zhao, C. Northe, and R. Meyer. "Symmetry-resolved entanglement for excited states and two entangling intervals in AdS3/CFT2". In: *JHEP* 2021.12 (2021), pp. 1–31.
 DOI: 10.1007/JHEP12(2021)104.
- S. Zhao, C. Northe, K. Weisenberger, and R. Meyer. "Charged Moments in W₃ Higher Spin Holography". In: JHEP 2022 (2022), p. 166. DOI: 10.1007/JHEP05(2022)166.
- [128] S. Fraenkel and M. Goldstein. "Symmetry resolved entanglement: exact results in 1D and beyond". In: J. Stat. Mech. 2020.3 (2020), p. 033106. DOI: 10.1088/1742-5468/ab7753.
- [129] S. Murciano, G. D. Giulio, and P. Calabrese. "Symmetry resolved entanglement in gapped integrable systems: a corner transfer matrix approach". In: *SciPost Phys.* 8 (2020), p. 46. DOI: 10.21468/SciPostPhys.8.3.046.
- [130] P. Calabrese, M. Collura, G. D. Giulio, and S. Murciano. "Full counting statistics in the gapped XXZ spin chain". In: *EPL* 129.6 (2020), p. 60007. DOI: 10.1209/0295-5075/129/60007.
- [131] G. Parez, R. Bonsignori, and P. Calabrese. "Quasiparticle dynamics of symmetry-resolved entanglement after a quench: Examples of conformal field theories and free fermions". In: *Phys. Rev. B* 103 (2021), p. L041104. DOI: 10.1103/PhysRevB.103.L041104.
- [132] M. T. Tan and S. Ryu. "Particle number fluctuations, Rényi entropy, and symmetry-resolved entanglement entropy in a two-dimensional Fermi gas from multidimensional bosonization". In: *Phys. Rev. B* 101 (2020), p. 235169. DOI: 10.1103/PhysRevB.101.235169.
- S. Murciano, P. Ruggiero, and P. Calabrese. "Symmetry resolved entanglement in two-dimensional systems via dimensional reduction". In: J. Stat. Mech. 2020.8 (2020), p. 083102. DOI: 10.1088/1742-5468/aba1e5.
- S. Fraenkel and M. Goldstein. "Entanglement Measures in a Nonequilibrium Steady State: Exact Results in One Dimension". In: SciPost Phys. 11 (2021), p. 85. DOI: 10.21468/SciPostPhys.11.4.085.

- [135] G. Parez, R. Bonsignori, and P. Calabrese. "Exact quench dynamics of symmetry resolved entanglement in a free fermion chain". In: J. Stat. Mech. 2021.9 (2021), p. 093102. DOI: 10. 1088/1742-5468/ac21d7.
- [136] G. Parez, R. Bonsignori, and P. Calabrese. "Dynamics of charge-imbalance-resolved entanglement negativity after a quench in a free-fermion model". In: J. Stat. Mech 2022.5 (2022), p. 053103. DOI: 10.1088/1742-5468/ac666c.
- [137] S. Scopa and D. X. Horváth. "Exact hydrodynamic description of symmetry-resolved Rényi entropies after a quantum quench". In: J. Stat. Mech. 2022.8 (2022), p. 083104. DOI: 10.1088/ 1742-5468/ac85eb.
- [138] X. Turkeshi, P. Ruggiero, V. Alba, and P. Calabrese. "Entanglement equipartition in critical random spin chains". In: *Phys. Rev. B* 102 (2020), p. 014455. DOI: 10.1103/PhysRevB.102.014455.
- [139] M. Kiefer-Emmanouilidis, R. Unanyan, M. Fleischhauer, and J. Sirker. "Evidence for Unbounded Growth of the Number Entropy in Many-Body Localized Phases". In: *Phys. Rev. Lett.* 124 (2020), p. 243601. DOI: 10.1103/PhysRevLett.124.243601.
- [140] M. Kiefer-Emmanouilidis, R. Unanyan, M. Fleischhauer, and J. Sirker. "Unlimited growth of particle fluctuations in many-body localized phases". In: Ann. Phys. 435 (2021), p. 168481.
 DOI: 10.1016/j.aop.2021.168481.
- K. Monkman and J. Sirker. "Operational entanglement of symmetry-protected topological edge states". In: *Phys. Rev. Research* 2 (2020), p. 043191. DOI: 10.1103/PhysRevResearch.2.
 043191.
- [142] E. Cornfeld, L. A. Landau, K. Shtengel, and E. Sela. "Entanglement spectroscopy of non-Abelian anyons: Reading off quantum dimensions of individual anyons". In: *Phys. Rev. B* 99 (2019), p. 115429. DOI: 10.1103/PhysRevB.99.115429.
- [143] D. Azses and E. Sela. "Symmetry-resolved entanglement in symmetry-protected topological phases". In: Phys. Rev. B 102 (2020), p. 235157. DOI: 10.1103/PhysRevB.102.235157.
- [144] O. A. Castro-Alvaredo, C. De Fazio, B. Doyon, and I. M. Szécsényi. "Entanglement Content of Quantum Particle Excitations III. Graph Partition Functions". In: J. Math. Phys. 60.8 (2019), p. 082301. DOI: 10.1063/1.5098892. arXiv: 1904.02615 [math-ph].
- [145] A. Jafarizadeh and M. A. Rajabpour. "Bipartite entanglement entropy of the excited states of free fermions and harmonic oscillators". In: *Phys. Rev. B* 100.16 (2019). DOI: 10.1103/ PhysRevB.100.165135.

- [146] J. Zhang and M. A. Rajabpour. "Excited state Rényi entropy and subsystem distance in two-dimensional non-compact bosonic theory. Part I. Single-particle states". In: *JHEP* 12 (2020), p. 160. DOI: 10.1007/JHEP12(2020)160. arXiv: 2009.00719 [hep-th].
- [147] J. Zhang and M. Rajabpour. "Universal Rényi entanglement entropy of quasiparticle excitations". In: EPL 135.6 (2021), p. 60001. DOI: 10.1209/0295-5075/ac130e.
- [148] J. Zhang and M. A. Rajabpour. "Corrections to universal Rényi entropy in quasiparticle excited states of quantum chains". In: J. Stat. Mech. 2109 (2021), p. 093101. DOI: 10.1088/1742-5468/ac1f28. arXiv: 2010.16348 [cond-mat.stat-mech].
- [149] J. Zhang and M. A. Rajabpour. "Excited state Rényi entropy and subsystem distance in two-dimensional non-compact bosonic theory. Part II. Multi-particle states". In: *JHEP* 08 (2021), p. 106. DOI: 10.1007/JHEP08(2021)106. arXiv: 2011.11006 [hep-th].
- [150] J. Zhang and M. A. Rajabpour. "Entanglement of magnon excitations in spin chains". In: JHEP 02 (2022), p. 072. DOI: 10.1007 / JHEP02(2022) 072. arXiv: 2109.12826 [cond-mat.stat-mech].
- [151] J. Angel-Ramelli. "Entanglement entropy of excited states in the quantum Lifshitz model". In: J. Stat. Mech. 2021.1 (2021), p. 013102. DOI: 10.1088/1742-5468/abcd35.
- [152] G. Mussardo and J. Viti. "The $\hbar \rightarrow 0$ Limit of the Entanglement Entropy". In: *Phys. Rev. A* 105.3 (2022), p. 032404. DOI: 10.1103/PhysRevA.105.032404. arXiv: 2112.06840 [quant-ph].
- [153] J. Mölter, T. Barthel, U. Schollwöck, and V. Alba. "Bound states and entanglement in the excited states of quantum spin chains". In: J. Stat. Mech. 2014.10 (2014), P10029. DOI: 10. 1088/1742-5468/2014/10/P10029.
- [154] F. C. Alcaraz, M. I. Berganza, and G. Sierra. "Entanglement of low-energy excitations in Conformal Field Theory". In: *Phys. Rev. Lett.* 106 (2011), p. 201601. DOI: 10.1103/PhysRevLett.106.201601. arXiv: 1101.2881 [cond-mat.stat-mech].
- [155] M. I. Berganza, F. C. Alcaraz, and G. Sierra. "Entanglement of excited states in critical spin chains". In: J. Stat. Mech. (2012), P01016. DOI: 10.1088/1742-5468/2012/01/P01016.
- [156] O. Castro-Alvaredo, B. Doyon, and E. Levi. "Arguments towards a *c*-theorem from branch-point twist fields". In: J. Phys. A: Math. Theor. 44 (2011), p. 492003. DOI: 10.1088/1751-8113/44/49/492003.
- [157] E. Levi. "Composite branch-point twist fields in the Ising model and their expectation values".
 In: J. Phys. A: Math. Theor. 45 (2012), p. 275401. DOI: 10.1088/1751-8113/45/27/275401.
 arXiv: 1204.1192 [hep-th].

- [158] D. Bianchini, O. Castro-Alvaredo, B. Doyon, E. Levi, and F. Ravanini. "Entanglement entropy of non-unitary conformal field theory". In: J. Phys. A: Math. Theor. 48 (2015), 04FT01. DOI: 10.1088/1751-8113/48/4/04FT01.
- B. Pozsgay and G. Takacs. "Form-factors in finite volume I: Form-factor bootstrap and truncated conformal space". In: Nucl. Phys. B 788 (2008), pp. 167–208. DOI: 10.1016/j.nuclphysb.2007.06.027.
- B. Pozsgay and G. Takacs. "Form factors in finite volume. II. Disconnected terms and finite temperature correlators". In: Nucl. Phys. B 788 (2008), pp. 209–251. DOI: 10.1016/j.nuclphysb.2007.07.008.
- [161] P. Fonseca and A. Zamolodchikov. Ward identities and integrable differential equations in the Ising field theory. hep-th/0309228. 2003. URL: https://doi.org/10.48550/arXiv.hepth/0309228.
- H. Casini, C. D. Fosco, and M. Huerta. "Entanglement and alpha entropies for a massive Dirac field in two dimensions". In: J. Stat. Mech. 0507 (2005), P07007. DOI: 10.1088/1742-5468/2005/07/P07007. arXiv: cond-mat/0505563.
- [163] H. Bethe. "Zur theorie der metalle: I. Eigenwerte und eigenfunktionen der linearen atomkette".
 In: Zeitschrift für Physik 71.3 (1931), pp. 205–226. DOI: urtheoriedermetalle: I. Eigenwerte.
- [164] C.-N. Yang and C. P. Yang. "Thermodynamics of a one-dimensional system of bosons with repulsive delta-function interaction". In: *Journal of Mathematical Physics* 10.7 (1969), pp. 1115–1122. DOI: 10.1063/1.1664947.
- [165] F. Franchini. An introduction to integrable techniques for one-dimensional quantum systems.
 Vol. 940. Springer, 2017. DOI: 10.1007/978-3-319-48487-7. arXiv: 1609.02100
 [cond-mat.stat-mech].
- [166] A. B. Zamolodchikov. "Integrable field theory from conformal field theory". In: Adv. Stud. Pure Math. 19 (1989), pp. 641–674. DOI: 10.1016/B978-0-12-385342-4.50022-6.
- [167] E. Witten. "APS Medal for Exceptional Achievement in Research: Invited article on entanglement properties of quantum field theory". In: *Rev. Mod. Phys.* 90.4 (2018), p. 045003.
 DOI: 10.1103/RevModPhys.90.045003.
- [168] L.-Y. Hung, R. C. Myers, and M. Smolkin. "Twist operators in higher dimensions". In: JHEP (2014), p. 178. DOI: 10.1007/JHEP10(2014)178.
- [169] A. Sveskoa. "Extending Charged Holographic Rényi Entropy". In: Class. Quant. Grav. 38 (2021), p. 13. DOI: 10.1088/1361-6382/abfed4.
- [170] J. Long. "Area law of connected correlation function in higher dimensional conformal field theory". In: JHEP (2021), p. 110. DOI: 10.1007/JHEP02(2021)110.

- [171] R. Dijkgraaf, C. Vafa, E. Verlinde, and H. Verlinde. "The operator algebra of orbifold models". In: Commun. Math. Phys. 123 (1989), pp. 485–526. DOI: 10.1007/BF01238812.
- [172] T. Dupic, B. Estienne, and Y. Ikhlef. "Entanglement entropies of minimal models from null-vectors". In: SciPost Phys. 4 (2018), p. 031. DOI: 10.21468/SciPostPhys.4.6.031.
- [173] I. Peschel. "Calculation of reduced density matrices from correlation functions". In: J. Phys. A: Math. Theor. 36 (2003), p. L205. DOI: 10.1088/0305-4470/36/14/101.
- [174] S.-A. Cheong and C. L. Henley. "Many-body density matrices for free fermions". In: *Phys. Rev.* B 69.7 (2004), p. 075111. DOI: 10.1103/PhysRevB.69.075111.
- [175] I. Peschel and V. Eisler. "Reduced density matrices and entanglement entropy in free lattice models". In: J. Phys. A: Math. Theor. 42.50 (2009), p. 504003. DOI: 10.1088/1751-8113/42/50/504003.
- K. Audenaert, J. Eisert, M. B. Plenio, and R. F. Werner. "Entanglement properties of the harmonic chain". In: *Phys. Rev.* A66.4 (2002), p. 042327. DOI: 10.1103/PhysRevA.66.042327.
 URL: https://link.aps.org/doi/10.1103/PhysRevA.66.042327.
- K. Życzkowski, P. Horodecki, A. Sanpera, and M. Lewenstein. "Volume of the set of separable states". In: *Phys. Rev.* A58.2 (1998), pp. 883–892. DOI: 10.1103/PhysRevA.58.883. URL: https://link.aps.org/doi/10.1103/PhysRevA.58.883.
- [178] J. Eisert and M. B. Plenio. "A comparison of entanglement measures". In: J. Mod. Opt. 46.1 (1999), pp. 145–154. DOI: 10.1080/09500349908231260.
- [179] G. Vidal and R. F. Werner. "Computable measure of entanglement". In: *Phys. Rev. A* 65.3 (2002), p. 032314. DOI: 10.1103/PhysRevA.65.032314.
- [180] J. Eisert. "Entanglement in quantum information theory". quant-ph/0610253. PhD thesis. 2006.
 URL: https://doi.org/10.48550/arXiv.quant-ph/0610253.
- [181] V. Eisler and Z. Zimborás. "Entanglement negativity in two-dimensional free lattice models".
 In: *Phys. Rev.* B93.11 (2016), p. 115148. DOI: 10.1103/PhysRevB.93.115148.
- [182] H. Shapourian, K. Shiozaki, and S. Ryu. "Partial time-reversal transformation and entanglement negativity in fermionic systems". In: *Phys. Rev. B* 95.16 (2017), p. 165101. DOI: 10.1103/PhysRevB.95.165101.
- [183] H. Shapourian and S. Ryu. "Entanglement negativity of fermions: Monotonicity, separability criterion, and classification of few-mode states". In: *Phys. Rev. A* 99.2 (2019), p. 022310. DOI: 10.1103/PhysRevA.99.022310.
- [184] H. Shapourian, K. Shiozaki, and S. Ryu. "Many-body topological invariants for fermionic symmetry-protected topological phases". In: *PRL* 118.21 (2017), p. 216402. DOI: 10.1103/PhysRevLett.118.216402.

- [185] H.-H. Chen. "Charged Rényi negativity of massless free bosons". In: JHEP 2022.2 (2022), pp. 1–27. DOI: 10.1007/JHEP02(2022)117.
- [186] H. Gaur and U. A. Yajnik. "Charge imbalance resolved Rényi negativity for free compact boson: Two disjoint interval case". In: JHEP 2023.2 (2023), pp. 1–26. DOI: 10.1007/JHEP02(2023)118.
- [187] P. Bouwknegt. Coset construction for winding subalgebras and applications. q-alg/9610013.
 1996. URL: https://doi.org/10.48550/arXiv.q-alg/9610013.
- [188] L. Borisov, M. B. Halpern, and C. Schweigert. "Systematic approach to cyclic orbifolds". In: Int. J. Mod. Phys. A 13 (1998), pp. 125–168. DOI: 0.1142/S0217751X98000044.
- [189] V. Eisler and Z. Zimborás. "On the partial transpose of fermionic Gaussian states". In: New Journal of Physics 17.5 (2015), p. 053048. DOI: 10.1088/1367-2630/17/5/053048.
- [190] J. Eisert, V. Eisler, and Z. Zimborás. "Entanglement negativity bounds for fermionic Gaussian states". In: *Phys. Rev. B* 97.16 (2018), p. 165123. DOI: 10.1103/PhysRevB.97.165123. URL: https://link.aps.org/doi/10.1103/PhysRevB.97.165123.
- [191] H. Shapourian and S. Ryu. "Finite-temperature entanglement negativity of free fermions". In: J. Stat. Mech 2019.4 (2019), p. 043106. DOI: 10.1088/1742-5468/ab11e0.
- [192] J. Zhang and M. A. Rajabpour. "Universal Rényi entanglement entropy of quasiparticle excitations". In: *EPL* 135.6 (2021), p. 60001. DOI: 10.1209/0295-5075/ac130e. arXiv: 2010.13973 [cond-mat.stat-mech].
- [193] J. Zhang and M. A. Rajabpour. "Subsystem distances between quasiparticle excited states". In: JHEP 07 (2022), p. 119. DOI: 10.1007/JHEP07(2022)119. arXiv: 2202.11448
 [cond-mat.stat-mech].
- [194] H.-H. Chen. "Dynamics of charge imbalance resolved negativity after a global quench in free scalar field theory". In: JHEP 2022.8 (2022), pp. 1–26. DOI: 10.1007/JHEP08(2022)146.
- [195] J. J. Tattersall. Elementary Number Theory in Nine Chapters. 2nd. Cambridge University Press, 2005.
- [196] G. E. Bergum and V. E. H. Jr. "Irreducibility of Lucas and generalized Lucas polynomials".
 In: *The Fibonacci Quarterly* 12 (1974), pp. 95–100.
- [197] P. E. Ricci. "Generalized Lucas polynomials and Fibonacci polynomials". In: Riv. Mat. Univ. Parma 4 (1995), pp. 137–146.
- [198] P. Di Francesco, P. Mathieu, and D. Senechal. Conformal Field Theory. Graduate Texts in Contemporary Physics. New York: Springer-Verlag, 1997. DOI: 10.1007/978-1-4612-2256-9.

- [199] O. A. Castro-Alvaredo and B. Doyon. "Bi-partite entanglement entropy in integrable models with backscattering". In: J. Phys. A: Math. Theor. 41 (2008), p. 275203. DOI: 10.1088/1751-8113/41/27/275203.
- [200] O. A. Castro-Alvaredo. "Massive corrections to entanglement in minimal E8 Toda field theory".
 In: SciPost Physics 2.1 (2017), p. 008. DOI: 10.21468/SciPostPhys.2.1.008.
- [201] D. Bianchini, O. Castro-Alvaredo, and B. Doyon. "Entanglement Entropy of Non-Unitary Integrable Quantum Field Theory". In: Nucl. Phys. B896 (2015), pp. 835–880. DOI: 10.1016/j.nuclphysb.2015.05.013.
- [202] G. Delfino, P. Simonetti, and J. L. Cardy. "Asymptotic factorisation of form-factors in two-dimensional quantum field theory". In: *Phys. Lett. B* 387 (1996), pp. 327–333. DOI: 10.1016/0370-2693(96)01035-0. arXiv: hep-th/9607046.
- [203] H. Babujian and M. Karowski. "Towards the construction of Wightman functions of integrable quantum field theories". In: Int. J. Mod. Phys. A 19S2 (2004), pp. 34–49. DOI: 10.1142/ S0217751X04020294.
- [204] C. Kittel and P. McEuen. Introduction to solid state physics. John Wiley & Sons, 2018.
- [205] J. F. Annett. Superconductivity, superfluids and condensates. Vol. 5. Oxford University Press, 2004.
- [206] J. Richter. "Zero-temperature magnetic ordering in the inhomogeneously frustrated quantum Heisenberg antiferromagnet on a square lattice". In: Phys. Rev. B 47.10 (1993), p. 5794. DOI: 10.1103/PhysRevB.47.5794.
- [207] S. Sachdev. "Quantum phase transitions". In: *Physics world* 12.4 (1999), p. 33. DOI: 10.1088/ 2058-7058/12/4/23.
- [208] G. Parez. "Symmetry-resolved Rényi fidelities and quantum phase transitions". In: *Phys. Rev.* B 106.23 (2022), p. 235101. DOI: 10.1103/PhysRevB.106.235101.
- [209] E. B. Vinberg. *Linear representations of groups*. Springer Science & Business Media, 1989.
- [210] L. Capizzi and V. Vitale. "A universal formula for the entanglement asymmetry of matrix product states". In: arXiv preprint arXiv:2310.01962 (2023). URL: https://doi.org/10.48550/arXiv.2310.01962.
- [211] N. D. Mermin and H. Wagner. "Absence of Ferromagnetism or Antiferromagnetism in Oneor Two-Dimensional Isotropic Heisenberg Models". In: *Phys. Rev. Lett.* 17.26 (1966), pp. 1307–1307. DOI: 10 . 1103 / PhysRevLett . 17 . 1307. URL: https://link.aps.org/doi/10.1103/PhysRevLett.17.1307.
- B. M. McCoy and T. T. Wu. The Two-Dimensional Ising Model. Cambridge, MA and London, England: Harvard University Press, 1973. DOI: doi:10.4159/harvard.9780674180758. URL: https://doi.org/10.4159/harvard.9780674180758.
- [213] G. Delfino. "Universal amplitude ratios in the two-dimensional Ising model". In: *Phys. Lett. B* 419 (1998). [Erratum: Phys.Lett.B 518, 330–330 (2001)], pp. 291–295. DOI: 10.1016/S0370-2693(97)01457-3. arXiv: hep-th/9710019.
- [214] O. A. Castro-Alvaredo and B. Doyon. "Permutation operators, entanglement entropy, and the XXZ spin chain in the limit $\Delta \rightarrow 1^{-}$ ". In: J. Stat. Mech. 1102 (2011), P02001. DOI: 10.1088/1742-5468/2011/02/P02001. arXiv: 1011.4706 [hep-th].
- [215] O. A. Castro-Alvaredo and B. Doyon. "Entanglement Entropy of Highly Degenerate States and Fractal Dimensions". In: *Phys. Rev. Lett.* 108.12 (2012), p. 120401. DOI: 10 . 1103 / PhysRevLett . 108 . 120401. URL: https://link.aps.org/doi/10.1103/PhysRevLett.108.120401.
- [216] K. Ohmori and Y. Tachikawa. "Physics at the entangling surface". In: J. Stat. Mech. 1504 (2015), P04010. DOI: 10.1088/1742-5468/2015/04/P04010. arXiv: 1406.4167 [hep-th].
- [217] O. A. Castro-Alvaredo and E. Levi. "Higher particle form factors of branch point twist fields in integrable quantum field theories". In: J. Phys. A: Math. Theor. 44.25 (2011), p. 255401. DOI: 10.1088/1751-8113/44/25/255401.
- [218] O. A. Castro-Alvaredo and D. X. Horvath. "Branch point twist field form factors in the sine-Gordon model I: Breather fusion and entanglement dynamics". In: *SciPost Physics* 10.6 (2021), p. 132. DOI: 10.21468/SciPostPhys.10.6.132.
- [219] G. Delfino and J. L. Cardy. "Universal amplitude ratios in the two-dimensional q-state Potts model and percolation from quantum field theory". In: Nucl. Phys. B 519.3 (1998), pp. 551–578.
 DOI: 10.1016/S0550-3213(98)00144-8.
- [220] L. Chim and A. Zamolodchikov. "Integrable field theory of the q-state Potts model with 0 < q < 4". In: International Journal of Modern Physics A 7.21 (1992), pp. 5317–5335. DOI: 10. 1142/S0217751X9200243X.
- [221] D. Voiculescu. "Symmetries of some reduced free product C*-algebras". In: vol. 1132. 2006, pp. 556–588. ISBN: 978-3-540-15643-7. DOI: 10.1007/BFb0074909.
- [222] G. Delfino, G. Mussardo, and P. Simonetti. "Correlation functions along a massless flow". In: Phys. Rev. D 51 (1995), pp. 6620–6624. DOI: 10.1103/PhysRevD.51.R6620. arXiv: hepth/9410117.

- [223] S. Ghoshal and A. B. Zamolodchikov. "Boundary S matrix and boundary state in two-dimensional integrable quantum field theory". In: Int. J. Mod. Phys. A 9 (1994).
 [Erratum: Int.J.Mod.Phys.A 9, 4353 (1994)], pp. 3841–3886. DOI: 10.1142/S0217751X94001552. arXiv: hep-th/9306002.
- [224] R. Bonsignori, L. Capizzi, and P. Panopoulos. "Boundary Symmetry Breaking in CFT and the string order parameter". In: *JHEP* 05 (2023), p. 027. DOI: 10.1007/JHEP05(2023)027. arXiv: 2301.08676 [hep-th].
- [225] H. A. Kramers and G. H. Wannier. "Statistics of the Two-Dimensional Ferromagnet. Part I". In: Phys. Rev. 60.3 (1941), pp. 252-262. DOI: 10.1103/PhysRev.60.252. URL: https: //link.aps.org/doi/10.1103/PhysRev.60.252.
- [226] D. Aasen, R. S. K. Mong, and P. Fendley. "Topological Defects on the Lattice I: The Ising model". In: J. Phys. A: Math. Theor. 49.35 (2016), p. 354001. DOI: 10.1088/1751-8113/49/ 35/354001. arXiv: 1601.07185 [cond-mat.stat-mech].
- [227] U. Schollwöck. "The density-matrix renormalization group in the age of matrix product states".
 In: Annals of Physics 326.1 (2011). January 2011 Special Issue, pp. 96–192. ISSN: 0003-4916.
 DOI: https://doi.org/10.1016/j.aop.2010.09.012.
- [228] J. Cardy. "Quantum Quenches to a Critical Point in One Dimension: some further results". In:
 J. Stat. Mech. 1602.2 (2016), p. 023103. DOI: 10.1088/1742-5468/2016/02/023103.
- [229] L. Mirsky. "A trace Inequality of John von Neumann". In: Monatsh. für Math. 79 (1975), pp. 303–306. DOI: 10.1007/BF01647331.
- [230] A. Müller-Hermes and D. Reeb. "Monotonicity of the Quantum Relative Entropy Under Positive Maps". In: Annales Henri Poincaré 18 (2015). DOI: 10.1007/s00023-017-0550-9.
- [231] P. Calabrese, J. Cardy, and I. Peschel. "Corrections to scaling for block entanglement in massive spin-chains". In: J. Stat. Mech. 1009 (2010), P09003. DOI: 10.1088/1742-5468/2010/09/P09003.
- [232] R. J. Baxter. *Exactly solved models in statistical mechanics*. Elsevier, 2016.
- [233] I. Affleck and A. W. W. Ludwig. "Universal noninteger "ground-state degeneracy" in critical quantum systems". In: *Phys. Rev. Lett.* 67.2 (1991), pp. 161–164. DOI: 10.1103/PhysRevLett. 67.161.